LOUISVILLE-SOUTHERN INDIANA OHIO RIVER BRIDGES



Toll Collection System
Roadside Back Office Component
Toll Collection System Contract

TABLE OF CONTENTS

1.0 CONTRACT AND CONTRACT DOCUMENTS	<u>3</u>
1.1. Certain Definitions	3
1.2. CONTRACT DOCUMENTS AND ORDER OF PRECEDENCE	3
1.3. COORDINATION AND INTENT OF CONTRACT DOCUMENTS	5
1.4. CONTRACTOR'S REVIEW OF CONTRACT DOCUMENTS	
1.5. Latest Version Governs	6
1.6. COMPUTATION OF TIME PERIODS; AGREEMENT TO CHANGE TIME PERIODS	<u> </u>
1.7. Interpretation of Contract Documents	<u>6</u>
1.8. Standard for Approvals	
1.9. FEDERAL REQUIREMENTS	<u>7</u>
2.0 CONTRACT PRICE	<u>7</u>
3.0 COMMENCEMENT AND CONTRACT TIME	<u>7</u>
3.1 TIME OF THE ESSENCE	7
3.1. TIME OF THE ESSENCE	887
3.3. COMPLETION DEADLINES	98
3.4. LIQUIDATED DAMAGES	111011
3.5. SCHEDULING OF DESIGN, CONSTRUCTION AND PAYMENT	111011
3.6. Prerequisites for Start of Construction	
3.7. RECOVERY SCHEDULE	
4.0 CONTRACT CHANGES	13 1213
4.1. CHANGE ORDERS GENERALLY	<u>131313</u>
4.2. Construction Change Directives	<u> 151415</u>
4.3. Joint Board-Initiated Change Orders	<u> 171416</u>
4.4. CONTRACTOR-REQUESTED CHANGE ORDERS	
4.5. CONTENTS OF CHANGE ORDERS	<u> 232023</u>
4.6. PRICING AND FORCE ACCOUNT; CERTAIN LIMITATIONS	25 2225
4.7. NECESSARY BASIC CONFIGURATION CHANGES	
4.8. CHANGE ORDERS FOR FORCE MAJEURE EVENTS AND HAZARDOUS MATERIAL	<u>s312830</u>
4.9. CHANGE ORDER RECORDS	343134
4.10. MATTERS NOT ELIGIBLE FOR CHANGE ORDERS AND WAIVER	
4.11. CHANGES NOT REQUIRING CHANGE ORDER	
4.12. No Release or Waiver	
5.0 VALUE ENGINEERING CHANGE PROPOSALS	38 3538
5.1. DESCRIPTION	<u>383538</u>
5.2. CONCEPTUAL VALUE ENGINEERING CHANGE PROPOSAL (CVEC)	39 3639
5.3. VALUE ENGINEERING CHANGE PROPOSAL (VECP)	40 3640
6.0 PAYMENTS	44 4044
6.1. MILESTONE PAYMENTS	
6.2. No RETAINAGE WITHHELD	

6.3. DEDUCTIONS, EXCLUSIONS AND LIMITATIONS ON PAYMENT; PAYMENT FOR	CERTAIN
<u>ITEMS 514751</u>	
6.4. FINAL PAYMENT	<u>534953</u>
6.5. PAYMENT AS FULL COMPENSATION	55 5155
6.6. FINAL PAYMENT OF MAINTENANCE PRICE	55 5255
6.7. ADMINISTRATION OF PAYMENT	56 5256
6.8. FUND AVAILABILITY	57 5357
6.9. PAYMENT TO SUBCONTRACTORS	57 5357
6.10. Cost Principles	58 5458
6.11. DISPUTES	<u>585458</u>
7.0 PROSECUTION, CONTROL, AND INSPECTION OF THE WORK	<u>585458</u>
7.1. AUTHORITY OF JOINT BOARD PERSONNEL	58 5458
7.2. CONFORMITY WITH PLANS AND SPECIFICATIONS	605660
7.3. REMOVAL OF DEFECTIVE AND UNAUTHORIZED WORK	
7.4. COOPERATION BY THE CONTRACTOR	62 5862
7.5. THE JOINT BOARD REVIEW AND INSPECTION GENERALLY	63 5963
7.5. THE JOINT BOARD REVIEW AND INSPECTION GENERALLY	64 606 4
7.7. INTENTIONALLY OMITTED	65 6165
7.8. LIMITATIONS OF OPERATIONS	
7.9. CHARACTER OF WORKERS, METHODS, AND EQUIPMENT	
7.10. SUBCONTRACTING	67 6367
7.11. No ESTOPPEL	70 6670
7.12. ACCEPTANCE OF NONCONFORMING WORK	71 6771
7.13. KEY PERSONNEL	
8.0 LIQUIDATED DAMAGES AND STIPULATED DAMAGES	
8.1. REASONABLE APPROXIMATIONS OF DAMAGES	736973
8.2 LIQUIDATED DAMAGES FOR DELAYS IN ACHIEVING COMPLETION	746974
8.2. LIQUIDATED DAMAGES FOR DELAYS IN ACHIEVING COMPLETION	767175
8.4. LIQUIDATED DAMAGES FOR LANE CLOSURES AND TOLL LANE CLOSURES	777276
8.5. OTHER LIQUIDATED DAMAGE PROVISIONS	
8.6. DISPUTES RELATING TO LIQUIDATED DAMAGES	
8.7. NO WAIVER OF ADDITIONAL DAMAGES EXCEPT AS SPECIFICALLY STATED	
8.8. STIPULATED DAMAGES	
8.9. PAYMENT; OFFSET; REDUCTION; WAIVER	827581
8.10. No Waiver of Additional Damages Except as Specifically Stat	ED 827581
9.0 COMPLETION AND ACCEPTANCE	
9.1. Substantial Completion	83 7582
9.2. PUNCH LIST ACCEPTANCE	86 7885
9.3. FORMAL ACCEPTANCE	86 7985
9.4. END OF MAINTENANCE TERM ACCEPTANCE	
9.5. ASSIGNMENT OF CAUSES OF ACTION	89 <mark>8288</mark>
10.0 SUSPENSIONS, TERMINATION FOR CONVENIENCE, AND DEFAUL	

<u>10.1.</u>	SUSPENSIONS	89 8288
10.2.	TERMINATION FOR CONVENIENCE	91 8390
<u>10.3.</u>	DEFAULT	<u>989097</u>
<u>10.4.</u>	ANTICIPATORY BREACH	104 96103
11.0 IND	EMNIFICATION	<u>10698104</u>
<u>11.1.</u>	INDEMNIFICATIONS BY THE CONTRACTOR	106 9810 4
11.2.	DEFENSE AND INDEMNIFICATION PROCEDURES	108 100107
11.3.	No Effect on Other Rights	
12.0 INS	URANCE AND BONDING REQUIREMENTS	<u>112104110</u>
12.1.	INSURANCE	123 115120
12.2.	PAYMENT, PERFORMANCE AND WARRANTY BONDS	123 115120
12.3.	No Relief of Liability	
13.0 DIS	PUTE RESOLUTION	
13.2.	PARTNERING AND ELEVATED NEGOTIATIONS	
13.3.	ADMINISTRATIVE CLAIMS PROCESS	128 11912 4
13.4.	HEARING PROCEDURE	129 121126
13.5 .	LIMITATIONS ON CONSEQUENTIAL DAMAGES	130 122127
14.0 RES	SPONSIBILITY FOR DESIGN; INFORMATION SUPPLIED	<u>131123128</u>
14.1.	STANDARD OF CARE	131 12312 8
14.2.	DESIGN, INSTALLATION AND INTEGRATION OBLIGATIONS	131 123128
14.3.	PROFESSIONAL LICENSING AND REGISTRATION	133 125130
14.4.	USE AND OWNERSHIP OF DOCUMENTS [134 12513 0
14.5.	INFORMATION SUPPLIED	137 12813 4
14.6.	RESPONSIBILITY FOR DESIGN	<u>138129134</u>
14.7.	No Liability Regarding Reference Documents	138 129134
14.8.	CONTRACTOR REPRESENTATIONS	138 12913 4
<u>14.9.</u>	NOTIFICATION OF THIRD-PARTY CLAIMS	<u>139130135</u>
15.0 OBI	LIGATIONS OF THE CONTRACTOR	<u>139130136</u>
15.1.	PERFORMANCE REQUIREMENTS	140 130136
15.2.	GENERAL OBLIGATIONS OF THE CONTRACTOR	140 131136
15.3.	REPRESENTATIONS, WARRANTIES AND COVENANTS	
15.4.	DESIGN REQUIREMENTS	145 136141
15.5 .	GOVERNMENTAL APPROVALS AND PERMITS	146 136142
<u>15.6.</u>	DESIGN AND ENGINEERING PERSONNEL.	<u>146137142</u>
<u>15.7.</u>	PROGRESSION OF WORK	
<u>15.8.</u>	CONTRACTOR RESPONSIBLE FOR MEANS AND METHODS	<u>146137142</u>
<u>15.9.</u>	SERVICES, MATERIALS AND EQUIPMENT	<u>147137143</u>
<u>15.10.</u>	LIMITATION ON USE OF SITE AND OTHER AREAS	
<u>15.11.</u>	SAFETY AND PROTECTION	<u>147138143</u>
<u>15.12.</u>	SAFETY REPRESENTATIVES	
<u>15.13.</u>	HAZARD COMMUNICATIONS PROGRAMS	
15.14.	EMERGENCIES	148 13914 4

<u>15.15.</u>	RELATED WORK AT SITE	148 139144
15.16.	NO RELEASE OF THE CONTRACTOR OBLIGATIONS	149 139145
15.17.	MAINTENANCE DURING CONSTRUCTION	149 140146
15.18.	MAINTENANCE OBLIGATIONS AFTER SYSTEM ACCEPTANCE	<u>150140146</u>
16.0 MIS	SCELLANEOUS PROVISIONS	
16.1.	PERSONAL LIABILITY OF PUBLIC OFFICIALS	154 144150
16.2.	No Waiver of Legal Rights	1541441 50
16.3.	POLICIES, PROCEDURES AND STANDARDS	155 145151
16.4.	CONFLICTS OF INTEREST	
16.5.	LAWS TO BE OBSERVED	155 145151
16.6.	REFERENCE STANDARDS	155 146152
16.7.	No THIRD PARTY BENEFICIARY	156146152
16.8.	CONTRACTOR'S RESPONSIBILITY FOR WORK	157 147153
16.9.	DOCUMENTS AND RECORDS	157 147153
16.10.	RETENTION OF RECORDS	162 152159
16.11.	INTEREST	163 153159
16.12.	SEVERABILITY	163 153159
16.13.	ENTIRE AGREEMENT	163 153159
17.0 RE	PRESENTATIVES	<u>163153159</u>
17.1.	THE JOINT BOARD'S DESIGNATED REPRESENTATIVE	163 153159
17.2.	CONTRACTOR'S DESIGNATED REPRESENTATIVE	
17.3.	CHANGING DESIGNATED REPRESENTATIVE	
<u>18.0 A</u>	CCESS TO PROJECT SITE; UTILITIES; ENVIRONMENTAL	COMPLIANCE
	<u>163153160</u>	
<u>18.1.</u>	ACCESS TO PROJECT SITE	163 154160
18.2.	UTILITIES	164 154160
18.3.	HAZARDOUS MATERIALS MANAGEMENT	164 154161
18.4.	ENVIRONMENTAL COMPLIANCE	166 156162
19.0 AD	DITIONAL FEDERAL AND STATE PROVISIONS	
10	.1. CERTIFICATION REGARDING USE OF CONTRACT FUNDS FO	D L OPPVING
<u>13</u>	166 156162	R LUBBTING
19		EDTIFICATION
<u>13</u>	167157163	RIFICATION
19.3.	BUY AMERICA	168157164
19.4.	PREVAILING WAGE RATES	168 <u>158164</u>
19.5.	CIVIL RIGHTS ACT OF 1964	
19.6.	NOTICE TO ALL BIDDERS	168 <u>158164</u>
19.7.	SECOND TIER SUBCONTRACTS	
	E PARTICIPATION	
<u>20.1.</u>	DBE PARTICIPATION PLAN COMMERCIALLY USEFUL FUNCTION	10 3133103
<u>20.2.</u>	MONTHLY REPORTS OF DRE PARTICIPATION	109 139100 170150166

20.4.	REPLACEMENT OF NON-PERFORMING DBES	170 160166
20.5.	WAIVER PROCESS FOR DBE PROJECT GOAL	<u> 170160167</u>
20.6.	SUBCONTRACTOR AGREEMENTS	<u> 172161168</u>
20.7.	GOOD FAITH REQUIREMENTS AND NONCOMPLIANCE	<u> 172162168</u>
20.8.	PROMPT PAYMENT	<u> 174164170</u>
20.9.	AFFIDAVIT OF SUBCONTRACTOR PAYMENT	<u>174164170</u>
21.0 WC	ORKFORCE/EEO	<u> 174164171</u>
21.1.	RECORDS AND REPORTS	17 4164171
21.2.	PREQUALIFICATION	
22.0 WA	ARRANTIES	<u> 175165172</u>
22.1.	WARRANTIES	175 165172
22.2.	WARRANTIES APPLICABILITY OF WARRANTIES TO RE-DONE WORK	177 167174
22.3.	SUBCONTRACTOR WARRANTIES	178 167174
22.4.	No LIMITATION OF LIABILITY	178 168175
22.5.	DAMAGES FOR BREACH OF WARRANTY	179 168175

This	Toll Co	llection S	Syster	n Roadside Ba	ack Offi	ce Co	mponent To	oll Collection Sy	/stem
Cont	ract is n	nade and	ente	red into this _	day	of		2013 bet	ween
the C	Common	wealth of	f Ken	tucky on behal	If of the	Joint	Board, actir	ng by and throu	gh its
agent, the Kentucky Transportation Cabinet, acting in its official capacity and hereinafter						nafter			
referred to as KYTC, as the entity designated to procure this Contract by resolution of									
<u>the</u>	Joint	Board	in	accordance	with	the	Interlocal	Agreement,	and
, hereinafter referred to as the Contractor with									
reference to the Exhibits hereto and the following recitals.									

RECITALS

- 1. The RBOC Component Project generally consists of the design, installation, testing, integration, construction and maintenance of the roadside and back office portions of the toll collection system for the Louisville Southern-Indiana Ohio River Bridges project, a design, construction and reconstruction project of two Ohio River crossings located in Louisville, Kentucky; Prospect, Kentucky; Jeffersonville, Indiana; and Clarksville, Indiana, commonly referred to as the East End Crossing and the Downtown Crossing. The Downtown Crossing is made up of two bridges, the Kennedy BridgeCrossing and the New Downtown Crossing-Bridge.
- 2. The Indiana Finance Authority has entered into that certain Public Private Agreement East End Crossing Louisville-Southern Indiana Ohio River Bridges Project), dated as of _____December 27, 2012, for the design, construction, finance, operation and maintenance of the East End Crossing with WVB East End Partners, LLC as the private developer of such project. Said Public Private Agreement provides that the Developer at all times shall coordinate and cooperate, and require its contractors to coordinate and cooperate, with Contractor in connection with its activities relating to the installation and testing of the TCS and the ‡tolling £equipment, and the Developer shall make the Project Site available to Contractor for installation and testing of the toll collection system not later than 120 days prior to the anticipated date of substantial completion of the Developer's construction of the East End BridgeCrossing. The Public Private Agreement provides that IFA shall commence making availability payments to the Developer, upon such substantial completion date, and that tolling shall commence as soon as the East End Crossing is open to traffic, which is expected to occur upon substantial completion.
- 3. KYTC has entered into that certain Design Build Contract dated as of December 28, 2012 for the design and construction of the Downtown Crossing with Walsh Construction Company as the design-builder. The Downtown Crossing Design-Build Contract provides that the Design-Builder shall coordinate and cooperate with the Contractor throughout the course of construction. It is currently anticipated that the New Downtown bridge will be substantially complete and open to traffic by April 1, 2016. The entire Downtown Crossing must be substantially complete by no later than December 10, 2016.
- 4. KYTC issued a three part Request for Proposals on behalf of the Joint Board on April 17, 2013. The three parts of the RFP included: (a) the RBOC Component Tolling Component One; (b) the Electronic Toll Collection Component –

Tolling Component Two; and (c) the Operations Services Component – Tolling Component Three.

- 5. KYTC received responses to the RFP, including the Contractor's Technical Proposal for the RBOC Component dated _____, 2013 —and the Contractor's Price Proposal.
- 6. The RFP evaluation committee determined that the Proposal was the response to the RFP that provided best value to the Joint Board with respect to the RBOC Component Tolling Component One.
- 7. The Joint Board approved the RFP evaluation committee's determination on _____ which ultimately led to the Joint Board awarding this Contract to the Contractor.
- 8. The parties intend for this Contract to be a lump sum contract obligating the Contractor to perform all work necessary to complete the Project by the Completion Deadlines specified in the Contract Documents for the Construction Price specified as a component of the Contract Price in the Price Proposal, and within the Contract time, subject only to certain specified limited exceptions. To allow the Joint Board to budget for the Project and to reduce the risk of cost overruns, this Contract includes restrictions affecting the Contractor's ability to make claims for an increase to the Contract Price or an extension of the Completion Deadlines. Except for changes expressly allowed under the Contract, the Contractor has agreed in this Contract to assume such responsibilities, risks, and restrictions and has reflected the assumption of such responsibilities, risks, and restrictions in the Contract Price.
- 9. If Contractor fails to complete the Project by the Substantial Completion Deadline, then the Joint Board will suffer substantial losses and damages, including but not limited to adverse financial consequences to be incurred by IFA under the Public Private Agreement for the East End Crossing, and by KYTC under the Design Build Contract for the Downtown Crossing Design-Build Contract. The Contract Documents therefore provide that Contractor shall pay the Joint Board <code>Liquidated dD</code> amages if such Substantial Completion Deadline is not met.
- 10. The AgreementContract provides for the Contractor to provide a one year warranty of the Work commencing for each Crossing upon the Substantial Completion Acceptance Date for such Crossing and continuing until the first anniversary of such Substantial Completion Acceptance for the Warranty Price for each Crossing and the System as a whole. The Warranty Price is included in the Contract Price indicated in the Price Proposal.
- 11. The AgreementContract further provides for Contractor to maintain the entire System for the applicable Maintenance Price throughout the Maintenance Term for the Maintenance Price for each Crossing and the System as a whole indicated in the Price Proposal. The Maintenance Price is included in the Contract Price indicated in the Price Proposal. The maintenance obligations and Performance Requirements during the Maintenance Term are the same as during the Warranty Period, however the Maintenance Price is intended to take into consideration any factory warranties on equipment that may expire during the Warranty Period.

12. The RBOC Component's accuracy and reliability is of paramount importance to the Joint Board's ability to fulfill its duty to finance the design, construction and reconstruction of the Ohio River Crossings and to operate and maintain such crossings, and the Joint Board will suffer substantial losses and damages if the RBOC Component does not meet the Performance Requirements. The Contract Documents provide that Contractor shall pay the Joint Board substantial Stipulated_Liquidated

NOW, THEREFORE, in consideration of the sums to be paid to the Contractor by the Joint Board, the foregoing promises and the covenants and agreements set forth herein, the Parties hereby agree as follows:

1.0 Contract and Contract Documents

1.1. Certain Definitions

<u>Exhibit A</u> hereto contains the meaning of various terms used in the Contract Documents.

1.2. Contract Documents and Order of Precedence

- (a) The Contract Documents consist of the following documents, including any exhibits thereto. The Contract Documents are intended to be complementary and describe and provide for a complete contract for design and implementation of the Project. Except as provided below, in the event of any conflict among the Contract Documents, the order of precedence, from highest to lowest, shall be as set forth below.
 - 1. Addenda and modification to this Ceontract, including eChange eOrders
 - 2. This Contract and attached exhibits
- 3. Technical Proposal elements that exceed the requirements of other Contract Documents, including but not limited to those included in Exhibit XXXB (Contractor's Commitments and Clarifications)
- 4. Request for Proposals issued ____ and any amendments or addenda thereto_(excluding all Reference Documents as indicated by _____)
- 5. As to Work done on the Kentucky side of the state line, KYTC supplemental specifications dated _____, and as to work done on the Indiana side of the state line, INDOT supplemental specifications dated ______.
- 6. As to work done on the Kentucky side of the state line, KYTC standard specifications dated_____ and as to work done on the Indiana side of the state line, INDOT, INDOT standard specifications dated _____.
- 7. Any other documents specifically incorporated by reference as a Contract Document
 - 8. The Proposal

The Contract Documents also include the following documents which may be delivered, prepared or issued after the effective date of this Contract:

Notice to Proceed

Contractor's DBE participation plan

The post RFP questions and responses published by KYTC on behalf of the Joint Board shall not be deemed to be part of the Contract Documents and shall not be relevant in interpreting the Contract Documents except to the extent they may clarify provisions otherwise considered ambiguous. Such responses will not act to alter or contravene the language of the Contract Documents to which they pertain, nor shall they act to relieve the Contractor of its obligation to comply with any laws, regulations or codes, but rather shall be considered clarifications as to the intent of the referenced Contract Document provisions.

- (b) Notwithstanding the foregoing order of precedence, in the event and to the extent that Exhibit BXX— expressly specifies that it is intended to supersede specific provisions of the Contract Documents, Exhibit BXX— shall control over the specified provisions. In determining whether a conflict exists between the Proposal and other Contract Documents, to the extent that the Proposal includes any statements, offers, terms, concepts and designs that can reasonably be interpreted as offers to provide higher quality items than otherwise required by the Contract Documents or to perform services in addition to those otherwise required, or otherwise contains terms which the Joint Board considers to be more advantageous than the requirements of the other Contract Documents, the Proposal shall not be considered in conflict with the other Contract Documents, and Contractor's obligations hereunder shall include compliance with all such statements, offers and terms.
- (c) For installation and construction-related standards, specifications and requirements, the order of precedence set forth in clause (a) shall apply, except that the Final Design Documents shall also be considered Contract Documents and shall be added following the Proposal in the order of precedence, provided that (i) specifications contained therein shall have precedence over plans, and (ii) no conflict shall be deemed to exist between the Final Design Documents and the other Contract Documents with respect to requirements of the Final Design Documents that The Joint Board determines are more beneficial than the requirements of the other Contract Documents; and (iii) any other Deviations contained in the Final Design Documents shall have priority over conflicting requirements of other Contract Documents only to the extent that the conflicts are specifically identified in the approval.
- (d) Portions of the Reference Documents are referenced in the Contract Documents for the purpose of defining requirements of the Contract Documents. The referenced information shall be deemed incorporated in the Contract Documents to the extent that it is so referenced, with the same order of priority as the Contract Document in which the reference occurs.
- (e) Additional details and more stringent requirements contained in a lower priority document will control unless the requirements of the lower priority document present an actual conflict with the requirements of the higher level document (i.e. it is not possible to comply with both requirements). Notwithstanding the order of

precedence among Contract Documents set forth in this <u>Section 1.2</u>, in the event of a conflict among any standard or specification applicable to the Project established by reference contained in the Contract Documents to a described publication, the Joint Board shall have the right to determine, in its sole discretion, which provision applies regardless of the order of precedence of the documents in which such standards are referenced. Contractor shall request in writing the Joint Board's determination respecting the order of precedence involving the referenced standards promptly upon becoming aware of any such conflict. In the event of any conflict, ambiguity or inconsistency between or among provisions in this Contract and any other Contract Document, or any applicable codes in effect at the effective date of this Contract, the provisions that require the most complete scope, highest quality, highest degree of safety or greatest value to the Joint Board shall prevail.

1.3. Coordination and Intent of Contract Documents

It is the intent of the Contract Documents, to describe a functionally complete Project to be designed and constructed in accordance with the Contract Documents. Contractor shall furnish or perform all Work that may reasonably be inferred from the Contract Documents or from prevailing custom or trade usage as being required to produce the indicated result, whether or not specifically called for, at no additional cost to the Joint Board or the States.

Each of the Contract Documents is an essential part of the Contract and what is required by one shall be as binding as if required by all. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the <u>wW</u>ork by the Contractor. Contractor shall perform all <u>wW</u>ork required by the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results

Where reference is made in this Contract to a provision of another Contract Document, the reference refers to that provision as amended or supplemented by other provisions of the Contract Documents.

In case of a discrepancy relative to other Contract Documents, the approved products list, the Indiana approved materials list or the Kentucky list of approved materials shall be regarded the same as supplemental specifications.

1.4. Contractor's Review of Contract Documents

Contractor acknowledges and agrees that it had the opportunity and obligation, prior to submission of its Proposal, to review the terms and conditions of the Contract Documents (including those Reference Documents that are referenced in the Contract Documents, and pursuant to Section 1.2(d) above, are considered Contract Documents) and to bring to the attention of the Joint Board any conflicts or ambiguities contained therein. Contractor further acknowledges and agrees that it has independently reviewed the Contract Documents with legal counsel, and that it has the requisite experience and sophistication to understand, interpret and agree to the particular

language of the provisions of the Contract Documents. Accordingly, in case iof an ambiguity in or dispute regarding the interpretation of the Contract Documents, they shall not be interpreted or construed against the Person which prepared them, and, instead, other rules of interpretation and construction shall be used. The Contractor shall not take advantage of any apparent error or omission in the Contract Documents. The Contractor shall immediately notify the Joint Board upon discovering such an error or omission. Contractor shall promptly report in writing to the Joint Board any conflict, error, ambiguity, or discrepancy in any document prepared by the Joint Board which Contractor discovers and shall obtain a written interpretation or clarification from the Joint Board before proceeding with any work affected thereby. The fact that the Contract Documents omit or misdescribe any details of any work which are necessary to carry out the intent of the Contract Documents, or which are customarily performed, shall not relieve Contractor from performing such omitted work (no matter how extensive) or misdescribed details of the work, and they shall be performed as if fully and correctly set forth and described in the Contract Documents, without entitlement to a Change Order hereunder except as specifically allowed under Section 4.

1.5. Latest Version Governs

The Parties agree that except as otherwise specified in the Contract Documents or directed by the Joint Board, the latest version of a particular Contract Document will apply over an earlier version of that Contract Document, and material and workmanship specified by the number, symbol or title of any standard established by reference to a described publication affecting the Project shall comply with the latest edition or revision thereof and amendments and supplements thereto in effect on the Proposal Date.

1.6. Computation of Time Periods; Agreement to Change Time Periods

References to "days" contained in the Contract Documents shall mean calendar days unless otherwise specified, provided that if the date to perform any act or give any notice specified in the Contract Documents (including the last date for performance or provision of notice "within" specified time period) falls on a non-working day, such act or notice may be timely performed on the next succeeding day which is a working day. Notwithstanding the foregoing, requirements contained in the Contract Documents relating to actions to be taken in the event of an emergency or—and any other requirements for which it is clear that performance is intended to occur on a nonworking day, shall be required to be performed as specified, even though the date in question may fall on a nonworking day. The term "working day" shall mean a day that is not a weekend and that is not an official state holiday in either Kentucky or Indiana. The Parties may mutually agree to increase or decrease any time periods or to otherwise modify any deadline in the Contract Documents.

1.7. Interpretation of Contract Documents

In the Contract Documents, where appropriate: the singular includes the plural and vice versa; references to statutes or regulations include all statutory or regulatory provisions consolidating, amending or replacing the statute or regulation referred to; unless

otherwise indicated references to Codes are to the codified laws of the State; the words "including," "includes" and "include" shall be deemed to be followed by the words "without limitation"; unless otherwise indicated references to sections, appendices or schedules are to this Contract; words such as "herein," "hereof" and "hereunder" shall refer to the entire document in which they are contained and not to any particular provision or section; words not otherwise defined which have well-known technical or construction industry meanings, are used in accordance with such recognized meanings; references to Persons include their respective permitted successors and assigns and, in the case of Persons representing Governmental Entities, Persons succeeding to their respective functions and capacities; and words of any gender used herein shall include each other gender where appropriate. Unless otherwise specified, lists contained in the Contract Documents defining the Project or the work shall not be deemed all-inclusive. On plans, working drawings, and standard plans, calculated dimensions shall take precedence over scaled dimensions.

1.8. Standard for Approvals

In all cases where approvals or consents are required to be provided by the Joint Board or Contractor hereunder, such approvals or consents shall not be withheld unreasonably except in cases where a different standard (such as sole discretion) is specified. In cases where sole discretion is specified the decision shall not be subject to $\underline{D\underline{d}}$ ispute resolution hereunder.

1.9. Federal Requirements

The <u>wW</u>ork to be performed under this Contract will be financed in part with federal funds and is, therefore, subject to federal statutes, rules and regulations applicable to work financed with federal funds, including <u>but not limited to</u>-the federal requirements set forth in <u>RFP Exhibit __this Contract</u>. In the event of any conflict between any applicable federal requirements and the other requirements of the Contract Documents, the federal requirements shall prevail, take precedence and be in force over and against any such conflicting provisions.

2.0 Contract Price

The Joint Board shall pay Contractor the Contract Price in current funds for Contractor's performance of the Contract.

3.0 Commencement and Contract Time

3.1. Time of the Essence

Contractor and the Joint Board agree that time is of the essence of this Contract and all Completion Deadlines and milestones of the Project.

3.2. Notice to Proceed; Contract Time; Date of Commencement

Authorization allowing Contractor to proceed with the Work hereunder shall be provided by the Joint Board's issuance of a Notice to Proceed. Subject to the satisfaction of the conditions below, the Joint Board shall issue the Notice to Proceed on or before August 16, 2013. Delay in the issuance of the Notice to Proceed shall entitle the Contractor to a day-for-day extension of the Completion Deadlines as provided in the Contract Documents except to the extent such delay is caused or attributable to the acts, omissions, negligence or misconduct of Contractor or any Contractor-Related Entity.

The following are conditions that must be satisfied by Contractor prior to issuance of the Notice to Proceed

- (a) Contractor shall have delivered to the Joint Board the Performance Bond required pursuant to <u>Section 12.2</u>;
- (b) Contractor shall have delivered to the Joint Board the Payment Bond required pursuant to <u>Section 12.2;</u>
- (c) Contractor shall have delivered to the Joint Board any Guaranty of Contractor's obligations pursuant to the Contract Documents that may be required by the Joint Board;
- (d) Contractor shall have provided to the Joint Board the insurance policies, certificates of insurance, riders to its existing insurance policies or other evidence reasonably required by the Joint Board to confirm the existence of all the insurance coverages required pursuant to <u>Section 12.1</u>;
- (e) Contractor has submitted all EPDs used in connection with the Project, Contractor and the Joint Board shall have jointly reviewed and indexed the same as set forth in <u>Section 16.9.1</u> and such EPDs are in form and substance satisfactory to the Joint Board : and
- (f) the Source Code Escrow shall be in place pursuant to Section 14.4.4 and shall be in full force and effect.
- (f)(g) the Contractor shall have obtained the Joint Board's written approval of the Contractor's initial submittals required pursuant to Section 3.12.3.1 of the RFP.

The Joint Board shall have no obligation to Contractor for any compensation whatsoever until the issuance of the Notice to Proceed, except for nonrefundable portions of insurance and bond premiums. Except as required to obtain the Joint Board's approval of the initial submittals required pursuant to Section 3.12.3.1 of the RFP, ‡the Contractor shall perform no work prior to the Notice to Proceed.

3.3. Completion Deadlines

3.3.1. Substantial Completion_-Acceptance Deadline(s)

Contractor shall achieve <u>Substantial Completion Acceptance of the Work at the Operations Center on or before the Operations Center Substantial Completion Acceptance Deadline, Substantial Completion Acceptance of the Work at a Walk-in Center on or before the Walk in Center Substantial Completion Acceptance Deadline applicable to a Walk-in Center, Substantial Completion—Acceptance of each Crossing on or before the applicable Substantial Completion—Acceptance Deadline, and Substantial Completion Acceptance of the entire System on or before the System Substantial Completion Acceptance Deadline. The Substantial Completion Acceptance Deadlines are as follows:</u>

- (a) Operations Center Substantial Completion Acceptance Deadline: [60] days3 months from prior to the date that the . Operations Center is made available to the Contractor for its Work first bridge is open to traffic.
- (b) Walk-in Center Substantial Completion Acceptance Deadline: [60] days 3 months 3 months prior to the date that the first bridge is open to trafficfrom the date that the Walk-in Center is made available to the Contractor for its Work.
- (cb) East End Crossing (Toll Zones EEC-1 and EEC-2) Substantial Completion Acceptance Deadline:: 120 days from the Tolling Infrastructure Turnover date as defined in the Public-Private Agreement.
- (b<u>c</u>) <u>Downtown Crossing (Downtown CrossingToll Zones DB-1R and R-1)</u> Substantial Completion Acceptance Deadline: :
- (i) Downtown Crossing Temporary Configuration Substantial Completion Acceptance Deadline: [date_]______[TBD]
- (iid) <u>Kennedy Crossing</u> (Kennedy Bridge Substantial Completion Acceptance DeadlineToll Zones KB-1R and R-2) Substantial Completion Acceptance Deadline: [date_]or days from the date the Kennedy Crossing Project Site is made available to the Contractor for the Work.] [TBD][Substantial Completion Deadlines for the toll operations will be provided by Addendum to this RFP.]
- (f) System Substantial Completion Acceptance Deadline: 30 days after the Substantial Completion acceptance Date for the last Crossing to achieve Substantial Completion Acceptance.
- <u>- (e) Downtown Crossing (Toll Zones DB-1 and R-1) Substantial Completion Acceptance Deadline: _____ days from the date that the Design Build Team [completes the pavement on the New Downtown Crossing?].</u>

(f) Downtown Crossing (Toll Zones DB-1, R-1, KB-1 and R-2) Substantial Completion Acceptance Deadline: days from the date that the Design Build Team [completes the pavement on the Kennedy Crossing (Toll Zones KB-1 and R-2)?]. :

(iiig) New Downtown Crossing BridgeSystem Substantial Completion Acceptance Deadline (Toll zones EEC-1, EEC-2, DB-1, R-1, KB-1, R-2, Operations Center and Walk-in Center(s): [date or days from]

Each such deadline, as it may be extended hereunder, is referred to herein as a Substantial Completion Acceptance Deadline.

3.3.2. Punch List Acceptance Deadline

Contractor shall achieve Punch List Acceptance of the Operations Center on or before the Operations Center Punch List Acceptance Deadline, Punch List Acceptance of each Walk-in Center on or before the applicable Walk-in Center Punch List Acceptance Deadline, Punch List Acceptance of each Crossing (or individual bridge as part of the Crossing) Crossing on or before the Punch List Acceptance Deadline for such Crossing and Punch List Acceptance of the System on or before the Punch List Acceptance Deadline for the System. The Punch List Acceptance Deadline for each element of the Work that has a Substantial Acceptance Crossing Completion Deadline shall be 30 Days after the applicable Substantial Completion Acceptance date. of such Crossing Each such deadline for achieving Punch List Acceptance, as it may be extended hereunder, is referred to herein as a Punch List Acceptance Deadline.

3.3.3. Formal Acceptance Deadline

The Formal Acceptance Deadlines for the Operations Center, each Walk-in Center each Crossing and the System are as specified below.

Contractor shall achieve Formal Acceptance of the Operations Center on or before 3 months—days after Substantial Completion Acceptance of the Operations Center.

Contractor shall achieve Formal Acceptance of each Walk-in Center on or before 3 months—days after the Substantial Completion Acceptance of such Walk-in Center.

<u>Contractor shall achieve Formal Acceptance of each Crossing (except for the Downtown Crossing Temporary Configuration) Crossing on or before the Formal Acceptance Deadline for such Crossing, as the case may be. The Formal Acceptance Deadline for each Crossing shall be before 60_days after Substantial Completion Acceptance of such Crossing, and.</u>

The Formal Acceptance Deadline for the System shall be 180 days after the later of the East End Crossing (Toll Zones EEC-1 and EEC-2) Substantial Completion Acceptance Date or the Downtown Crossing (Toll Zones DB-1, R-1, KB-1 and R-2) New Downtown Crossing Bridge-Substantial- Completion -Acceptance Date.

Each such deadline for achieving Formal Acceptance, as it may be extended hereunder, is referred to herein as a Formal Acceptance Deadline.

3.3.4. No time Extensions

Except as otherwise specifically provided in <u>Section 4</u>, the Joint Board shall have no obligation to extend a Completion Deadline and Contractor shall not be relieved of its obligation to comply with the Project Schedule and to achieve Substantial Completion Acceptance, Punch List Acceptance and Formal Acceptance by the applicable Completion Deadlines for any reason.

3.3.5. Waiver of Statute of Limitations

To the extent the statute of limitations on any claims against the Contractor beginstatute of limitations on any claims against the Contractor begins to run at Substantial Completion Acceptance, the Contractor agrees to waive these rights and begin calculating the statute of limitations from the time of System Formal Acceptance.

3.4. Liquidated Damages

Failure by Contractor to meet the Completion Deadlines and milestone dates shall entitle the Joint Board to liquidated damages as specified in <u>Section 8</u> of this <u>agreementContract.</u>

3.5. Scheduling of Design, Construction and Payment

3.5.1. Project Schedule

Work shall be undertaken and completed in accordance with the Project Schedule. The Project Schedule shall be used by the parties for planning and monitoring the progress of the Work and as the basis for determining the amount of payments to be made to Contractor. Contractor shall incorporate the project milestones and the Completion Deadlines into its proposed Project Schedule. If Contractor fails to provide an acceptable Project Schedule within 90 Days after issuance of the Notice to Proceed, Contractor shall have no right to receive payments until such time as Contractor has prepared and the Joint Board has approved the Project Schedule.

3.5.2. Float

All Float contained in the Project Schedule, as initially approved or generated thereafter, shall be considered a Project resource available to either party or both Parties as needed to achieve schedule milestones, interim completion dates and/or Completion Deadlines. All Float shall be shown as such in the Project Schedule on each affected schedule path. Identification of (or failure to identify) Float on the Project-sSchedule. Once identified, Contractor shall monitor, account for and maintain Float in accordance with Critical Path methodology.

3.6. Prerequisites for Start of Construction

Contractor shall not start construction (or recommence construction following any suspension) of any portion of the Project prior to occurrence of all the following events, except with the prior written approval of the Joint Board, in its sole discretion, and Contractor shall commence such construction promptly following occurrence of such events:

The Joint Board shall have issued the Notice to Proceed.

The Joint Board shall have approved the Project Schedule.

The Joint Board shall have provided written acceptance of the updated civil-construction quality control plan, quality management program, and if hazardous materials are identified in connection with the Project, the Contractor must develop a the hazardous materials management plan relating to such work. Any hazardous materials plan must be consistent with those plans submitted by the Design Build Team and the Developer.

All requirements of the construction quality program and quality management program, which are a condition to construction and installation, shall have been met.

The Joint Board shall have approved all applicable Design Documents and received all Construction Documents relating to such work.

All Governmental Approvals necessary for construction of the applicable portion of the Project shall have been obtained and all conditions of such Governmental Approvals, which are a prerequisite to commencement of such construction, shall have been performed and satisfied.

All insurance policies, bonds and letters of credit required to be delivered to the Joint Board hereunder prior to commencement of construction shall have been received and approved by the Joint Board and shall remain in full force and effect.

All necessary rights of access for such portion of the Project shall have been obtained.

All pre-construction environmental surveys and mitigation shall have been completed as required for the area(s) proposed for construction, and Contractor shall have performed all other survey work and delivered all notices required by the Contract Documents to be delivered prior to commencement of construction on such portion of the Project.

3.7. Recovery Schedule

If at any time, the Work on any Critical Path item is delayed, other than solely for a reason for which Contractor is entitled to a time extension pursuant to <u>Section 4</u>, for a period which exceeds the greater of either 30 days in the aggregate or that number of days in the aggregate equal to 5% of the days remaining until a Completion Deadline, then Contractor shall prepare and submit to the Joint Board for review and approval with the next Monthly Progress Report a Recovery Schedule demonstrating Contractor's proposed plan to regain lost schedule progress and to achieve the original contractual milestones in accordance with this Contract, including Substantial Completion

Acceptance by the applicable Substantial Completion Acceptance Deadline, Punch List Acceptance by the applicable Punch List Acceptance Deadline and Formal Acceptance by the applicable Formal Acceptance Deadline.

The Joint Board shall notify Contractor within 14 Days after receipt of each such Recovery Schedule whether the Recovery Schedule is deemed accepted or rejected. Within 7 Days after any rejection by the Joint Board of the Recovery Schedule, Contractor will resubmit a revised Recovery Schedule incorporating the Joint Board's comments. When the Joint Board accepts Contractor's Recovery Schedule, Contractor shall, within five Days after the Joint Board's acceptance, incorporate and fully include such schedule into the applicable Project Schedule, deliver the same to the Joint Board and proceed in accordance with the approved Recovery Schedule. The Contractor shall continue Work during the Recovery Schedule approval process.

All costs incurred by Contractor in preparing, implementing and achieving the Recovery Schedule shall be borne by Contractor and shall not result in a change to the Contract Price, except to the extent that a change in the Contract Price is permitted in accordance with <u>Section 4</u>.

If Contractor fails to provide an acceptable Recovery Schedule as required herein, Contractor shall have no right to receive Milestone Payments until such time as Contractor has prepared and the Joint Board has approved such Recovery Schedule.

4.0 CONTRACT CHANGES

This <u>Section 4</u> sets forth the requirements for obtaining all Change Orders under this Contract. Contractor hereby acknowledges and agrees that the Contract Price constitutes full compensation for performance of all of the work for the Project, subject only to those exceptions specified in this <u>Section 4</u>, and that the Joint Board is subject to constraints limiting its ability to increase the Contract Price or extend any Completion Deadline. Contractor unconditionally and irrevocably waives the right to any claim for a time extension or for any monetary compensation in addition to the applicable Price and other compensation specified in this Contract, except in accordance with this <u>Section 4</u>. To the extent that any other provision of this Contract expressly provides for a Change Order to be issued, such provision is incorporated into and subject to this Section 4.

The Contract may be amended or modified only by (1) a Change Order (which upon execution by the Joint Board shall become a supplemental agreement), or (2) a Construction Change Directive. Adjustments of the Contract Price on account of changes in the work may be determined by either an agreed upon Change Order, which upon execution by the Joint Board shall become a supplemental agreement as provided in <u>Section 4.1</u>, or a Construction Change Directive as provided in <u>Section 4.2</u>.

4.1. Change Orders Generally

A Change Order is a written amendment to the terms and conditions of the Contract Documents issued in accordance with this <u>Section 4</u>. The Joint Board may, at any time and from time to time, without notice to any Surety, authorize and/or require changes in

the Work within the general scope of this Contract pursuant to a Change Order. For the purpose of this <u>Section 4</u>, any direction to perform work shall be considered within the general scope of this Contract if it is related to the Project; any direction to delete or modify work shall be considered within the general scope unless as a result this Contract would no longer be considered a design-build contract for the Project of the nature described in the RFP. Contractor shall have no obligation to perform any work outside the general scope of this Contract, except on terms mutually acceptable to the Joint Board and Contractor. The Joint Board may issue a unilateral Change Order as specified in <u>Section 4.3</u>. Change Orders may be requested by the Contractor only pursuant to <u>Section 4.4</u>. A Change Order shall not be effective for any purpose unless signed by the Joint Board. Change Orders may be issued for the following purposes (or combination thereof):

- a change in the ₩Work;
- 2. adjustment in the Contract Price;
- adjustment in a Completion Deadline;
- 4. revision of other terms and conditions of the Contract Documents.

The change order document shall be prepared by the Joint Board and executed by the Contractor. Once executed by the Contractor, the change order <u>document</u> will be presented to the Joint Board to execute. Upon execution by the Joint Board, the <u>Cehange eOrder</u> shall be considered a supplemental agreement to the <u>eContract</u> and only then a Contract Document.

If the Joint Board requests a proposal for a change in the work from the Contractor and subsequently elects not to proceed with the change, a Change Order shall be processed to reimburse the Contractor for any costs it reasonably incurred for estimating services, design services or preparation of proposed revisions to the Contract Documents, as provided in <u>Section 4.3</u>.

Methods used in determining adjustments to the Contract Price may include those listed in Section 4.2.

Adjustments to a Completion Deadline or milestone deadline, if any, shall also be included in any Change Order along with the adjustment to the Contract Price. In order to warrant an extension of a Completion Deadline, the Contractor must show that the work affected by the change was on the Critical Path of the Contractor's current schedule.

The Contractor must provide the Joint Board with Kentucky's form TC63-32 within ten (10) days of becoming aware of any item of work the Contractor considers extra work, over and above what is described in the Contract Documents or for any other item the Contractor feels warrants a Change Order affecting either the Contract Price or a Completion Deadline or milestone deadline, and before performing said work, as discussed more fully in <u>Section 4.4.2</u>. Kentucky Form TC63-32 shall also be provided within ten (10) days when the Joint Board issues a Construction Change Directive, if the

Contractor has not already submitted Kentucky Form TC63-32 for the same item. Submission of the TC63-32 form shall be considered an absolute condition precedent to maintaining a claim for extra work.

Failure to provide Form TC63-32 within ten (10) days of the Contractor becoming aware of the changed work shall be deemed a waiver by the Contractor to any claim for an increase in the Contract Price or extension of a Completion Deadline associated with the changed work.

Quantity Overrun, Under-run Adjustments

The Contractor shall be responsible for the calculation of various quantities necessary to design and construct the Project. Therefore, adjustment to the Contract Price due to quantity overrun or under-run shall not be permitted.

The Contractor further agrees that the Contract Price provides full and complete compensation for the under-run or overrun quantities, including any and all unreimbursed expenses, loss of expected reimburse, loss of anticipated profits, delay, inefficiency, and all other costs.

This Project shall be constructed using English Units. The Contractor is directed to use only those Standard Drawings and Standard Specifications that are in English Units. No additional compensation or payment shall be made for any work that may be required to convert from Metric Units to English unites or provide a substitute equivalent material.

4.2. Quantity Overrun, Under-run Adjustments

The Contractor shall be responsible for the calculation of various quantities necessary to design and construct the Project. Therefore, adjustment to the Contract Price due to quantity overrun or under-run shall not be permitted.

The Contractor further agrees that the Contract Price provides full and complete compensation for the under-run or overrun quantities, including any and all unreimbursed expenses, loss of expected reimbursement, loss of anticipated profits, delay, inefficiency, and all other costs.

This Project shall be constructed using English Units. The Contractor is directed to use only those Standard Drawings and Standard Specifications that are in English Units. No additional compensation or payment shall be made for any work that may be required to convert from Metric Units to English Units or to provide a substitute equivalent material.

Construction Change Directives

A Construction Change Directive is a written order signed by the Contractor directing a change in the Work prior to agreement on adjustment, if any, in the Contract Price or Contract Time, or both. The Joint Board may by Construction Change Directive, at any

time, without invalidating the Contract, and without notice to or releasing the surety, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other alterations or revisions. A Construction Change Directive developed by the Joint Board shall include a description of the change in Work, will state the basis for determining compensation, if any, and will state the Joint Board's proposed changes in the Contract Price and Contract Time, if any.

If the Construction Change Directive provides for an adjustment to the Contract Price, the adjustment shall be based on one of the following methods:

- 1. mutual acceptance of a lump sum price that is properly itemized by sufficient substantiating data to permit evaluation on an open book basis, which may be based on the pricing contained in the EPDs, where applicable, as well as Subcontractors' bid prices;
- 2. cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
 - 3. as provided below in Section 4.6.

Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed as directed in the Construction Change Directive pending execution of a formal Change Order (or if the Construction Change Directive states the work is within Contractor's original scope of work, Contractor shall proceed with the work as directed but shall have the right pursuant to Section 4.4 to request that the Joint Board issue a Change Order with respect thereto by delivery of a Potential Change Order (PCO) Notice. Contractor shall advise the Joint Board of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Price or Contract Time. A Construction Change Directive signed by the Contractor indicates the agreement of the Contractor therewith, including adjustment in Contract Price and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

Contractor shall not be entitled to additional compensation or time extension for any Work performed prior to receipt of a Construction Change Directive or Change Order, and Contractor acknowledges that it will be at risk if it elects to proceed with Work in advance of receipt of the same. In addition to provisions of a PCO Notice and subsequent Change Order request pursuant to Section 4.4, receipt of a Construction Change Directive from the Joint Board shall be a condition precedent to Contractor's right to make a Claim that a Joint Board-Directed Change has occurred. The fact that a Construction Change Directive was issued shall not be considered evidence that in fact a Joint Board-Directed Change occurred. The determination whether a Joint Board-Directed Change in fact occurred shall be based on an analysis of the original requirements of the Contract Documents and a determination whether the Construction Change Directive in fact constituted a change in those requirements.

For contract adjustments that reduce the Scope of Work, anticipated profits on the eliminated Work shall be deducted from the Contract Price along with the cost

adjustment for the revised Work. Before performing the changed work, the Contractor shall reach agreement with the Joint Board concerning the basis for the Contract Price adjustment. Absent an agreement, the Joint Board will pay for the changed Work on a force account basis as described below.

4.2.4.3. Joint Board-Initiated Change Orders

This <u>Section 4.3</u> concerns (a) Change Orders issued by the Joint Board following a Request for Change Proposal, and (b) Change Orders unilaterally issued by the Joint Board.

4.2.1.4.3.1. Request for Change Proposal

4.2.1.1.4.3.1.1. If the Joint Board desires to issue a Joint Board-Directed Change or to evaluate whether to initiate such a change, then the Joint Board may, at its discretion, issue a Request for Change Proposal. A Construction Change Directive may also constitute a Request for Change Proposal.

4.2.1.2.4.3.1.2. Within five working days after Contractor's receipt of a Request for Change Proposal, or such longer period as may be mutually agreed to by the Joint Board and Contractor, the Joint Board and Contractor shall consult to define the proposed scope of the change. Within five working days after the initial consultation, or such longer period as may be mutually agreed to by the Joint Board and Contractor, the Joint Board and Contractor shall consult concerning the estimated cost and time impacts.

4.2.1.3.4.3.1.3. Within five working days after the second consultation described in Section 4.3.1.2, the Joint Board shall notify Contractor whether the Joint Board (a) wishes to issue a Change Order, (b) wishes to request Contractor to provide a cost and schedule proposal on KYTC Form TC63-32 as discussed at the meeting, (c) wishes to request Contractor to prepare a modified work plan for the change and a cost and schedule proposal based on the modified plan, or (d) no longer wishes to issue a Change Order. The Joint Board may at any time, in its sole discretion, require Contractor to provide two alternative cost and schedule Proposals, one of which shall provide for a time extension and any additional costs permitted hereunder, and the other of which shall show all Acceleration Costs associated with meeting the original Completion Deadlines, as well as any additional costs permitted hereunder.

4.2.1.4.4.3.1.4. If so requested, Contractor shall, within ten working days after receipt of the notification described in Section 4.3.1.3, or such longer period as may be mutually agreed to by the Joint Board and Contractor, prepare and submit to the Joint Board for review and approval by the Joint Board a cost and schedule proposal (in the format provided by the Joint Board) for the requested change, complying with all applicable requirements of this Section 4. and incorporating and fully reflecting all requests made by the Joint Board. Contractor shall bear the cost of developing the cost and schedule proposal, including any modifications thereto requested by the Joint

Board, except that costs of design work required for preparation of plans or exhibits necessary to the cost and schedule proposal, as pre-authorized by the Joint Board, may be included in the Change Order as reimbursable items. If the Change Order is approved, such design costs will be included within the Change Order, otherwise, they shall be separately reimbursed through a separate Change Order.

<u>4.3.1.5.</u> If Contractor and the Joint Board agree that a change in the requirements relating to the Work has occurred but disagree as to whether the change justifies additional compensation or time or disagree as to the amount of any change to be made to the Contract Price or Completion Deadlines, the Joint Board may, in its sole discretion, order Contractor to proceed with the performance of the work in question notwithstanding such disagreement. Such order may, at the Joint Board's option, be in the form of: (a) a Force Account Change Order as provided in <u>Section 4.6.6</u> or (b) a Change Order Directive under <u>Section 4.3.2</u>.

4.2.2.4.3.2. Unilateral Change Orders - Construction Change Directives

The Joint Board may issue a <u>unilateral</u> Change Order at any time, regardless of whether it has issued a Request for Change Proposal. Contractor shall be entitled to compensation in accordance with <u>Section 4</u> for additional work which is required to be performed as the result of any unilateral Change Order, and shall have the right to submit the issue of entitlement to an extension of the Completion Deadlines to Dispute Resolution in accordance with <u>Section 12</u>. For deductive unilateral Change Orders, the Change Order may contain a Contract Price deduction deemed appropriate by the Joint Board, and Contractor shall have the right to submit the amount of such Contract Price deduction to Dispute Resolution in accordance with <u>Section 12</u>.

A Construction Change Directive is a written order signed by the Joint Board directing a change in the Work prior to agreement on adjustment, if any, in the Contract Price or Contract Time, or both. The Joint Board may by Construction Change Directive, at any time, without invalidating the Contract, and without notice to or releasing the surety, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other alterations or revisions. A Construction Change Directive developed by the Joint Board shall include a description of the change in Work, will state the basis for determining compensation, if any, and will state the Joint Board's proposed changes in the Contract Price and Contract Time, if any.

If the Construction Change Directive provides for an adjustment to the Contract Price, the adjustment shall be based on one of the following methods:

5. mutual acceptance of a lump sum price that is properly itemized by sufficient substantiating data to permit evaluation on an open book basis, which may be based on the pricing contained in the EPDs, where applicable, as well as Subcontractors' bid prices;

- 6. cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
 - 7. as provided below in Section 4.6.

Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed as directed in the Construction Change Directive pending execution of a formal Change Order (or if the Construction Change Directive states the work is within Contractor's original Scope of Work, Contractor shall proceed with the work as directed but shall have the right pursuant to Section 4.4 to request that the Joint Board issue a Change Order with respect thereto by delivery of a Potential Change Order (PCO) Notice. Contractor shall advise the Joint Board of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Price or Contract Time. A Construction Change Directive signed by the Contractor indicates the agreement of the Contractor therewith, including adjustment in Contract Price and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

Contractor shall not be entitled to additional compensation or time extension for any Work performed prior to receipt of a Construction Change Directive or Change Order, and Contractor acknowledges that it will be at risk if it elects to proceed with Work in advance of receipt of the same. In addition to provisions of a PCO Notice and subsequent Change Order request pursuant to Section 4.4, receipt of a Construction Change Directive from the Joint Board shall be a condition precedent to Contractor's right to make a Claim that a Joint Board-Directed Change has occurred. The fact that a Construction Change Directive was issued shall not be considered evidence that in fact a Joint Board-Directed Change occurred. The determination whether a Joint Board-Directed Change in fact occurred shall be based on an analysis of the original requirements of the Contract Documents and a determination whether the Construction Change Directive in fact constituted a change in those requirements.

For contract adjustments that reduce the Scope of Work, anticipated profits on the eliminated Work shall be deducted from the Contract Price along with the cost adjustment for the revised Work. Before performing the changed work, the Contractor shall reach agreement with the Joint Board concerning the basis for the Contract Price adjustment. Absent an agreement, the Joint Board will pay for the changed Work on a force account basis as described below in Section 4.6.

4.3.4.4. Contractor-Requested Change Orders

4.3.1.4.4.1. Eligible Changes

Contractor may request a Change Order to extend a Completion Deadline only for delays directly attributable to one or more of the following events or circumstances which change the duration of a Critical Path:

(a) Force Majeure Events;

- (b) Joint Board-Caused Delays;
- (c) delays relating to Differing Site Conditions, discovery of Hazardous Materials, and/or Changes in Law, to the extent permitted by Section 4.8; and
- (d) delays relating to Necessary Basic Configuration Changes, to the extent permitted by Section 4.7.

Contractor may request a Change Order to increase the Contract Price only for increased costs of performance of the work as follows:

- (a) additional costs directly attributable to additional Work resulting from the Joint Board-Directed Changes and the Joint Board-Caused Delays for which the Joint Board has not submitted a Change Order or a Request for Change Proposal;
- (b) additional costs directly attributable to Necessary Basic Configuration Changes, to the extent provided in <u>Section 4.7_;</u>
- (c) additional costs relating to Differing Site Conditions, Hazardous Materials, and Force Majeure Events, to the extent provided in Section 4.8; or
- (d) additional costs directly attributable to uncovering, removing and restoring work, to the extent provided in <u>Section 7.1</u>.

Contractor's entitlement to a Change Order for eligible changes is subject to the restrictions and limitations contained in this <u>Section 4</u> and furthermore is subject to Contractor's compliance with all notification and other requirements identified herein. Contractor shall initiate the Change Order process by delivery of a PCO Notice as described in <u>Section 4.4.2</u>, followed by submittal of a Request for Change Order and supporting documentation to the Joint Board.

4.3.2.4.4.2. Procedures

The requirements set forth in this <u>Section 4.4.2</u> constitute conditions precedent to Contractor's entitlement to request and receive a Change Order except those involving a Request for Change Proposal. Contractor understands that it shall be forever barred from recovering against the Joint Board under this <u>Section 4</u> if it fails to give notice of any act, or failure to act, by the Joint Board or any of its representatives or the happening of any occurrence pursuant to a proper PCO Notice, and thereafter complies with the remaining requirements of this <u>Section 4.4</u>.

4.3.2.1.4.4.2.1. Delivery of PCO Notices

Contractor acknowledges the importance of providing prompt notification to the Joint Board upon occurrence of anything entitling Contractor to a Change Order under <u>Section 4.4.1</u>. Among other things, such notification allows the Joint Board to take action to mitigate adverse impacts. Such notification must be delivered as promptly as

possible after the occurrence of such event or situation, through a PCO Notice as described in Section 4.4.2.

4.3.2.4.4.2.2. PCO Notices

The term "PCO Notice" shall mean a notice delivered by Contractor, meeting the requirements set forth below, stating that an event or situation has occurred within the scope of <u>Section 4.34.1</u> and stating which subsection thereof is applicable. The first notice shall be labeled "PCO Notice No. 1" and subsequent notices shall be numbered sequentially.

The PCO Notice shall: (a) include a copy of completed Kentucky Form TC63-32 submitted in accordance with <u>Section 4.1</u>, (b) state in detail the facts underlying the anticipated Request for Change Order, the reasons why Contractor believes additional compensation or time will or may be due and the date of occurrence, (c) state the name, title, and activity of each project management consultant and Joint Board representative knowledgeable of the facts underlying the anticipated Request for Change Order, (d) identify any documents and the substance of any oral communication involved in the facts underlying the anticipated Request for Change Order, (e) state in detail the basis for necessary accelerated schedule performance, if applicable, (f) state in detail the basis that the work is not required by this Contract, if applicable, (g) identify particular elements of Contract performance for which additional compensation may be sought under this <u>Section 4.4.2</u>, (h) identify any potential Critical Path impacts, and (i) provide an estimate of the time within which a response to the notice is required to minimize cost, delay or disruption of performance.

If the Request for Change Order relates to a decision which this Contract leaves to the discretion of a Person or as to which this Contract provides that such Person's decision is final, the PCO Notice shall set out in detail all facts supporting Contractor's objection to the decision, including all facts supporting any contention that the decision was capricious or arbitrary or is not supported by substantial evidence.

Any adjustments made to this Contract shall not include increased costs or time extensions for delay resulting from Contractor's failure to timely provide requested additional information under this <u>Section 4.4.2.2</u>.

4.3.2.3.4.4.2.3. Waiver

4.4.2.3.1 Each PCO Notice shall be delivered as promptly as possible after the occurrence of such event or situation. If any PCO Notice is delivered later than ten days after Contractor first discovered (or should have discovered in the exercise of reasonable prudence) the occurrence described therein, Contractor shall be deemed to have waived (a) the right to collect any costs incurred prior to the date of delivery of the PCO Notice, and (b) the right to seek an extension of any Completion Deadline with respect to any delay in a Critical Path which accrued prior to the date of delivery of the written notice. Furthermore, if any PCO Notice concerns any hazardous material condition or material, Contractor shall be deemed to have waived the right to collect any

and all costs incurred in connection therewith to the extent that the Joint Board is not afforded the opportunity to inspect such material or condition before it is disturbed.

4.4.2.3.2 In addition to the limitations set forth in <u>Section 4.4.2.3.1</u>, Contractor's failure to provide a PCO Notice within 60 days after Contractor first discovered (or should have discovered in the exercise of reasonable prudence) the occurrence of a given event or situation shall preclude Contractor from any relief, unless Contractor can show, based on a preponderance of the evidence, that (a) the Joint Board was not materially prejudiced by the lack of notice, or (b) the Joint Board's designated representative specified in accordance with <u>Section 17.1</u> had actual knowledge, prior to the expiration of the 60-day period, of the event or situation and that Contractor believed it was entitled to a Change Order with respect thereto.

4.3.2.4.4.4.2.4. Delivery of Request for Change Order

Contractor shall deliver all Requests for Change Order under this <u>Section 4.4</u> to the Joint Board within 30 days after delivery of the PCO Notice, or such longer period of time as may be allowed in writing by the Joint Board. The Joint Board may require design and construction costs to be covered by separate Requests for Change Order. If Contractor requests a time extension, then the Joint Board, in its sole discretion, may require Contractor to provide two alternative Requests for Change Order, one of which shall provide for a time extension and any additional costs permitted hereunder, and the other of which shall show all Acceleration Costs associated with meeting the original Completion Deadlines, as well as any additional costs permitted hereunder. If it is not feasible to recover to the original Completion Deadline or if Contractor believes that the costs associated with such a recovery are prohibitive, then Contractor shall recommend a date to be shown in the alternative Change Order form.

Contractor shall furnish, when requested by the Joint Board or its designee, such further information and details as may be required to determine the facts or contentions involved. Contractor agrees that it shall give the Joint Board or its designee access to any and all of Contractor's books, records and other materials relating to the work, and shall cause its Subcontractors to do the same, so that the Joint Board or its designee can investigate the basis for such proposed Change Order. Contractor shall provide the Joint Board with a monthly update to all outstanding Requests for Change Order describing the status of all previously unfulfilled requirements and stating any changes in projections previously delivered to the Joint Board, expenditures to date and time anticipated for completion of the activities for which the time extension is claimed. The Joint Board may reject the Request for Change Order at any point in the process. The Joint Board's failure to respond to a complete Request for Change Order within 15 working days of delivery of the request shall not be deemed an acceptance of such request, and the Contractor shall have the burden of following up with the Joint Board on the status of any such Request for Change Order.

4.3.2.5.4.4.2.5. Review of Subcontractor Claims

Each Request for Change Order involving Subcontractor Work, and each update to an incomplete Change Order request involving such Work shall include a summary of Contractor's analysis of all Subcontractor claims components and shall include a certification signed by Contractor's Project Manager stating that Contractor has investigated the basis for the Subcontractor's claims and has determined that all such claims are justified as to entitlement and amount of money and/or time requested, has reviewed and verified the adequacy of all back-up documentation to be placed in escrow pursuant to Section 16.9.1, and has no reason to believe and does not believe that the factual basis for the Subcontractor's claim is falsely represented. Any Request for Change Order involving Subcontractor Work which is not accompanied by such analysis and certification shall be considered incomplete.

4.3.3.4.4.3. Performance of Disputed Work

If the Joint Board refuses to issue a Change Order based on Contractor's request, Contractor shall nevertheless perform all work as specified by Change Order Directive, and shall have the right to submit the issue to Dispute Resolution pursuant to <u>Article 13</u>. Contractor shall maintain and deliver to the Joint Board, upon request, contemporaneous records, meeting the requirements of <u>Section 4.9</u>, for all work performed which Contractor believes constitutes extra work (including non-construction work), until all disputes regarding entitlement or cost of such work are resolved.

4.4.4.5. Contents of Change Orders

4.4.1.4.5.1. Form of Change Order

Each Cost and Schedule Proposal and Request for Change Order shall be prepared in form acceptable to the Joint Board, and shall meet all applicable requirements of this Section 4. Where practical, Change Orders shall be on a lump sum basis.

4.4.2.4.5.2. Scope of Work, Cost Estimate, Delay Analysis and Other Supporting Documentation

Contractor shall prepare a scope of work, cost estimate, delay analysis and other information as required by this Section 4.5.2 for each cost and schedule proposal and Request for Change Order.

4.4.2.1.4.5.2.1. Scope of Work

The scope of work shall describe in detail satisfactory to the Joint Board all activities associated with the Change Order, including a description of additions, deletions and modifications to the existing requirements of the Contract Documents.

4.4.2.2.4.5.2.2. Cost Estimate

The cost estimate shall set out the estimated costs in such a way that a fair evaluation can be made. It shall include a breakdown for labor, materials, equipment and markups for overhead and profit, unless the Joint Board agrees otherwise. If the work is to be performed by Subcontractors and if the work is sufficiently defined to obtain

Subcontractor quotes, Contractor shall obtain quotes (with breakdowns showing cost of labor, materials, equipment and markups for overhead and profit) on the Subcontractor's stationery and shall include such quotes as back-up for Contractor's estimate. No markup shall be allowed in excess of the amounts allowed under <u>Sections 4.6.1 and 4.6.7</u>. Contractor shall identify all conditions with respect to prices or other aspects of the cost estimate, such as pricing contingent on firm orders being made by a certain date or the occurrence or non-occurrence of an event.

4.4.2.3.4.5.2.3. Delay Analysis

If Contractor claims that such event, situation or change affects a Critical Path, it shall provide an impacted delay analysis indicating all activities represented or affected by the change, with activity numbers, durations, predecessor and successor activities, resources and cost, and with a narrative report, in form satisfactory to the Joint Board. which compares the proposed new schedule to the current approved Project Schedule. If the Joint Board, its members, or another Joint Board or member contractor is the sole source of the delay to the Contractor's prosecution of the work, the Contractor shall be entitled to an adjustment in the Contract Time, provided that the affected portion of Contractor's Work is shown as a Critical Path task on the Contractor's most recent schedule. In such cases, the Contractor shall also be entitled to an increase in the Contract Price to compensate the Contractor for its extended on-site overhead for the period of delay caused solely by the Joint Board, its members or another Joint Board or member contractor, as well as the Contractor's costs for idle equipment and idle manpower, provided that such costs cannot be mitigated, such as by use of such manpower and equipment elsewhere on the Project or on other projects. Such overhead shall be measured by the Contractor's cost-loaded CPM schedule as last approved by the Joint Board. The Contractor shall make no claims for any other types of delay damages, and shall not be entitled to an adjustment in the Contract Price, for lost productivity during ongoing Work or disruptions in the Work. A claim for damages due to delays caused solely by the Joint Board, its members, or another Joint Board or member contractor must be made in accordance with Section 4.4.2 above, including submission of the Kentucky TC 63-32 form within ten days of the start of such delay. Failure to timely submit the required written notice shall be a waiver of any such claim. The remedies discussed above for delays due to the sole fault of the Joint Board, its member, or an Agency Contractor shall not be available if a concurrent delay, caused by the Contractor's own actions, or those for whom the Contractor is responsible, is experienced by the Contractor for the same time period. However, any concurrent delay caused by the Contractor shall only nullify the Contractor's recovery rights as to the actual period of overlap with an otherwise compensable delay caused solely by the Joint Board.

4.4.2.4.4.5.2.4. Other Supporting Documentation

Contractor shall provide such other supporting documentation as may be required by the Joint Board.

4.4.3.4.5.3. Justification

All Requests for Change Orders shall include an attachment containing a detailed narrative justification therefor, describing the circumstances underlying the proposed change, identifying the specific provision(s) of Section 4 which permit a Change Order to be issued, and describing the data and documents which establish the necessity and amount of such proposed change.

4.4.4.4.5.4. Contractor Representation

Each Change Order shall be accompanied by a certification under penalty of perjury, in form acceptable to the Joint Board, executed by Contractor and stating that (a) the amount of time and/or compensation requested is justified as to entitlement and amount, (b) the amount of time and/or compensation requested includes all known and anticipated impacts or amounts, direct, indirect and consequential, which may be incurred as a result of the event or matter giving rise to such proposed change, and (c) the cost and pricing data forming the basis for the Change Order is complete, accurate and current. Each Change Order involving work by a Subcontractor for which pricing data is required to be provided under Section 16.9.1 shall include a statement that the Subcontractor pricing data has been provided and shall include a copy of the certification by the Subcontractor that the information is complete, accurate and current.

4.5.4.6. Pricing and Force Account; Certain Limitations

4.5.1.4.6.1. Detailed Cost Proposal

Contractor may be required to submit a detailed cost proposal: (a) showing all impacts on this Contract from Work additions, deletions and modifications shown in the Change Order being priced; and (b) setting out the proposed costs in such a way that a fair evaluation can be made. When the Change Order adds Work to Contractor's scope, the increase in the Contract Price shall be negotiated based on estimates or actual costs of labor, material and equipment. When the Change Order deletes Work from Contractor's scope, the amount of the reduction in the Contract Price shall be based upon an estimate including a bill of material, a breakdown of labor and equipment costs, on an Open-Book Basis, and to the extent applicable, the pricing in the EPDs and Subcontractor bid prices. Markup for profit and overhead consistent with Section -4.6.7 shall apply for added and deleted Work Change Orders

4.5.2.4.6.2. Identification of Conditions

Contractor shall identify all conditions with respect to prices or other aspects of the proposal, such as pricing contingent on firm orders being made by a certain date or the occurrence or nonoccurrence of an event.

4.5.3.4.6.3. Contents

A negotiated Change Order shall specify costs, scheduling requirements, time extensions and all costs of any nature arising out of the work covered by the Change Order. Notwithstanding the foregoing, the parties may mutually agree to use a multiple-step process involving issuance of a Change Order which includes an estimated construction and installation cost and which provides for a revised Change Order to be

issued after a certain design level has been reached, thus allowing a refinement and further definition of the estimated construction and installation cost.

4.5.4.4.6.4. Unit Priced Change Orders

Unit prices shall be deemed to include all costs for labor, material, overhead and profit, and shall not be subject to change regardless of any change in the estimated quantities. Unit-priced Change Orders shall initially include an estimated increase in the Contract Price based on estimated quantities. Upon final determination of the quantities, the Joint Board will issue a modified Change Order setting forth the final adjustment to the applicable Price.

4.5.5.4.6.5. All-Inclusive Change Orders

All Change Orders submitted by Contractor shall be all-inclusive, comprehensive and complete and shall not include any conditions with respect to pricing or schedule.

4.5.6.4.6.6. Force Account Change Orders

If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Price, the method and the adjustment shall be determined by the Joint Board on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Price, a reasonable allowance for overhead and profit, subject to the limitations in this Section 4.6. In such case, all costs associated with the increased portion of the Work shall be priced in accordance with this Section. The Contractor shall keep and present, in such form as the Joint Board may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs allowable for the purposes of this Section shall be limited as indicated in Section 4.6.7.

4.5.7.4.6.7. Force Account Work

When the Contractor and the Joint Board cannot agree to a lump sum price for extra work, the Joint Board may direct the Contractor to perform such work on a force account basis. The Joint Board will pay for extra work on a force account basis as outlined hereinafter, only when all items of work are agreed to in writing before the Contractor begins the work.

4.5.7.1.4.6.7.1. Labor

For all labor and for foremen in direct charge of the specific operations, the Joint Board will pay the Contractor:

- 1. The actual cost of wages paid, but at rates not to exceed those for comparable labor currently employed on the Project, as the Joint Board Representatives determine;
- 2. An amount equal to the sum of the products of established labor burden percentages and the actual cost of wages. The amounts determined by the established

labor burden percentages constitute full compensation for the cost of workers' compensation insurance, social security taxes, unemployment compensation insurance, public liability insurance; and any other taxes or insurance which are added to labor costs; and

3. An amount equal to 25 percent of the actual cost of wages and the other costs identified above. This amount is full compensation for office overhead and general superintendence.

4.5.7.2.4.6.7.2. Professional Services

For any change-order related direct costs involving professional services performed by Contractor's own employees, the Joint Board will pay the lesser of (1) the actual costs to Contractor of the additional work performed by those professionals required by the change order, plus _____%, or (2) the costs of such additional work using the hourly professional services rates utilized in the Contractor's Proposal, without additional markup.

4.5.7.3.4.6.7.3. Materials

For all materials that the Contractor incorporates into the work and the Joint Board Representatives accept, the Joint Board will pay the actual cost of such material, including transportation charges and sales taxes, to which the Joint Board will add a sum equal to 15 percent.

4.5.7.4.4.6.7.4. **Equipment and Tools**

For any machinery or special equipment that the Joint Board Representatives have authorized for use and the Contractor has used, the Joint Board will pay the rental rate stated on the rental company invoice for the actual agreed time and rate that such equipment is required on the work and will add an amount equal to 15 percent of the rental sum as full compensation for fuel, lubricants, and filters.

The Joint Board will pay for equipment that the Contractor is already using on the Project, and which is not obtained specifically for the force account work based on an hourly rate. The Joint Board will determine the hourly rate by taking the Blue Book monthly rental rate, adjusted for age and geographic region, dividing it by 176 and adding the Blue Book estimated operational cost. The Joint Board will pay rental rates for equipment required to be on standby at one half the normal rate, excluding operational cost, and pay for standby time for a maximum of eight hours per day and 40 hours per week.

The Joint Board Representatives will measure the rental of equipment by time in hours of actual working time and the necessary traveling time of the equipment within the limits of the Project, unless the Joint Board Representatives have ordered special equipment in connection with force account work, in which case the Joint Board Representatives will also include travel time and transportation to the Project.

Joint Board will not pay rental rates or percentages for the use of small tools and manual equipment.

4.5.7.5.4.6.7.5. Subcontracted Work

To the extent that any Change Order is intended to compensate Contractor for the cost of work performed by Subcontractors, the Change Order shall provide for compensation equal to (1) the actual cost to Contractor of such work (which shall be charged by the Subcontractor on a force account basis in accordance with this Section 4.6, unless otherwise approved in writing by the Joint board), plus (2) 5% of such cost. The 5% markup for subcontracted work shall not apply to (i) Subcontracts with Affiliates; or (ii) Subcontracts with Suppliers.

4.5.7.6.4.6.7.6. Other Direct Costs

For any justified change-related direct cost not covered by the categories of costs contained in Sections 14.6.7, Contractor shall accept as full payment therefor an amount equal to the actual cost to Contractor for such direct cost item. Without additional mark-up, back-up documentation supporting each cost item for this category shall be provided by Contractor in writing to the Joint Board prior to any payment authorization being granted.

4.5.7.7.4.6.7.7. Bonds.

The Joint Board will pay an amount equal to the actual cost of any increase in the cost of the bond to the Contractor..

4.5.7.8.4.6.7.8. Records and Statements.

The Contractor shall compare all records of force account work with the Joint Board Representatives at the end of each day. The Contractor shall prepare all force account records on suitable forms that the Joint Board Representatives will provide for this purpose. The contents of records of force account work shall comply with Section 4.9.

When both additions and credits covering related work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

Pending final determination of the total cost of a Construction Change Directive to the Joint Board, amounts not in Dispute for such changes in the Work shall be included in applications for payment accompanied by a Change Order indicating the parties' agreement with part or all of such costs. For any portion of such cost that remains in dispute, the Joint Board shall make an interim determination for purposes of monthly payment for those costs. That determination of cost shall adjust the Contract Price on the same basis as a Change Order, subject to the right of the Contractor to disagree and assert a claim in accordance with Section 4.5.

When the Joint Board and the Contractor reach agreement concerning the adjustments in the Contract Price and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and shall be recorded by preparation and execution of an appropriate Change Order.

4.5.8.4.6.8. Limitation on Price Increases

Any increase in the Contract Price allowed hereunder shall exclude: (a) costs caused by the breach of contract or fault or negligence, or act or failure to act by any Contractor-Related Entity; (b) costs to the extent that they are unnecessary or could reasonably be avoided by Contractor, including by resequencing, reallocating or redeploying its forces to other portions of the work or to other activities unrelated to the work; and (c) costs for remediation of any Nonconforming Work. Costs incurred for the purpose of mitigating damages as described in clause (b) above, and not otherwise disallowed hereunder, would be reimbursable.

4.5.9.4.6.9. Limitation on Delay and Disruption Damages 4.5.9.1.4.6.9.1. Acceleration Costs; Delay and Disruption Damages

Acceleration Costs shall be compensable hereunder only with respect to Change Orders issued by the Joint Board as an alternative to allowing an extension of a Completion Deadline as contemplated by Section 4.4.1 and 4.5.2.3. Other delay and disruption damages shall be compensable hereunder only in the case of delays which qualify as the Joint Board-Caused Delays entitling Contractor to an extension of a Completion Deadline. Without limiting the generality of the foregoing, costs of rearranging Contractor's work plan to accommodate the Joint Board-Directed Changes not associated with an extension of a Completion Deadline shall not be compensable hereunder.

4.5.9.2.4.6.9.2. Other Limitations

Delay and disruption damages shall be limited to direct costs directly attributable to the delays described in <u>Section 4.6.9.1</u> and markups thereon in accordance with <u>Section 4.6.7</u> and any additional field office and jobsite overhead costs incurred by Agreement or directly attributable to such delays. In addition, before Contractor may obtain any increase in a Price to compensate for additional or extended overhead, Acceleration Costs or other damages relating to delay, Contractor shall have demonstrated to the Joint Board's satisfaction that:

- (a) its schedule which defines the affected Critical Path in fact set forth a reasonable method for completion of the Work; and
- (b) the change in the Work or other event or situation which is the subject of the requested Change Order has caused or will result in an identifiable and measurable disruption of the Work which impacted the Critical Path activity (i.e. consumed all available Float and extended the time required to achieve Substantial Completion Acceptance, Punch List Acceptance or Formal Acceptance beyond the applicable Completion Deadline); and

- (c) the delay or damage was not due to any breach of contract or fault or negligence, or act or failure to act of any Contractor-Related Entity, and could not reasonably have been avoided by Contractor, including by resequencing, reallocating or redeploying its forces to other portions of the Work or other activities unrelated to the Work (provided that the Joint Board has agreed to reimburse Contractor for additional costs reasonably incurred in connection with such reallocation or redeployment); and
- (d) the delay for which compensation is sought is not concurrent with any delay for which Contractor is responsible hereunder; and
- (e) Contractor has suffered or will suffer actual costs due to such delay, each of which costs shall be documented in a manner satisfactory to the Joint Board.

4.5.10.4.6.10. Limitation on Time Extensions

Any extension of a Completion Deadline allowed hereunder shall exclude any delay to the extent that it (a) did not impact a Critical Path, (b) was due to the fault or negligence, or act or failure to act of any Contractor-Related Entity, (c) is concurrent with any other unrelated delay to a Critical Path that is Contractor's responsibility hereunder, or (d) could reasonably have been avoided by Contractor, including by resequencing, reallocating or redeploying its forces to other portions of the work (provided that if the request for extension involves a Joint Board-Caused Delay, Contractor shall be entitled to a time extension unless the Joint Board shall have agreed, if requested to do so, to reimburse Contractor for its costs incurred, if any, in resequencing, reallocating or redeploying its forces). Contractor shall be required to demonstrate to the Joint Board's satisfaction that the change in the Work or other event or situation which is the subject of the Request for Change Order seeking a change in a Completion Deadline has caused or will result in an identifiable and measurable disruption of the Work which has impacted the Critical Path activity (i.e. consumed all available Float and extended the time required to achieve Substantial Completion Acceptance, Punch List Acceptance or Formal Acceptance beyond the applicable Completion Deadline).

4.6.4.7. Necessary Basic Configuration Changes

4.6.1.4.7.1. Notwithstanding the fact that this Contract generally obligates Contractor to undertake all Work necessary to complete the Project without an increase in the Contract Price, this Section 4.7 provides for an increase in the Contract Price to be made in conjunction with Necessary Basic Configuration Changes, and such changes shall be considered a Joint Board-Directed Change for which the Joint Board has not submitted a Change Order or a Request for Change Proposal. If Contractor commenced any construction work affected by the change prior to delivery of an appropriate PCO Notice, the Change Order shall allow the Joint Board a credit for the cost of any unnecessary work performed and/or shall exclude any additional costs associated with redoing the work already performed.

4.6.2.4.7.2. Contractor shall be responsible for any cost increases and/or delays which affect the duration of a Critical Path resulting from changes in requirements and obligations of Contractor due to Errors in the Owner Design Documents other than

those which require a Necessary Basic Configuration Change. In such event, no change in the Work shall be deemed to have occurred and no Change Order shall be issued for any such cost increases and/or delays.

4.7.4.8. Change Orders for Differing Site Conditions, Force Majeure Events and Hazardous Materials

4.7.1. Differing Site Conditions

Subject to the restrictions and limitations set forth in this <u>Section 4</u>, Contractor shall be entitled to a Change Order for certain additional costs which are directly attributable to any Differing Site Conditions to the extent permitted in this <u>Section 4.8.1</u>. No time extension shall be available with respect to Differing Site Conditions, and no delay damages shall be recovered. To the extent that additional costs are incurred due to changes in Contractor's obligations relating to the Work resulting from the existence of Differing Site Conditions and which are not reimbursed by insurance proceeds, the Joint Board shall be fully responsible for any additional costs directly attributable to changes in Contractor's obligations hereunder resulting from the existence of Differing Site Conditions, and a Change Order shall be issued to compensate Contractor for such additional costs.

During progress of the work, if Differing Site Conditions are encountered, Contractor shall immediately notify the Joint Board thereof telephonically or in person, to be followed immediately by written notification. Contractor shall be responsible for determining the appropriate action to be undertaken, subject to concurrence by the Joint Board. In the event that any Governmental Approvals specify a procedure to be followed, Contractor shall follow the procedure set forth in the Governmental Approvals. If the discovery of Differing Site Conditions necessitates a change in the design of the Project, such change shall be submitted to the Joint Board for concurrence in accordance with Scope of Work contained in RFP Section ______.

Contractor hereby acknowledges and agrees that it has assumed all risks with respect to the need to work around locations impacted by Differing Site Conditions. Contractor shall bear the burden of proving that a Differing Site Condition exists and that it could not reasonably have worked around the Differing Site Condition so as to avoid additional cost.

Each request for a Change Order relating to a Differing Site Condition shall be accompanied by a statement signed by a qualified professional setting forth all relevant assumptions made by Contractor with respect to the condition of the Project Site, justifying the basis for such assumptions, explaining exactly how the existing conditions differ from those assumptions, and stating the efforts undertaken by Contractor to find alternative design or construction solutions to eliminate or minimize the problem and the associated costs. No time extension or costs will be allowed in connection with any work stoppage in affected areas during the investigation period described above.

4.7.2.4.8.1. Force Majeure Events

Subject to the limitations contained in, and upon Contractor's fulfillment of all applicable requirements of, this <u>Section 4</u>, the Contractor shall be entitled to relief for its failure to perform its contract obligations if such failure is due to certain circumstances or events that are beyond Contractor's control. These include but are not limited to Acts of God (which includes floods as provided below, fires, earthquakes, tornados, or other natural disasters), wars, hostilities (regardless of whether war is declared), terrorist activities, strikes that materially impact the work, or interruption or failures of critical utilities that are not wholly or partially attributable to the Contractor or its consultants, Subcontractors, or lower-tier subconsultants or Subcontractors and Suppliers.

If the Contractor asserts a Force Majeure Event as an excuse for failure to perform its obligations, then the Contractor must prove that it took reasonable steps to minimize the delay or damages caused by the foreseeable events flowing from the Force Majeure Event, substantially fulfilled all non-excused obligations, and that the Joint Board was timely notified of the likelihood or actual occurrence of the Force Majeure Event.

In the event of a Force Majeure Event, the Contractor shall be entitled only to an extension of the Contract Time associated with the Force Majeure Event and to additional costs incurred arising directly from Force Majeure Events (excluding Acceleration Costs or delay and disruption damages other than for any Force Majeure Events which are included in the definition of Joint Board-Caused Delay), and not to any adjustment of the Contract Price. In order for the Contractor to be granted an extension of the Contract Time or a change in the Contract Price for additional costs, the Contractor must follow the Change Order procedures in Section 4, and prove that the Force Majeure Event delays activities are on the Critical Path and that it is not possible to work around the problem.

The Contractor shall not be entitled to an adjustment in Contract Price or Contract Time for delays within the control of the Contractor. Delays attributable to and within the control of a Subcontractor or Supplier shall be deemed to be delays within the control of the Contractor.

The Joint Board will not allow any extension of time for weather or resulting conditions, except for delays caused by those weather events described in this Section 4.8.21.

If the Contractor asserts that a flooding event is an excuse for failure to perform its obligations, the Contractor shall employ all measures to minimize or eliminate Project delays due to high water. Work which is most susceptible to high water shall be scheduled during time frames when low water levels in the river can be reasonably expected.

The Contractor shall notify the Joint Board immediately when the mean water elevation rises above 425 Feet (ORD) as measured at the upper gage of the McAlpine Lock and Dam and these conditions materially affect the prosecution of the Work. The Joint Board Representatives will promptly review the Project conditions and Contractor Work underway to determine whether the day should be designated as a "Flood Day." The Joint Board will only consider an extension of time if the Contractor clearly

demonstrates that the number of Flood Days exceeds 60 calendar days for the life of the Project.

4.7.3.4.8.2. Hazardous Materials Management

If compensation is payable to Contractor pursuant to <u>Section 178.3</u> with respect to Hazardous Materials Management, the amount of the Change Order shall either be a negotiated amount acceptable to the Parties, or 100% of the Reimbursable Hazardous Materials Costs for the Work in question, subject to the limitations set forth in this <u>Section 4.8.3</u>. Contractor shall not be entitled to a Change Order for additional compensation or extension of time with respect to the Hazardous Materials Management responsibilities set forth in <u>Sections 178.3.3</u>, <u>178.3.4</u> and <u>178.3.5</u>.

Entitlement to compensation or a time extension shall be limited to work performed pursuant to Contractor's Hazardous Materials Management Plan, and plans and reports regarding Site investigation for such Hazardous Materials as approved by the Joint Board, in writing. These plans shall not conflict or interfere with the similar plans submitted by the Design-Builder or the Developer. No compensation or time extension shall be allowed with respect to (a) immaterial quantities of Hazardous Materials, (b) any Hazardous Materials that could have been avoided by reasonable design modifications or construction techniques (which design modifications or construction techniques would have been consistent with applicable Law), (c) any costs that could have been avoided, or (d) any Hazardous Materials encountered during or in connection with the demolition of buildings, fixtures or other improvements on any parcels within the Project Site.

4.7.3.1.4.8.2.1. Determination of Reimbursable Amount

Contractor shall be deemed to have waived the right to collect any and all costs incurred in connection with any Hazardous Materials Management and any right to obtain an extension of a Completion Deadline if the Joint Board is not provided written notice of the discovery of Hazardous Materials and afforded the opportunity to inspect sites containing Hazardous Materials before any action is taken which would inhibit the Joint Board's ability to ascertain, based on a site inspection, the nature and extent of the materials. In the event of an emergency involving Hazardous Materials, Contractor may take such limited actions as are required by Law without advance notice to the Joint Board, but shall provide such notice immediately thereafter (which in no event shall be more than 2 hours after the incident by phone and 24 hours after the incident by written notice).

In cases involving reimbursement for Hazardous Materials Management under this <u>Section 4.8.3</u>, allowable costs shall be limited to the incremental costs incurred after completion of the testing process to determine whether Hazardous Materials are present, associated with the Hazardous Materials at issue (deducting any avoided costs such as the cost of disposal that would have been incurred had Hazardous Materials not been present). Investigating and characterizing are included in the Contract Price and Contractor shall not be entitled to additional compensation therefor. Contractor shall take all reasonable steps to minimize any such costs. Compensation shall be

allowed only to the extent that Contractor demonstrates to the Joint Board's reasonable satisfaction that (a) the Hazardous Materials Management could not have been avoided by reasonable design modifications or construction techniques and (b) Contractor's plan for the Hazardous Materials Management represents the approach which is most beneficial to the Project and the public. Contractor shall provide the Joint Board with such information, analyses and certificates as may be requested by the Joint Board in order to enable a determination regarding eligibility for payment.

4.7.3.2.4.8.2.2. Time Extensions

Contractor shall not be entitled to an extension of any Completion Deadline with regard to any need to investigate or characterize any Hazardous Materials, regardless of the total quantities. If Contractor encounters Hazardous Materials within the ROW Limits which, due to no fault of Contractor, results in delays to its Critical Path, then the Joint Board shall bear the risk of such delay (excluding those conditions for which Contractor has agreed to be responsible as described in <u>Section 11.1.1(7)</u>).

4.7.3.3.4.8.2.3. Limitations on Change Orders

Contractor shall have no right to receive any compensation for any Hazardous Materials Management resulting from a situation described in Section 11.1.1(7).

4.7.3.4.4.8.2.4. Insurance Proceeds

If the cost of any Hazardous Materials Management is covered by the insurance described in Article 512, Contractor shall be entitled to reimbursement of its costs from proceeds of insurance and self-insurance, up to the limits of the applicable policy, less any deductibles which shall be Contractor's responsibility. To the extent that such proceeds are available, Contractor shall not be entitled to payment hereunder on any other basis for such Hazardous Materials Management.

4.8.4.9. Change Order Records

Contractor shall maintain its records in such a manner as to provide a clear distinction between the direct costs of Work for which it is entitled (or for which it believes it is entitled) to an increase in the Contract Price and the costs of other operations. Contractor shall contemporaneously collect, record in writing, segregate and preserve (a) separate daily occurrence logs as required under Scope of Work in RFP Section ______, together with all other data necessary to determine the costs of all Work which is the subject of a Change Order or a requested Change Order, specifically including costs associated with design Work, and (b) all data necessary to show the actual impact (if any) of the change on each Critical Path with respect to all Work which is the subject of a Change Order or a proposed Change Order, if the impact on the Project Schedule is in dispute. Such data shall be provided to any dispute resolvers, the Joint Board and its Authorized Representatives as directed by the Joint Board, on forms approved by the Joint Board. The cost of furnishing such reports is included in Contractor's predetermined overhead and profit markups.

4.8.1.4.9.1. Daily Work Reports and Data Collection

Contractor shall furnish the Joint Board completed daily work reports for each day's Work which is to be paid for on a Force Account basis. The daily Force Account Work reports shall be detailed as follows:

- (a) Name, classification, date, daily hours, total hours, rate, and extension for each worker (including both construction and non-construction personnel) for whom reimbursement is requested.
- (b) Designation, dates, daily hours, total hours, rental rate, and extension for each unit of machinery and equipment.
 - (c) Quantities of materials, prices, and extensions.
- (d) Transportation of materials. The reports shall also state the total costs to date for the Force Account Work.

4.8.2.4.9.2. Supplier's Invoices

Materials charges shall be substantiated by valid copies of Supplier's invoices. Such invoices shall be submitted with the daily Force Account Work reports, or if not available, they shall be submitted with subsequent daily Force Account Work reports. Should said Supplier's invoices not be submitted within 60 days after the date of delivery of the materials, the Joint Board shall have the right to establish the cost of such materials at the lowest current wholesale prices at which such materials are available, in the quantities concerned, delivered to the location of Work, less any discounts available.

4.8.3.4.9.3. Execution of Reports

All Force Account Change Order reports shall be signed by Contractor's Project Manager.

4.8.4.4.9.4. Adjustment

The Joint Board will compare its records with the completed daily Force Account Work reports furnished by Contractor and make any necessary adjustments. When these daily Force Account Work reports are agreed upon and signed by both parties, said reports shall become the basis of payment for the work performed, but shall not preclude subsequent adjustment based on a later audit. Contractor's cost records pertaining to work paid for on a Force Account basis shall be open, during all regular business hours, to inspection or audit by representatives of the Joint Board during the life of this Contract and for a period of not less than five years after the Formal Acceptance Date, and Contractor shall retain such records for that period. Where payment for materials or labor is based on the cost thereof to any Person other than Contractor, Contractor shall make every reasonable effort to insure that the cost records of each such other Person will be open to inspection and audit by representatives of the Joint Board on the same terms and conditions as the cost records of Contractor. Payment for such costs may be deleted if the records of such third parties are not made

available to the Joint Board's representatives. If an audit is to be commenced more than 60 days after the System Formal Acceptance Date, Contractor will be given a reasonable notice of the time when such audit is to begin.

4.9.4.10. Matters Not Eligible for Change Orders and Waiver

Contractor acknowledges and agrees that no increase in a Price or extension of a Completion Deadline is available except in circumstances expressly provided for herein, that such price increase and time extension shall be available only as provided in this Section 4 and that Contractor shall bear full responsibility for the consequences of all other events and circumstances. Matters which are Contractor's exclusive responsibility include the following:

- (a) Errors in the Design Documents and Construction Documents (including Errors therein traceable to Errors in the Owner Design Documents, subject only to the right to a Change Order to the extent permitted by <u>Sections 4.7 or 4.8</u>);
- (b) any design changes requested by the Joint Board as part of the process of approving the Design Documents for consistency with the requirements of the Contract Documents, the Governmental Approvals and/or applicable Laws;
- (c) defective or incorrect schedules of Work or changes in the planned sequence of performance of the Work (unless arising from causes which otherwise give rise to a right to a Change Order);
- (d) action or inaction of Contractor's employees, any member of the Contractor-Related Entity or Subcontractors (unless arising from causes which otherwise give rise to a right to a Change Order);
- (e) action or inaction of adjoining property owners or the Joint Board's other contractors (unless arising from causes which otherwise give rise to a right to a Change Order);
 - (f) groundwater levels or subsurface moisture content;
- (g) untimely delivery of equipment or material, or unavailability or defectiveness or increases in costs of material, equipment or products specified by the Contract Documents, except to the extent resulting from a Force Majeure Event;
- (h) any costs covered by insurance proceeds received by (or on behalf of) Contractor;
- (i) correction of Nonconforming Work and review and acceptance thereof by the Joint Board (including rejected design submittals);
- (j) failure by any Contractor-Related Entity to comply with the requirements of the Contract Documents, Governmental Approvals or Laws;

- (k) delays not on a Critical Path;
- (I) any suspensions, terminations, interruptions, denials, non-renewals of, or delays in issuance of a Governmental Approval that is required to be obtained by Contractor, any failure to obtain such Governmental Approval, and compliance with the terms and conditions of all Governmental Approvals-;
- (m) any situations (other than Force Majeure Events) which, while not within one of the categories delineated above, were or should have been anticipated because such situations are referred to elsewhere in this Contract or arise out of the nature of the Work:
- (n) fluctuations and changes in the price of labor, materials, equipment, Hardware, Software, machinery or any other item required to perform and complete the Work; and
- (o) all other events beyond the control of the Joint Board for which the Joint Board has not expressly agreed to assume liability hereunder.

Contractor hereby assumes responsibility for all such matters, and acknowledges and agrees that assumption by Contractor of responsibility for such risks, and the consequences, costs, and delays resulting therefrom, is reasonable under the circumstances of this Contract and that contingencies included in the Contract Price Proposal in Contractor's sole judgment, constitute sufficient consideration for its acceptance and assumption of said risks and responsibilities.

CONTRACTOR HEREBY EXPRESSLY WAIVES ALL RIGHTS TO ASSERT ANY AND ALL CLAIMS BASED ON ANY CHANGE IN THE WORK, DELAY, SUSPENSION OR ACCELERATION (INCLUDING ANY CONSTRUCTIVE CHANGE, DELAY, SUSPENSION OR ACCELERATION) FOR WHICH CONTRACTOR FAILED TO PROVIDE PROPER AND TIMELY NOTICE OR FAILED TO PROVIDE A TIMELY REQUEST FOR CHANGE ORDER, AND AGREES THAT IT SHALL BE ENTITLED TO NO COMPENSATION OR DAMAGES WHATSOEVER IN CONNECTION WITH THE WORK EXCEPT TO THE EXTENT THAT THE CONTRACT DOCUMENTS EXPRESSLY SPECIFY THAT CONTRACTOR IS ENTITLED TO A CHANGE ORDER OR OTHER COMPENSATION OR DAMAGES.

CONTRACTOR FURTHER AGREES THAT THE JOINT BOARD, THE JOINT BOARD MEMBERS, JOINT BOARD CONTRACTORS, AND CONSULTANTS EMPLOYED BY THE JOINT BOARD OR THE JOINT BOARD MEMBERS SHALL NOT BE LIABLE TO THE CONTRACTOR FOR ANY CLAIMS, COSTS, LOSSES, OR DAMAGES (INCLUDING BUT NOT LIMITED TO ALL FEES AND CHARGES OF ENGINEERS, ARCHITECTS, ATTORNEYS, AND OTHER PROFESSIONALS AND ALL COURT OR ARBITRATION OR OTHER DISPUTE RESOLUTION COSTS) SUSTAINED BY THE CONTRACTOR ON OR IN CONNECTION WITH ANY OTHER PROJECT OR ANTICIPATED PROJECT.

4.10.4.11. Changes Not Requiring Change Order

Changes in the Work which have no net cost effect on the Contract Price may be approved in writing by the Joint Board as a Deviation, and in such event shall not require a Change Order. The Joint Board shall have authority to order minor changes in the Work not involving adjustment in the Contract Price or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes shall be effected by written order and shall be binding on the Contractor. The Contractor shall carry out such written orders promptly. Any other change in the requirements of the Contract Documents shall require either a Construction Change Directive or a Change Order.

4.11.4.12. No Release or Waiver

No extension of time granted hereunder shall release Contractor's Surety from its obligations. Work shall continue and be carried on in accordance with all the provisions of this Contract and this Contract shall be and shall remain in full force and effect during the continuance and until Formal Acceptance unless formally suspended or terminated by the Joint Board in accordance with the terms hereof. Permitting Contractor to finish the Work or any part thereof after a Completion Deadline, or the making of payments to Contractor after such date, shall not constitute a waiver on the part of the Joint Board of any rights under this Contract.

Neither the grant of an extension of time beyond the date fixed for the completion of any part of the Work, nor the performance and acceptance of any part of the Work or materials specified by this Contract after a Completion Deadline, shall be deemed to be a waiver by the Joint Board of its right to terminate this Contract for abandonment or failure to complete within the time specified (as it may have been extended) or to impose and deduct damages as may be provided.

No course of conduct or dealings between the parties or express or implied acceptance of alterations or additions to the Work, and no claim that the Joint Board has been unjustly enriched shall be the basis for any claim, request for additional compensation or extension of a Completion Deadline. Further, Contractor shall undertake, at its risk, work included in any request, order or other authorization issued by a person in excess of that person's authority as provided herein, or included in any oral request. Contractor shall be deemed to have performed such work as a volunteer and at its sole cost. In addition, the Joint Board may require Contractor to remove or otherwise undo any such work, at Contractor's sole cost.

5.0 Value Engineering Change Proposals

5.1. Description

A Value Engineering Change Proposal (VECP) is a cost reduction plan that is initiated, developed, and submitted by the Contractor to the Joint Board for modifying the Contract requirements after award.

A VECP requires changes in the Contract that result in savings to the Joint Board without impairing essential functions and characteristics of the facility while maintaining

and meeting all design and Joint Board policies, environmental requirements, and project commitments. Essential functions and characteristics include but are not limited to service life, reliability, economy of operation, ease of maintenance, standardized features, safety, satisfaction of customer needs, and special design requirements. Contractor, when developing a VECP, must address environmental requirements and similar concerns as part of the VECP.

The "direct cost savings" is the difference of the "construction savings" generated by implementing the VECP minus reasonable "design costs" associated with the VECP. The "construction savings" is the difference between what it would cost to complete all the Contract work without implementing the VECP and the cost to complete all the Contract work if the VECP is implemented. This includes any changes to quantities or unit prices across the entire Contract if affected by the VECP. If the estimated cost to complete all the Work without implementing the VECP differs from the estimated cost included in the Proposal Price, supporting documentation to explain the variance shall be provided. Reimbursable "design costs" are specific to engineering changes (examples: design changes, plan sheet revisions, and quantity estimating). Expenditures toward proposal preparation (examples: scheduling, documentation, cost analysis, material research, etc.) are not reimbursable.

Indirect cost savings (time, user delay, railroad force account costs, inspection costs, etc.), although considered when reviewing the merits of the VECP, are not reimbursed. A VECP may alter the Project Schedule and milestone dates, which, in turn, could affect time-related Contract provisions.

VECPs that reduce the time to complete the Project, and only result in indirect cost savings, may be accepted based on the mutual benefit derived.

A VECP proposing a total savings of less than \$_____ (written out \$) will normally not be considered unless there are additional non-monetary benefits to the Joint Board.

The provisions of this <u>Article 5</u> will not apply unless the Contractor identifies the submission as a VECP.

5.2. Conceptual Value Engineering Change Proposal (CVEC)

The Contractor may submit for review a Conceptual Value Engineering Change Proposal (CVECP) provided that it contains enough information to clearly define the work involved and the benefits to be realized. The CVECP should outline the general technical concepts associated with the VECP and the estimated direct cost savings which may result. The CVECP should contain sufficient information to provide concept evaluation and review. The conceptual VECP should include the following, at a minimum:

- 1. Conceptual plans;
- 2. An initial estimate of costs which should include sufficient information to determine the reasonableness of the CVECP;

- 3. The most recently approved Project Schedule showing the impact of the VECP on the Project Schedule. The Project Schedule shall include the time required to: develop a formal VECP, if required; approve the Change Order to incorporate the required changes into the Contract; order, fabricate, and deliver long lead material; and obtain or modify any Environmental Approvals or other required approvals. In addition, the Contractor must indicate the latest date that the CVECP and the VECP Change Order must be approved to not affect the currently approved Project Schedule. Should the Joint Board find that insufficient time is available for review and processing, it may reject the CVECP solely on such basis. If the Joint Board fails to respond to the CVECP by the date specified, the Contractor shall consider the CVECP rejected and will have no claims against the Joint Board as a result thereof.
- 4. A description of any previous use or testing of the CVECP on another project of the Joint Board or elsewhere and the conditions or results therewith. The Contractor shall submit the technical aspects of the CVECP in sufficient detail as to enable reviewers to determine the suitability of the VECP from an engineering perspective. If the technology is new, test information must be provided to the Joint Board's satisfaction.

Written notification by the Joint Board that the review has been completed and that the CVECP appears favorable merely indicates that the engineering and plan development may continue for submittal of the formal VECP and is not authorization for any construction work associated with the CVECP (including the purchase of any materials unique to the CVECP) to begin.

5.3. Value Engineering Change Proposal (VECP)

5.3.1. Documentation

VECPs will be processed in the same manner as prescribed for any other alteration of the Contract requiring a eChange eOrder or supplemental agreement. At a minimum, the following information shall be submitted by Contractor with each VECP:

- 1. A detailed description of the difference between the existing eContract requirements and the proposed changes and the comparative advantages and disadvantages of each including considerations of service life, economy of operation, ease of maintenance, traffic flow, safety, desired appearance, and increase or reduction of environmental impacts.
- 2. A complete set of plans and construction details when necessary, showing proposed revisions to the existing Contract requirements.
- 3. A detailed cost estimate for performing the work under the existing Contract and under the proposed change. The Contractor shall include items, units, and quantities. The Contractor shall include in the price all costs for labor, materials, supplies, equipment, tools, and all incidentals required for the complete incorporation of the option into the work. The Contractor shall present the direct cost savings to be achieved from the proposed changes.

- 4. An engineering analysis including Plans, computations, and other documents necessary for evaluation by the Joint Board;
- 5. A description of any previous use or testing of the CVECP on another project of the Joint Board or elsewhere and the conditions or results therewith. The Contractor shall submit the technical aspects of the CVECP in sufficient detail as to enable reviewers to determine the suitability of the VECP from an engineering perspective. If the technology is new, test information must be provided to the Joint Board's satisfaction.
- 6. A statement of the deadline for adopting the VECP in order to obtain the maximum cost reduction during the remainder of the Contract noting any effect on project milestones or other Completion Deadlines and any other schedule impacts.
- 7. A prediction of any effects the proposed changes would have on items such as maintenance of traffic, conformance with environmental commitments, and Joint Board costs other than construction, such as future maintenance, operating costs, and life cycle costs.
 - 8. A statement of the effect the VECP would have on the Contract Time.

In preparing VECPs, the Contractor must perform an independent examination of the affected WorkProject Site. The Joint Board shall rely exclusively upon the accuracy of the engineering data upon which the VECP is based. The Joint Board is not required to perform additional investigations, cross checks, or Site examinations. Adoption of a Contractor's VECP shall not be construed to alleviate or reduce the Contractor's full and absolute liability if the VECP upon implementation fails to satisfactorily perform because of the Contractor's use of inaccurate or incomplete engineering data or because of the Contractor's failure to adequately investigate and examine the affected construction Project Site.

5.3.2. Submission and Review

VECPs submitted by the Contractor will be processed as expeditiously as possible; however, the Joint Board will not be liable for any delay in acting upon VECPs submitted. The Contractor may withdraw, wholly or in part, any VECP not accepted by the Joint Board within the time specified in <u>Section 5.2(3)</u>.

The Joint Board will determine the acceptability of a VECP and the estimated net cost savings to be realized from adoption of all or any part of the VECP. Within 45 calendar days after receipt of the required minimum information, as specified in <u>Section 5.3.1</u>, the Joint Board will review and provide notification of whether the VECP is accepted or rejected. The Joint Board reserves the right to reject any VECP. The approval of the CVECP in no way obligates the Joint Board to accept the formal VECP. If a VECP is rejected, the Contractor will be required to complete the Contract in accordance with the Contract Documents

The Contractor acknowledges the following:

- 1. The Contractor has no claim to additional costs or delays, including development costs; loss of anticipated profits; or increased material or labor costs if the VECP is rejected.
- 2. The Joint Board has sole authority in determining the acceptance of any VECP.
- 3. The Joint Board reserves the right to reject all unacceptable work resulting from an approved VECP and can require that rejected work be removed and reconstructed under the original contract.
- 4. The VECP will be disqualified if requests for additional information are not immediately met.

5.3.3. Acceptance

The decision of The Joint Board to accept or reject a VECP shall be final and will not be subject to the provisions of <u>Article 13</u>, Dispute Resolution, or any other remedy. The Joint Board reserves the right not to consider any VECP. An informal meeting with the Joint Board may be requested by the Contractor. The Contractor acknowledges that no claim may be made arising from rejection of a VECP.

5.3.4. Notification

The Contractor will be notified in writing of the Joint Board's decision to accept or reject each VECP submitted under these provisions. If a VECP is accepted in whole or in part, such acceptance will be by letter from the Joint Board. If the approval is conditional, the letter will note those conditions. If a VECP is accepted, the Joint Board will execute a supplemental agreement that incorporates the necessary contract modifications. The Joint Board reserves the right to include in the supplemental agreement any conditions it deems appropriate for consideration, approval, and implementation of the VECP. The Contractor shall perform all work according to the existing terms of this Contract until the Joint Board executes a supplemental agreement.

The Joint Board's approval of a VECP voids any restrictions that the Contractor had imposed on the use or disclosure of the information that the Contractor included in the VECP, and the Joint Board then has the right to use, duplicate, and disclose, in whole or in part, any data necessary to implement any portion of the VECP on this project and all other Joint Board projects.

5.3.5. Contract Time

The Joint Board may adjust the Contract Time for any time savings realized by implementing a VECP in accordance with the statement provided by the Contractor pursuant to Section 5.3.1(8) or as otherwise agreed to by the Joint Board and the Contractor. The Joint Board may grant additional Contract Time when specified in a Supplemental Agreement which accepts a VECP.

5.3.6. Measurement

The Joint Board will measure the quantities for all revised Work specified in the Change Order or supplemental agreement in accordance with Section 109 of the Kentucky

Standard Specifications for Road and Bridge Construction, 2012 Edition. For purposes of this reference, where the Standard Specifications refer to the "Engineer," it shall mean the Joint Board Representatives.

Third party review costs incurred by the Joint Board, when substantiated by invoice, shall be subtracted from the computed net savings of the VECP.

5.3.7. Payments

Any changes to the Contract Price and Contract Time by virtue of the supplemental agreement accepting a VECP shall be immediately incorporated by the Contractor into its Project Schedule. The Joint Board will make payment for all completed and accepted revised work specified in the change order or supplemental agreement as provided in Article 6.

The Joint Board and the Contractor will share the savings from a VECP with half for the benefit of the Joint Board and half for the benefit of the Contractor. If the Joint Board accepts a VECP, the Contract Price shall be adjusted in accordance with the following:

For VECPs which reduce the Contractor's costs, the Contract Price shall be reduced by an amount equal to the sum of: (a) 100% of any additional costs incurred by the Joint Board, including the costs incurred in reviewing the VECP and any impact the VECP may have on Project revenue, but excluding the amounts due to the Contractor resulting from the VECP (excluding any impact on the Contract Price itself), plus (b) 50% of estimated net savings. For such VECPs, the term "estimated net savings" shall mean: (i) the difference between the cost of performing the work according to the Contract Documents and the actual cost to perform the work, as modified by the VECP, less (ii) the actual costs of studying and preparing the VECP as substantiated by Contractor and approved by the Joint Board in writing in accordance with the change procedures set forth herein, less (iii) the costs in clause (a) above. Design-Builder's profit shall not be considered part of the cost.

For VECPs that result in an increase in the Contractor's costs, the Contract Price shall be increased by an amount equal to the sum of: (a) 100% of any additional costs incurred by Contractor and approved by the Joint Board, plus (b) 50% of estimated net savings. For such VECPs, the term "estimated net savings" shall mean (i) the amount of any savings in the Joint Board's costs resulting from the VECP (taking into consideration the costs incurred in reviewing the VECP and any impact the VECP may have on project revenue), less (ii) the actual costs of studying and preparing the VECP as substantiated by Contractor and approved by the Joint Board in writing in accordance with the Order-on-Contract procedures set forth herein, less (iii) the costs in clause (a) above. Contractor's profit shall not be considered part of the cost.

Contractor is not entitled to share in either collateral or future contract savings. The term "collateral savings" means those measurable net reductions in the Joint Board's costs of operation resulting from the VECP, including costs of maintenance by the Joint Board or any third party, logistics, the Joint Board-furnished property and future costs associated with the Project. The term "future contract savings" shall mean reductions in

the cost of performance of future construction contracts for essentially the same item resulting from a VECP submitted by Contractor.

In the event that Contractor proceeds with a Contractor-requested Change Order that the Joint Board believes should be characterized as a VECP, and it is later determined through the Dispute resolution process that the change meets the technical qualifications for a VECP, the Contract Price shall be reduced by an amount equal to the sum of: (a) 100% of any additional costs incurred by the Joint Board resulting from the VECP plus (b) 75% of estimated net savings.

The Joint Board's Project Manager will be the sole judge of the estimated net savings in construction cost and costs incurred by the Joint Board resulting from the adoption of all or any portion of a VECP.

The Contractor's share will be considered full compensation to the Contractor for effecting all changes pursuant to the supplemental agreement stemming from the VECP.

In the event of the Joint Board's acceptance of a direct cost savings CVECP, and the Contractor is directed to proceed with the VECP implementation steps, and acceptance of the formal VECP is not reached, reimbursement of the implementation costs will be limited to 50%.

A cost reduction plan that is initiated and developed by the Joint Board shall not result in a sharing of savings with the Contractor. However, the Joint Board will reimburse the Contractor for actual additional design costs incurred by the Contractor relating to studying or pricing possible value engineering measures not originating with the Contractor and requested by the Joint Board in an amount agreed to by e \underline{C} hange e \underline{O} rder.

6.0 Payments

The following process shall apply to invoicing and payment:

6.1. Milestone Payments

Based upon applications for payment submitted to KYTC by the Contractor as provided in this <u>Article 6</u>, KYTC will forward the invoice to the Joint Board Representatives for review and approval. After approval, KYTC and IFA shall make payments in accordance with the Schedule of Values after the Contractor completes milestones. The Contractor shall accept payments from both KYTC and IFA.

6.1.1. Schedule of Values

Within 60 days of the award of the Contract, the Contractor shall submit a Schedule of Values establishing values for each Project milestone. This Schedule of Values will be reviewed and approved by the Joint Board. No payment applications by the Contractor shall be considered until the Contractor's proposed Schedule of Values has been approved in writing by the Joint Board.

Upon the Joint Board's approval of the Schedule of Values, the Contractor shall place these milestones in the Project's cost loaded schedule. The Contractor shall evenly disperse overhead and profit to each milestone over the duration of the Project. The total of all cost loaded milestones, shall total to 100 percent of the Contract Price excluding Warranty Price and Maintenance Price.

Each cost loaded milestone shall have a detailed breakdown of the quantities necessary for testing frequency purposes for each of the various kinds of work and the units of measure. Entries are informational only and are non-calculating. Quantities shall be entered for each activity under a column heading "Quantities." Units of measure shall be entered for each activity under a column heading "Units of Measure."

6.1.2. Applications for Payment

When the Contractor believes it has completed a project milestone as defined in RFP Section ____Price Proposal Table VI.1 as such milestones are defined in the general sScope of wWork of the RFP and the Schedule of Values, the Contractor shall prepare an Application for Payment. Contractor may submit an application for milestone payments not more frequently than monthly. To request a milestone payment, Contractor shall deliver to KYTC one hard original copy and one electronic copy of an Application for Payment in the form attached hereto as Exhibit _C and meeting all requirements specified herein except as otherwise approved in writing by Joint Board. Each Application for Payment shall be executed by a designated and Authorized Representative of Contractor appointed by Contractor to have such authority in accordance with this Contract. Contractor acknowledges that the Joint Board may obtain funding for portions of the Work from the federal government, local agencies and other third parties, and Contractor agrees to segregate Applications for Payment for all such Work in a format reasonably requested by the Joint Board and with detail and information as reasonably requested by the Joint Board. Each Application for Payment shall be organized to account for applicable reimbursement requirements, including an allocation for any reimbursements with respect to applicable insurance and/or bonds, and to facilitate the reimbursement process.

Concurrent with its submittal of an Application for Payment to the Joint Board, Contractor shall submit to KYTC at the address specified in the Scope of Work a complete copy of the Application for Payment, and an invoice of the total amount specified in the Application for Payment.

After receiving a complete Application for Payment, KYTC will forward the Application for Payment to the Joint Board Representatives, who will review the application and determine whether to approve the Application for Payment based upon the work being complete and of sufficient quality. If approved for payment both KYTC and IFA will <u>each</u> generate a check to the Contractor for 50% of the total amount of the Application for Payment approved.

6.1.2.1. Contents of Application for Payment

Each Application for Payment must contain the following items:

- (a) Application for Payment cover sheet;
- (b) Description of the status of all completed Milestones, as of the date of the Application for Payment;
- (c) Payments which are then due in accordance with the Milestone Payment Structure in Exhibit N, as of the date of the Application for Payment;
- (d) A monthly progress report(s) as of the date of the Application for Payment in the format described in Scope of Work Section ; in accordance with RFP Section 3.12.3.1.
- (e) Certification by the Contractor that all work which is the subject of the Application for Payment fully complies with the requirements of the Contract Documents subject to any exceptions identified in the certification;
- (f) A report of personnel hours since the prior Application for Payment (including a list setting forth all Key Personnel and the hours which they worked during this period);
- (g) Application for Payment data sheet(s) and supporting documents, as required by the Joint Board to support and substantiate the amount requested (based on quantities and unit prices for unit priced work), and based on labor, equipment and materials for force account Change Orders,
- (h) DBE utilization report in a format reasonably satisfactory to the Joint Board:
 - (i) An approved and updated Project Schedule;
- (i)(j) A list of all Software and related work product that has been completed during the relevant time period and Certification by the Contractor that such documents have been deposited in the Source Code Escrow as provided in Section 14.4.4; and
 - (i)(k) Such other items as the Joint Board reasonably requests.
- (k)(I) In addition, no Application for Payment shall be considered complete unless it: (1) describes in detail the status of completion as it relates to the Project; (2) sets forth in detail the related payments which are then due in accordance with the Project Schedule, as of the end of most recent prior Application for Payment, including detailed itemization for any payments with respect to applicable insurance and/or bonds; (3) in the case of amounts to be paid on a unit price basis, includes invoices, receipts or other evidence establishing the number of units delivered; (4) in the case of amounts invoiced on a Force Account basis, includes supporting documentation; and (5) sets forth in detail the amounts paid to Subcontractors (including Suppliers and sub-Subcontractors) from the payments made by the Joint Board to Contractor with respect to the most recent prior Application for Payment, including executed unconditional waivers of claims with respect to all amounts so paid.

6.1.2.2. Application for Payment Cover Sheet Contents

The Application for Payment cover sheet shall include the following:

- (a) Project number and title;
- (b) Request number (numbered consecutively starting with "1");
- (c) Total amount earned to date for the Project; and
- (d) Authorized signature, title of signer, and date of signature.

6.1.2.3. Certification by QC Personnel

Each Application for Payment shall include a certificate in the form included in <u>Exhib</u>it <u>—C</u>, certifying that:

- (a) All work, including that of designers, Subcontractors, including Suppliers and fabricators, which is the subject of the Application for Payment has been checked and/or inspected by the Contractor's Quality Manager (with respect to all work);
- (b) Except as specifically noted in the certification, all work which is the subject of the Application for Payment conforms to the requirements of the Contract Documents, the Governmental Approvals and applicable Laws;
- (c) The Civil Construction Quality Plan and the Quality Management PQuality Control Plan and Quality Management Plan lan[define] or delete reference if not applicable] and all of the measures and procedures provided therein arise functioning properly and are being followed in all respects;
- (d) The percentages identified for completion of construction and System integration percentages indicated are accurate and correct; and
- (e) The milestone to which the Application for Payment relates has been achieved.

6.1.2.4. Warranty Price Payments

The Warranty Price for the Operations Center shall be paid in arrears, in equal monthly installments commencing upon Substantial Completion Acceptance. The Warranty Price for each Crossing shall be paid in arrears, in equal monthly installments. Payment of the Warranty Price for the East End Crossing shall commence commencing—upon Substantial Completion Acceptance of each Crossing—the East End—Crossing—and ending on the first anniversary of one year from the date of Formal Acceptance of the Toll Collection Systemsuch date. Payment of the Warranty Price for the Downtown Crossings (Toll Zones DB-1R and R-1) shall commence upon Substantial Completion Acceptance of such Crossingsthe Downtown Crossing (Toll Zones DB-1R and R-1), payment of the Warranty Price for the Kennedy Crossing (Toll Zones KB-1R and R-2)

shall commence upon Substantial Completion Acceptance of the Kennedy Crossing (Toll Zones KB-1R and R-2), and payment of the Warranty Price with respect to both bridges comprising the Downtownfor such Crossings shall continue until the first anniversary of Substantial Completion Acceptance of the Downtown Crossing (Toll Zones DB-1, R-1, KB-1 and R-2). The Warranty Price for any components of the System not accepted as part of a Crossing Substantial Completion Acceptance shall commence upon the System Substantial Completion Acceptance Date and shall continue until the first anniversary of such date. In all cases, payment of the Warranty Price for the System shall be subject to any amounts withheld pursuant to the Contract Documents.

6.1.2.5. Maintenance Price Payments

The annual Maintenance Price shall be paid in arrears, in equal monthly installments, commencing for each Crossing (on a per lane/per month basis as indicated in the Price Proposal) upon expiration of the Warranty Period for such Crossing, and for the System as a whole, on the expiration of the Warranty Period for any components of the System not accepted as party of a Crossing Substantial Completion Aacceptance and continuing for the duration of the Maintenance Term, as it may be extended by the Joint Board's exercise of any Maintenance Term Option, subject to any amounts withheld pursuant to the Contract Documents.

6.1.2.6. Delivery of Maintenance Price Application for Payment

Within 30 days of the last day of each month, Contractor shall deliver to the Joint Board one copy of a Maintenance Price Application for Payment in the form attached hereto as Exhibit D— and meeting all requirements specified herein except as otherwise approved in writing by the Joint Board. Each Maintenance Price Application for Payment shall be executed by a designated and Authorized Representative of Contractor appointed by Contractor to have such authority in accordance with this Contract. Contractor acknowledges that the Joint Board may obtain funding for portions of the work from the federal government, local agencies and other third parties, and Contractor agrees to segregate Maintenance Price Application for Payments for all such work in a format reasonably requested by the Joint Board and with detail and information as reasonably requested by the Joint Board. Each Maintenance Price Application for Payment shall be organized to account for applicable reimbursement requirements and to facilitate the reimbursement process.

6.1.2.7. Contents of Maintenance Price Application for Payment

Each Maintenance Price Application for Payment must contain the following items:

- (a) Maintenance Price Application for Payment cover sheet;
- (b) Monthly Maintenance Reports as described in Section of the Scope of Workrequired pursuant to RFP Appendix B Section B.23;

- (c) Certifications in form acceptable to the Joint Board by the Project Manager and Maintenance Manager that the work conforms to the approved and updated (if applicable) Maintenance Plan;
- (d) Monthly report of personnel hours (including a list setting forth all Key Personnel and the hours which they worked during this period);
- (e) Maintenance Price Application for Payment data sheet(s), documents and summary that support and substantiate the Performance Requirements have been met; and
- (f) Submitted and approved System updates, As-Built Documents and changes to the maintenance RBOC maintenance manual if applicable; and
- (f)(g) A list of all Software and related work product that has been completed during the relevant time period and Certification by the Contractor that such documents have been deposited in the Source Code Escrow as provided in Section 14.4.4.

6.1.2.8. Report of Personnel Hours

With each Application for Payment and Maintenance Price Application for Payment, Contractor shall report the total monthly labor hours for design, construction and maintenance personnel used in connection with the Work which is the subject of the Application for Payment or Maintenance Price Application for Payment.

6.1.2.9. Application for Payment and Maintenance Price Application for Payment Data Sheets

Application for Payment and Maintenance Price Application for Payment data sheets shall be subdivided into Contractor designated Project elements and shall be attached to a Project wide report and Application for Payment/Maintenance Price Application for Payment data sheet. It is the Joint Board's intent to base payments for Work other than Maintenance Work on the achievement of milestones as provided in the Milestone Payment Structure (except as expressly set forth in this Contract), except that cost plus or unit price Change Order work or items to be paid from an allowance may be paid based upon measured quantities. Contractor shall present the format of the Application for Payment/Maintenance Price Application for Payment data sheets for the Joint Board approval at least 20 working days prior to the submittal of the first Application for Payment and Maintenance Price Application for Payment. Once the Application for Payment format has been approved by the Joint Board, the format shall not change without the Joint Board's prior written approval. Upon commencement of the Maintenance Term, the annual price for Maintenance Work will be paid in equivalent monthly installments; provided, however, that the Joint Board intends to pay the annual cost of the Warranty Bond, the Maintenance Payment Bond, the Maintenance Performance Bond and insurance premiums incurred for the relevant year of the Maintenance Term during such year of the Maintenance Term on an actual cost basis as incurred, without markups, profit or overhead, provided, however, that Contractor shall not be entitled to reimbursement for any increase in the actual cost of bond or insurance premiums attributable to claims or loss experience of any Contractor-Related Entity, whether under an insurance policy or bond required for the Project or in connection with any unrelated work or activity of Contractor-Related Entities. Contractor shall bear the burden of proving that increases in such costs are not the result of adverse changes in Contractor's financial condition, Contractor risk rating or experience or the acts, omissions, negligence or misconduct of Contractor or any member of the Contractor-Related Entity.

6.1.3. Timing of Applications and Payments

The Contractor shall submit an Application for Payment within 30 days from the date that it believes it has completed a Project milestone. The Contractor agrees that the conditions regarding the date on which an application is timely and properly submitted satisfy prompt payment requirements of the Kentucky Fairness in Construction Act and all such other laws as may be applicable the {insert reference to relevant Indiana law, if any]. Within fifteen working days after the Joint Board's receipt of a complete Application for Payment or Maintenance Price Application for Payment, the Joint Board Representatives will review the Application for Payment or Maintenance Price Application for Payment, as applicable, and all attachments and certificates thereto for conformity with the requirements of the Contract Documents, and shall notify Contractor and each of KYTC and IFA of the amount approved for payment and the reason for disapproval of any remaining invoiced amounts or of any other information set forth in the Application for Payment or Maintenance Price Application for Payment. Contractor may include such disapproved amounts in the next month's Application for Payment after correction of the deficiencies noted by the Joint Board and satisfaction of the requirements of the Contract Documents related thereto. Within 15 working days after the Joint Board's approval of an Application for Payment or Maintenance Price Application for Payment, as applicable, but in any event not later than 30 working days after the Joint Board's receipt of a complete Application for Payment or Maintenance Price Application for Payment, as applicable, KYTC and IFA each shall pay Contractor 50% of the amount approved for payment in respect of such request, less 50% of any amounts which the Joint Board is otherwise entitled to withhold or deduct. With respect to any Change Order Work to be performed on a Force Account basis, in no event shall Contractor be entitled to (a) payment for any activity in excess of the value of the activity times the completion percentage of such activity (for non-unit priced work), or (b) aggregate payments hereunder in excess of the overall completion percentage for the Project times the Contract Price (for non-unit-priced \wWork).

6.1.4. The Joint Board Reliance on Information Supplied by the Contractor

In taking action on the Contractor's applications for payment, the Joint Board (and each of KYTC and IFA) shall be entitled to rely on the accuracy and completeness of the information furnished by the Contractor and shall not be deemed to have made a detailed examination, audit, or arithmetic verification of the documentation submitted; to have made exhaustive or continuous on-site inspections; or to have made examinations to ascertain how or for what purposes the Contractor has used amounts previously paid on account of the <u>agreementContract</u>. Such examinations, audits and verifications, if required by the Joint Board, will be performed by the Joint Board's accountants or the

accountants of the Joint Board members States' Parties, acting in the sole interest of the Joint Board.

6.2. No Retainage Withheld

KYTC and IFA will not withhold retainage from the Contractor as a percentage of the milestones completed to date. The Contractor likewise shall not withhold retainage on any Subcontractors and shall provide contractual language sufficient to ensure that no lower-tier Subcontractors withhold retainage on contractors and Suppliers.

6.3. Deductions, Exclusions and Limitations on Payment; Payment for Certain Items

6.3.1. Deductions

The Joint Board may deduct from any payment due and payable by it to the Contractor under this Contract the following:

- (a) Any Joint Board or third party Losses for which Contractor is responsible hereunder or any Liquidated Damages, Stipulated Damages or Delay Reporting Damages which have accrued as of the date of the Application for Payment or which are anticipated to accrue based on the Substantial Completion, Punch List Acceptance and Formal Acceptance dates shown in the current Project Schedule or the applicable submission deadline, as the case may be;
- (b) If a notice to stop payment, claim or Lien is filed with the Joint Board due to the Contractor's failure to pay for labor or materials used in the work, money due for such labor or materials will be withheld from payment to the Contractor;
- (c) Any sums expended by or owing to the Joint Board as a result of Contractor's failure to maintain the as-built drawings,
- (d) Any sums expended by the Joint Board in performing any of Contractor's obligations under this Contract which Contractor has failed to perform, and
- (e) Any other sums which the Joint Board is entitled to recover from Contractor under the terms of this Contract, including, without limitation, pursuant to Section 11.1.

The failure by the Joint Board to deduct any such amounts from any payment due and payable by it to Contractor under this Contract shall not constitute a waiver of the Joint Board's right to such amounts.

All amounts owing by Contractor to the Joint Board under this Contract shall earn interest from the date on which such amount is owing at the lesser of (i) 12% per annum or (ii) the maximum rate allowable under applicable Law.

6.3.2. Unincorporated Materials

The Joint Board will not pay for materials not yet incorporated in the work other than precast concrete, reinforcing steel, structural steel, precast concrete members, stone, gravel, sand or other non-perishable materials that will be permanently incorporated in the work unless all of the following conditions are met:

Material shall be (1) delivered to the Project Site, (2) delivered to Contractor and promptly stored by Contractor in bonded storage at a location approved by the Joint Board in its sole discretion, or (3) stored at a Supplier's fabrication site, which must be a bonded commercial location approved by the Joint Board, in its sole discretion. Contractor shall submit certified bills for such materials with the invoice, as a condition to payment for such materials. The Joint Board shall allow only such portion of the amount represented by these bills as, in its sole opinion, is consistent with the reasonable cost of such materials. If such materials are stored at any site not approved by the Joint Board, Contractor shall accept responsibility for and pay all personal and property taxes that may be levied against the Joint Board by any state or subdivision thereof on account of such storage of such material.

All such materials so accepted shall be and become the property of the Joint Board or its designee(s). Contractor at its own cost shall promptly execute, acknowledge and deliver to the Joint Board proper bills of sale or other instruments in writing in a form acceptable to the Joint Board conveying and assuring to the Joint Board title to such material included in any invoice, free and clear of all Liens. Contractor, at its own cost, shall conspicuously mark such material as the property of the Joint Board, shall not permit such materials to become commingled with non-Joint Board-owned property and shall take such other steps, if any, as the Joint Board may require or regard as necessary to vest title to such material in the Joint Board free and clear of Liens.

The cost and charges for material included in an invoice but which is subsequently lost, damaged or unsatisfactory may be deducted from succeeding invoices if the Joint Board, in its sole discretion, determines that is appropriate after considering the availability of insurance coverage and Contractor's actions to replace the lost, damaged or unsatisfactory items.

Payment for material furnished and delivered as indicated in this <u>Section 6.3.2</u> will not exceed the amount paid by Contractor as evidenced by a bill of sale supported by paid invoice.

6.3.3. Payments for Bond and Insurance Premiums and As-Built Documents

With respect to the NTP, the portion of the Contract Price allocable to bond and insurance premiums (for the insurance policies required to be in place following the NTP through Formal Acceptance) shall be payable to reimburse Contractor for bond and insurance premiums actually paid, without markup, profit or overhead, not to exceed the line item for such premiums set forth in the Proposal. Any excess portion of the line item for such premiums set forth in the Proposal shall be payable following Substantial Completion Acceptance. Insurance premiums for the insurance policies required for the Warranty Period and the premiums paid for the four-year Maintenance Performance Bond and the Maintenance Payment Bond shall be payable to reimburse Contractor for the bond premiums actually paid, without markup, profit or overhead, not to exceed the line item for such premiums set forth in the Proposal. Any excess portion of the line item for such premiums set forth in the Proposal shall be payable following expiration of the first two years of the Maintenance Term.

Except as set forth in this <u>Section 6.3.3</u>, payment to reimburse Contractor for bond and insurance premiums shall be made for costs actually incurred by Contractor, without markup, profit or overhead; provided, however, that the Joint Board shall not be responsible for the payment of any increases or adjustments in bond or insurance premiums resulting from adverse changes in Contractor's financial condition, Contractor's risk rating or experience or the acts, omissions, negligence or misconduct of Contractor or any member of the Contractor-Related Entity.

The amount payable for as-built documents acceptable to the Joint Board shall equal 1% of the Contract Price. Contractor shall not be entitled to payment for the last 1% of the Contract Price until acceptable as-built documents have been delivered to the Joint Board.

6.3.4. Equipment

With the exception of work done on a Force Account basis, the Joint Board shall not pay for direct costs of equipment. Costs of equipment, whether new, used or rented, and to the extent not included in the mobilization payments under <u>Section 6.2</u>, shall be allocated to and paid for as part of the activities with which the equipment is associated, in a manner which is consistent with the requirements of Section 6.1.4.

6.4. Final Payment

6.4.1. Within 180 days after the Joint Board has completed final inspection and made System Formal Acceptance of the Work, the Contractor shall submit a final Application for Payment for the Project, showing the final cost of all Work performed, and all deductions from the final amount for <code>Liquidated dDamages</code> and any other deductions provided for in this Contract and any amounts owing from Change Orders. In addition to meeting all other requirements for Applications for Payment hereunder, the Final Application for Payment shall list all outstanding PCO Notices, stating the amount at issue associated with each such notice. A Final Application for Payment shall be accompanied by (a) evidence regarding the status of all existing or threatened claims, Liens and stop notices of Subcontractors and laborers against Contractor or against the Joint Board, (b) consent of any Guarantors and Surety to such Final Payment, (c) such

other documentation as the Joint Board may reasonably require; and (d) the release described in <u>Section 6.4.4</u>, executed by Contractor. Prior applications and payments shall be subject to correction in the Final Application for Payment. PCO Notices filed concurrently with a Final <u>Draw</u> Application for Payment must be otherwise timely and meet all requirements under <u>Sections 4 and 13</u>.

- **6.4.2.** If the Final Application for Payment shows no existing or threatened claims, Liens and stop notices of Subcontractors, Utility Owners, railroads or laborers against Contractor or against the Joint Board, and provided such Final Application for Payment has been approved, the Joint Board, in exchange for an executed release meeting the requirements of Section 6.4 and otherwise satisfactory in form and content to the Joint Board, will pay the entire sum found due on the approved Final Application for Payment, less the amount of any Losses that have accrued as of the date of the Final Payment of the Contract Price, the costs to complete or remediate uncompleted Work or Nonconforming Work and any other deductions permitted under Section 6.3.1 and 6.3.2 above.
- **6.4.3.** If a Final Application for Payment lists any existing or threatened claims, Liens and stop notices of Subcontractors, laborers, or Utility Owners or railroads against Contractor or against the Joint Board, or if any is thereafter filed, the Joint Board may withhold from the payment of the amounts set forth on the approved Final Application for Payment such amount as the Joint Board deems advisable to cover any amounts owing or which may become owing to the Joint Board by Contractor, including the amount of any Losses that have accrued as of the date of the Final Payment of the Contract Price and any other deductions permitted under <u>Section 6.3.1 and 6.3.2</u> above, the costs to complete or remediate uncompleted Work or Nonconforming Work, and the amount of any existing or threatened claims, Liens and stop notices of Subcontractors, laborers, Utility Owners and railroads against Contractor or against the Joint Board.
- **6.4.4.** The executed release from Contractor shall be from any and all claims arising from the Work, and shall release and waive any claims against the Indemnified Parties, excluding only those matters identified in any PCO Notices listed as outstanding in the Final Application for Payment. The release shall be accompanied by an affidavit from Contractor certifying that:
- (a) all Work has been performed in strict accordance with the requirements of the Contract Documents;
- (b) Contractor has resolved any claims made by Subcontractors, Utility Owners, laborers, railroads and others against Contractor, the Joint Board or the Project;
- (c) Contractor has no reason to believe that any Person has a valid claim against Contractor, the Joint Board or the Project which has not been communicated in writing by Contractor to the Joint Board as of the date of the certificate; and
- (d) all required guarantees, letters of credit, Warranties, insurance, and the Payment Bond and the Performance Bond are in full force and effect; and

(d)(e) complete, accurate and current copies of all documents that are required to be in the Software Source Code Escrow have been delivered to the Software Source Code Escrow Holder.

Each release and the affidavit shall survive a Final Payment.

- **6.4.5.** All prior Application for Payments shall be subject to correction in the Final Application for Payment.
- **6.4.6.** The Joint Board will review Contractor's proposed Final Application for Payment, and changes or corrections will be forwarded to Contractor for correction within 20 working days. The Joint Board shall pay any undisputed amounts, less any Losses that have accrued as of the date of the Final Payment of the Contract Price, the costs to complete or remediate uncompleted Work or Nonconforming Work and any other deductions permitted under <u>Section 6.3.1 and 6.3.2</u> above, within 30 Days after its approval of such amounts on the application for Final Payment of the Contract Price, but not earlier than the date of Formal Acceptance.

The Contractor's acceptance of the final ppayment constitutes a release to the Joint Board and a waiver of any and all claims against the same, except those previously made in writing and identified by the Contractor as unsettled at the time of Final Application for Payment.

6.5. Payment as Full Compensation

Contractor acknowledges and agrees that, subject only to Contractor's rights under Section 4, the Contract Price includes (a) all designs, equipment, materials, Hardware, Software, labor, insurance, bond premiums, home office, jobsite and other overhead, profit and services relating to Contractor's performance of its obligations under the Contract Documents (including all work, equipment, Hardware, Software, supplies, materials, labor and services provided by Subcontractors and intellectual property rights necessary to perform the Work); (b) performance of each and every portion of the Work; (c) the cost of obtaining all Governmental Approvals; (d) all costs of compliance with and maintenance of the Governmental Approvals and compliance with Laws to the extent applicable to the Project; (e) payment of any taxes, duties, permit and other fees and/or royalties imposed with respect to the work and any equipment, materials, labor or services included therein; and (f) compensation for all obligations, risks and contingencies assigned to Contractor under the Contract Documents. The Contractor shall receive and accept the compensation provided for in the Contract as full payment for furnishing all materials and for performing all work under the Contract, including changes in work, materials, or plans as provided herein, in a complete and acceptable manner; for all risk, loss, damage, or expense of whatever character arising out of the nature of the work or the prosecution thereof; and for all expenses incurred in consequence of the suspension or discontinuance of the Work as specified under the Contract. The Joint Board's payment of any application for payment does not relieve the Contractor of any obligation to make good any defective work or material.

6.6. Final Payment of Maintenance Price

Final Maintenance Payment will be made as follows:

On or about the date that the Maintenance Term terminates, Contractor shall prepare and submit a proposed Final Maintenance Price Application for Payment to KYTC for the Final Maintenance Payment, showing the proposed total amount due Contractor, including any amounts owing from Change Orders and the portion of the Maintenance Price identified as the price for "End of Term Maintenance Activities". In addition to meeting all other requirements for Maintenance Price Application for Payments hereunder, the Final Maintenance Payment Application for Payment shall list all outstanding PCO Notices, stating the amount at issue associated with each such notice. A Final Maintenance Price Application for Payment shall be accompanied by (a) evidence regarding the status of all existing or threatened claims, Liens and stop notices of Subcontractors and laborers against Contractor or against the Joint Board, (b) consent of any Guarantors and Surety to such Final Maintenance Payment, (c) a list of any System equipment that has unexpired warranty records, accompanied by the service records with respect to such equipment; (d) such other documentation as the Joint Board may reasonably require; and (e) the release described in Section 6.4 executed by Contractor. Prior applications and payments shall be subject to correction in the Final Maintenance Payment Application for Payment. PCO Notices filed concurrently with a Final Maintenance Payment Application for Payment must be otherwise timely and meet all requirements under Sections 4 and 13.

The provisions of <u>Sections 4.4.2 and 6.4</u>, above, shall apply with respect to the Final Maintenance Payment.

The Final Maintenance Payment shall also be subject to Contractor's certification in the release required by Section 6.4 that Contractor has satisfied all the conditions to the end of the Maintenance Term set forth in Section ——B11.8 of the Scope of Work. The Joint Board will review Contractor's proposed Final Maintenance Price Application for Payment, and changes or corrections will be forwarded to Contractor for correction within 20 working days. The Joint Board shall pay any undisputed amounts, less any Losses that have accrued as of the date of the Final Maintenance Payment, the costs to complete or remediate uncompleted Work or Nonconforming Work and any other deductions permitted under Section 6.4 above, within 30 Days after its approval of such amounts on the application for the Final Maintenance Payment, but not earlier than the end of the Maintenance Term.

6.7. Administration of Payment

After the Joint Board Representatives have approved an Application for Payment from the Contractor, they will forward their approval of the application for payment to a KYTC designee and an IFA designee and each State shall pay 50% of the amount approved as provided in Section 6.1.

If either state fails to pay the designated amount within 30 days from the date the application for payment is received the Contractor shall notify both states, in writing immediately.

Upon Contractor's written request of not less than 30 days and approval by the Joint Board Representatives, the States may forward the Contractor's payment checks directly to any person or entity designated by the Contractor or include any person or entity as a joint payee on the Contractor's payment checks, however, in no event shall such approval and action obligate the Joint Board to anyone other than the Contractor and the Contractor shall remain responsible or fulfillment of all Contract obligations.

6.8. Fund Availability

Financial obligations of the Joint Board are contingent upon the funds for that purpose being appropriated, budgeted, allocated, and otherwise made available by the State of Indiana and the Commonwealth of Kentucky. The Contract is funded in part with federal funds. The Contract is, therefore subject to and contingent upon the continuing availability of federal funds.

If funds to effect payment are not available, the Joint Board shall provide written notification to the Contractor. If the Contract is terminated due to an unavailability of funding it shall be deemed a termination for convenience pursuant to Section 910.2. Contractor agrees to take back any affected deliverables not yet delivered, terminate any services supplied to the Joint Board, and relieve the Joint Board of any further obligation. The Joint Board shall remit payment for deliverables and services accepted prior to the date of the aforesaid termination notice in conformance with ethe payment terms established above.

6.9. Payment to Subcontractors

No later than 10 Days after receipt of payment from the Joint Board, Contractor shall promptly pay each Subcontractor, out of the amount paid to Contractor on account of such Subcontractor's portion of the Work, the amount to which such Subcontractor is entitled, and any other offsets and deductions provided in the Subcontract or by law. No later than 10 Days after satisfactory completion of all Work to be performed by a Subcontractor, including provision of appropriate releases, certificates and other evidence of the Subcontractor's compliance with all applicable requirements of the Contract Documents, Contractor shall return all moneys withheld in retention from the Subcontractor. Such payment shall be made promptly following satisfaction of the foregoing requirements, even if Work to be performed by Contractor or other Subcontractors is not completed and has not been accepted. Each Subcontract shall require the Subcontractor to make payments to sub-Subcontractors and Suppliers in a similar manner.

For the purpose of this <u>Section 6.8</u>, satisfactory completion shall have been accomplished when:

(a) the Subcontractor has fulfilled the Subcontract requirements and the requirements under the Contract Documents for the subcontracted Work, including the submission of all submittals required by the Subcontract and Contract Documents; and

(b) the Work performed by the Subcontractor has been inspected and approved in accordance with the Contract Documents and the final quantities of the Subcontractor's work have been determined and agreed upon.

The inspection and approval of a Subcontractor's work does not eliminate or impair the Contractor's responsibility for the Work. Any delay or postponement of payments to Subcontractors from the above-referenced time frames may occur only for good cause following written approval by the Joint Board. Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to sub-Subcontractors in a similar manner. The Joint Board shall have no obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law. Interest on late payments to Subcontractors shall be Contractor's responsibility, and shall not be a part of the Contract Price.

6.10. Cost Principles

If applicable, in order to be reimbursed with federal funds, Contractor shall comply with the cost principles established in OMB Circular A-87 that specify that all reimbursed costs are allowable, reasonable, and allocable to the Project.

6.11. Disputes

Failure by the Joint Board to pay any amount in Dispute shall not alleviate, diminish or modify in any respect Contractor's obligation to perform under the Contract Documents, including Contractor's obligation to achieve Formal Acceptance and all Work in accordance with the Contract Documents, and Contractor shall not cease or slow down its performance under the Contract Documents on account of any such amount in dispute. Any Dispute regarding such payment shall be resolved pursuant to <u>Section 123</u>. Contractor shall proceed as directed by the Joint Board pending resolution of the dispute. Upon resolution of any such dispute, each party shall promptly pay to other any amount owing.

7.0 Prosecution, Control, and Inspection of the Work

7.1. Authority of Joint Board Personnel

7.1.1. Joint Board Representatives

The Joint Board will, within a reasonable time, decide all questions regarding the quality and acceptability of materials furnished, work performed, manner of performance and the rate of progress of the work; all interpretation of the Contract Documents; and the acceptable fulfillment of the work performed by the Contractor. The Joint Board shall designate two individuals ("Joint Board Representatives") that shall be the only two individuals with authority to approve a Change Order or Construction Change Directive. These individuals may seek expert advice as needed to make such approvals.

The Joint Board Representatives will determine whether Project milestones have been met that the Joint Board will pay for under the Contract, and such decision and estimate will be final and conclusive. The Contractor may appeal the decision of the Joint Board Representatives by procedures outlined in Article 13. The Joint Board Representatives will answer any questions as to the meaning of this Contract, or any ambiguity as to the wording of this Contract and give all directions and explanations necessary to make definite any of the provisions of this Contract, or necessary to complete or give them due effect. Because the Contractor is charged with both design and construction of the Project, the Joint Board Representatives' acceptance of any materials or work performed shall not remove from the Contractor the responsibility for the adequacy of the Project's design. The references in this section to the Joint Board Representatives' answering of any questions as to the meaning of this Contract refer only to Contractor's responsibilities under this Contract. Neither the Joint Board nor the Joint Board Representatives are responsible for the soundness or competency of the design as performed by the Contractor, or the means and methods of construction planned or utilized by the Contractor. Review of the Contractor's design shall only be in the limited context of reviewing conformance of the design to the Contract Documents.

The Contractor may request and the Joint Board Representatives will provide written instructions concerning any significant item's conformance to the Project Scope of Work.

7.1.2. Authority of Inspectors

Inspectors employed or contracted by the Joint Board are authorized to inspect all work performed and materials furnished and performance of verification testing.

The inspector will:

- 1. Keep the Joint Board Representatives informed as to the progress of the Work and the manner in which it is being done; and
- 2. Report whenever it appears that the materials furnished and the work performed fail to fulfill the requirements of the Contract Documents;

Any failure by the inspector to call attention to Contractor's failure to comply with Contractor's own design shall not give rise to any liability by the Joint Board Representatives or the Joint Board.

Such inspection may extend to all or any part of the work and to the preparation, fabrication, or manufacture of the materials furnished. Such inspection shall not relieve the Contractor from any obligation to furnish acceptable materials or to perform all work strictly in accordance with the requirements of the Contract Documents. The inspector is not authorized to alter or waive provisions of the Contract. The inspector is not authorized to issue instructions contrary to this Contract or to act as foreman for the Contractor. Any advice that inspectors may give the Contractor shall not be construed as binding the Joint Board in any way or as releasing the Contractor from the fulfillment of the terms of the Contract.

However, the inspector has the authority to reject work or materials until any questions at issue are referred to and as the Joint Board Representatives decide.

7.1.3. Inspection of Work and Owner Verification Testing

The Contractor shall provide the Joint Board Representatives and the Joint Board's inspectors' access to all materials and each part or detail of the Work and furnish the Joint Board Representatives with such information and assistance as required to make a complete and detailed inspection. Such inspection may include preparation and manufacture of the materials at the plant and verification testing.

Contractor shall inform the Joint Board in writing of any part of the work which is about to be covered and offer a full and adequate opportunity to the Joint Board to inspect and test such part of the work before it is covered. The Contractor shall remove or uncover such portions of the finished work as directed. After examination, the Contractor shall restore said portions of the work to the standard required by the Contract Documents. When any unit of government or political subdivision or any railroad corporation pays a portion of the cost of the work covered by this Contract or is otherwise authorized to inspect the work, the Contractor shall provide access to its respective representatives to inspect the work. Such inspection in no way makes any unit of government or political subdivision or any railroad corporation a party to this Contract and in no way interferes with the rights of either party hereunder. If the Work exposed or examined is not in conformance with the requirements of the Contract Documents, then uncovering, removing and restoring the Work and recovery of any delay to any Critical Path occasioned thereby shall be at Contractor's cost and Contractor shall not be entitled to any time extension. Furthermore, any Work done or materials used without adequate notice to and opportunity for prior inspection by the Joint Board (if applicable) or without inspection in accordance with Scope of Work RFP Sections ____3.12.4.6 and —4.1.12.3 may be ordered uncovered, removed or restored at Contractor's cost and without a time extension, even if the work proves acceptable and conforming after uncovering. Except with respect to work done or materials used as described in the foregoing sentences of this Section 7.1.3, if Work exposed or examined under this Section 7.1.3 is in conformance with the requirements of the Contract Documents, then any delay in any Critical Path from uncovering, removing and restoring Work shall be considered a Joint Board-Caused Delay, and Contractor shall be entitled to a Change Order for the cost of such efforts and recovery of any delay to any Critical Path occasioned thereby. Refer to Section 7.2 for provisions regarding payments owing by Contractor to the Joint Board if the Joint Board agrees (in its sole discretion) to accept certain Nonconforming Work.

7.2. Conformity with Plans and Specifications

The Contractor shall perform all Work and furnish all materials in conformity with the material requirements of the Contract Documents. Where definite tolerances are specified in this Contract by the Joint Board, the Joint Board will use such tolerances to establish the limits of reasonably close conformity. Where tolerances are not specified in this Contract, the Joint Board Representatives shall determine the limits of conformity in each individual case.

When the Joint Board Representatives find that the materials, or the finished product in which the materials are used, are not within conformity with the requirements of the

Contract Documents, but finds that reasonably acceptable work has been produced, they shall then make a determination whether to accept the work in place. If the Joint Board agrees to accept any Nonconforming Work without requiring it to be fully corrected, the Joint Board shall be entitled to reimbursement of a portion of the Contract Price in accordance with <u>Section 7.12</u>.

Acceptance of the work in place shall not be construed as binding the Joint Board in any way or as releasing the Contractor from any terms of the Contract Documents. When the Joint Board Representatives find that either the materials, the finished product in which the materials are used, or the work performed are not in conformity with this Contract and have resulted in an inferior or unsatisfactory product, the Contractor shall remove, replace, or correct the work and materials at no additional expense to the Joint Board as provided in <u>Section 7.3</u>.

The values and standards indicated in the Contract Documents are to be considered as the requirements to be complied with in the performance of the Work. The purpose of any specified tolerance range is to accommodate occasional minor variations from the median zone that are unavoidable for practical reasons. When a maximum or minimum value is specified, the production and processing of the material and the performance of the work shall not be preponderantly of borderline quality or dimension.

When referenced standards, such as those promulgated by AASHTO, ASTM, or other recognized organizations, or the Joint Board's or KYTC's or INDOT's own Specifications, Standard Drawings, or similar documents are revised subsequent to the date of the advertisement for RFP, the Contractor shall furnish materials or perform work conforming to the latest edition at the time the work is done. The Joint Board Representatives may, in their sole discretion, approve such a request lif the material or work is deemed to be equal to or better than originally required. This revision shall be viewed as an authorized change. Such revisions shall not be deemed to warrant a eChange eOrder except pursuant to the process outlined in Section 4 herein. In the event of any Ddispute, the Joint Board will select the referenced standard current at the date of advertisement for RFP or the standard specifically referenced in this Contract to determine the Ddispute.

7.3. Removal of Defective and Unauthorized Work

The Contractor shall remedy, or remove and replace in an acceptable manner, at no expense to the Joint Board, all work that has been rejected or does not conform to the requirements of the Contract Documents. The Joint Board will consider any work done contrary to instructions, or any work performed beyond the lines and grades specified in the Plans or as given, except as herein provided, or any work performed without a change order being issued by the Joint Board Representatives, as unauthorized and at no expense to the Joint Board.

Unacceptable work, whether the result of poor workmanship, use of defective materials, damage through carelessness, or any other cause found to exist prior to the formal acceptance of the work, shall be removed immediately and replaced in an acceptable

manner by the Contractor such that it complies with the requirements of the Contract Documents.

Should the Contractor decline or neglect to begin the removal and the replacement of any defective work or remove any unauthorized work within the amount of time stated in a written notice to do so, the Joint Board may retain all monies due or which may become due the Contractor in relation to the defective work or in relation to the costs of its correction, until the requirements of these Contract Documents have been met. When deemed best by the Joint Board Representatives, the Joint Board shall employ the necessary labor to make good or remove such defective or unauthorized work and deduct the cost from any monies due or to become due the Contractor.

7.4. Cooperation by the Contractor

The Contractor shall maintain copies of the construction documents at the site of the work at all times and furnish copies to superintendents, foremen, and inspectors. The Contractor shall require each superintendent and foreman to have with him on the Site, at all times, a copy of that part of the Contract Documents applying to the work he is directing. The Contractor shall be present or have a representative present on the Project at all times, when construction is in progress, to receive and carry out such instructions as the Joint Board Representatives may give. The superintendent must be capable of reading and understanding the plans and specifications and be experienced in the type of work being performed. This Authorized Representative or superintendent shall have full authority to execute orders or directions without delay and shall promptly supply such materials, equipment, tools, labor, and incidentals as may be required.

Such superintendence shall be furnished irrespective of the amount of work sublet. The Contractor shall provide reasonable facilities to enable the Joint Board Representatives to inspect the workmanship and materials entering into the work and cooperate in setting and preserving survey stakes, bench marks, etc., and in all other things necessary to satisfactorily complete the work as contemplated.

Where new work is to be fitted to old work or with work provided by the Design-Build Contractor or the Developer, the Contractor shall check all dimensions and conditions in the field prior to ordering material and assume responsibility for fit of new Work to other work.

Contractor is required to complete a weekly survey to collect feedback from all designated staff to form a holistic view of the project. Contract shall submit the project schedule as a .csv file with format to be provided for collection of operational project data.

The Contractor recognizes that other contractors employed by the Joint Board members or other entities, including but not limited to the Design-Build Contractor and the Developer, must perform work in the Project right of way during the period of this Contract. The Contractor is responsible for being familiar with the terms of the Public-

Private Agreement and the Design-Build Contract, and shall plan its work and coordinate its effort to be consistent with the terms applicable to the Developer's and the Design-Build Contractor's efforts under such agreements. The Contractor shall coordinate with the Joint Board with regards to work being performed by other contractors within the Project limits so as to prevent any delay to Design-Build Contractor's schedule, the Developer's schedule, or the Contractor's Project Schedule. When the Joint Board lets separate contracts within the limits of any one project or for adjacent projects, the Contractor shall conduct the Work so as not to interfere with or hinder the progress or completion of the work being performed by other contractors. The Contractor shall cooperate with contractors working on the same project or adjacent projects. In case of a Delispute with other contractors, the Joint Board Representatives (or their designees) shall referee and make a final and binding decision. In refereeing disputes related to a particular Crossing, the Joint Board shall be represented by the designated representative of the IFA for disputes with respect to the East End Crossing, and by the designated representative of KYTC for disputes with respect to the Downtown Crossing. The Contractor consents, at KYTC's option, to join the dispute resolution proceedings as outlined in the Design-Build Contract and the Public Private Agreement when disputes involve the Contractor and are initiated by IFA or the Developer under the Public-Private Agreement or are initiated by KYTC or the Design-Build Contractor under the Design-Build Contract, as applicable.

As far as possible, the Contractor shall arrange the work and place and dispose of the materials being used so as not to interfere with the operations of the other contractors within the limits of the same project or on adjacent projects.

The Contractor shall join the work with that of the other contractors, the Design-Builder, and the Developer in an acceptable manner and perform it in proper sequence with the work of the other contractors.

7.5. The Joint Board Review and Inspection Generally

The Work shall be performed in accordance with the Contract Documents and the designs prepared by the Contractor and approved by the Joint Board during the design phases of the Project pursuant to <u>Article 14</u> and <u>Article 15</u> hereof.

It is Contractor's responsibility to provide work of such a nature to develop a finished product in accordance with the Contract Documents, the design and construction documents and specifications developed by the Contractor after review and comment by the Joint Board. The Contractor shall verify all dimensions in the field before developing the design documents and be responsible for the satisfactory quality and completion of the <u>wW</u>ork regardless of the level of the Joint Board inspection or concurrence.

The Work shall be subject to the Joint Board's review and written comments before its implementation, and shall not be thereafter amended or altered without the prior approval of the applicable Joint Board Representatives and subsequent review and written comment by the Joint Board.

The Contractor shall not be relieved of its obligation to perform the work in accordance with the Contract Documents, or any of its other obligations under the Contract Documents, by oversight, spot checks, audits, reviews, tests, inspections, acceptances or approvals by any persons, or by any failure of any person to take such action. The oversight, spot checks, audits, reviews, tests, inspections, acceptances and approvals by any person do not constitute Formal Acceptance of the particular material or work, or waiver of any legal or equitable right with respect thereto. The Joint Board may reject or require the Contractor to remedy any Nonconforming Work and/or identify additional work which must be done to bring the Project into compliance with Contract requirements, whether or not previous oversight, spot checks, audits, reviews, tests, inspections, acceptances or approvals were conducted by any Person. The Joint Board's review of design documents for construction as described in the Contract Documents shall constitute review and comment only, and not be deemed to relieve the Contractor of liability for the design.

7.6. Final Inspection and Formal Acceptance of Work

The Joint Board will not consider the work on a Crossing complete and will not make final payment with respect to a Crossing, the Operations Center or a Walk-in Center until the Contractor clears the right-of-way, borrow pits, and all ground the Contractor occupies in connection with the work of all rubbish, equipment, excess materials, and temporary structures. The Contractor shall place rubbish and all waste materials of whatever nature, other than hazardous materials, on either public or private property in a location out of view from the roadway and in a manner acceptable to the Joint Board that does not present an unsightly appearance. The Contractor shall restore in an acceptable manner all property, both public and private, that was damaged in the prosecution of the work. The Contractor shall leave all space under structures unobstructed and in such condition that drift shall not collect and induce scouring.

The Contractor shall follow the procedures for Substantial Completion in <u>Section 9.1</u>. Following the attainment of Substantial Completion Acceptance for a Crossing, the <u>Operations Center or a Walk-in Center</u>, the Contractor shall proceed with the completion of punch list and other items such as striping, seeding, tree planating and any remaining work necessary for final completion of such Crossing, <u>Operations Center or Walk-in Center</u>, and shall notify the Joint Board Representatives when the Project work for thea Crossing, the <u>Operations Center or a Walk-in Center</u> is near final completion. The Joint Board Representatives will then advise in writing all punch list or other items that remain unsatisfactory. When these items are complete to the Joint Board Representative's satisfaction, the Joint Board Representatives will call the Project complete with respect to such Crossing. <u>Operations Center or Walk-in Center</u> and issue a project completion notice.

The Joint Board and other appropriate agencies, such as FHWA, will complete final inspections on all items of work for Formal Acceptance of the Crossing in accordance with <u>Section 9.3</u> within 90 calendar days of the date of the Joint Board Representative's issuance of the project completion notice. The Joint Board will make individual final inspections on particular groups of work items such as structures, electrical, grade and

drain, and surface. The Joint Board may, at its election, make final inspections before the Project is called complete on items of work that have been completed. The Joint Board Representatives will issue written final inspection reports for items of work upon completion of each final inspection. The reports will include a list of all uncompleted work and required corrective work. The Joint Board Representatives will issue a comprehensive final inspection report that will include all inspection reports. The Contractor shall complete all items of uncompleted work and all required corrective work listed in the final inspection reports within 90 calendar days of receiving the engineer's comprehensive final inspection report.

The Contractor shall submit required as-built drawings, project documentation, and required information on materials and equipment incorporated into the Project, considering them as uncompleted work or required corrective work. The Contractor shall deposit any additional Software Source Code and other required documents in the Source Code Escrow and provide the Joint Board with an inventory of such deposit.

If there is a dispute regarding any of the items listed as uncompleted work or required corrective work on any of the final inspection reports, the Contractor shall submit in writing a letter of dispute to the Joint Board Representatives within 30 days of receipt of the report. The Joint Board will respond back in writing to the letter of dispute within 21 days of receiving the letter.

If there is still a dispute, proceed according to <u>Section 13</u>. When the dispute does not apply to all items of Work in the report, complete the items not in dispute as specified herein. Subject to <u>Section 16.8</u>, the Joint Board will make Formal Acceptance of the Project when the Joint Board determines that the Contractor has met all of the conditions to Formal Acceptance in <u>Section 9.3.-</u> Formal acceptance shall not be construed as a waiver by the Joint Board of any legal rights should the Contractor's performance of the design and construction duties performed under this Contract be found to be defective.

7.7. Intentionally Omitted

7.8. Limitations of Operations

The Contractor shall limit operations so that its efforts do not unduly interfere with the work of the Design-Build Contractor and the Developer. The Contractor shall take care not to cause undue inconvenience to the traveling public. In the prosecution of the Work, the Contractor shall confer with the Joint Board Representatives regarding the points for the start of operations. If the Contractor is not conducting ongoing, active work in areas which have been blocked to traffic, notwithstanding any monetary amounts assessed to the Contractor under the Contract Documents for such restriction of traffic, the Joint Board Representatives may instruct the Contractor to modify the points in which those portions of Contractor's operations occur, so as to minimize traffic congestion or interruption. When operations have temperature or seasonal limitations, the Contractor shall schedule the work to comply with any specification, manufacturer, or Supplier requirements, taking into account obligation to coordinate efforts with the

Developer and the Design-Builder, and schedule work consistent with the requirements of the Public-Private Agreement and the Design-Build Contract. When ordered by the Joint Board, the Contractor shall open any or all sections to travel, whether the Work is completed or not.

As identified in the Contract Documents, the Contractor must at all times restrict its lane closures so as to provide for two lanes of traffic, at minimum, in each direction on all Interstates involved in the Project. See Project ScopeExhibit XXXRFP Appendix N and Section 8.4.1 for additional details and the requirement that Contractor pay Liquidated Damages with respect to certain lane closures. as provided in Section 8.4.1

7.9. Character of Workers, Methods, and Equipment

7.9.1. General.

The Contractor shall employ, at all times, sufficient labor and equipment for prosecuting the work to full completion in the manner and time required by the Contract.

The Contractor shall employ workers that have sufficient skill and experience to properly perform the work assigned to them. The Contractor shall employ workers engaged in special work or skilled work that have sufficient experience in such work and in the operation of the equipment required to perform all work properly and satisfactorily.

If the Joint Board representatives judge that any person employed by the Contractor does not perform the work in a proper and skillful manner or is intemperate or disorderly, at the written request of the Joint Board Representatives, the Contractor shall remove such person from the Project and shall not employ such person again in any portion of the work without the approval of the Joint Board Representatives. Should the Contractor fail to remove such person or persons as directed, or fail to furnish suitable and sufficient personnel for the proper prosecution of the work, the Joint Board Representatives may suspend the work by written notice until the Contractor complies with such orders.

The Contractor shall use only equipment of sufficient size and in such mechanical condition as to conform to the requirements of the work and to produce a satisfactory quality of work. The Contraction shall use equipment that does not harm the roadway, adjacent property, or other highways.

When a weight or weight range is specified for compaction equipment, the Contractor shall use equipment that has a plate or sign attached showing its weight, or minimum and maximum weights when applicable. In lieu of the plate or signs, the Contractor may weigh the equipment on scales certified by the Division of Weights and Measures before using the equipment on the Project.

When the methods and equipment that are used to accomplish the construction are not prescribed in the Contract, the Contractor shall use any methods or equipment that will to the satisfaction of the Engineer, accomplish the Contract work in a manner conforming to the Contract.

When the Contract specifies certain methods and equipment, the Contractor shall use such methods and equipment unless the Joint Board Representatives authorize others. The Contractor shall request approval from the Joint Board Representatives to use a method or type of equipment other than those specified in the Contract. The Contractor shall make the request in writing and include a full description of the methods and equipment proposed and an explanation of the reasons for desiring to make the change. If the Joint Board Representatives' approval is granted, the Contractor is fully responsible for producing Work according to the Contract. If, after trial use of the substituted methods or equipment, the Joint Board Representatives determine that the work produced does not conform to the Contract, Contractor shall discontinue the use of the substitute method or equipment and complete the remaining construction with the specified methods and equipment. The Contractor shall remove the deficient work and replace it with work of specified quality, or take such other corrective action as the Joint Board Representatives may direct, at no expense to the Joint Board. The Joint Board Representatives shall not change the basis of payment or the Contract time for the construction items involved as a result of authorizing a change in methods or equipment.

7.10. Subcontracting

The Contractor shall not subcontract, sell, transfer, assign, or otherwise dispose of this Contract or any portion of this Contract, or of the right, title, or interest therein, without the Joint Board's written consent. When the Joint Board gives such consent, the Joint Board will allow the Contractor to subcontract a portion of the Work. Subcontractors may, with the Joint Board's consent, subcontract a portion of the Work to lower tiered Subcontractors. However, the Contractor shall remain responsible for all work and shall maintain a supervisory role over the entire Work and Project. Except for Work elements for which no prequalification requirement exists, all Subcontractors shall be prequalified to perform the Work subcontracted to them as provided in the Construction Procurement and Professional Services sections of the Joint Board's website, as applicable to such Subcontractor.

The Joint Board's written consent to subcontract, assign, or otherwise dispose of any portion of this Contract does not, under any circumstances, relieve the Contractor of its liabilities and obligations under the Contract Documents. The Joint Board will make transactions only with the Contractor. The Joint Board will recognize Subcontractors only in the similar capacity of employees or workers of the Contractor who are subject to the same requirements as to character and competence as specified in <u>Section 7.9</u>.

The Contractor shall be fully responsible to the Joint Board for all acts and omissions of the Subcontractors, Suppliers, joint-venturers, partners, consultants and other individuals or entities performing or furnishing any of the Work (sometimes referred to herein as "the "Contractor-Related Entities") just as the Contractor is responsible for Contractor's own acts and omissions. The retention of Subcontractors by Contractor will not relieve Contractor of its responsibility hereunder or for the quality of the Work or materials provided by it. Contractor will at all times be held fully responsible to the Joint Board for the actions, omissions, negligence, willful misconduct, or breach of applicable

Law or contract by its Subcontractors and persons employed by them and no Subcontract entered into by Contractor will impose any obligation or liability upon the Joint Board to any such Subcontractor or any of its employees. Nothing in this Contract will create any contractual relationship between the Joint Board and any Subcontractor of Contractor.

Each instrument evidencing any agreement of Contractor with any Subcontractor shall provide, in terms and in form and substance satisfactory to the Joint Board that: (a) the rights of Contractor under such instrument are assigned to the Joint Board contingent only upon delivery of written request from the Joint Board or its successor or assign following default by Contractor or termination or expiration of this Contract; and (b) all warranties (express and implied) of such Subcontract shall inure to the benefit of the Joint Board.

Contractor shall provide the Joint Board with a list of all Subcontracts with each monthly update, shall allow the Joint Board access to all Subcontracts and records regarding Subcontracts and shall deliver to the Joint Board, within 10 Days after execution, copies of all Major Subcontracts and, within 10 Days after receipt of a request from the Joint Board, copies of all other agreements or documents as may be requested.

The Contractor shall be solely responsible for scheduling and coordinating Subcontractors, Suppliers, and other individuals and entities performing or furnishing any of the Work under a direct or indirect contract with the Contractor. The Contractor shall require all Subcontractors, Suppliers, and such other individuals and entities performing or furnishing any of the Work to communicate with the Joint Board through the Contractor.

All work performed for the Contractor by a Subcontractor or Supplier will be pursuant to an appropriate subcontract between the Contractor and the Subcontractor or Supplier which specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract Documents for the benefit of the Joint Board, with the only exceptions being that the Contractor may require lesser insurance coverage limits and/or the absence of surety bond requirements equal to those the Joint Board requires of the Contractor. The following specific requirements shall apply to all Subcontracts:

(a) Contractor shall, prior to soliciting any bids for performance of work or labor or rendering of services relating to the design or construction of any portion of the Project or for special fabrication, assembly and installation of any portion of the Work, submit to the Joint Board for its review and approval a procedure for the conduct of the bidding and approval process applicable to Subcontracts. Contractor may use procedures set forth in the Joint Board Standards or may submit alternative procedures to the Joint Board for approval. Contractor shall not enter into any Subcontracts except in accordance with the foregoing procedure. Once Contractor has entered into any Subcontract, Contractor shall not have the right to make any substitution of such Subcontractor except with the Joint Board's prior written approval.

- (b) As soon as Contractor has identified a potential Subcontractor, but in no event less than 30 Days prior to the scheduled initiation of work by such proposed Subcontractor, Contractor shall notify the Joint Board in writing of the name, address, phone number and contact name of such Subcontractor.
- (c) Each Subcontract shall include terms and conditions sufficient to ensure compliance by the Subcontractor with the requirements of the Contract Documents, and shall include those terms that are specifically required by the Contract Documents to be included therein. All Subcontracts, including Subcontracts with Suppliers, shall incorporate terms substantially similar to those contained in this Contract, specifically including an agreement by the Subcontractor to participate in any dispute proceeding pursuant to Section 13, if such participation is requested by either the Joint Board or Contractor, and a requirement to allow audits by the Joint Board.

Each Subcontract shall:

- (a) Set forth a standard of professional responsibility or a standard for commercial practice equal to the requirements of the Contract Documents and prudent industry standards for work of similar scope and scale and shall set forth effective procedures for claims and change orders.
- (b) Require the Subcontractor to carry out its scope of work in accordance with the Contract Documents, the Governmental Approvals and applicable Law, including the applicable requirements of the DBE Performance Plan.
- (c) Include a covenant to maintain all licenses required by applicable Law, and set forth warranties, guaranties and liability provisions of the contracting party in accordance with good commercial practice for work of similar scope and scale.
- (d) Be fully assignable without cost to the Joint Board, such assignability to include the benefit of all Subcontractor warranties, indemnities, guarantees and professional responsibility and include express requirements that: (i) it will maintain usual and customary books and records for the type and scope of operations of business in which it is engaged (e.g., constructor, equipment supplier, designer, service provider); (ii) permit audit thereof by Contractor, and provide progress reports to Contractor appropriate for the type of work it is performing sufficient to enable Contractor to provide the reports it is required to furnish the Joint Board under this Contract; and (iii) allow the Joint Board to assume the benefit of Contractor's rights with liability only for those remaining obligations of Contractor accruing after the date of assumption by the Joint Board. No such assignment shall release or relieve Contractor from its obligations or liabilities under the assigned Subcontract.
- (e) Not be assignable by the Subcontractor to any party other than the Joint Board (or its assignee) without Contractor's prior written consent.
- (f) With respect to any Subcontract which, when aggregated with all Subcontracts between Contractor and such Subcontractor for the same Fiscal Year, is in excess of \$250,000: (i) be terminable by the Subcontractor only for cause; and (ii)

include an indemnity from the Subcontractor in favor of Contractor and the Indemnified Parties against any and all Losses arising out of, related to or associated with, the actions, omissions, negligence, willful misconduct, or breach of applicable Law or contract by the Subcontractor or any of its officers, employees, agents or representatives.

- (g) Expressly require the Subcontractor to participate in meetings between Contractor and the Joint Board, upon the Joint Board's request, concerning matters pertaining to such Subcontract or its work, provided that all direction to such Subcontractor shall be provided by Contractor, and provided further that nothing in this clause (g) shall limit the authority of the Joint Board to give such direction or take such action which, in its sole opinion, is necessary to remove an immediate and present threat to the safety of life or property.
- (h) Expressly provide that all Liens, claims and charges of the Subcontractor and its Subcontractors at any time shall not attach to any interest of the States in the Project or the Final ROW.
- (i) Be consistent in all other respects with the terms and conditions of this Contract to the extent such terms and conditions are applicable to the scope of work of such Subcontractors, and include all provisions required by this Contract.

Contractor shall not amend any Subcontract with respect to any of the foregoing matters without the prior written consent of the Joint Board. All Subcontracts with Affiliates shall be on terms no less favorable (as determined by the Joint Board) to Contractor than to non-Affiliates of the Subcontractor. Contractor shall not enter into any Subcontracts with any Person then debarred or suspended from submitting bids by any agency of the State.

7.11. No Estoppel

Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents, or any of its Warranty, maintenance or indemnity obligations, as the result of oversight, spot checks, audits, reviews, tests or inspections performed by any Persons, approvals or acceptances made by any Persons, or any failure of any Person to take such action. Subject to statutory limitations on liability, the Joint Board shall not be precluded or estopped, by any measurement, estimate or certificate made either before or after Formal Acceptance and payment therefore, from showing that any such measurement, estimate or certificate is incorrectly made or untrue, or from showing the true amount and character of the work performed and materials furnished by the Contractor, or from showing that the work or materials do not conform in fact to the requirements of the Contract Documents. Notwithstanding any such measurement, estimate or certificate, or payment made in accordance therewith, the Joint Board shall not be precluded or estopped from (a) correcting such measurement, estimate or certificate and (b) recovering from the Contractor and the surety(ies) the damages the Joint Board may sustain by reason of Contractor's failure to comply with the terms of the Contract Documents, including recovery of any overpayment sustained due to such

failure. In addition, any waiver by the Joint Board of any breach by the Contractor shall not be held to be a waiver of any other or subsequent breach.

7.12. Acceptance of Nonconforming Work

Contractor acknowledges and agrees that the Joint Board shall have sole discretion regarding acceptance or rejection of Nonconforming Work and shall have sole discretion with regard to the amount payable in connection therewith, provided, however, that the Joint Board may in its sole discretion reject any civil construction Work that the Engineer for each Crossing determines is Nonconforming. Subject to the foregoing, the Joint Board may, in its sole discretion, accept any Nonconforming Work without requiring it to be fully corrected, and shall be entitled to a pay adjustment (or reimbursement of a portion of the Contract Price, if applicable). In certain events, it may not be possible for the Nonconforming Work to be made to conform to the requirements of the Contract Documents, including, without limiting the foregoing, Contractor's failure to perform such items during a required time period. In general, the pay adjustment (or reimbursement) shall equal, at the Joint Board's election, an amount equal to the greater of (a) 100% of Contractor's cost savings associated with its failure to perform the work in accordance with the Contract requirements, or (b) the amount of the Contract Price allocated to such Work, plus (c) the amount deemed appropriate by the Joint Board to provide compensation for impacts to affected parties such as future maintenance and/or other costs relating to the Nonconforming Work. Such reimbursement shall be payable to the Joint Board within 10 Days after Contractor's receipt of an invoice therefore.

7.13. Key Personnel

The Contract Documents identifyRFP Section 4.1.12.5 identifies certain job categories of Key Personnel for the Project. Contractor shall not change, or permit any change in, any Key Personnel without the prior written consent of the Joint Board.

Contractor shall designate in writing who shall have onsite field and office authority to represent and act for Contractor. An Authorized Representative shall be present at the jobsite at all times while work is actually in progress at the jobsite. Contractor shall provide phone and pager numbers for all Key Personnel. The Joint Board requires the ability to contact the following Key Personnel 24 hours per Day, seven days per week: Program Manager; installation manager (from the date of issuance of an NTP to the date of Final Acceptance); and maintenance manager (during the Maintenance Term).

Contractor acknowledges and agrees that the award of this Contract by the Joint Board to Contractor was based, in large part, on the qualifications and experience of the personnel listed in the Proposal and Contractor's commitment that such individuals would be available to undertake and perform the work. Contractor represents, warrants and covenants that such individuals are available for and will fulfill the roles identified for them in the Proposal in connection with the work. Unless otherwise agreed to by the Joint Board in writing, individuals filling Key Personnel roles shall be available for the work and shall maintain active involvement in the prosecution and performance of the

work. In addition to the foregoing, the Joint Board reserves the right to require a 100% time commitment per position from any Key Personnel if the Joint Board, in its sole discretion, determines that such personnel are not devoting sufficient time to the prosecution and performance of the work.

[Subject to the other terms, provisions and requirements of the Contract Documents (including the requirements for quality control and quality assurance and the right of the Joint Board to require a 100% time commitment per position as set forth in this Section 7.13), the Joint Board acknowledges that an individual may fill more than one Key Personnel role. If an individual filling one or more Key Personnel roles is are not available for the Work and does not maintain active involvement in the prosecution and performance of the Work, Contractor acknowledges that the Joint Board, the Work and the Project will suffer significant and substantial Losses and that it is impracticable and extremely difficult to ascertain and determine the actual Losses which would accrue to the Joint Board in such event. Therefore, if certain Key Personnel are not available or not actively involved in the prosecution and performance of the work, as determined by the Joint Board, in its sole discretion, Contractor agrees to pay the Joint Board a liquidated amount as indicated in RFP Section 4.1.12.5, provided, however, that the Joint Board shall provide the Contractor with 10 days prior notice and opportunity to cure before assessing such Liquidated Damages.

._for each position held by such individual equal to the annual amount set forth below for each such position, prorated monthly for the period (for greater certainty, rounded up to the next whole month) during which such individual is not available or not actively involved in the prosecution and performance of the work, as deemed compensation to the Joint Board for such Losses:

POSITION LIQUIDATED AMOUNT

Program Manager \$100,000

Maintenance Manager \$50,000

All other positions noted with an asterisk in Exhibit \$25,000

Contractor understands and agrees that any damages payable in accordance with this Section 7.43—13 and RFP Section 4.1.12.5 are in the nature of Liquidated Damages damages and not a penalty and that such sums are reasonable under the circumstances existing as of the Effective Date. The Joint Board shall have the right to deduct or draw upon any amount owed by Contractor to the Joint Board hereunder from any amounts owed by the Joint Board to Contractor, including from any bond held by the Joint Board under this Contract. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction or draw. Notwithstanding the foregoing, Contractor shall not be liable for Liquidated dDamages under this Section 7.13 if (i) Contractor removes or replaces such personnel at the direction of the Joint Board; (ii) such individual is unavailable due to death, retirement, injury or no longer being employed by the applicable Contractor-Related Entity

(provided that moving to an affiliated company shall not be considered grounds for avoiding Liquidated Ddamages), or (iii) such individual is unavailable due to the Joint Board's failure to issue the NTP by August 16, 2013 for a reason other than the fault, act or omission of any Contractor-Related Entity; provided, however, in such case, Contractor shall promptly propose to the Joint Board a replacement for such personnel, which individual shall be subject to the Joint Board's review and written consent, and shall replace such personnel as approved by the Joint Board. If the NTP has not been issued by the specified date, Contractor shall have 30 Days after issuance of the NTP to identify any change in Key Personnel without incurring any Liquidated dDamages. Following any Joint Board-approved substitution or replacement of a Key Personnel pursuant to the terms hereof, the new individual shall be considered a Key Personnel for all purposes under this Contract, including the provisions of this Section 7.13 relative to Liquidated dDamages.

Contractor acknowledges and agrees that the Key Personnel positions are of critical importance to The Joint Board and the Project. In addition to the approval rights of the Joint Board set forth in this Section 7.13 and the Liquidated Damages set forth in this Section 7.13 and RFP Section 4.1.12.5, if an individual in certain Key Personnel positions delineated on Exhibit —E leaves that position for a reason other than as set forth in clauses (i)-(iii) of Section 7.13, RFP Section 4.1.12.5 the Joint Board shall have the unilateral right to terminate this Contract without further liability to Contractor, except for payment for such work that has been completed and accepted as of the date of termination, unless Contractor provides the Joint Board a replacement acceptable to the Joint Board within 30 Days after the earlier of (i) the date on which such individual has left his/her position; or (ii) Contractor or the Joint Board becomes aware that such individual intends to leave his/her position. In connection with a termination under this Section 7.13, Contractor shall not be entitled to any payment of profit, overhead or any settlement costs, but shall only be entitled to the payment of Contractor's actual costs incurred and not previously paid for by the Joint Board in connection with work which has been completed and accepted as of the termination date.

Any position on the Contractor's Project organizational chart or within the Contractor's organization structure that is above that of a Key Personnel position for which Liquidated dDamages may apply will be deemed to be a Key Personnel position and, for purposes of liquidated damagesLiquidated Damages under this Section 7.13, shall be at the level which is immediately higher than the Key Personnel immediately below that position (e.g., an individual that reports to the Maintenance Manager level but is higher than the other Key Personnel level would be considered a Maintenance Manager for this purpose).

8.0 Liquidated Damages and Stipulated Damages

8.1. Reasonable Approximations of Damages

The Joint Board and the Contractor agree that time is of the essence of this Contract, and that there will be sustained by the Joint Board considerable monetary damages in the event of failure or delay in the completion of the Work, which damages will be impracticable and extremely difficult to definitely ascertain or establish Contractor

further understands and agrees that if Contractor fails to submit a pPerformance aAudit rReport in accordance with RFP Appendix BScope of Work Section —23, fails to perform the Work in a manner that prevents unscheduled closure of traffic lanes, or otherwise fails to meet the Performance Requirements specified in RFP Appendix BSection 23, the Joint Board will suffer substantial Losses that will be impracticable and extremely difficult to definitely ascertain or establish. Therefore, the Joint Board agrees to accept, and the Contractor agrees to pay, Liquidated Damages in the amounts specified herein as compensation for damages associated with any delay in meeting a Completion Deadline, and Delay Reporting Liquidated Damages as compensation for any Losses associated with any damages associated with delay in submitting a performance audit, and the sums stated below and in the other Contract Documents as Liquidated Damages for lane closures and failures to meet Performance Requirements, all of which are hereby agreed upon by the parties as reasonable approximations of such damages.

8.2. Liquidated Damages for Delays in Achieving Completion

The Joint Board will rely upon toll revenues from the completed Downtown Crossing and the East End Crossing for the funding for this Project. The Contractor shall be liable to the Joint Board for <code>Liquidated dD</code> amages for each and every calendar day after a Deadline that Contractor fails to reach Substantial Completion Acceptance, Punch List Acceptance and/or Formal Acceptance of a Crossing or the System, in the following amounts:

- 1) \$ [TBD]per day for failure to achieve Substantial Completion

 Acceptance of the Operations Center by the Operations Center

 Substantial Completion Acceptance Deadline.
- 2) \$ [TBD]per day for failure to achieve Substantial Completion

 Acceptance of a Walk-in Center by the applicable Walk-in Center

 Substantial Completion Deadline.
- \$ per day for failure to achieve East End Crossing (Toll Zones EEC-1 and EEC-2) Substantial Completion Acceptance by the East End Crossing (Toll Zones EEC-1 and EEC-2) Substantial Completion Deadline, \$ per day for failure to achieve East End Crossing (Toll Zones EEC-1 and EEC-2) Punch List Acceptance by the East End Crossing (Toll Zones EEC-1 and EEC-2) Punch List Acceptance Deadline, and \$ per day for failure to achieve East End Crossing (Toll Zones EEC-1 and EEC-2) Formal Acceptance by the East End Crossing (Toll Zones EEC-1 and EEC-2) Formal Acceptance Deadline.
- 4) \$ per day for failure to achieve Downtown Crossing (Toll Zones DB-R1 and R-1) Substantial Completion Acceptance by the Downtown Crossing (Toll Zones DB-R1 and R-1) Substantial Completion Acceptance Deadline, \$ per day for failure to reach Downtown Crossing (Toll Zones DB-R1 and R-1) Punch List Acceptance by the

<u>Downtown Crossing (Toll Zones DB-R1 and R-1) Punch List Acceptance</u>
<u>Deadline; and \$ _____per day ...for failure to reach Downtown Crossing</u>
(Toll Zones DB-R1 and R-1) Final Acceptance by the Downtown Crossing
(Toll Zones DB-R1 and R-1) Final Acceptance Deadline.

- 5) \$ per day for failure to achieve Kennedy Crossing (Toll Zones KB-R1 and R-2) Substantial Completion Acceptance by the Kennedy Crossing (Toll Zones KB-R1 and R-2) Substantial Completion Acceptance Deadline, \$ per day for failure to reach Kennedy Crossing (Toll Zones KB-R1 and R-2) Punch List Acceptance by the Kennedy Crossing (Toll Zones KB-R1 and R-2) Punch List Acceptance Deadline; and \$ per day for failure to reach Kennedy Crossing (Toll Zones KB-R1 and R-2) Final Acceptance by the Kennedy Crossing (Toll Zones KB-R1 and R-2) Final Acceptance Deadline.[Liquidated Damages for failure to achieve the Completion Deadlines for the bridge crossings will be specified by Addendum to this RFP.]
- 6) \$ per day for failure to achieve System Substantial Completion Acceptance by the System Substantial Completion Acceptance Deadline.
- 6) \$_____ per day for failure to achieve Downtown Crossing (Toll Zones DB-1 and R-1) Substantial Completion Acceptance by the Downtown Crossing (Toll Zones DB-1 and R-1) Substantial Completion Acceptance Deadline].
- 7) \$ per day for failure to achieve Downtown Crossing (Toll Zones DB-1, R-1, KB-1 and R-2) Substantial Completion Acceptance by the Downtown Crossing (Toll Zones DB-1, R-1, KB-1 and R-2) Substantial Completion Acceptance Deadline.
- 7) \$ per day for failure to achieve System Substantial Completion
 Acceptance by the System Substantial Completion Acceptance Deadline,
 \$ per day for failure to achieve System Punch List Acceptance by
 the System Punch List Acceptance Deadline, and \$ per day for
 failure to achieve System Formal Acceptance by the System Formal
 Acceptance Deadline,...

Said amounts shall in no event be considered a penalty or otherwise than as liquidated damages_Liquidated Damages. The Contractor agrees that any accrued lLiquidated dDamages may be deducted from any current or final estimate after the day said lLiquidated dDamages begin to accrue. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction. Said lLiquidated dDamages shall continue with respect to each Crossing Crossing and the System until the Contractor achieves the relevant Completion Acceptance Deadline as determined by the Joint Board. If during any period the Contractor has failed to meet multiple deadlines with respect to a Crossing Crossing or the System subject to liquidated damages under this Section 8.2, the lLiquidated dDamage amounts specified above with respect to such CrossingCrossing or with respect to the System shall not be

considered cumulative. In such event \bot iquidated d amages shall be assessed at the highest relevant daily rate.

8.3. Delay Reporting Liquidated Damages

8.3.1. Delay Reporting Liquidated Damages Generally

Contractor acknowledges and agrees that the Delay Reporting Liquidated Damages are intended to compensate the Joint Board solely for Contractor's failure to submit a performance audit-reports in accordance with Scope of WorkRFP Appendix B Section ——B23 and shall not excuse Contractor from liability from any other breach of requirements of the Contract Documents, including any failure of the System to meet Performance Requirements. Contractor shall not be liable for actual damages in addition to the Delay Reporting Liquidated Damages for Contractor's failure to submit a performance audit-report in accordance with RFP Appendix B Section B23.1.36, 23.2.22 and/or 23.2.23 Scope of Work Section ——. The fact that the Joint Board has agreed to accept Delay Reporting Liquidated Damages as compensation for its damages associated with any delay in meeting such report submission requirements shall not preclude the Joint Board from exercising its other rights and remedies respecting the delay set forth elsewhere in this Contract, other than the right to collect other damages due solely to the delay.

8.3.2. Amount of Delay Reporting Liquidated Damages

Contractor acknowledges and agrees that because of the unique nature of the Project, and the fact that the continual performance auditing of the System is essential to the Joint Board's ability to ensure performance of the System in conformance with Performance Requirements, it is not possible to ascertain and determine the actual Losses which would accrue to the Joint Board and the public from such failure. Therefore, commencing upon the date immediately following the submission deadline for the performance audit reports set forth in Scope of Work Section — RFP Appendix B Section B23, Contractor shall pay to the Joint Board Delay Reporting Damages in athe dollar amounts equal to \$____ per Dayspecified in RFP Appendix B, Section B23.1.36 ,B23.2.22 and B23.2.23 as damages for delay in providing certain reports. It is understood and agreed by Contractor that any Delay Reporting Liquidated Damages payable in accordance with this Section 8.3 and the RFP are in the nature of liquidated damages and not a penalty and that such sums are reasonable in light of the anticipated or actual harm caused by the breach, the difficulties of the proof of loss, and the inconvenience or nonfeasibility of otherwise obtaining an adequate remedy. Contractor further acknowledges and agrees that Delay Reporting Liquidated Damages may be owing even though no Event of Default has occurred.

8.4. Liquidated Damages for Lane Closures and Toll Lane Closures

- **8.4.1.** Liquidated damages shall be assessed for lane closures and toll lane closures, whether occurring before or after Substantial Completion or Formal Acceptance, as follows:
 - 1) For any lane closure caused by any <u>defect in the Work or</u> action or failure to act by Contractor that provides fewer open lanes than the number specified in "Lanes Required to be Maintained for Traffic Control" in [theRFP Appendix N Technical Specifications?] for each crossingCrossing, liquidated damages will be assessed as follows:
 - a) East End Crossing including approaches thereto:

 \$____5,000 per {30 minute} period or any fraction thereof.
 - b) Downtown Crossing including approaches thereto:
 - (i) Downtown Crossing Temporary Configuration: \$_____ per [minute].
 - (ii) Kennedy Crossing: \$_____5,000 per [30 minute] period or any fraction thereof.
 - (iii) New Downtown Crossing: \$_____5,000 per [30_minute] period or any fraction thereof

In no event shall a lane closure indicated above occur during the weekday peak hours as defined by Section 8.4.2.

- Once tolling has commenced, fFor inability to collect electronic tolls during weekday peak hours as a result of any action or failure to act by the Contractor or any Subcontractor (that is, electronic tolling capability is entirely shut down on all toll lanes onin a CrossingToll Zone), liquidated damages will be assessed as follows:
 - a) East End Crossing: \$_____ per [hour].
 - (i) Toll Zone EEC-1: \$30,000 per hour or any fraction thereof.
 - ii) Toll Zone EEC-2: \$30,000 per hour or any fraction thereof.
 - b) Downtown Crossing:
 - (i) Downtown Crossing Temporary Configuration Toll Zone R-1: \$_____25,000 per [hour] or any fraction threreof.
 - (ii) Kennedy CrossingToll Zone R-2: \$____25,000 per {hour} or any fraction thereof.

- (iii) New Downtown CrossingToll Zone DB-1 or DB-R1: \$_____25,000 per {hour} or any fraction thereof.
- (iv) Toll Zone KB-1 or KB-R1: \$25,000 per hour or any fraction thereof.
- Once tolling has commenced, Ffor inability to collect electronic tolls during weekday nonpeak hours and weekend days as a result of any action or failure to act by the Contractor or any Subcontractor (that is, electronic tolling capability is entirely shut down on all toll lanes on a Crossing), Liquidated &Damages will be assessed as follows:
 - a) East End Crossing: \$_____per [hour].
 - (i) Toll Zone EEC-1: \$10,000 per hour or any fraction thereof.
 - ii) Toll Zone EEC-2: \$10,000 per hour or any fraction thereof.
 - b) Downtown Crossing:
 - (i) Toll Zone R-1: \$7,500 per hour or any fraction thereof.
 - (ii) Toll Zone R-2: \$7,500 per hour or any fraction thereof.
 - (iii) Toll Zone DB-1 or DB-R1: \$7,500 per hour or any fraction thereof.
 - (iv) Toll Zone KB-1 or KB-R1: \$7,500 per hour or any fraction thereof.(i) Downtown Crossing Temporary Configuration: \$_____ per [hour].
 - (ii) Kennedy Crossing: \$ per [hour].
 - (iii) New Downtown Crossing: \$_____ per [hour].

8.4.2.

For purposes of calculating liquidated damages for toll-revenue impacts under items 2 and 3 of Section 8.4.1, weekdays are defined as Monday through Friday, excluding holidays, and peak hours are defined as 6:00 a.m. to 11:00 a.m., and 3:00 p.m. to 7:00 p.m. A five minute grace period will be allowed before Liquidated Damages are assessed for toll revenue impacts under items 2 and 3 of this Section 8.4; but the grace period shall not apply in determining whether the impact period exceeds one hour, so that two hours of Liquidated Damages will apply if the impact lasts more than 60 minutes.

Ramp and lane closure time is specified in calendar days. The Joint Board Representatives will charge every calendar day, beginning with the calendar day

following the date of the beginning of work, including all Saturdays, Sundays, holidays, and non-working days, to the Contract

The fact that the Joint Board has agreed to accept Liquidated Damages as compensation for its damages associated with subparagraphs 1 through 3 above shall not preclude the Joint Board from exercising any of its rights and remedies respecting such breach under the Contract, other than the right to collect damages for such breach.

In the event of a toll lane closure and assessment of the Liquidated Damages specified in this Section 8.4, such Liquidated Damages shall be in lieu of any other Liquidated Damages that may be specified for the particular Performance Requirement failure specified in RFP Appendix B, Sections 23.1 or 23.2. that may give rise to such closure.

8.5. Other Liquidated Damage Provisions

Except as specified in Section 8.4, The damages provided in this Section shall be in addition to and independent of those amounts charged to the Contractor pursuant to other sections of the Contract, including but not limited to those specified in RFP Appendix B, Section B.23 and Section B.24., —The Joint Board and the Contractor hereby agree to the specific standards and Performance Requirements set forth in thise agreement Contract Documents. Contractor further agrees to pay the Liquidated Damages specified in RFP Appendix B, Sections B23.1 and B23.2 for failure to meet the Performance Requirements, and that such Liquidated Damages are a reasonable approximation of the Losses that the Joint Board is likely to incur in the event of such failures.

The Contractor shall not be liable for <code>Liquidated dD</code> amages when, in the opinion of the Joint Board, incidents or delays result directly from causes beyond the control and without the fault or negligence of the Contractor. If the Joint Board elects not to impose <code>Liquidated dD</code> amages in a particular instance said decision shall not be construed as a waiver of the Joint Board's right to pursue future assessments for failure to meet project milestones or <code>pPerformance standardsRequirements</code>, nor construed to limit any additional remedies available to the Joint Board.

8.6. Disputes Relating to Liquidated Damages

There are multiple contractors working for the Joint Board and the Joint Board members both on the toll collection system and to design and construct the <code>BridgesCrossings</code> and roadways, and the Operations Center and Walk-in Centers. In the event that <code>\Liquidated dD</code> amages are assessed against the Contractor and the Contractor believes that these damages should instead be assigned to a different contractor on the project, or the Contractor in some other way disputes an assessment of <code>\Liquidated dD</code> amages, those disputes shall be resolved pursuant to <code>Section 13</code>. The Contractor shall pay the assessed amount pursuant to <code>Section 8—.</code> In the event it is determined, through the dispute resolution procedures, that <code>\Liquidated dD</code> amages were wrongly assessed, the <code>Joint Board shall return the <code>\Liquidated Dd</code> amages payment to the Contractor.</code>

If another contractor is assessed liquidated damages but asserts, through the dispute resolution process that the Contractor should have been assessed these damages instead, the Contractor shall participate in the dispute resolution process. If through that process, it is determined that the Contractor was responsible for the liquidated damages, the Contractor shall immediately pay the assessed damages pursuant to Section 8.

8.7. No Waiver of Additional Damages except as Specifically Stated

The <code>!_Liquidated dD</code> amages described above shall be the Joint Board's agreed compensation for damages incurred for the specified failures to perform. The Joint Board reserves its rights as to other damages for any other breach of the Contract by the Contractor, and specifically for direct damages—and stipulated damages. Consequential damages, however, shall be limited in the manner discussed in Sections 13.5. In the event that <code>!Liquidated dD</code> amages are disallowed for any reason, the Joint Board shall be entitled to actual damages including any and all consequential or incidental damages.

8.8. Stipulated Damage[Reserved.s]

8.8.1. Stipulated Damages Generally

Contractor understands and agrees that if the Performance Requirements are not met at any time after Substantial Completion Acceptance of a Crossing (including the Downtown Crossing Temporary Configuration) or the System, the Joint Board will suffer substantial Losses. Contractor agrees that it shall be liable for all such Losses to the extent set forth in this Section 8.8 and elsewhere in this Contract. Contractor and the Joint Board have agreed to stipulate to a process to determine the amount of damages payable for such failure. Contractor acknowledges and agrees that such Stipulated Damages are intended to compensate the Joint Board solely for its damages caused by the failure to meet the Performance Requirements, and shall not excuse Contractor from liability to correct any defects in the Project or from any other breach of requirements of the Contract Documents. The Joint Board agrees to accept Stipulated Damages as its sole compensation for damages caused by such failure, provided that the Joint Board shall not be precluded from exercising its other rights and remedies respecting such failure, including requiring Contractor to make adjustments to the Project that will cause it to meet the Performance Requirements after the Joint Board's written notice to Contractor of the failure.

8.8.2. Amount of Stipulated Damages

Contractor acknowledges and agrees that because of the unique nature of the Project, and the fact that performance of the System in conformance with the Performance Requirements is essential to Joint Board's ability to collect toll revenue to enable it to continue to finance, construct, operate and maintain and improve the Kentucky and Indiana highway systems, it is not possible to ascertain and determine the actual Losses, including lost toll revenues, which would accrue to Joint Board and the public from such failure. Therefore, commencing upon each Substantial Completion

Acceptance Date, Contractor shall pay to Joint Board an amount determined as follows (the "Stipulated Damages") as deemed compensation to Joint Board for lost toll revenues.

(a) With respect to any failure to meet the Performance Requirements set forth in the Technical Provisions for any Crossing or the Downtown Crossing Temporary Configuration, including those set forth in Section of the Technical Provisions, the parties hereby agree that the Stipulated Damages payable for lost toll revenues shall be calculated based upon a comparison of the number of transactions identified by the System during the period of failure to meet the Performance Requirements and the number of transactions identified by the System during a comparable prior period determined by the Joint Board; provided, however, that if the Crossing does not have at least 6 months of operating history from which to compare, Stipulated Damages shall be determined by reference to the traffic and revenue study estimates for the period of time in question. For example, if, during the period in which the System fails to meet the Performance Requirements, the System identifies 100 transactions, and during a comparable prior period the System identifies 1000 transactions, the Stipulated Damages for the failure to meet the Performance Requirements for the period in question would be the value of the difference in the number of transactions, i.e., an amount equal to 900 multiplied by the toll rate(s) applicable to such transactions.

(b) If during any calendar month, the mean time to respond to a maintenance call or the mean time to complete repairs exceeds the time stated in Scope of Work Section, then Contractor shall pay Joint Board \$____ dollars per hour on the Downtown Crossing and \$____ per hour on the East End Crossing. Expressed as a formula, Stipulated Damages for Contractor's failure to meet the required mean time to respond to a maintenance call or to complete repairs with the required mean time to complete repairs shall equal:

[(TTR RTTR)NRI x \$___] = damages for failure to meet required mean time to respond/repair

where:

TTR = Total Time to Respond to Maintenance Calls/Perform Repairs;

RTTR = Required Mean Time to Respond to Maintenance Calls/Perform Repairs;

NRI = Number of Response Incidents (with each component, lane, or subsystem affected counting as one incident).

Contractor may be liable for Stipulated Damages as a result of either or both of the foregoing calculations. In no event shall the foregoing calculations ever be a negative number or result in Contractor being entitled to additional payment from the Joint Board (e.g., as a result of a faster repair then required). Contractor understands and agrees that any Stipulated Damages payable in accordance with this <u>Section 8.8.2</u> are in the nature of stipulated damages and not a penalty and that the methodology for determining such sums was established based on the parties' agreement that the

amounts so determined will constitute a reasonable approximation of the actual damages from lost toll revenues that the Joint Board will accrue as a result of the circumstances giving rise to such Stipulated Damages. Contractor further acknowledges and agrees that Stipulated Damages may be owing even though no Event of Default has occurred.

8.9. Payment; Offset; Reduction; Waiver

When Liquidated Damages or Stipulated Damages are assessed pursuant to any section of this Contract, regardless of any pending disputes regarding the assessment, the Contractor, at the written direction of the Joint Board, shall pay the indicated amounts within 15 days after Contractor's receipt of an invoice therefor from the Joint Board. Amounts due the Joint Board as Liquidated Damages or Stipulated Damages, if not paid within 15 days of notification of assessment, may be deducted by the Joint Board from any money payable to the Contractor pursuant to this agreement, and/or by tendering or drawing upon the Performance Bond, Warranty Bond or Maintenance Performance Bond provided pursuant to Section 12.2. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction or draw.

Permitting or requiring Contractor to continue and finish the work or any part thereof after a Completion Deadline, notwithstanding any failure to meet the Performance Requirements, or after a submission deadline, as applicable, shall not act as a waiver of the Joint Board's right to receive Liquidated Damages, or Delay Reporting Liquidated Damages or Stipulated Damages—hereunder or any rights or remedies otherwise available to the Joint Board.

8.10. No Waiver of Additional Damages Except as Specifically Stated

The Liquidated Damages and Stipulated Damages described above shall be the Joint Board's agreed compensation for damages incurred if the Contractor breaches the Contract in the particular ways listed. The Joint Board reserves its rights as to other damages for any other breach of this Contract by the Contractor, and specifically for direct damages.

8.11. Limitation of Contractor's Liability

To the extent permitted by applicable Law, the Joint Board will not seek to recover damages (including actual, indirect, special, consequential, multiple or punitive damages) from Contractor resulting from breach of this Agreement (whether arising in contract, negligence or other tort, or any other theory of law) in excess of the sum of (a) all those costs reasonably incurred by the Joint Board or any party acting on the Joint Board's behalf in completing or correcting the Work or having the Work completed or corrected by another Person, plus (b) an amount equal to \$ (which amount shall specifically include any Liquidated Damages and Delay Reporting Damages paid pursuant to this Section 8); plus (c) any amounts paid by Contractor which are covered by insurance proceeds; plus (d) all Losses incurred by any Indemnified Party relating to or arising out of any illegal activities, fraud, criminal

conduct, gross negligence or willful misconduct on the part of any Contractor-Related Entity; and plus (e) work required or arising under the Warranties; plus (e) Contractor's indemnification obligations under Section 11.

- 9.0 Completion and Acceptance
- 9.1. Substantial Completion
- 9.1.1. General Requirements

9.1.1.1. Operations Center and Walk-in Center Substantial Completion

Substantial Completion of the Operations Center and/or a Walk-in Center is the stage in the progress of the Work at the Operations Center or Walk-in Center when (i) the Work related to the Operations Center or the Walk-in Center is sufficiently complete that all elements may be used for the intended purpose, (ii) Substantial Completion testing of the Work is successfully completed, (iii) no further Work remains in order to permit the Operations Center or the Walk-in Center to open to the public for business, (iv) and the System installed at the Operations Center or Walk-in Center functions accurately and reliably.

9.1.1.2. Crossing and System Substantial Completion

Substantial Completion of a Crossing is the stage in the progress of the Work on a Crossing when the Project is sufficiently complete in accordance with the Contract Documents so that the Joint Board can occupy or use all elements of the Project as it pertains to such Crossing for its intended use, including commencing tolling of the Crossing, and the stage of Work on the System where the System can be operated continuously in compliance with the Performance Requirements without interruption. For this Project, Substantial Completion shall be deemed to occur with respect to a Crossing, and such use possible when (i) all required Substantial Completion testing of the Work for such Crossing is successfully completed, (ii) the Work related to such Crossing is completed, (iii) no further Work remains to be performed by the Contractor in order that the Project Site for such Crossing safely may be opened to public and private vehicular traffic, and (iv) such Project Work for such Crossing functions accurately and reliably in accordance with the Performance Requirements. Substantial Completion of the System shall be deemed to occur when Substantial Completion Acceptance has occurred for all of the Crossings, and the entire System may be operated as an integrated System on a continuous basis without interruption in compliance with the Performance Requirements.

9.1.2. Notice and Inspection

9.1.2.1. Initial Notice

The Contractor shall include on its schedule the Substantial Completion Deadline for the Operations Center, each Walk-in Center, each Crossing and the System, and shall update the schedule periodically to indicate the date that Contractor anticipates that each such Substantial Completion for each Crossing and the System will be achieved. Approximately 30 days from the each date when the Contractor anticipates that the wWork will be substantially complete for the Operations Center, each Walk-in Center, each Crossing (or interim configuration)part of the Downtown Crossing, including the Downtown Crossing Temporary Configuration, approximately 30__ days from the date when Contractor anticipates that the work will be substantially complete for the East End Crossing, and approximately 30__ days from the date when Contractor anticipates that the work for the System will be substantially complete, the Contractor shall prepare and submit to the Joint Board a comprehensive list of items to be completed or corrected in order to achieve Substantial Completion for the relevant Crossing or the System, portion of the Work, as appropriate. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

Upon receipt of Contractor's list, the Joint Board shall make a preliminary inspection to determine items to be completed or corrected in order to achieve Substantial Completion. The Joint Board shall provide to the Contractor within 10— days following the date Contractor's list is presented, the Joint Board's list of items that must be completed or corrected in order to achieve Substantial Completion.

9.1.2.2. Notice of Substantial Completion

Contractor shall provide written notice to the Joint Board when all of the following have occurred with respect to the Operations Center, a Walk-in Center, or a Crossing:

- (a) Contractor has completed all Work except for Punch List items, final cleanup and other items included in the requirements for Formal Acceptance;
- (b) Contractor has ensured that all such Work has been performed in accordance with the requirements of the Contract Documents, the Governmental Approvals and applicable Law;
- (c) The Project may be operated without damage to the System or any other property on or off the Project Site, and without injury to any Person;
- (d) The Work has been installed such that the The Project (i) can be safely opened to public use, (ii) is fully signed and striped, and (iii) has all safety appurtenances installed;
- (e) All remaining Punch List work can be completed with no impact to other contractors and traffic. If any lane closures are required to complete the Punch List items, Contractor shall be only entitled to close lanes between ___ p.m. and ___ a.m. and through traffic will be maintained by Contractor at all times;

- (f) Contractor has furnished to the Joint Board (i) the necessary preliminary as-built drawings associated with completed Work, (ii) the as-built drawings and documents required for conducting inspections and/or testing, and (iii) the applicable as-built drawings and documents for maintenance and operations;
- (g) Contractor has furnished to the Joint Board a certification from Contractor's Quality Manager, in form and substance satisfactory to the Joint Board, certifying conformity of the construction with the Design Documents;
- (h) Contractor has acquired all spare parts, components and equipment as described in Scope of Work RFP Appendix B Section ____B.2.2 and [and ---?]; and
- (i) Contractor has completed all inspection, testing and verification as described in Scope of Work <u>Sections</u> ____, including any additional inspection and testing following any repairs, modifications and/or replacements required thereunder and has provided the Joint Board with all documentation and other items required by Scope of Work <u>Appendix B</u>, <u>Sections</u> ____ and ___.

The Joint Board will then conduct such inspections, verifications, surveys and/or testing as it deems desirable. If such inspections, verifications, surveys and/or tests disclose that any of work does not meet the requirements of the Contract Documents, the Joint Board will promptly advise Contractor as to any Errors in the Work necessary to be corrected as a condition to Substantial Completion Acceptance for the particular Operations Center, Walk-in Center, Crossing or the System, and as to any Errors which may be corrected as Punch List items. Upon correction of the Errors identified as a prerequisite to Substantial Completion Acceptance, Contractor shall provide written notification to the Joint Board, and the Joint Board will conduct additional inspections, verifications, surveys and/or testing as it deems desirable. This procedure shall be repeated until the Joint Board finds that all prerequisites to Substantial Completion Acceptance of the Project for the relevant Crossing or the System have been met.

9.1.2.3. Certificate of Substantial Completion

After providing a notice of Substantial Completion for a Crossing or the System to the Joint Board, and after the Joint Board inspects the Work, the Contractor shall prepare for the Joint Board's signature a Certificate of Substantial Completion with respect to the relevant Crossing or the System which, when signed by the Joint Board, shall establish (1) the date of Substantial Completion of the Work for such Crossing or System, (2) responsibilities between the Joint Board and the Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and (3) the time within which the Contractor shall finish all items necessary for final completion on the Punch List accompanying the acknowledgement. If the Joint Board determines that the Work is not Substantially Complete, the Joint Board shall notify the Contractor in writing of those items necessary to achieve Substantial Completion Acceptance. When the Joint Board determines that (a) all conditions set forth in Section 9.1.2.2 have been satisfied; (b) all Errors identified as prerequisites to Substantial Completion Acceptance have been corrected; (c) all inspection and testing as described in Scope of Work Section _____ have

been completed; (d) the Contractor and the Joint Board have agreed upon a Punch List; and (e) any other conditions precedent to Substantial Completion Acceptance expressly set forth in this Contract have occurred, the Joint Board shall sign the Certificate of Substantial Completion for the relevant Crossing or System. Warranties required by the Contract Documents shall commence for each <u>portion of the Work Crossing</u> on the Substantial Completion Acceptance Date for such <u>Crossingportion of the Work</u>, and on the System Substantial Completion Acceptance Date, for components of the System that are not already under warranty as part of <u>a Crossing,another Substantial Completion Acceptance</u>, unless otherwise provided in the Certificate of Substantial Completion.

In the event of a Dispute regarding whether the Contractor's <u>wW</u>ork is substantially complete, the Dispute shall be resolved pursuant to <u>Section 13</u>.

9.2. Punch List Acceptance

9.2.1. Requirements

Contractor shall provide written notice to the Joint Board when all of the items of work listed on the Punch List for a Crossing or for the System, as applicable, have been completed.

9.2.2. Certificate of Punch List Acceptance

The Joint Board will issue a Certificate of Punch List Acceptance for the Operations Center, a Walk-in Center, a Crossing or for the System as soon as reasonably practicable after: (a) Contractor has delivered to the Joint Board the relevant written notice described in Section 9.2.1 for such Crossing or the System; (b) all inspection and testing of Punch List www. ork as described in Scope of Work RFP Appendix B Section shall have been completed; and (c) the Joint Board finds that all Errors identified as prerequisites to Punch List Acceptance have been corrected.

9.3. Formal Acceptance

Promptly after Substantial Completion <u>Acceptance</u> of <u>the Operations Center</u>, a Walk-in <u>Center</u>, a Crossing and of the System, Contractor shall perform all of the work, if any, which was deferred in connection with the Substantial Completion of such Crossing or System, and shall satisfy all of its other obligations under the Contract Documents, the Governmental Approvals and applicable Laws, including ensuring that all equipment, materials, facilities, improvements, structures and components have been properly adjusted and tested. Contractor shall proceed with the completion of punch list items and other items such as striping, seeding, tree planting and any remaining work necessary for final completion and Formal Acceptance. The Joint Board will issue a Certificate of Formal Acceptance as soon as reasonably practicable after it finds that all of the following conditions be been satisfied:

(a) The Joint Board has issued a Certificate of Substantial Completion and a Certificate of Punch List Acceptance for the relevant Operations Center, Walk-in Center, Crossing or the System;

- (b) The Joint Board has received all Design Documents, As-Built Documents, Final ROW record maps, surveys, maintenance manuals, electronic files, test data, Software licenses, and other deliverables required under the Contract Documents and a Source Code Escrow has been established for any Software Source Code, executables and other documents that isare required to be deposited into a Source Code Escrow pursuant to Section 14.4.4 and such Source Code has been deposited;
- (c) All special tools, equipment, furnishings, Hardware, Software and supplies purchased and/or used by Contractor as provided in the Contract Documents (provided, however, that prior to expiration of this Contract, Contractor may retain possession of any such items that may be used in connection with the Maintenance Work) shall have been delivered to the Joint Board free and clear of Liens;
- (d) All personnel, construction supplies, equipment, waste materials, rubbish and temporary facilities of each Contractor-Related Entity shall have been removed from the- Project Site, Contractor shall restore and repair all damage or injury to public or private property arising from the Work with respect to the relevant Operations Center, Walk-in Center, or Crossing and/or such removal to the satisfaction of the Joint Board, the Project Site shall be in good working order and condition; and Contractor shall leave all space under structures unobstructed and in such condition that drift shall not collect and induce scouring;
- (e) Contractor shall have delivered to the Joint Board a certification representing that there are no outstanding claims of Contractor or claims, Liens or stop notices of any Subcontractor, laborer, Utility Owner or railroads with respect to the Work on such Operations Center, Walk-in Center, Crossing or the System, other than any previously submitted unresolved claims of Contractor and any claims, Liens or stop notices of a Subcontractor, laborer, Utility Owner or railroad being contested by Contractor (in which event the certification shall include a list of all such matters with such detail as is requested by the Joint Board and, with respect to all Subcontractor, laborer, Utility Owner and railroad claims, Liens and stop notices, shall include a representation by Contractor that it is diligently and in good faith contesting such matters by appropriate legal proceedings which shall operate to prevent the enforcement or collection of the same). For purposes of such certificate, the term "claim" shall include all matters or facts which may give rise to a claim;
- (f) Contractor shall have satisfactorily completed all additional inspection and testing following any repairs, modifications and/or replacements required;
- (g) The Punch List items for the relevant <u>Operations Center, Walk-in Center,</u> Crossing or the System shall have been completed to the satisfaction of the Joint Board, all of Contractor's other obligations under the Contract Documents, the Governmental Approvals and applicable Laws, other than obligations which by their nature are required to be performed after such Final Acceptance, shall have been satisfied in full or waived in writing by the Joint Board;

- (h) Contractor shall have completed the items set forth as conditions to Formal Acceptance of the Operations Center, Walk-in Center, Crossing or the System in the Scope of Work; and
- (i) Contractor shall have finalized and closed out all Governmental Approvals for the Operations Center, Walk-in Center, Crossing or the System, as applicable.
- (j) <u>For System Formal Acceptance</u>, <u>Tthe System shall have operated continuously</u> on a Project-wide basis for 90 consecutive days without interruption at levels meeting the Performance Requirements.

Formal Acceptance will not prevent the Joint Board from correcting any measurement, estimate, or certificate made before or after completion of the Work nor shall it prevent the Joint Board from recovering from Contractor, its Surety(ies) or other provider of performance security or any combination of the foregoing, overpayment sustained for failure of Contractor to fulfill the obligations under the Contract Documents. A waiver on the part of the Joint Board of any breach of any part of the Contract by Contractor shall not be held to be a waiver of any other or subsequent breach. The occurrence of Formal Acceptance of the Operations Center, a Walk-in Center, a Crossing or the System shall not relieve Contractor of any of its continuing obligations under the Contract Documents, including Warranty and maintenance obligations, or constitute any assumption of liability by the Joint Board.

9.4. End of Maintenance Term Acceptance

9.4.1. Requirements

In connection with the expiration of the Maintenance Term, Contractor shall provide written notice to the Joint Board when all of the following have occurred:

- (a) Contractor has supplied and the Joint Board has accepted revised As-Built Documents reflecting any and all changes and modifications that may have occurred during the Maintenance Term;
- (b) Contractor has supplied and the Joint Board has approved and accepted the complement of spare parts and tools required to properly maintain the System;
- (c) Contractor has supplied and the Joint Board has approved and accepted maintenance manuals revised as necessary to reflect any and all changes and modifications that may have been made to the System since the manuals were originally provided;
- (d) Contractor has supplied and the Joint Board has approved and accepted completed Maintenance Records and Maintenance Reports;
- (e) All punch list items relating to the end of the Maintenance Term have been corrected by Contractor and approved and accepted by the Joint Board;

- (f) A System Performance Audit –has been successfully completed to the Joint Board's satisfaction;
- (g) If not already located on the relevant <u>State's States'</u> property and if requested by the Joint Board, Contractor has delivered and installed, and the Joint Board has approved and accepted, all MOMS Software and Hardware at a location designated by the Joint Board; and
- (h) Contractor has trained the Joint Board's designated maintenance staff to the Joint Board's satisfaction.

9.4.2. Certificate of End of Maintenance Term Acceptance

The Joint Board will issue a Certificate of End of Maintenance Term Acceptance as soon as reasonably practicable after: (a) the Joint Board finds that all conditions set forth in Section 9.4.1 have been satisfied; (b) the Joint Board finds that all Errors identified as prerequisites to End of Maintenance Term Acceptance have been corrected; (c) all inspection and testing as described in the Scope of Work shall have been completed; and (d) any other conditions precedent to End of Maintenance Term expressly set forth in this Contract have occurred.

9.5. Assignment of Causes of Action

Contractor shall assign to the Joint Board all right, title and interest in and to all claims and causes of action it may have under <u>Section 4</u> of the Clayton Act (15 U.S.C. Sec. 15), arising from purchases of goods, services, equipment, Hardware, Software or materials pursuant to the Contract Documents or any Subcontract. This assignment shall be made and become effective at the time the Joint Board tenders the applicable Final Payment of the Contract Price with respect to <u>the Operations Center</u>, a Walk-in <u>Center</u>, a Crossing or the System, as appropriate, to Contractor, without further acknowledgment by the Parties.

9.6. Final Completion and Final Payment

The Project shall be considered finally complete when System Formal Acceptance of the entirety of the Project has been made in accordance with <u>Section 9.3</u>. Upon System Formal Acceptance, the Joint Board shall certify final completion and process Contractor's final payment application in accordance with Section 6.

10.0 Suspensions, Termination for Convenience, and Default

10.1. Suspensions

10.1.1. Suspensions for Convenience

The Joint Board may, at any time and for any reason, by written notice, order the Contractor to suspend all or any part of the work required under the Contract Documents for the period of time that the Joint Board deems appropriate for the convenience of the Joint Board. The Contractor shall comply with any such written

suspension order. The Contractor shall immediately recommence the work upon receipt of written notice from the Joint Board directing the Contractor to resume work. Contractor's sole remedy for such suspensions cumulatively totaling 30 days or less, shall be an extension of the Substantial Completion Deadline, but only to the extent that the suspensions delay the Critical Path and the Contractor has properly notified the Joint Board in writing. Adjustments of the Contract Price and the Substantial Completion Deadlines shall be available for any such suspensions cumulatively totaling more than 1530 days, but only to the extent that such suspensions delay the Critical Path, the Contractor has properly notified the Joint Board in writing as required by Section 4.4.2.1 and such suspensions are not caused or attributable to the acts, omissions, negligence or misconduct of Contractor or any Contractor-Related Entity. Any adjustment to the Contract Price shall be limited to actual direct costs reasonably incurred and not any delay and disruption damages.

The Contractor will not be entitled to an adjustment in the Contract Price or Substantial Completion Deadline to the extent that the performance would have been suspended or delayed by any other cause, or for which an adjustment is provided under any other term of this Contract.

10.1.2. Suspensions for Cause

The Joint Board has the authority to suspend the work by written order, wholly or in part, for Contractor's failure to:

- 1. Correct conditions unsafe for the Project personnel or the general public;
- 2. Comply with any Governmental Approval, governmental rule or otherwise carry out the requirements of the Contract Documents;
 - 3. Carry out appropriate orders of the Joint Board;
 - 4. Supply sufficient skilled workers or suitable materials or equipment;
- 5. Furnish or perform the work in such as way that the completed work will conform to the Contract Documents;
- 6. Comply with requirements for developing and implementing the designs and plans required by the Contract Documents;
- 7. Remove and replace unsuitable personnel as reasonably directed by the Joint Board;
- 8. Provide proof of required insurance coverage and performance security; or
 - 9. Comply with environmental requirements.

The Contractor shall comply with any such written suspension order. The Contractor shall recommence the work upon receipt of written notice from the Joint Board directing the Contractor to resume the work. Except as provided in Section 10.1.3 below, Tthe Contractor shall not be entitled to an adjustment of the Contract Price or Contract Time for suspensions for cause. However, if the Joint Board orders suspension of work on one of the foregoing grounds but it is finally determined under the dispute resolution

procedures of this Contract that such grounds did not exist, it shall be treated as a suspension for the Joint Board's convenience under <u>Section 10.1.1</u>.

The Joint Board's right to suspend the work will not give rise to any duty on the part of the Joint Board to exercise this right for the benefit of the Contractor or any other party.

10.1.3. Responsibilities of the Contractor during Suspension Periods

During periods that work is suspended, the Contractor shall continue to be responsible for the work and shall prevent damage or injury to the Project, provide for drainage and shall erect necessary temporary structures, signs or other facilities required to maintain the Project. Additionally, the Contractor shall continue other work that has been or can be performed on-Site or off-Site during the period that work is suspended. The Contractor shall be entitled to reimbursement for the cost of any temporary structures, signs or other physical facilities required to maintain the Project if such suspension is for the Joint Board's convenience.

10.2. Termination for Convenience

10.2.1. Contingent on Funding

This Contract is contingent upon the continued availability of appropriated funding. If the funding for the Project becomes unavailable for any reason, including the Indiana or Kentucky General Assembly's failure to appropriate the funding, by operation of law or as a result of a reduction in Federal funding, this Contract may be terminated, the Project may be cancelled, the timeline may be extended or the scope of the Project may be amended by the Joint Board, either in whole or in part. Project cancellation, extension, or amendment because of an interruption in the appropriated funding shall not be a default or breach of this Contract by the Joint Board nor may such cancellation, extension, or amendment give rise to any claim against the Joint Board except for payment as provided in this Section 10.

10.2.2. No Obligation to Issue NTP

The Contractor acknowledges and agrees that notwithstanding the date for issuance of the Notice to Proceed specified in <u>Section 3.2</u> hereof, the Joint Board has no obligation to issue a Notice to Proceed hereunder, and further agrees that unless and until a Notice to Proceed is issued, the Joint Board shall have no liability to the Contractor hereunder except any non-refundable portions of insurance or payment and performance bond premiums the Contractor was required to provide through the date of termination.

10.2.3. Notice of Termination or Partial Termination

The Joint Board may, at any time, terminate this Contract and the performance of the work by the Contractor in whole or in part, if the Joint Board determines, in its sole discretion, that a termination is in the Joint Board's best interest, regardless of the availabileity of funding. The Contractor waives any claim that a termination must be made in "good faith", or in the alternative, to the extent that "good faith" is required,

acknowledges that any termination made in the Joint Board's general interest is in good faith.

The Joint Board shall terminate by delivering to Contractor a written Notice of Termination for Convenience or Notice of Partial Termination for Convenience specifying the extent of termination and its effective date. Termination (or partial termination) of this Contract shall not relieve any surety of its obligation for any claims arising out of the work performed.

10.2.4. Transition Plan

Within three days after receipt of a Notice of Termination for Convenience or Notice of Partial Termination for Convenience, the Contractor shall meet and confer with the Joint Board for the purpose of developing an interim transition plan for the orderly transition of the terminated work, demobilization and transfer of the Project to the Joint Board. The Parties shall use diligent efforts to complete preparation of the interim transition plan within 15 days after the date the Contractor receives such notice of termination. The Parties shall use diligent efforts to complete a final transition plan within 30 days after such date. The transition plan shall be in form and substance acceptable to the Joint Board in its good faith discretion and shall include and be consistent with the other provisions and procedures set forth herein, all of which provisions and procedures the Contractor shall immediately follow, regardless of any delay in preparation or acceptance of the transition plan.

10.2.5. Contractor's Responsibilities after Receipt of Notice of Termination

After receipt of a Notice of Termination for Convenience or Notice of Partial Termination for Convenience, and except as otherwise directed by the Joint Board, the Contractor shall immediately proceed with the following obligations, regardless of any delay in determining or adjusting any amounts due under this <u>Section 10</u>:

- 1. Stop work as specified in the notice;
- 2. Notify all affected Subcontractors that this Contract is being terminated and that their subcontracts (including orders for materials, services or facilities) are not to be further performed unless otherwise authorized in writing by the Joint Board;
- 3. Place no further subcontracts or orders for materials, services or facilities except as necessary to complete the continued portion of the work, if any, or for mitigation of damages;
- 4. Unless instructed otherwise by the Joint Board, terminate all subcontracts to the extent they relate to the work terminated;
- 5. To the extent directed by the Joint Board, execute and deliver to the Joint Board written assignments, in form and substance reasonably acceptable to the Joint Board, of all of Contractor's right, title, and interest in and to: (a) Subcontracts that relate to the terminated work, provided the Joint Board assumes in writing all of Contractor's obligations thereunder that arise after the effective date of the termination, and (b) all assignable warranties, claims and causes of action held by the Contractor against

Subcontractors and other third parties in connection with the terminated work, to the extent such work is adversely affected by any Subcontractor or other third party breach of warranty, contract or other legal obligation;

- 6. Subject to the prior written approval of the Joint Board, settle all outstanding liabilities and all termination settlement proposals arising from termination of subcontracts;
- No later than 30 days from the effective date of termination, unless extended in writing by the Joint Board upon written request of the Contractor within this 30-day period, provide the Joint Board with an inventory list of all materials, supplies, equipment, Hardware and Software previously produced, purchased or ordered from Suppliers for use in the work and not yet used in the work, including its storage location, as well as any documentation or other property required to be delivered hereunder which is either in the process of development or previously completed but not yet delivered to the Joint Board, and such other information as the Joint Board may request; and transfer title (or, with respect to any third party licensed Software, assign to the Joint Board all of Contractor's and any Contractor-Related Entity's license to such Software, or obtain a direct license in the name of the Joint Board for such software on the same terms) and deliver to the Joint Board through bills of sale or other documents of title, assignment or license, as applicable, as directed by the Joint Board, (a) the work in process, completed work, supplies, equipment, any Hardware, Software and other material produced or acquired for the work terminated, and (b) the design documents, construction documents and all other completed or partially completed drawings (including plans, elevations, sections, details and diagrams), specifications, records, samples, information and other property that would have been required to be furnished to the Joint Board if the work had been completed;
- 8. Complete performance in accordance with the Contract Documents of all work not terminated;
- 9. Take all action that may be necessary, or that the Joint Board may direct, for the safety, protection and preservation of (a) the public, including public and private vehicular movement, (b) the work and (c) equipment, machinery, materials, Hardware, Software and property related to the Project that is in the possession of the Contractor and in which the Joint Board has or may acquire an interest;
- 10. As authorized by the Joint Board in writing, use its best efforts to sell at reasonable prices any property of the types referred to in Section 10.2.5(7); provided, however, that the Contractor (a) is not required to extend credit to any purchaser, and (b) may acquire the property under the conditions prescribed and at prices accepted by the Joint Board. The proceeds of any transfer or disposition will be applied to reduce any payments to be made by the Joint Board under the Contract Documents or paid in any other manner directed by the Joint Board;
- 11. If requested by the Joint Board, withdraw from the portions of the Project Site designated by the Joint Board and remove such materials, equipment, tools and instruments used by, and any debris or waste materials generated by, the Contractor and any Subcontractor in the performance of the work as the Joint Board may direct; and

12. Take other reasonable actions directed by the Joint Board.

10.2.6. Responsibility after Notice of Termination

The Contractor shall continue to be responsible for damage to materials after issuance of the Notice of Termination for Convenience, except as follows:

- 1. Contractor's responsibility for damage to materials for which partial payment has been made as provided herein shall terminate when the Joint Board Representatives certify that those materials have been stored in the manner and at the locations directed by the Joint Board. The Joint Board shall make its assessment of whether the materials have been stored in the manner and at the locations as directed by the Joint Board, and provide such certification or notice of the reason that it will not make such certification, within 30 days of receipt of notice from the Contractor that the Contractor has stored the materials in the manner and at the locations directed by the Joint Board.
- 2. Contractor's responsibility for damage to materials purchased by the Joint Board subsequent to the issuance of the notice that this Contract is to be terminated shall terminate when title and delivery of those materials has been taken by the Joint Board.

10.2.7. Acceptance of Completed Work by the Joint Board

When the Joint Board Representative determines that the Contractor has completed the work directed to be completed prior to termination and such other work as may have been ordered to secure the Project for termination, the Joint Board Representatives will recommend that the Joint Board formally accept such work, and immediately upon and after the acceptance by the Joint Board, the Contractor will not be required to perform any further work thereon and shall be relieved of the contractual responsibilities for injury to persons or property which occurs after the formal acceptance of such work by the Joint Board.

10.2.8. Settlement Proposal

After receipt of a notice of termination for convenience or notice of partial termination for convenience, the Contractor shall submit a final termination settlement proposal to the Joint Board in the form and with the certification prescribed by the Joint Board. The Contractor shall submit the proposal no later than 120 days from the effective date of termination unless the Contractor has requested a time extension in writing within such 120-day period and the Joint Board has agreed in writing to allow such an extension. Contractor's termination settlement proposal shall then be reviewed by the Joint Board and acted upon, returned with comments, or rejected. If the Contractor fails to submit the proposal within the time allowed, the Joint Board may determine, on the basis of information available, the amount, if any, due the Contractor because of the termination and shall pay the Contractor the amount so determined and Contractor shall be bound by the Joint Board's determination.

10.2.9. Amount of Negotiated Termination Settlement

The Contractor and the Joint Board may agree, as provided in <u>Section 10.2.8</u>, upon the whole or any part of the amount or amounts to be paid to the Contractor by reason of the total or partial termination of work for convenience pursuant to this <u>Section 10.2</u>.

Such negotiated settlement shall include a reasonable allowance for profit solely on work which has been completed as of the termination date and subsequently inspected and accepted by the Joint Board. The Contractor shall not be entitled to overhead or profit on work not performed. Such agreed amount or amounts, exclusive of settlement costs, shall not exceed the total Contract Price as reduced by the amount of payments otherwise made and the Contract Price of work not terminated. Upon determination of the settlement amount this Contract will be amended accordingly, and the Contractor will be paid the agreed amount as described in this Section 10.2.9. Nothing in Section 10.2.10, prescribing the amount to be paid to the Contractor in the event that the Contractor and the Joint Board fail to agree upon the whole amount to be paid to the Contractor by reason of the termination of work pursuant to this Section 10.2, shall be deemed to limit, restrict or otherwise determine or affect the amount or amounts which may be agreed upon to be paid to the Contractor pursuant to this Section 10.2.9. The Joint Board's execution and delivery of any settlement agreement shall not affect any of its rights under the Contract Documents with respect to completed work, relieve the Contractor from its obligations with respect thereto, including warranties, or affect the rights of the Joint Board or the Contractor's obligations under the performance bond and payment bond as to such completed or nonterminated work.

10.2.10. Determination of Settlement Amount if Negotiations Fail

If the Contractor and the Joint Board fail to agree upon the whole amount to be paid the Contractor by reason of the termination of work for convenience pursuant to this <u>Section 10</u>, the amount payable (exclusive of interest charges) shall be determined by the Joint Board in accordance with the following, but without duplication of any items or of any amounts agreed upon in accordance with Sections 10.2.8 and 10.2.9:

- 1. The Joint Board will pay the Contractor the sum of the following amounts for work performed prior to the <u>effective</u> date of the <u>Notice of Ttermination</u> for <u>Cconvenience or Notice of Ppartial tTermination for Cconvenience as set forth in the Notice of Termination for Convenience or Notice of Partial Termination for Convenience, as the case may be:</u>
- (a) All amounts due for work completed by the Contractor and accepted by the Joint Board on the basis of agreed prices for the activities completed as defined in the sSchedule of yValues approved for the Project, plus reasonable costs and jobsite overhead for and accounting for any refunds payable with respect to insurance premiums, deposits or similar items, as established to the Joint Board's satisfaction. In determining the reasonable cost, deductions will be made for the cost of materials to be retained by the Contractor, amounts realized by the sale of materials, and for other appropriate credits against the cost of the work.

Deductions will also be made, when this Contract is terminated as the result of a Force Majeure Event, for the cost of materials damaged by the "occurrence." When, in the

opinion of the Joint Board Representative, the cost of a Contract item of work is excessively high due to costs incurred to remedy or replace defective or rejected work, the reasonable cost to be allowed will be the estimated reasonable cost of performing that work in compliance with the requirements of the Contract Documents and the excessive actual cost shall be disallowed. No profit or overhead shall be allowed for unperformed portions of the work.

- (b) The cost of settling and paying claims arising out of the timely termination of work under subcontracts, exclusive of the amounts paid or payable on account of supplies or materials delivered or services furnished by the Subcontractor prior to the effective date of the Notice of termination for Convenience or Notice of Partial Termination for Convenience of work under the Contract, which amounts shall be included in the cost on account of which payment is made under clause (a) above;
- (c) The reasonable out-of-pocket cost (including reasonable jobsite overhead) of the preservation and protection of property incurred pursuant to <u>Section 10.2.9</u> and any other reasonable out-of-pocket cost (including jobsite overhead) incidental to termination of work under this Contract, including the reasonable cost to the Contractor of handling material returned to the vendor, delivered to the Joint Board or otherwise disposed of as directed by the Joint Board, and including a reasonable allowance for Contractor's administrative costs in determining the amount payable due to termination of this Contract.
- 2. The Contractor acknowledges and agrees that it shall not be entitled to any compensation in excess of the value of the work performed (determined as provided in Section 10.2.10(1)) plus its settlement costs, and that items such as lost or anticipated profits, unabsorbed overhead and opportunity costs shall not be recoverable by it upon total or partial termination of this Contract. The total amount to be paid to the Contractor, exclusive of costs described in Sections 10.2.10(1)(b) and (c), may not exceed the total Contract Price less the amount of payments previously made.

Furthermore, in the event that any refund is payable with respect to insurance or bond premiums, deposits or other items which were previously passed through to the Joint Board by the Contractor (and paid by KYTC and IFA as provided herein), such refund shall be paid in equal amounts directly to KYTC and IFA as directed by the Joint Board, the Joint Board or otherwise credited to the Joint Board as directed by the Joint Board. Except for normal spoilage, and except to the extent that the Joint Board will have otherwise expressly assumed the risk of loss, there will be excluded from the amounts payable to the Contractor under Section 10.2.10(1), the fair value, as determined by the Joint Board, of equipment, machinery, materials and property which is destroyed, lost, stolen, or damaged so as to become undeliverable to the Joint Board, or sold pursuant to Section 10.2.5.10. Upon determination of the amount of the termination payment, this Contract shall be amended to reflect the agreed termination payment, the Contractor shall be paid the agreed amount, and the Contract Price shall be reduced to reflect the reduced scope of work.

3. If a termination hereunder is partial, the Contractor may file a proposal with the Joint Board for an equitable adjustment of the Contract Price for the continued

portion of the Contract. Any proposal by the Contractor for an equitable adjustment under this clause shall be requested within 90 days from the effective date of termination unless extended in writing by the Joint Board. The amount of any such adjustment as may be agreed upon shall be set forth in an amendment to this Contract.

- 4. The amount otherwise due the Contractor under this <u>Section 10.2</u> shall be reduced by (a) the amount of any claim which the Joint Board may have against any Contractor-Related Entity in connection with the Contract, (b) the agreed price for, or the proceeds of sale, of property, materials, supplies or other things acquired by the Contractor or sold, pursuant to the provisions of this <u>Section 10.2</u>, and not otherwise recovered by or credited to the Joint Board, (c) all unliquidated advance or other payments made to or on behalf of the Contractor applicable to the terminated portion of the work or Contract, (d) amounts that the Joint Board deems advisable, in its sole discretion, to retain to cover any existing or threatened claims, liens and stop notices relating to the Project, (e) the cost of repairing any Nonconforming Work and (f) any amounts due or payable by the Contractor to the Joint Board.
- 5. The Joint Board may from time to time, under such terms and conditions as it may prescribe and in its sole discretion, make partial payments on account against costs incurred by the Contractor in connection with the terminated portion of this Contract, whenever in the opinion of the Joint Board the aggregate of such payments shall be within the amount to which the Contractor will be entitled hereunder. If the total of such payments is in excess of the amount finally agreed or determined to be due under this <u>Section 10.2</u>, such excess shall be payable by the Contractor to the Joint Board upon demand together with interest as provided in this Contract.
- 6. Provisions shall be included in each subcontract (at all tiers) regarding terminations for convenience, allowing such terminations to be passed through to the Subcontractors and establishing terms and conditions relating thereto, including procedures for determining the amount payable to the Subcontractor upon a termination, consistent with this <u>Section 10.2</u>.
- 7. Each Major Subcontract shall provide that, in the event of a termination for convenience by the Joint Board, the Subcontractor will not be entitled to any anticipatory or unearned profit on work terminated or partly terminated, or to any payment which constitutes consequential damages on account of the termination or partial termination.

10.2.11. No Waiver; Release

Anything contained in the Contract Documents to the contrary notwithstanding, a termination under this <u>Section 10.2</u> shall not waive any right or claim to damages which the Joint Board may have and the Joint Board may pursue any cause of action which it may have by law, in equity or under the Contract Documents.

The Joint Board's payment to the Contractor of the amounts required under this <u>Section 10.2</u> shall constitute full and final satisfaction of, and upon payment the Joint Board shall be forever released and discharged from, any and all claims, causes of action, suits, demands and losses, known or unknown, suspected or unsuspected, that the Contractor may have against the Joint Board arising out of or relating to the terminated

work. Upon such payment, the Contractor shall execute and deliver to the Joint Board all such releases and discharges as the Joint Board may reasonably require to confirm the foregoing, but no such written release and discharge shall be necessary to give effect to the foregoing satisfaction and release.

Under no circumstances shall Contractor be entitled to anticipatory or unearned profits or consequential, special, indirect or other damages as a result of a termination or partial termination under this <u>Section 10</u>. The payment to Contractor determined in accordance with this <u>Section 10</u> constitutes Contractor's exclusive remedy for a termination hereunder.

10.2.12. Dispute Resolution

The failure of the Parties to agree on amounts due under this <u>Section 10.2</u> shall be a dispute to be resolved in accordance with Section 13.

10.3. Default

10.3.1. Events of Default By the Contractor

The Contractor shall be in breach under this Contract upon the occurrence of any one or more of the following events or conditions:

- 1. The Contractor fails to promptly begin the work under the Contract Documents following issuance of a Notice to Proceed authorizing such work;
- 2. The Contractor fails to perform the work with sufficient resources to ensure the prompt completion thereof;
- 3. The Contractor fails to perform the work in accordance with the Contract Documents, refuses to remove and replace rejected materials or Nonconforming Work or unacceptable work, fails to remove and replace workers as directed by the Joint Board, fails to achieve Substantial Completion Acceptance of the Operations Center, a Walk-in Center, a Crossing or the System within at least 30 days following the relevant Substantial Completion Acceptance Deadline, or to achieve Formal Acceptance within at least 30 days following the relevant Formal Acceptance Completion Deadline;
- 4. The Contractor discontinues or suspends, ceases, stops or abandons the prosecution of the work, as a whole or for an individual Crossing, or fails to continuously and diligently prosecute the work as a whole or for the Operations Center, a Walk-in Center, or an individual Crossing (exclusive of work stoppage due to (i) termination by the Joint Board, (ii) compliance with stoppage requirements due to contaminated materials, or (iii) a Force Majeure Event or suspension by the Joint Board;
- 5. The Contractor fails to resume performance of work which has been suspended or stopped, within a reasonable time after receipt of notice from the Joint Board to do so or (if applicable) after cessation of the event preventing performance;
- 6. The Contractor materially fails to timely observe or perform or cause to be observed or performed any other material covenant, agreement, warranty, obligation,

term or condition required to be observed or performed by the Contractor under the Contract Documents;

- 7. The Contractor fails to obtain, provide and maintain in full force and effect any insurance, bonds or guarantees as and when required under this Contract for the benefit of the relevant parties, or fails to comply with any requirement of this Contract pertaining to the amount, terms or coverage of the same;
- 8. The Contractor fails to obtain, provide and maintain in full force and effect the required payment bond and performance bond as and when required under this Contract;
- 9. The Contractor makes or attempts to make or suffers a voluntary or involuntary assignment or transfer of the Contract Documents or any right or interest therein:
- 10. Subject to the Joint Board paying the Contractor pursuant to <u>Section 6</u>, the Contractor fails, absent a valid dispute, to make payment when due for labor, equipment or materials in accordance with its agreements with Subcontractors and applicable Law, fails to comply with any law or Governmental Approvals; or fails reasonably to comply with the instructions of the Joint Board consistent with the Contract Documents;
- 11. The Contractor fails to discharge or obtain a stay within 10 Days of any final judgment(s) or order for the payment of money against it in excess of \$100,000 in the aggregate arising out of the prosecution of the work (provided that, for purposes hereof, posting of a bond in the amount of 125 percent of such judgment or order or the filing of an appeal with respect to such judgment or order shall be deemed an effective stay);
- 12. The Contractor does not comply with public safety and public convenience requirements of this Contract or fails to correct any safety hazards promptly;
- 13. The Contractor or any Contractor-Related Entity shall have become insolvent, generally does not pay its debts as they become due, admits in writing its inability to pay its debts, or makes an assignment for the benefit of creditors;
- 14. Insolvency, receivership, reorganization or bankruptcy proceedings shall have been commenced by or against the Contractor or any major Contractor-related entity and not dismissed within 60 days;
- 15. Any representation or warranty made by the Contractor in the Contract Documents or in any certificate, schedule, instrument or other document delivered as part of the proposal or pursuant to the Contract Documents shall have been false or materially misleading when made or becomes inaccurate in any material respect during the contract term;
 - 16. The Contractor is a party to fraud in any way relating to the Project; or
- 17. Any final judgment is issued holding Contractor or any Guarantor liable for an amount in excess of \$100,000 based on a finding of intentional or reckless misconduct or violation of a state or federal false claims act: or

- 18. voluntary Contractor commences а case seeking liquidation, reorganization or other relief with respect to itself or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect; seeks the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its assets; becomes insolvent, or generally does not pay its debts as they become due; admits in writing its inability to pay its debts; makes an assignment for the benefit of creditors; or takes any action to authorize any of the foregoing; or any of the foregoing acts or events shall occur with respect to any of Contractor's partners, members or joint venturers, any material Subcontractors, or any Surety, Guarantor or letter of credit Bank; or
 - 19. A Persistent Breach occurs; or
- 20. Any Guarantor revokes or attempts to revoke its obligations under its guarantee or otherwise takes the position that such instrument is not longer in full force and effect; or
- 21. The Contractor otherwise materially fails to comply with the terms of the Contract Documents.

10.3.2. Right to Cure

Except for a breach declared under Section 10.1.3(20), the Joint Board agrees to allow the Contractor and surety 15 days written notice and opportunity to cure any breach before declaring an Event of Default, provided that the notice and cure period shall only be three days for a breach under Sections 10.3.1(7), 10.3.1(8) and 10.3.1(12) and that there shall be no such notice and opportunity to cure for any breach which by its nature cannot be cured. If a breach for which a 15-day cure period is provided is curable but by its nature cannot be cured within 15 days, as determined by the Joint Board, the Joint Board agrees not to declare an Event of Default provided that the Contractor commences such cure within such 15-day period and thereafter diligently prosecutes such cure to completion; provided, however, that in no event will such cure period exceed 60 days in total. Notwithstanding the foregoing, if the Joint Board believes a condition affecting the Project poses an immediate and imminent danger to public health or safety, the Joint Board may, without notice and without awaiting lapse of any cure period, rectify the condition at Contractor's cost, and so long as the Joint Board undertakes such action in good faith, even if under a mistaken belief in the occurrence of such default, such action shall not expose the Joint Board to liability to the Contractor, except in the case of the Joint Board's gross negligence, recklessness, willful misconduct or bad faith, and shall not entitle the Contractor to any other remedy, it being acknowledged that the Joint Board has a paramount public interest in providing and maintaining safe public use of and access to the Project. The Joint Board's good faith determination of the existence of such danger shall be deemed conclusive in the absence of clear and convincing evidence to the contrary.

With respect to a breach declared under <u>Sections 10.3.1(19)</u>, the Joint Board may declare an anticipatory breach under <u>Section 10.3.1(2)</u>, (3) or (15), in which case the provisions of <u>Section 10.3.3</u> shall apply.

10.3.3. Rights of the Joint Board

If an Event of Default occurs, then, in addition to all other rights and remedies provided by law or equity or available under this Contract or otherwise, including the rights to recover Liquidated Damages and Stipulated Damages and to seek recourse against the Payment Bond, Performance Bond, Maintenance Payment Bond, Maintenance Performance Bond, Warranty Bond and/or other performance security, the Joint Board shall have the following rights and remedies, without further notice, and without prejudice to any of its other rights or remedies and without waiving or releasing the Contractor from any obligations, and the Contractor shall have the following obligations (as applicable):

- 1. The Joint Board may order the Contractor to suspend or discontinue the work or any portion of the work;
- 2. The Joint Board may terminate this Contract or a portion thereof, including the Contractor's rights of entry upon, possession, control, operation and maintenance of the Project, in which case, the provisions of <u>Section 10.2</u> shall apply;
- 3. If and as directed by the Joint Board, the Contractor shall withdraw from the <u>Project</u> Site; and shall remove such materials, equipment, tools and instruments used by, and any debris or waste materials generated by, any Contractor-Related Entity in the performance of the work;
- 4. The Contractor shall deliver to the Joint Board possession of any or all of Contractor's facilities located on the Project Site as well as any or all design documents, construction documents and all other completed or partially completed drawings (including plans, elevations, sections, details and diagrams), specifications, records, information, schedules, samples, shop drawings, Hardware, Software, electronic files and other documents, and facilities related to the Project that the Joint Board deems necessary for completion of the work;
- 5. If the Joint Board requests the Contractor to assign to the Joint Board specific Subcontracts, the Contractor shall confirm the assignment of those Subcontracts to the Joint Board and shall terminate, at its cost, all other Subcontracts;
- 6. The Joint Board may deduct from any amounts payable by the Joint Board to the Contractor such amounts payable by the Contractor to the Joint Board, including reimbursements owing, Liquidated Damages, Stipulated Damages, an amount the Joint Board deems advisable to cover any existing or threatened claims, Liens and stop notices of Subcontractors, laborers, or Utility Owners against Contractor or against the Joint Board, the amount of any Losses that have accrued, the cost to complete or remediate uncompleted work or Nonconforming Work or other amounts or damages that the Joint Board has determined are or may be payable to the Joint Board under the Contract Documents, and the Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction;
- 7. The Joint Board shall have the right, but not the obligation, to pay such amount and/or perform such act as may then be required;
- 8. The Joint Board, without incurring any liability to the Contractor, shall have the rights (i) to take the performance of all or a portion of the work from the Contractor (either with or without the use of Contractor's materials, equipment, tools and

instruments) and enter into an agreement with another entity for the completion of such work; or (ii) to use such other methods, as in the opinion of the Joint Board, will be required for the completion of the Project; and/or

- 9. If the Joint Board exercises any right to perform any obligations of the Contractor, in the exercise of such right the Joint Board may, but is not obligated to, among other things: (i) perform or attempt to perform, or cause to be performed, such work; (ii) spend such sums as the Joint Board deems necessary and reasonable to employ and pay such architects, engineers, consultants and contractors and obtain materials and equipment, Hardware and Software as may be required for the purpose of completing such work; (iii) execute all applications, certificates and other documents as may be required for completing the work; (iv) modify or terminate any contractual arrangements; (v) take any and all other actions which it may in its sole discretion consider necessary to complete the work; and (vi) prosecute and defend any action or proceeding incident to the work.
- 10. The Joint Board shall also have the right, but not the obligation, to require the Contractor to comply with the obligations in <u>Section 10.2.5</u>.

10.3.4. Liability of the Contractor/Occurrence of an Event of Default

If an Event of Default has occurred, the Contractor and surety and each Guarantor shall be jointly and severally liable to the Joint Board (in addition to any other damages under the Contract Documents other than those costs intended to be covered by Liquidated Damages and Stipulated Damages that have already been paid hereunder) for all costs reasonably incurred by the Joint Board or any party acting on the Joint Board's behalf in completing the work or having the work completed by another entity (including any reprocurement costs, throw away costs for unused portions of the completed work, increased financing costs and increased costs payable to the Design-Build contractor and Developer). The preceding sentence shall expressly include all Warranty Work, Maintenance Work and work to be performed during the Warranty Period and Maintenance Term. Upon the occurrence of an event of default, the Joint Board shall be entitled to withhold all or any portion of further payments to the Contractor until such time as the Joint Board is able to determine how much (if any) remains owing to the Contractor. Promptly upon such determination, the Joint Board shall notify the Contractor in writing of the amount, if any, that the Contractor shall pay the Joint Board or that the Joint Board shall pay the Contractor with respect thereto. All costs and charges incurred by the Joint Board, including attorneys', accountants' and expert witness fees and costs, together with the cost of completing the work under the Contract Documents, will be deducted from any moneys due or which may become due to the Contractor. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction. If such expense exceeds the sum which would have been payable under the Contract, then the Contractor and its surety(ies) shall be liable and shall pay to the Joint Board the amount of such excess. If the Surety or Guarantor fails to pay such amount immediately upon the Joint Board's demand, then the Joint Board shall be entitled to collect interest from the Surety or Guarantor on the amounts the Joint Board is required to pay in excess of the remaining balance of the Contract Price. The interest rate which the Surety and each Guarantor shall pay shall be the lesser of (a) 12% per annum or (b) the maximum rate allowable under applicable

Law. The interest rate shall accrue on all amounts the Joint Board has had to pay excess of the remaining balance of the Contract Price from the date of the Joint Board payment.

10.3.5. Assurance of Future Performance

It is recognized that if a breach or Event of Default occurs, such event could impair or frustrate Contractor's performance of the work.

Accordingly, upon the occurrence of any such event, the Joint Board is entitled to request the Contractor, or its successor in interest, to provide adequate assurance of future performance in accordance with the terms and conditions hereof. Failure to comply with such request within 10 days of the Joint Board's delivery of the request shall entitle the Joint Board to terminate the Contract. Pending receipt of adequate assurance of performance and actual performance in accordance therewith, the Joint Board shall be entitled to proceed with the work with its own forces or with other contractors on a Force Account or other appropriate basis, the cost of which will be credited against and deducted from the Joint Board's payment obligations hereunder. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction. The foregoing shall be in addition to all other rights and remedies provided by law or equity and such rights and remedies as are otherwise available under the Contract, the Payment Bond—, the Performance Bond, the Maintenance Bonds and the Warranty Bonds.

10.3.6. Alternative to Terminating the Contract and Completing the work

In lieu of the provisions of this <u>Section 10.3</u> for terminating this Contract and completing the work, the Joint Board may pay the Contractor for the parts already done according to the provisions of the Contract Documents and may treat the parts remaining undone as if they had never been included or contemplated by this Contract. No claim under this provision will be allowed the Contractor for prospective profits on, or any other compensation relating to, work uncompleted by the Contractor.

10.3.7. Termination Deemed to Constitute a Termination for Convenience

If this Contract is terminated for grounds which are later determined not to justify a termination for default, such termination shall be deemed to constitute a termination for convenience pursuant to Section 10.2.

10.3.8. Damages Resulting From Contractor's Breach or Failure to Perform

If the Joint Board suffers Losses or other damages as a result of Contractor's breach or failure to perform an obligation under the Contract Documents, then the Joint Board shall be entitled to recovery of such Losses and other damages from the Contractor regardless of whether the breach or failure that gives rise to the Losses and other damages ripens into an Event of Default.

10.3.9. Cumulative Remedies

The exercise or beginning of the exercise by the Joint Board of any one or more rights or remedies under this <u>Section 10.3</u> shall not preclude the simultaneous or later exercise by the Joint Board of any or all other rights or remedies, each of which shall be cumulative.

10.3.10. Continued Liability of the Contractor and Surety

The Contractor and surety shall not be relieved of liability for continuing Liquidated Damages or Stipulated Damages on account of a default by the Contractor hereunder or by the Joint Board's declaration of an Event of Default, or by actions taken by the Joint Board under this Section 10.3.

10.3.11. Contractor's Right to Stop Work If Undisputed Payment Is Not Made

The Contractor shall have the right to stop work if the Joint Board fails to make an undisputed payment due under this Contract within <u>4590</u> days after receipt of notice of nonpayment. Any such work stoppage shall be considered a suspension. The Contractor shall not have the right to terminate this Contract for default as the result of any failure by the Joint Board to make an undisputed payment due hereunder, but the Contractor shall have the right to declare a termination for convenience if such nonpayment continues for more than 150 days after the Joint Board's receipt of written notice of nonpayment from Contractor.

10.3.12. Notice and Opportunity to Cure Other Types of the Joint Board Breaches

In the event of any breach of this Contract by the Joint Board other than a failure to make payments to the Contractor, the Contractor shall provide to the Joint Board a written notice describing the breach and the opportunity to cure such breach. The Joint Board shall be entitled to 60 days to cure any such breach from the date it receives the notice; provided that if such breach is capable of cure but by its nature cannot be cured within 60 days, the Joint Board shall have a reasonable period of time as may be reasonably necessary under the circumstances to cure the breach so long as the Joint Board commences such cure within such 60-day period and thereafter diligently prosecutes such cure to completion. The Contractor shall have no right to exercise any remedies to which it may be entitled at law or in equity until the foregoing notice is delivered and the foregoing cure period lapses without cure of the breach.

10.4. Anticipatory Breach

Contractor recognizes and acknowledges that a pattern or practice of continuing, repeated or numerous breaches or failures to perform by Contractor, even if individual instances are not material or are eventually cured, will undermine the confidence and trust essential to the success of this Contract and will have a material, cumulative adverse impact on the value of this Contract to the Joint Board. Contractor acknowledges and agrees that the measures for determining the existence of such a pattern or practice described in the definition of Persistent Breach are a fair and appropriate objective basis to conclude that such a pattern or practice will continue.

Accordingly, in the event the Joint Board issues a notice under <u>Section 10.3.2</u> with respect to a breach under <u>Section 10.3.1(19)</u> for a Persistent Breach, the Joint Board shall have the right to declare an anticipatory breach of this Contract by Contractor. So long as the circumstances under the definition of Persistent Breach have occurred, any such declaration of anticipatory breach shall be valid, conclusive and binding, and such breach shall be deemed material even if such items comprising the Persistent Breach shall have been cured.

Contractor shall have a reasonable period of time, in no event to exceed the time period for each stated below (measured from the date the Joint Board issues notice of anticipatory breach), to fully and completely deliver all, and not less than all, of the following assurances of performance, which Contractor agrees and acknowledges are (i) the minimum necessary to tender adequate assurance of performance and (ii) reasonable, fair and appropriate to bring to a halt the pattern and practice of continuing, repeated and numerous breaches and failure to perform:

- (a) Full and complete cure of all outstanding Contractor defaults, to be completed no later than 60 Days (provided, however, that the foregoing shall not modify or limit the Joint Board's rights to declare an Event of Default or exercise rights and remedies with respect thereto or to other breaches under <u>Section 10.3</u> that have no or a shorter cure period than 60 Days);
- (b) Any new Key Personnel, to the extent required by the Joint Board, each replacement to be acceptable to the Joint Board in its sole discretion, to be completed no later than 60 Days;
- (c) A revised and restated TSSQP, and CQP,Quality Management PlanControl and Assurance Plan meeting best industry standards and practices, in form and content acceptable to the Joint Board in its good faith discretion, to be completed and delivered no later than 60 Days. As part of the revised plans, the Joint Board may, in its sole discretion, increase the level of the Joint Board involvement and oversight, at the sole cost and expense of Contractor and such involvement shall not be cause for any relief or rights on the part of Contractor under the Contract Documents or otherwise;
- (d) Replacement of each Subcontractor that the Joint Board reasonably determines is or was a material source of any continuing, repetitive or chronic breach or failure to perform (including any Subcontractor with responsibility for quality assurance or quality control), with a substitute Subcontractor acceptable to the Joint Board in its good faith discretion, to be completed no later than 60 Days; and
- (e) Notwithstanding any limitation on the maximum amount of the performance bonds set forth under <u>Section 12.2</u> or otherwise, Contractor shall increase the bonded sums by 100%, to be completed within 15 Days or if unable to so increase the bonded sums, deposit cash collateral with the Joint Board.

If for any reason Contractor fails to complete any element of the assurances of performance described in this Section 10.4 within the applicable time period, the same

shall constitute an uncured material Event of Default. Thereupon, the Joint Board, without further notice and cure period, shall be able to exercise all rights and remedies under Section 10.3 and otherwise under this Contract, at law or in equity.

Nothing contained in this Section 10.4 shall modify, alter, discharge or release Contractor from any obligations to pay Liquidated Damages, Stipulated Damages or other compensation under this Contract

11.0 Indemnification

11.1. Indemnifications by the Contractor

11.1.1. General Provisions

Subject to <u>Section 11.1.3</u>, the Contractor shall release, defend, indemnify and hold harmless the Commonwealth of Kentucky, KPTIA, KYTC, the Louisville/Jefferson County Metro Government, the State of Indiana, Clark County, Indiana, INDOT, Indiana Finance Authority, the Bridges Authority, the Tolling Body and the Joint Board, and all their respective officers, agents, consultants, and employees, their successors and assigns, and their respective board members, council members, officers, directors, agents and employees (collectively referred to as the "Indemnified Parties") from and against any and all claims, causes of action, suits, judgments, investigations, legal or administrative proceedings, Losses, costs, penalties, fines, damages, losses, liabilities and response costs, including any injury to or death of persons or damage to or loss of property, and including penalties, fines, attorneys', accountants' and expert witness fees and costs incurred in connection with the enforcement of this indemnity, arising out of, relating to or resulting from the following, (each an "Indemnified Claim"):

- 1. The breach or alleged breach of any of the Contract Documents by any Contractor-Related Entity; and/or
- 2. The failure or alleged failure by any Contractor-Related Entity to comply with the Governmental Approvals, any applicable environmental laws or other governmental rules (including governmental rules regarding hazardous materials management); and/or
- 3. Any actual or alleged patent or copyright infringement or other allegedly improper appropriation or use of trade secrets, patents, proprietary information, knowhow, copyright rights or inventions in performance of the work, or arising out of any use in connection with the Project of methods, processes, Software, designs, information, or other items furnished or communicated to the Joint Board or another Indemnified Party pursuant to the Contract; provided that this indemnity shall not apply to any infringement directly resulting from a method, process or design specifically required by the Contract Documents and by the Joint Board's failure to comply with specific written instructions regarding use provided to the Joint Board by the Contractor, and further provided that if infringement can be avoided by modification to the allegedly infringing article, the Joint Board agrees to allow such modification, at Contractor's sole cost and expense, unless the form, fit or function of the allegedly infringing article or the System is, in the Joint Board's sole determination, adversely affected; and/or

- 4. Actual or alleged act, error, omission, negligence, breach or misconduct of any Contractor-related entity in or associated with performance of the work; and/or
- 5. Any and all claims by any governmental entity or taxing authority claiming taxes based on gross receipts, purchases or sales, the use of any property or income of the Contractor or any of its Subcontractors or any of their respective agents, officers or employees with respect to any payment for the work made to or earned by any Contractor-related entity; and/or
- 6. Any and all stop notices and/or liens filed in connection with the work, including all expenses and attorneys', accountants' and expert witness fees and costs incurred in discharging any stop notice or lien and any other liability to Subcontractors for failure to pay sums due for their work or services, provided that the Joint Board is not in default in undisputed payments owing to Contractor with respect to such work; and/or
- 7. Any actual or alleged spill or release or threatened spill or release of a Hazardous Material (i) which was brought onto the Site by any Contractor-Related Entity, or (ii) attributable to the negligence, willful misconduct, or breach of contract, Governmental Approval or governmental rule by any Contractor-Related Entity; and/or
- 8. Any claim or assertion by any contractor of inconvenience, disruption, delay or loss caused by interference by any Contractor-Related Entity with or hindering the progress or completion of work being performed by other contractors, or failure of any Contractor-Related Entity to cooperate reasonably with other contractors in accordance therewith; and/or
- 9. Any actual or alleged Contractor-Related Entity's breach of or failure to perform an obligation that the Joint Board owes to a third person, including governmental entities, under law or under any agreement between the Joint Board and a third person, where the Joint Board has delegated performance of the obligation to the Contractor under the Contract Documents or (ii) the acts or omissions of any Contractor-Related Entity which render the Joint Board unable to perform or abide by an obligation that the Joint Board owes to a third person, including governmental entities, under any agreement between the Joint Board and a third person, where the agreement was expressly disclosed to the Contractor; and/or
- 10. Inverse condemnation, trespass, nuisance or similar taking of or harm to real property by reason of: (i) the failure of any Contractor-Related Entity to comply with good industry practices, requirements of the Contract Documents, project management plan or Governmental Approvals respecting control and mitigation of construction activities and construction impacts, (ii) the intentional misconduct or negligence of any Contractor-Related Entity, or (iii) the actual physical entry onto or encroachment upon another's property by any Contractor-Related Entity.

11.1.2. Errors in Owner Design Documents and/or Reference Documents

Subject to <u>Section 11.1.3</u>, Contractor shall release, defend, indemniting and hold harmless the Indemnified Parties from and against any and all Losses, claims, causes of action, suits, judgments, investigations, legal or administrative proceedings, costs penalties, fines, damages, losses, liabilities and response costs, including any injury to or death of persons or damage to or loss of property, and including penalties, fines,

attorneys', accountants' and expert witness fees and costs, arising out of, relating to or resulting from errors, omissions, inconsistencies or other defects in the Project and/or System design, installation, integration or construction, regardless of whether such errors, omissions, inconsistencies or defects were also included in any Joint Board design documents or reference documents. Contractor hereby acknowledges and agrees that it is Contractor's obligation to cause the Project to be designed and to construct the Project in accordance with the Contract Documents and that the Indemnified Parties are fully entitled to rely on Contractor's performance of such obligation. Contractor further agrees that any certificate, review and/or approval by the Joint Board and/or others hereunder shall not relieve the Contractor of any of its obligations under the Contract Documents or in any way diminish its liability for performance of such obligations or its obligations under this Section 11.

11.1.3. Restrictions on Indemnities

The following restriction shall apply to the indemnities set forth in <u>Sections 11.1.1 and 11.1.2</u>: With respect to any loss, damage or cost which is not of the type covered by insurance to be provided hereunder, Contractor's indemnity obligation shall not extend to any loss, damage or cost to the extent that such loss, damage or cost was caused by the active negligence or willful misconduct of such Indemnified Party or its agents, servants or independent contractors who are directly responsible to such Indemnified Party.

Such indemnities shall not inure to the benefit of an Indemnified Party so as to impose liability on the Contractor for the active negligence of the Joint Board, or to relieve the Joint Board of liability for such active negligence.

11.1.4. Not Limited by Workers' Compensation, Disability, or Employee Benefits Laws

In claims by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under this <u>Section 11.1</u> shall not be limited by a limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or a Subcontractor under workers' compensation, disability benefit or other employee benefits laws.

11.2. Defense and Indemnification Procedures

11.2.1. The Joint Board's Notice of Indemnified Claim

If any of the Indemnified Parties receives notice of a claim, action, suit or other matter covered under <u>Section 11.1</u> or otherwise has actual knowledge of an Indemnified Claim that it believes is within the scope of the indemnities under <u>Section 11.1</u>, the Joint Board shall by writing as soon as practicable after receipt of the Indemnified Claim: (a) inform the Contractor of the Indemnified Claim, (b) send to the Contractor a copy of all written materials the Joint Board has received asserting such Indemnified Claim, and (c) notify the Contractor that should no insurer accept defense of the Indemnified Claim, the Indemnified Party will conduct its own defense unless the Contractor accepts the tender

of the Indemnified Claim in accordance with <u>Section 11.2.3</u>. As soon as practicable after the Contractor receives notice of an Indemnified Claim or otherwise has actual knowledge of an Indemnified Claim, it shall tender the Indemnified Claim in writing to the insurers under all potentially applicable insurance policies. The Joint Board and other Indemnified Parties also shall have the right to tender such Indemnified Claims to such insurers.

11.2.2. Cooperation in the Insured Defense

If the insurer under any applicable insurance policy accepts the tender of defense, of any claim, action or suit or other matter covered under <u>Section 11.1</u> or otherwise covered under such policy, the <u>Joint Board Indemnified Party</u> and <u>The Contractor shall cooperate in the defense as required by the insurance policy.</u>

11.2.3. Rejection of Defense By Insurer/Reservation of Rights by Insurer

If the insurer(s) under potentially applicable insurance policy(ies) refuses to provide defense of any claim, action, suit or either other matter covered under Section 11.1 or otherwise, then Section 11.2.4 shall apply. If the insurer(s) under potentially applicable insurance policy(ies), agree(s) to provide a defense of any claim, action, suit or other matter covered under Section 11.1 or otherwise, or that the insurer(s) accept(s) the defense subject to a "reservation of rights," Section 11.2.2 shall apply until such time that the insurer(s) may subsequently deny further defense and for coverage for such claim, action, suit or other matter, where-upon Section 11.2.4 shall apply.

11.2.4. Tender of Defense to the Contractor

If the defense is tendered to the Contractor, then within 30 days after receipt of the tender it shall notify the Indemnified Party whether it has tendered the matter to an insurer and (if not tendered to an insurer or if the insurer has rejected the tender) shall deliver a written notice stating that the Contractor:

- 1. Accepts the tender of defense and confirms that the Indemnified Claim is subject to full indemnification hereunder without any "reservation of rights" to deny or disclaim full indemnification thereafter;
- 2. Accepts the tender of defense but with a "reservation of rights" in whole or in part; or
- 3. Rejects the tender of defense based on a determination that it is not required to indemnify against the Indemnified Claim under the terms of this Contract.

11.2.5. Acceptance of Tender/Control of Defense by the Contractor

If the Contractor accepts the tender of defense under <u>Section 10.2.4(1)</u>, the Contractor shall have the right to select legal counsel for the Indemnified Party, subject to reasonable approval by the Indemnified Party, and the Contractor shall otherwise control the defense of such Indemnified Claim, including settlement, and bear the fees and costs of defending and settling such Indemnified Claim. During such defense:

- 1. The Contractor shall fully and regularly inform the Indemnified Party of the progress of the defense and of any settlement discussions; and
- 2. The Indemnified Party shall fully cooperate in said defense, provide to the Contractor all materials and access to personnel it requests as necessary for defense, preparation and trial and which or who are under the control of or reasonably available to the Indemnified Party, and maintain the confidentiality of all communications between it and the Contractor concerning such defense.

11.2.6. Control of Defense by Indemnified Party

If the Contractor responds to the tender of defense as specified in <u>Section 11.2.4(2)</u> or <u>11.2.4(3)</u>, the Indemnified Party shall be entitled to select its own legal counsel and otherwise control the defense of such Indemnified Claim, including settlement.

11.2.7. Control of Defense if Conflict

The Indemnified Party may assume its own defense by delivering to the Contractor written notice of such election and the reasons therefore, if the Indemnified Party, at the time it gives notice of the Indemnified Claim or at any time thereafter, reasonably determines that:

- 1. A conflict exists between it and the Contractor which prevents or potentially prevents the Contractor from presenting a full and effective defense;
- 2. The Contractor is otherwise not providing an effective defense in connection with the Indemnified Claim; or
- 3. The Contractor lacks the financial capacity to satisfy potential liability or to provide an effective defense.

11.2.8. Reimbursement of Expenses

If the Indemnified Party is entitled and elects to conduct its own defense pursuant hereto of an Indemnified Claim for which it is entitled to indemnification, the Contractor shall reimburse on a current basis all reasonable costs and expenses the Indemnified Party incurs in investigating and defending. In the event the Indemnified Party is entitled to and elects to conduct its own defense, then:

- 1. In the case of a defense conducted under <u>Section 11.2.4(1)</u>, it shall have the right to settle or compromise the Indemnified Claim with Contractor's prior written consent, which shall not be unreasonably with-held or delayed;
- 2. In the case of a defense conducted under <u>Section 11.2.4(2)</u>, it shall have the right to settle or compromise the Indemnified Claim with Contractor's prior written consent, which shall not be unreasonably withheld or delayed, or with approval of the court or arbitrator following reasonable notice to the Contractor and opportunity to be heard and without prejudice to the Indemnified Party's rights to be indemnified by the Contractor; and
- 3. In the case of a defense conducted under <u>Section 11.2.4(3)</u>, it shall have the right to settle or compromise the Indemnified Claim without Contractor's prior

consent (but with prior notice to the Contractor) and without prejudice to its rights to be indemnified by the Contractor.

11.2.9. Management of Claim Involving Shared Responsibility

The Parties acknowledge that while <u>Section 11.1</u> contemplates that the Contractor will have responsibility for certain Indemnified Claims and liabilities arising out of its obligations to indemnify, defend and hold harmless, circumstances may arise in which there is no insurance coverage and in which there may be shared liability of the Parties with respect to such Indemnified Claims and liabilities. In such case, where either Party believes an Indemnified Claim or liability may entail shared responsibility and that principles of comparative negligence and indemnity are applicable, it shall confer with the other Party on management of the Indemnified Claim or liability in question. If the Parties cannot agree on an approach to representation in the matter in question, each shall arrange to represent itself and to bear its own costs in connection therewith pending the outcome of such matter. Within 30 days subsequent to the final, nonappealable resolution of the matter in question, whether by arbitration or by judicial proceedings, the Parties shall adjust the costs of defense, including reimbursement of reasonable attorneys' fees and other litigation and defense costs, in accordance with the indemnification arrangements of this Section 11.2, and consistent with the outcome of such proceedings concerning the respective liabilities of the Parties on the Indemnified Claim.

11.2.10. Additional Factors Regarding Defense of Suits

In determining responsibilities and obligations for defending suits pursuant to this <u>Section 11.2</u>, specific consideration shall be given to the following factors: (a) the party performing the activity in question; (b) the location of the activity and incident; (c) contractual arrangements then governing the performance of the activity; and (d) allegations of respective fault contained in the Indemnified Claim.

11.3. No Effect on Other Rights

The foregoing obligations shall not be construed to negate, abridge, or reduce other rights or obligations, which would otherwise exist in favor of an Indemnified Party hereunder.

No rights of the Joint Board described in <u>Section 11.1.1</u> above, no exercise or failure to exercise such rights, and no certificates or statements by the Joint Board regarding completion or acceptance, shall:

- (a) relieve Contractor of its responsibility for the selection and the competent performance of all Contractor-Related Entities;
- (b) relieve Contractor of any of its obligations or liabilities under the Contract Documents;
- (c) be deemed or construed to waive any of the Joint Board's rights and remedies under the Contract Documents, applicable Law or in equity; or

(d) be deemed or construed as any kind of representation or warranty, express or implied, by the Joint Board.

12.0 Insurance and Bonding Requirements

12.1. Insurance

12.1.1. INSURANCE General Insurance Requirements

Without limiting Contractor's indemnification of the Indemnified Parties, and prior to commencement of Work, Contractor shall obtain, at its own expense, and continuously maintain in full force and effect, the insurance coverages specified in this Section 12.1. Coverage shall be maintained through Completion of the Work, or such longer or shorter time as may be specifically provided in this Section 12.1.

All insurance required hereunder shall be procured from insurance companies with an A.M. Best and Company rating level of A-: VI or better and authorized or approved to do business in the States of Kentucky and Indiana, or as otherwise approved by KYTC. All limits of insurance set forth below are in U.S. dollars. Each policy of insurance of the type and amounts described below shall in a form satisfactory to KYTC.

12.1.2. Workers Compensation and Employer's Liability Insurance

During all phases of the Project, Contractor shall obtain and maintain and shall require all Subcontractors of all tiers to obtain and maintain, a policy or policies of insurance providing workers' compensation statutory benefits and employer's liability in conformance with the laws of the States of Kentucky and Indiana. Employer's liability limits shall be no less than \$1 million each accident, each employee, and policy limit, if scheduled under the excess or umbrella liability policies to reach a total of \$25,000,000. The workers' compensation policies shall provide the following:

- 1. A waiver of subrogation in favor of the Indemnified parties;
- 2. A provision extending coverage to all states operations;
- 3. A voluntary compensation endorsement;
- 4. Coverage for liability under the United States Longshore and Harbor Workers' Compensation Act, as appropriate, by adding a Longshore and Harbor Workers' Compensation Act coverage endorsement (WC 00 01 06;
- 5. Coverage for liability under Title 46 of the United States Code § 688 ("Jones Act") on an "if any" basis or as otherwise appropriate; and
- 6. An endorsement extending the policy to cover the liability of the insureds under the Federal Employer's Liability Act on an "if any" basis or as otherwise appropriate.

12.1.3. Commercial General Liability Insurance

Contractor shall obtain and maintain a policy or policies of commercial general liability insurance for bodily injury, property damage, personal injury and advertising injury. Coverage shall be written on an occurrence form that shall provide coverage at least as broad as the coverage provided by Insurance Services Office (ISO) form CG 00 01.

There shall be no endorsement or modification of the CGL limiting the scope of coverage for liability assumed under an insured contract.

The commercial general liability insurance coverage shall have limits of not less than \$1,000,000 per occurrence, \$2,000,000 general aggregate and \$2,000,000 products/completed operations aggregate. Completed operations coverage shall extend for as long as there is any exposure to liability under a statute of repose or any other applicable statute either through continuous maintenance of completed operations coverage in Contractor's corporate insurance program, including an endorsement providing completed operations coverage for additional insureds, or by purchase of extended completed operations coverage.

12.1.4. Automobile Liability Insurance

Contractor shall obtain and maintain and shall require all subcontractors of all tiers to obtain and maintain a business auto liability policy or policies. Each policy shall cover accidental death, bodily injury and property damage liability arising from the ownership, maintenance, or use of all vehicles connected with performance of the Work, including loading and unloading. Contractor auto liability policies shall cover "any auto" (symbol 1) or shall be specifically endorsed to include liability coverage on an excess basis for vehicles owned or operated by Contractor's subcontractors. Policies shall include as an insured anyone liable for the conduct of an insured as defined in the policy, or shall add as insureds, the Indemnified Parties. For any contractor of any tier, including Contractor, who will be involved in any way with the transportation of Hazardous Materials using its own vehicles, pollution liability coverage at least as broad as that provided under the ISO pollution liability-broadened coverage for covered autos endorsement (CA 99 48) shall be provided and the automobile liability insurance policies shall be endorsed to include Motor Carrier Act Endorsement-Hazardous materials clean up (MCS-90) with a sublimit of no less than \$1,000,000.

Contractor's automobile liability coverage shall have a combined single limit per policy period of not less than \$25,000,000 and may be scheduled under the excess or umbrella liability policies to achieve the desired limit. Excess or umbrella policies shall cover "any auto" (symbol 1) or shall be specifically endorsed to include liability coverage on an excess basis for vehicles owned or operated by Contractor's subcontractors.

12.1.5. Umbrella Or Excess Liability Insurance

Contractor shall obtain and maintain an umbrella or excess liability insurance policy to provide a total liability limits of not less than \$25,000,000, that will provide bodily injury, personal injury and property damage liability coverage at least as broad as the primary coverages set forth above including commercial general liability and employer's liability in excess of the amounts set forth above.

Such policy or policies shall include the following terms and conditions:

- Policies shall contain a drop down feature requiring the policy to respond in the event that any primary insurance limits are exhausted or for occurrences covered by an umbrella policy but not covered in the underlying insurance;
- 2. Policies shall provide coverage at least as broad as found in the underlying primary policies; and
- 3. There shall be no "contractors limitation" endorsements that have not been reviewed and approved by the Indemnified Parties or their representatives.

The Indemnified Parties shall be included as insureds on the excess policy including coverage extension to all insureds for completed operations.

12.1.6. Pollution Liability Insurance

If the Work includes excavation, utility relocation, or similar types of construction activity, Contractor shall obtain and maintain contractor's pollution liability (CPL) insurance with a total limit of liability of no less than \$5,000,000 per loss and \$5,000,000 in the aggregate.

The CPL shall be obtained on an occurrence basis for a policy term inclusive of the entire period of construction. The CPL shall provide for a minimum 3-year extended reporting period upon completion of the Work.

The CPL policy shall include coverage for investigation, removal, and remediation costs including monitoring or disposal of contaminated soil, surface water, groundwater or other contamination to the extent required by environmental laws caused by pollution conditions resulting from covered operations; third-party bodily injury and property damage, provided that the third-party property damage liability coverage includes loss of use of damaged property or of property that has not been physically injured or destroyed, resulting from pollution conditions caused by construction operations. The policy shall have no exclusions or limitations for loss occurring over water including but not limited to a navigable waterway.

Coverage as required in this paragraph shall apply to sudden and non-sudden pollution conditions resulting from the escape or release of smoke, vapors, fumes, acids, alkalis, toxic chemicals, liquids, or gases, waste materials, or other irritants, contaminants, or pollutants. The CPL shall also provide coverage for transportation and off-Site disposal of materials.

The Indemnified Parties shall also be insureds. The policy shall not contain any provision or exclusion (including any so-called "insured versus insured" exclusion or "cross-liability" exclusion) the effect of which would be to prevent, bar, or otherwise preclude any insured or additional insured under the policy from making a claim which would otherwise be covered by such policy on the grounds that the claim is brought by

an insured or additional insured against an insured or additional insured under the policy.

12.1.7. Professional Liability Insurance

During all phases of the Project, Contractor shall obtain and maintain or cause others, as appropriate, to obtain and maintain professional liability insurance for design professional services covering professional services performed in connection with this Agreement, with limits not less than \$5,000,000 per claim and in the aggregate.

No self-insured retention for Contractor shall exceed \$250,000 without prior written approval from KYTC, in its good faith discretion. Coverage shall apply specifically to professional activities performed under the Contract Documents. The policy(ies) shall have a retroactive date consistent with the inception of the first date of design or project or construction management activities, and no later than the date on which the RFP was issued.

Contractor agrees to maintain this required coverage for a period of no less than three years after Substantial Completion or to purchase an extended reporting period for no less than three years after Substantial Completion.

12.1.8. Railroad Protective Liability Insurance

Contractor shall provide any coverage as may be required by any railroad as a condition of the railroad's consent for entry into railroad facilities or property. Such policy shall be effective during the period any Work is being performed within 50 feet of any railroad right of way.

12.1.9. Watercraft Liability Insurance

If appropriate, Contractor shall provide, or cause to be provided, liability insurance covering the ownership, use, maintenance, loading or unloading of watercraft related to the performance of Contractor's Work or any other operations contemplated under this Agreement. Policies shall provide a limit of not less than \$10,000,000 per occurrence or higher limits as may be required by KYTC, in all cases where any watercraft is used on the Project that is owned, leased, hired, or chartered by any Contractor-Related Entity. Coverage may be provided through any combination of commercial general liability, marine general liability, or protection and indemnity insurance provided that all watercraft operation is covered, regardless of watercraft size. Such coverage may be arranged in any combination of primary and excess policies, all of which shall include the Indemnified Parties as insureds and shall explicitly waive subrogation against the Indemnified Parties.

12.1.10. Cyber Liability Insurance

"Cyber liability" insurance, including privacy liability coverage and business income coverage, with limits of not less than \$10,000,000 per incident and in the aggregate. If written on a "claims made" basis, policy shall include prior acts at least as far in the past as the effective date of this agreement. Coverage shall apply to both electronic and physical breaches and to employee data as well as customer data. Information in the

care, custody, or control of vendors shall be covered, including coverage for "cloud" systems or for data transferred by a third party. Coverage shall apply to accidental losses as well as breaches perpetrated by outsiders or employees. Regulatory fines and penalties shall be covered. The policy shall contain no requirement that all data be encrypted. Any business interruption coverage waiting period shall be no greater than 12 hours.

12.1.11. Crime Insurance

<u>Crime (Blanket Fidelity) with limits of not less than \$5,000,000 covering Employee</u> Dishonesty, Forgery and Alteration.

12.1.12. Property Insurance (Course of Construction)

Upon commencement of construction and with approval of KYTC, Contractor shall obtain and maintain a policy of property insurance for the Project as specified below. Coverage may be provided on a builder's risk or contractor's installation form. The policy may be provided under a corporate master program policy or a stand-alone policy as long as all of the requirements of this are met. The insureds shall be Contractor, all Subcontractors (excluding those solely responsible for design Work) of any tier, KYTC, and the Indemnified Parties. Coverage extended to KYTC and the Indemnified Parties shall NOT be limited by use of the phrase "as their interests may appear."

The course of construction property insurance policy shall be written on an "all risk" basis. Such insurance shall be on a replacement cost basis using a completed value form reasonably acceptable to KYTC to ensure adequacy of terms and sublimits.

The policy shall cover all property, materials, supplies, machinery and equipment that are part of or related to the portions or elements of the Project, and the works of improvement, including permanent and temporary works and materials, and including items intended for incorporation into the work. The policy shall not cover tools or equipment used by any contractor to perform their work. Such tools and equipment are the sole responsibility of the contractor who owns or uses such tools.

The policy shall provide a "severability of interests provision," or "multiple insured's clause" or similar wording that the policy shall apply to each insured as if a separate policy had been issued to each insured except as to limits.

The policy shall not contain any coinsurance provision. The policy may include a "waiting period" deductible of no more than 60 days. No liquidated damages shall be deducted from the insurance recovery unless such damages are actually collected. Coverage shall include payment for expenses incurred for the purpose of reducing any loss amount to the extent such loss amount is actually reduced. Coverage shall include.

The policy shall include coverage for:

- 1. any ensuing loss from faulty workmanship, nonconforming work, error, omission or deficiency in design or specifications;
 - 2. machinery accidents and operational testing, if applicable;
- 3. removal of debris, with a sub-limit of 20% of the loss and insuring the buildings, structures, machinery, equipment, materials, facilities, fixtures and all other properties constituting a part of the Project;
- 4. transit, including ocean marine coverage (unless insured by the Supplier or through a separate marine cargo policy), with sub-limits sufficient to insure the full replacement value of any key equipment item;
- 5. sub-limits sufficient to insure the full replacement value of any property or equipment stored either on or off the Site;
- 6. loss of revenue resulting from delayed opening and expense to reduce loss;
 - 7. collapse;
 - 8. terrorism;
 - 9. plans, blueprints and specifications;
- 10. demolition and increased cost of construction as required by law or ordinance;
- 11. loss caused by action of civil authority following loss due to an insured peril;
- 12. soft costs expense (including costs of governmental approvals, mitigation costs, attorneys' fees, and other fees and costs associated with such damage or loss or replacement thereof); and
- 13. delay coverage, with no sublimit, for loss of revenue due to physical damage to the Project, which shall apply only for the benefit of the Indemnified Parties. Valuation shall be based on estimated lost toll revenues, with no policy sublimit, with a period of indemnity limited to a period of no less than one year commencing on the date Substantial CompletionFormal Acceptance would have occurred absent the loss occurrence.

There shall be no coinsurance penalty provision in any such policy. Deductibles or self-insured retentions shall be no greater than 5% of the total value insured at the time of loss subject to a minimum deductible no greater than \$250,000. All deductibles or self-insured retentions shall be the responsibility of Contractor.

12.1.13. General Requirements, Conditions, And Agreements Pertaining To Insurance

12.1.13.1. Premiums, Deductibles and Self-Insured Retentions.

Contractor shall be responsible for payment of premiums for all insurance required under this Section 12.1. KYTC and the Indemnified Parties have no obligation to pay any premium. Contractor further agrees that for each claim, suit or action made against insurance provided hereunder, with respect to all matters for which Contractor is responsible hereunder, Contractor shall be solely responsible for all deductibles or self-insured retentions. Any self-insured retentions maintained by Contractor over \$250,000 must be declared and approved by KYTC. At the option of KYTC, the insurer shall either reduce or eliminate such self insured retentions with respect to KYTC, and the other Indemnified Parties; or KYTC in its good faith discretion, may require posting of collateral by Contractor guaranteeing payment of losses and related investigations, claims administration and defense expenses.

12.1.13.2.Evidence of Insurance.

Concurrently with Contractor's execution hereof or on such later date on which coverage is required to be provided hereunder, Contractor shall deliver to KYTC evidence of coverage required to be provided by Contractor under this Section 12.1, including any certificates of insurance and endorsements, such as additional insured endorsements, used to satisfy the terms of this Section 12.1. KYTC shall have no duty to pay or perform under the Contract until such evidence of insurance, in compliance with all requirements of this Section 12.1 has been provided.

12.1.13.3. Enforcement of Contract Provisions (non estoppel).

Contractor acknowledges and agrees that any actual or alleged failure on the part of KYTC to inform Contractor of non-compliance with any requirement imposes no additional obligations on KYTC nor does it waive any rights hereunder.

12.1.13.4. Renewal Policies.

Contractor shall promptly deliver to KYTC evidence of insurance with respect to each renewal policy, as necessary to demonstrate the maintenance of the required insurance coverages for the terms specified herein. Such evidence shall be delivered to KYTC not less than 15 days prior to the expiration date of any policy, or such shorter period as approved in advance by KYTC.

12.1.13.5. Policy Endorsements and Waivers

All insurance policies required hereunder shall contain, or be endorsed to comply with, the following provisions:

1. For claims covered by the insurance specified herein, said insurance coverage shall be primary insurance with respect to the insureds, additional insureds, and their respective members, directors, officers, employees, agents, and consultants, except for coverage that by its nature cannot be written as primary. Any insurance or self-insurance beyond that

- specified in this Contract that is maintained by an Indemnified Party, their directors, officers, employees, agents, and consultants shall be excess of such insurance and shall not contribute with it.
- 2. Liability insurance shall apply separately to each insured and additional insured against whom a claim is made or suit is brought, except with respect to the limits of the insurer's liability. If Contractor's liability policies do not contain the standard Insurance Services Office separation of insureds provision, or a substantially similar clause, such policies shall be endorsed to provide cross-liability coverage.
- Policies shall not be suspended, voided, canceled, or reduced in coverage or in limits except after 30 days prior written notice has been given to KYTC by Contractor or its insurance broker or insurer.
- 4. All endorsements adding insureds to required commercial general liability policies shall provide additional insureds with coverage for "completed operations," or a separate endorsement providing such coverage must be added to the policy.
- 5. Each policy shall provide coverage on an "occurrence" basis and not a "claims made" basis (with the exception of professional liability or cyber liability) and no policy issued on an occurrence basis shall have any sunset clause requiring reporting within a specified period of time except as specified for pollution liability policies.

12.1.13.6. Waivers.

Contractor waives all rights against the Indemnified Parties and each of their agents and employees and their respective members, directors, officers, employees, agents and. Contractor shall require all Subcontractors to provide similar waivers in writing each in favor of all other parties enumerated above. Workers' compensation/employers liability policies shall include by endorsement or otherwise, a waiver of any right of subrogation against the Indemnified Parties.

12.1.13.7.Changes in Insurance Requirements.

KYTC shall notify Contractor in writing of any changes in the requirements applicable to insurance to be provided by Contractor. Except as set forth otherwise in this Agreement, any additional cost from such change shall be paid by the Joint Board and any reduction in cost shall reduce the Contract Price pursuant to a Change Order.

12.1.13.8. No Recourse.

There shall be no recourse against the Indemnified Parties for payment of premiums or other amounts with respect to the insurance to be provided by Contractor hereunder.

12.1.13.9. Non-Limitation of Insurance Requirements.

The insurance coverage provided and limits required hereunder are minimum requirements and are not intended to limit Contractor's indemnification obligations under this Section 12.1, nor do the indemnity obligations limit the rights of the insured parties to the coverage afforded by their insured status. Requirements of specific coverage features or limits contained in this Section 12.1 are not intended as a limitation on coverage, limits or other requirements, or a waiver of any coverage normally provided by insurance. Specific reference to a given coverage feature is for purposes of clarification only as it pertains to a given issue and is not intended by any party or insured to be all inclusive, or to the exclusion of other coverage, or a waiver of any type. With the exception of any insurance required by a railroad, liability coverage will not be limited to the specific location designated as the Site. No liability policy shall contain any provision or definition that would serve to eliminate so-called "third-party-over action" claims, including any exclusion for bodily injury to an employee of the insured or of any contractor or subcontractor.

12.1.13.10. Commercial Unavailability of Required Coverages.

If, through no fault of Contractor, any of the coverages required in this Section 12.1 (or any of the required terms of such coverages, including policy limits) become unavailable or are available only with commercially unreasonable premiums, KYTC will consider in good faith alternative insurance packages and programs proposed by Contractor, with the goal of reaching agreement on a package providing coverage equivalent to that specified herein. Contractor must demonstrate to KYTC's satisfaction that it has used diligent efforts in the global insurance markets to place the required insurance coverages, and shall advise KYTC of the specific results of those efforts. Contractor shall not be entitled to any increase in the Contract Price for increased costs resulting from the unavailability of coverage and the requirement to provide acceptable alternatives. KYTC shall be entitled to a reduction in the Contract Price if KYTC, in its sole discretion, agrees to accept alternative policies providing less than equivalent coverage.

12.1.13.11. Notice and Prosecution of Claims.

KYTC shall have the right, but not the obligation, to submit, on behalf of any of the Indemnified Parties, claims and tenders of defense and indemnity under applicable insurance policies. Unless otherwise directed by KYTC in writing with respect to insurance claims involving any of the Indemnified Parties, Contractor shall be responsible for reporting and processing all potential claims or tenders for defense and indemnity under the appropriate insurance policies. Contractor agrees to report timely to the insurer(s) under such policies any and all matters which may give rise to an insurance claim by and to promptly and diligently pursue such insurance claims in accordance with the claims procedures specified in such policies, whether for defense or indemnity or both. Contractor shall enforce all legal rights against the insurer under the applicable insurance policies and applicable Governmental Rules in order to collect thereon, including pursuing necessary litigation and enforcement of judgments.

Contractor shall immediately notify KYTC, and thereafter keep KYTC fully informed, of any incident, potential claim, claim or other matter of which Contractor becomes aware that involves or could conceivably involve an Indemnified Party as a defendant. Contractor will cooperate with KYTC, and shall require its liability insurers to agree in writing to work with KYTC to assure compliance with all regarding timely response to claims.

KYTC agrees to promptly notify Contractor of incidents, potential claims against an Indemnified Party, and matters of which KYTC is actually aware which may give rise to an insurance claim or to a right of defense and indemnification under this Section 12.1. Delivery of any such notice will constitute a tender of defense of the claim to Contractor and the insurer under any applicable Insurance Policies, subject to KYTC's or an Indemnified Party's rights to control its own defense to the extent provided in this Section 12.1 or by applicable Governmental Rules. The Indemnified Parties shall cooperate with Contractor as necessary for Contractor to fulfill its duties hereunder, including providing Contractor a copy of all written materials an Indemnified Party receives asserting a claim against the Indemnified Party that is subject to defense by an insurer under an Insurance Policy or by Contractor under this Section 12.1.

If, in any instance, Contractor has breached its obligations respecting insurance coverage set forth in the Contract Documents or is unable to enforce and collect any such insurance for failure to assert claims in accordance with the terms of the Insurance Policies or to prosecute claims diligently, then for purposes of determining damages resulting from the breach or inability to enforce or collect, on or determining reductions in compensation due from an Indemnified Party to Contractor, Contractor shall be treated as if it has elected to self-insure up to the full amount of insurance coverage that would have been available had Contractor performed such obligations or not committed such failure. Nothing in this Section 12.1 shall be construed to treat Contractor as electing to self-insure where Contractor is unable to collect due to the bankruptcy or insolvency of any insurer, which at the time the insurance policy is written meets the rating qualifications set forth in this Section 12.1.

In the event that an insurer providing any of the insurance policies becomes the subject of bankruptcy proceedings, becomes insolvent, or is the subject of an order or directive

limiting its business activities given by any Governmental Entity, including the State of Kentucky or State of Indiana Department of Insurance, Contractor shall exercise best efforts to promptly, and at its sole cost and expense, secure alternative coverage in compliance with the insurance requirements contained in this Section 12.1 so as to avoid any lapse in insurance coverage.

12.1.13.12. Commencement of Work.

Contractor shall not commence Work under this Contract until it has obtained the insurance required under this Section 12.1, has furnished original evidence of insurance for the required coverage as required under this Section 12.1, and such insurance has been approved by KYTC, nor shall Contractor allow any Subcontractor to commence work under its Subcontract until the insurance required of the Subcontractor has been obtained and approved by Contractor.

12.1.13.13. Contractor's Failure to Comply.

If Contractor or any Subcontractor fails to provide and maintain insurance as required herein, then KYTC shall have the right but not the obligation, to purchase such insurance or to suspend Contractor's right to proceed until proper evidence of insurance is provided. Any amounts paid by KYTC (plus an administrative charge equal to 10% of the cost) shall, at KYTC's sole option, be deducted from amounts payable to Contractor or reimbursed by Contractor upon demand, plus interest thereon from the date of payment by KYTC to the reimbursement date, at the lesser of (a) 10% per annum or (b) the maximum rate allowable under applicable Governmental Rules. The Joint Board shall notify the Contractor promptly of any such deduction. Nothing herein shall preclude KYTC from exercising its rights and remedies under this Section 12.1 as a result of the failure of Contractor or any Subcontractor to satisfy the obligations of this Section 12.1.

If on account of Contractor's failure to comply with the provisions of this Section 12.1, any Indemnified Party is adjudged to be responsible for all or any portion of a judgment, loss or settlement (through admission or stipulation by Contractor or court decision) that would have been covered by insurance but for non-compliance with this Section 12.1, then any loss or damage it shall sustain by reason thereof shall be borne by Contractor, and Contractor shall immediately pay the same to the Indemnified Party, upon receipt of written demand therefor and evidence of such loss or damage.

12.1.13.14. Subcontractor Insurance Requirements.

Contractor shall cause each Subcontractor to provide insurance that complies with requirements for Contractor-provided insurance set forth in this Section 12.1 in circumstances where Subcontractor acts or omissions are not covered by Contractor-provided insurance, including automobile liability. Except a otherwise specified in this Section 12.1, Contractor shall have sole responsibility for determining the limits of coverage required to be obtained by Subcontractors, which determination shall be made in accordance with reasonable and prudent business practices. Contractor shall cause each such Subcontractor to include the Indemnified Parties as additional insureds under

such Subcontractor's general liability and excess liability insurance policies. If requested by KYTC, Contractor shall promptly provide certificates of insurance evidencing coverage for each Subcontractor. KYTC shall have the right to contact the Subcontractors directly in order to verify the required coverage.

12.1.13.15. Disclaimer.

Contractor and each Subcontractor shall have the responsibility to make sure that their insurance programs fit their particular needs, and it is their responsibility to arrange for and secure any insurance coverage that they deem advisable, whether or not specified herein.

KYTC on behalf of the Joint Board makes no representation or warranty that the coverage, limits of liability or other terms specified for the insurance policies to be carried pursuant to this Section 12.1 are adequate to protect Contractor against its undertakings under the Contract Documents or its liability to any third party or preclude KYTC from taking any actions as are available to it under the Contract or otherwise at law. KYTC shall not be limited to the amount of the insurance premium not paid in the proof of any damages it may claim against Contractor arising out of or by reason of failure of Contractor to provide and keep in force the insurance policies required by and on the terms of this Section 12.1, but KYTC shall instead be entitled to recover the full amount of damages available.

Builder's Risk

11.4. Insurance

The Contractor shall provide liability insurance and property insurance as provided in Exhibit ____. [INSURANCE ADVISOR SHOULD REVIEW INSURANCE AS INSURANCE FOR TOLL SYSTEMS IS NOT TYPICAL CONSTRUCTION INSURANCE]

11.5.12.2. Payment, Performance and Warranty Bonds

As a part of the consideration and to assure the faithful performance of this Contract in every respect, the Contractor shall provide payment, performance and warranty bonds securing Contractor's obligations hereunder, each in an amount set forth herein, and shall maintain such bonds in full force and effect as described below. Such bonds, when approved by the Joint Board, shall be for the use and benefit of the Joint Board, and each person furnishing materials, labor and supplies for use in the performance of this Contract. Upon request, the Contractor shall also provide copies of such bonds to its Subcontractors and other entities providing labor, equipment, or materials on the Project.

- <u>Language</u> When the Contractor delivers the executed Contract to the Joint Board, the Contractor shall also deliver to the Joint Board a performance bond in the form of <u>Exhibit CF</u> (a "Performance Bond") for the Work covered by the NTP. Such bond shall be in an amount of 100% of the Contract Price <u>(excluding the Maintenance Price)</u>. Subject to <u>Section 142.2.3</u>, the Joint Board will <u>release reduce the amount of the Performance Bond by the amount of the Contract Price relating solely to the Operations Center, a Walk-in Center or [all Toll Zones within] a Crossing, as the case may be (a) upon expiration of the Warranty <u>termPeriod</u>, for such Operations Center, Walk-in Center or Crossing, provided that no outstanding claims are then pending or threatened against Contractor hereunder, or (b) upon satisfaction of the conditions in <u>Section 124.2.3</u>.</u>
- As a condition to issuance of the NTP, Contractor shall provide, and continuously maintain in place for the benefit of the Joint Board, a payment bond in the form of Exhibit BG (a "Payment Bond") for the work covered by the NTP. Such bond shall be in an amount of 100% of the Contract Price. Subject to Section 142.2.3, the Joint Board will release the reduce the amount of the Payment Bond by the amount of the Contract Price relating solely to the Operations Center, a Walk-in Center, or [all Toll Zones within a] Crossing, as the case may be (a) upon receipt of (i) evidence satisfactory to the Joint Board that all Persons eligible to file a claim against the bond have been fully paid, and (ii) unconditional releases of Liens and stop notices from all Subcontractors who filed preliminary notice of a claim against the bond, (b) upon expiration of the statutory period for Subcontractors to file a claim against the bond if no claims have been filed, or (c) upon satisfaction of the conditions in Section 142.2.3.
- After Formal Acceptance of the Operations Center, a Walk-in Center or a Crossing, subject to the requirements herein, Contractor may obtain a reduction in the amount of the Performance Bond by providing a warranty bond, or such other security as is approved by the Joint Board, in its sole discretion, which shall guarantee performance of the work required to be performed during the Warranty termPeriod for such Operations Center, Walk-in Center or Crossing, and which shall also constitute a payment bond guaranteeing payment to Persons performing such work ("Warranty The Warranty Bond shall be (i) in an amount equal to 20% of the applicable amount of the Contract Price attributable solely to the Operations Center, Walk-in Center or Crossing, as the case may be, (ii) substantially in the form attached hereto as Exhibit —H (with such modifications as the Joint Board approves in writing, in its sole discretion), and (iii) released upon satisfaction of the conditions in Section 142.2.1(a) and Section 142.2.2(a) or (b). Alternatively, subject to the requirements herein, after any Formal Acceptance of the Operations Center, a Walk-in Center or a Crossing, as the case may be, the Contractor may obtain a reduction in the amount of the Performance Bond and the Payment Bond by an amount equal to eighty percent (80%) of the amount of the Contract Price attributable -solely to the Operations Center, Walk-in Center or Crossing, as the case may be.

12.2.4. As a condition to Formal Acceptance, Contractor shall furnish the Joint Board with (a) a maintenance performance bond in the form of Exhibit I ——(with such modifications as the Joint Board approves in writing, in its sole discretion) (a "Maintenance Performance Bond"), and (b) a maintenance payment bond in the form of Exhibit —J (with such modifications as the Joint Board approves in writing, in its sole discretion) (a "Maintenance Payment Bond"). The Maintenance Performance Bond shall be in an amount equal to the greater of (i) 100% of the aggregate twofour (24) year Maintenance Price attributable to the relevant Operations Center, Walk-in Center or . The Maintenance Payment Bond shall be in an amount Crossing, and (ii) \$ equal to 100% of the aggregate twofour (24) year Maintenance Price for the Project.relevant Operations Center, Walk-in Center or Crossing After the initial Maintenance Performance Bond and Maintenance Payment Bond are issued, as a condition precedent to subsequent Final Acceptances, in lieu of providing separate Maintenance Performance Bonds and Maintenance Payment Bonds for each additional Walk-in Center or Crossing, Contractor may provide Riders to the existing Maintenance Performance Bond and Maintenance Payment Bond. Maintenance Performance Bonds (or Riders, as the case may be) shall be in an amount equal to the greater of (i) 100% of the aggregate four (4) year Maintenance Price for the relevant Operations Center, Walkin Center and/or Crossing, and (ii) \$_____. Maintenance Payment Bonds (or Riders, as the case may be) shall be in an amount equal to 100% of the aggregate four (4) year Maintenance Price. Contractor's obligation to maintain and provide a current Maintenance Performance Bond and Maintenance Payment Bond shall continue throughout the Maintenance Term, but the Joint Board will accept Maintenance Performance Bonds and Maintenance Payment Bonds with a stated term of at least two (2) years with a statement set forth in the applicable bond that it shall be renewable annually in accordance with the Surety's customary renewal practices. Provided that Contractor has paid the Joint Board any applicable Stipulated Liquidated Damages, compensation for Losses and any other amounts that are payable to the Joint Board under the Contract Documents, the Maintenance Performance Bond shall be released after the satisfaction of all conditions to Final Maintenance Payment of the Maintenance Price pursuant to Section 6 at the end of the Maintenance Term. Upon expiration of the Maintenance Term, the Joint Board will release the Maintenance Payment Bond (i) upon receipt of (A) evidence satisfactory to the Joint Board that all Persons eligible to file a claim against the bond have been fully paid and (B) unconditional releases of Liens and stop notices from all Subcontractors who filed preliminary notice of a claim against the bond, or (ii) upon expiration of the statutory period for Subcontractors to file a claim against the bond if no claims have been filed.

Each bond required hereunder shall be issued by a Surety authorized to do business in the State with a rating of at least AVIII by "Best & Company" or rated in the top two categories by two nationally recognized rating agencies, or as otherwise approved by the Joint Board in its sole discretion. If any bond previously provided becomes ineffective, or if the Surety that provided the bond no longer meets the requirements hereof, Contractor shall provide a replacement bond in the same form issued by a surety meeting the foregoing requirements, or other assurance satisfactory

to the Joint Board in its sole discretion. If a Price is increased in connection with a Change Order, the Joint Board may, in its sole discretion, require a corresponding proportionate increase in the amount of each bond or alternative security. If notice of any change affecting the general scope of the work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Time) is required by the provisions of any bond to be given to a surety, the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

12.2.6. No draw under any Performance Bond, Payment Bond, Maintenance Performance Bond or Maintenance Payment Bond shall preclude, inhibit, alter or modify the Joint Board's rights to exercise any and all other rights and remedies available under the Contract Documents, at law or in equity.

41.6.12.3. No Relief of Liability

Notwithstanding any other provision set forth in the Contract Documents, performance by a Surety or Guarantor of any of the obligations of Contractor shall not relieve Contractor of any of its obligations hereunder.

12.013.0 Dispute Resolution

12.1.1. <u>Dispute Resolution Procedures</u>

Notwithstanding any other available process (which the Contractor hereby waives in consideration of the following resolution procedures provided), all claims between the Joint Board and the Contractor or its Subcontractors relating to this Contract or the Project that cannot be resolved through the elevated negotiations procedure discussed below shall follow the administrative process outlined below in Section 13.4, as the sole remedy for any claim by the Contractor. The administrative process is based upon 603 KAR 2:015 Sections 9 and 10 and in accordance with KRS Chapter 13B. The Contractor shall continue the work and adhere to the Project Schedule during all Delisputes or disagreements with the Joint Board. No work shall be delayed or postponed pending resolution of any Delisputes or disagreements, except as the Contractor and the Joint Board may otherwise agree in writing.

Should Contractor's actions be implicated or alleged to be implicated as a causational factor in any claims by Walsh Construction Company on the Downtown Crossing or WVB Partners on the East End Crossing, then Contractor agrees, at the election of the Joint Board, to participate in the dispute resolution process outlined in Article 12 of the DBT Agreement between KYTC and Walsh or Article 19 of the Public-Private Agreement between WVB Partners and IFA. Contractor agrees to be bound by any determinations under Article 12.4 and 12.5 of the DBT Agreement, and/or Article 19.6.4.2 of the Public-Private Agreement that find the Contractor liable or responsible in whole or in part for claims presented to KYTC or IFA by Walsh or WVB Partners. At the

request of the Joint Board, in the event of claims by Walsh or WVB as discussed above, Contractor will also participate, at its own expense, in the non-binding Dispute Review Board process found in Article 12.3 of the DBT Agreement, or the non-binding arbitration process found in Article 19.6.4.1 of the PPA Agreement.

12.2. 13.2. Partnering and Elevated Negotiations

12.2.1.13.2.1. Partnering

The Joint Board intends to encourage the use of an extensive partnering program among the Joint Board, the Contractor, its Subcontractors, FHWA, other contractors employed by the Joint Board or its members, and other stakeholders, where appropriate. The partnering relationship will be structured to draw on the strengths of each organization to identify and achieve reciprocal goals. The objectives include effective and efficient Project performance and completion on schedule, within budget and in accordance with the Contract Documents.

The Joint Board anticipates a full-time partnering effort involving the executive management, Project management, Project staff and others. Participation is encouraged. Any cost associated with effectuating partnering shall cause no change to the Contract Price.

It is the intent of the parties that the <u>Ddispute</u> resolution provisions contained in <u>Sections 13.3 and 13.4</u> shall apply only in the event that the Joint Board's and the Contractor's issue resolution efforts through partnering, and elevated negotiations, are not successful. The <u>Ddispute</u> resolution provisions set forth in <u>Section 13.3</u> shall apply to all disputes arising out of the work that are not resolved by the parties through the partnering process, except as expressly provided to the contrary in the Contract Documents.

12.2.2.13.2.2. Elevated Negotiations

Both the Joint Board and the Contractor shall designate senior representatives for the purposes of this Section. If a claim cannot be resolved by the Parties' representatives through partnering discussions or via Section 4 above, then, upon the request of either party, the Contractor's senior representative and the Joint Board's senior representative shall meet as soon as conveniently possible, but in no case later than fourteen (14) days after such a request is made, to attempt to resolve such claim. Prior to any meetings between such representatives, the parties will exchange relevant information that will assist the parties in resolving the claim and, if applicable, make available any independent expert opinion. Subject to the requirements of the Kentucky Open Records Act, any statements made or documents or materials prepared by the partnering facilitator shall not be admissible or discoverable in any other Delispute resolution proceeding. All negotiations and discussions pursuant to this Section 13.2, including any opinions provided by any technical experts, shall be deemed compromise and settlement negotiations, and therefore inadmissible for the purposes of all evidentiary rules under applicable Law.

12.3.13.3. Administrative Claims Process

[DISCUSS INDIANA ROLE IN 13.3 AND 13.4]

In the event that the Parties do not resolve their <u>Ddisputes</u> following the elevated negotiations in <u>Section 13.2</u> above, any claims by the Contractor shall follow the process outlined in this <u>Section 13.3</u>. Claims by the Joint Board may be resolved through litigation in the Franklin County Circuit Court. The following applies to all claims not resolved through elevated negotiations:

- 1. The Joint Board shall not consider a claim for extra work unless the Contractor has submitted Kentucky form TC 63-32, "Notice of Changed Condition and Disagreement" to the Joint Board Representative before beginning the disputed work. Submittal of such form shall not guarantee that claim for extra work will be granted unless a Change Order or Construction Change Directive is approved pursuant to Section 4.
- 2. Any other claim not referenced in subsection (1) of this section that a the Contractor shall possess against the Joint Board for compensation shall be submitted in writing on form TC 63-32, "Notice of Changed Condition and Disagreement", to the Joint Board Representatives within ten (10) days of the date of which the Contractor knew or should have known of the existence of said claim. Any claim presented after said ten (10) days shall be waived and not be considered for payment by the Joint Board. After receipt of TC 63-32, "Notice of Changed Condition and Disagreement", the Joint Board shall respond to the Contractor with Kentucky form TC 63-33, "Acknowledgement of Notice of Changed Condition and Disagreement".
- 3. If a the Contractor has a contract claim or requests relief from the Joint Board, the Contractor shall exhaust the administrative process within the Joint Board as set forth below prior to requesting an administrative hearing.
- (a) For claims involving extra work, the Contractor shall submit his claim in writing, setting forth the amount in <u>Dd</u>ispute, the basis of the claim and any supporting documentation of said claim to the Joint Board Representatives not later than thirty (30) days after receipt of the "Final Inspection and Formal Acceptance Report of Completed Construction", Form TC 63-44. This provision shall not act to relieve the Contractor of its responsibility to comply with any notice requirements of <u>Section 4.2 or 4.3</u>.
- (b) For claim <u>Dd</u>isputes involving final quantities and payments, the Contractor shall submit his claim in writing setting forth the amount in <u>Dd</u>ispute, the basis of the claim and any supporting documentation of said claim to the Joint Board Representatives not later than sixty (60) days from the date of the "Final Release" (Form TC 63-34) sent by the Joint Board.
- (c) The Joint Board Representatives and the Contractor shall attempt to resolve the <u>Ddispute</u>. The Joint Board Representatives shall have sixty (60) days from the date of receipt of written notice of a formal claim to resolve the <u>Ddispute</u>.

- (d) Should the claim not be resolved by the Joint Board Representatives, then the claim shall be submitted to the Joint Board, who shall have ninety (90) days to make the final determination.
- (e) If the matter is not resolved by the Joint Board Representatives and the Contractor prior to making a final determination on the matter, the Joint Board, shall convene an informal settlement conference with the Contractor for the purpose of either settling the Delispute or identifying the issues which need resolution. If the settlement conference is unsuccessful, the Joint Board, shall notify the Contractor in writing of the Joint Board's decision regarding Contractor's claim. Said notification shall inform the Contractor of his rights to an administrative hearing pursuant to Section 2.[?]KRS Chapter 13B.
- (f) Should the Joint Board Representatives or the Joint Board, fail to render a decision within the time limits set forth in this administrative regulation, said inaction shall be deemed a denial of the claim by the Joint Board and the Contractor may proceed with the administrative hearing process pursuant to Section 10 of this administrative regulation[?].KRS Chapter 13B. Further, should the Joint Board, fail to render a decision in the time frame previously stated, the Joint Board shall bear all costs associated with the hearing officer.
- (g) The Contractor shall request an administrative hearing pursuant to KRS Chapter 13B within thirty (30) days of the date of the notification of the decision by the Joint Board.
- (h) Upon the agreement in writing of both parties, the parties may engage in formal nonbinding mediation of the <u>Dd</u>ispute with a mediator agreeable to both parties. The parties participating in the mediation shall each pay one-half (1/2) of the costs associated with the mediator.
- (i) If mediation is agreed upon by the parties, the formal administrative process that culminates with a KRS Chapter 13B hearing and all time limits therein shall be stayed until the Joint Board or the Contractor submits written notice to the other that they are terminating the mediation process. The time limits previously stayed shall commence to run upon the date of the written notice.

12.4. 13.4. Hearing Procedure

- (a) A request for an administrative hearing pursuant to the provisions of this administrative regulation[?] sKRS Chapter 13B shall be in writing and mailed to KYTC, Secretary's Office, 200 Metro Street, Frankfort, Kentucky 40622.
- (b) Upon receipt of a request for an administrative hearing, the KYTC shall forward the request to the KYTC Office of General Counsel and Legislative Affairs for proceedings in accordance with the provisions of KRS Chapter 13B. Failure to submit a written request for an administrative hearing to KYTC within thirty (30) days of the date of the decision by the Joint Board, shall be ground to summarily deny the request for hearing and the decision by the Joint Board, shall stand.

- (c) If a hearing is commenced, the hearing examiner shall prepare and submit his report with a recommendation to the Kentucky Secretary of Transportation through the Office of General Counsel and Legislative Affairs.
- (d) The secretary, after receiving the report and recommendation of the hearing examiner, may accept the report and recommendation in its entirety, or reject or modify any or all of the findings and recommendations of the hearing examiner as set out in a final order pursuant to KRS Chapter 13B.
- (e) The Contractor shall be notified by Final Order of the Kentucky Secretary's decision.
 - (f) The Contractor shall have appeal rights pursuant to KRS Chapter 13B.
- (g) The Contractor shall reimburse the Joint Board one-half (1/2) of the expenses of the hearing officer within thirty (30) days after the date of the entry of the final order by the Kentucky Secretary of Transportation. The Joint Board may withhold any sum owed a Contractor on a current or future project that the Joint is owed for the costs and/or expenses of the hearing officer."

After all administrative remedies have been exhausted as outlined above which along with proper written notice of the claim are conditions precedent to any other relief, the Contractor shall have the right of appeal with the Franklin County Circuit Court of the Commonwealth of Kentucky. All questions as to the execution, validity, interpretation, construction and performance of this Contract shall be governed by the laws of the Commonwealth of Kentucky. Furthermore, the parties hereto agree that any legal action which is brought on the basis of this Contract shall be filed in the Jefferson County Circuit Court of the Commonwealth of Kentucky.

12.5.13.5. Limitations on Consequential Damages

Under no circumstances shall the Contractor be entitled to consequential damages arising out of the performance (or failure to perform) of the Joint Board, and the Contractor releases the Joint Board from any such liability.

Consequential damages shall be recoverable by the Joint Board only in the following limited categories:

- 1. Any component of the Liquidated Damages and Stipulated Damages amounts provided in the Contract Documents that may be consequential in nature.
- 2. Any component of the Liquidated Damages relating to lane closures that may be consequential in naturelLoss of toll revenues, or damages payable byte a States' Party under the Design-Build Contract or the Public-Private Agreement that arises directly as a result of Contractor's negligent or intentional delay of or interference with the Design-Builder's or the Developer's respective performance under the Design-Build Contract and the Public-Private Agreement .

- 3. Any damage that may be deemed "consequential" but is nonetheless covered by insurance purchased by the Contractor or others on the Project, or for which this Contract requires the Contractor to purchase insurance coverage, is not waived up to the limits of such coverage. However, any consequential damages outside the insurance coverage limits required by this Contract are waived except as provided in this <u>Section 13.5</u>.
- 4. The Joint Board does not waive consequential damages for any injuries suffered by the Joint Board due to fraud, intentional misconduct, or criminal acts by the Contractor or entities for whom the Contractor is responsible.

The term "consequential damages" shall mean those special, indirect or incidental damages that do not flow directly and immediately from an injurious act but that result indirectly from an action or failure to act, such as revenue losses, loss of use, cost of capital, debt service, loss of profit on related contracts, administrative costs, loss of bonding capacity, lost opportunity, claims of taxpayers and other indirect damage. Liabilities, costs and losses incurred by either party due to failure by the other to procure and maintain insurance policies required hereunder, as well as any amounts that this Contract expressly states are to be reimbursed from one party to the other (including interest, late charges, fees, penalties, and similar charges), shall be considered direct damages.

This provision shall apply to limit liability under actions brought under any theory of law, including actions in tort (including negligence) as well as in contract. The statute of limitations for any breach of the Acceptance for the Project. This provision shall not relieve the Contractor of its obligation to timely comply with all notice provisions for claims, nor shall it alter the requirements of the Contractor to comply with Sections 13.1-13.2 or the administrative claims process in Section 13.3 above.

13.014.0 Responsibility for Design; Information Supplied 13.1.14.1. Standard of Care

The standard of care for all design services performed or furnished by the Contractor under this Contract will be the care and skill ordinarily used by members of the subject profession practicing under similar conditions at the same time and in the same locality.

13.2.14.2. Design, Installation and Integration Obligations

As indicated throughout the Contract Documents, Contractor's obligations under this Contract are for the full design, installation, integration and construction of the Project for the Contract Price. While the Joint Board will review and comment on Contractor's design for the Project, such review and comment by the Joint Board is limited only to general conformance with the Joint Board's program requirements for the Project and the requirements of the Contract Documents. The Contractor remains at all times responsible and obligated to the Joint Board for the provision of a complete design which meets the standard of care and complies with all applicable Laws, regulations, and required specifications. Review, comment, or approval of any kind by the Joint

Board shall not alter Contractor's responsibility for such design, and the failure by the Joint Board to identify any errors shall not give rise to any liability on the Joint Board's behalf.

13.2.1.14.2.1. Design Review Process and Compliance with Final Design Documents

13.2.1.1. 14.2.1.1. Contractor shall furnish the Design Documents to the Joint Board in accordance with the design review submittal and certification process set forth in Scope of WorkRFP Sections ______. 3.12.3.3.

13.2.1.2.14.2.1.2. The Joint Board will review submittals as required by the Contract Documents. The Joint Board's review of any such submittals as required by the Contract Documents will be only to determine if the items covered by the submittals will, after installation or incorporation in the construction, conform to the information given in the Contract Documents and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. The Joint Board's review will not extend to means, methods, techniques, sequences, or procedures of construction or to safety precautions or programs incident thereto, although the Joint Board may comment on such matters. The review of a separate item as such will not indicate approval of the assembly in which the item functions. The Joint Board's review of submittals shall not relieve the Contractor from responsibility for any variation from the requirements of the Contract Documents.

13.2.1.3.14.2.1.3. Contractor shall respond to the comments provided by the Joint Board and make modifications to the Design Documents based on such comments in accordance with Scope of Work Sections—the design review process in the RFP. Contractor acknowledges that comments may be provided which reflect concerns regarding operability or preferences of the commenter or which otherwise do not directly relate to specific requirements of the Contract Documents. Contractor agrees to undertake reasonable efforts to accommodate or otherwise resolve any such comments through the review process described in Scope of Work Section—the RFP. The foregoing shall in no way be deemed to obligate Contractor to incorporate any comments that would result in a significant disruption to its schedule or a significant increase in its costs, except pursuant to a Joint Board-Directed Change.

13.2.1.4.14.2.1.4. Contractor shall construct, fabricate, assemble, provide, install, integrate, test and verify all aspects of the Project in accordance with the Final Design Documents and Construction Documents. The Final Design Documents may be changed only with prior written approval of the Joint Board in its sole discretion. Contractor may make minor modifications to the Construction Documents without prior written approval of the Joint Board, but must deliver the modifications to the Joint Board in advance of performance of the Work.

13.2.2.14.2.2. Contractor Responsibility for Design

Contractor agrees that it has full responsibility for the design of the Project and that Contractor will furnish the design of the Project, regardless of the fact that with respect

- to certain Project Segments, Owner Design Documents, including, in certain circumstances, a Schematic Design, may be provided to Contractor as a preliminary basis for Contractor's design. Contractor specifically acknowledges and agrees that:
- (a) Contractor is not entitled to rely on (i) any Owner Design Documents except as specified in <u>Section 14.2.2</u>, (ii) the Reference Documents or (iii) any other documents or information provided by the Joint Board, except to the extent specifically permitted in the Contract Documents.
- (b) Contractor is responsible for correcting any Errors in the Owner Design Documents through the design and/or construction process without any increase in any Price or extension of a Completion Deadline for the relevant Work, subject only to the right to a Change Order with respect to any Necessary Basic Configuration Changes to the extent permitted by Section 14.8.
- (c) The Joint Board's liability for Errors in any Owner Design Documents is limited to its obligations relating to Necessary Basic Configuration Changes, and is subject to the requirements and limitations of <u>Section 14</u>.
- (d) Contractor's warranties and indemnities hereunder cover Errors in the Project even though they may be related to Errors in the Owner Design Documents.
- (e) Contractor is responsible for verifying all technical information contained in the RFP Documents or otherwise provided by the Joint Board with respect to a Project Segment.

13.2.3.14.2.3. Owner Design Documents

Contractor may rely on the Basic Configuration elements as shown on any Owner Design Documents as representing a feasible design solution for the Project and that it is feasible to develop the Project within the Schematic ROW limits identified in the Owner Design Documents, and shall have the right to obtain a Change Order for Necessary Basic Configuration Changes as provided in <u>Section 4.7</u>.

Contractor acknowledges that any Owner Design Documents are preliminary and subject to refinement through the Final Design process, and that Contractor's entitlement to an increase in a Price or time extension in connection with any changes in Owner Design Documents is limited to Necessary Basic Configuration Changes.

13.3. 14.3. Professional Licensing and Registration

Prior to beginning any work on the Project:

1. An Engineer of Record performing work in Kentucky shall be a Licensed Professional Engineer in the Commonwealth and an Engineer of Record performing work in Indiana shall be a Registered Professional Engineer in Indiana. The land surveyor, if any, shall be a Licensed Professional Land Surveyor in the Commonwealth and a Registered Professional Land Surveyor in Indiana.

4.2. The Joint Board does not intend to contract for, pay for, or receive any design services which are in violation of any professional licensing laws, and by execution of this Contract, the Contractor acknowledges that the Joint Board has no such intent. It is the intent of the Parties that the Contractor is fully responsible for furnishing the design of the Project, although the fully licensed design firm(s) or individuals designated herein will perform the design services required by the Contract Documents. Any references in the Contract Documents to Contractor's responsibilities or obligations to "perform" the design portions of the work shall be deemed to mean that the Contractor shall "furnish" the design for the Project.

13.4.14.4. Use and Ownership of Documents [

13.4.1. 14.4.1. Ownership of Documents

Upon preparation or receipt thereof –the Joint Board (or its designee) shall receive ownership of the property rights, except for copyrights in Software, of all design-related documents, drawings, specifications, <u>Software and</u> Software licenses, electronic data and information prepared, provided or procured by the Contractor, its design partners, as well as all consultants and Subcontractors of any tier, for the Project ("Design Documents"). Construction Documents shall become the Joint Board's property upon delivery to the Joint Board, and other documents prepared or obtained by Contractor in connection with performance of its obligations under the Contract Documents, including studies, manuals, as-built drawings, technical and other reports and the like, shall become the property of the Joint Board upon Contractor's preparation or receipt thereof. The Joint Board may use, reproduce and make derivative works from said Design Documents for subsequent renovation or augmentation of the work. The Contractor grants the Joint Board any permissions or licenses pursuant to copyright or other intellectual property laws to effectuate this transfer of ownership.

Contractor grants the Joint Board and each of its member States' and Party and their agencies (as an exception to the transfer and assignment provided in the preceding paragraph of this Section 14.4), an irrevocable, permanent, nonexclusive, world-wide royalty-free right and license to use, execute, reproduce, display, perform, and distribute internally copies of, and prepare derivative works based upon, the Software, the System design and technology and/or any other System idea, concept or deliverable, and the right to authorize third parties to do any of the foregoing, subject to the execution of a commercially reasonable non-disclosure agreement by such third party and further subject to the terms of this Agreement and any separate and specific licensing agreement entered into between the Joint Board and the Contractor. The foregoing licenses and rights shall be used solely as needed (i) to operate, maintain and support the System and the Project and at any toll bridge or road or similar or related facilities in which such Software, System design and technology and any other System idea, concept and deliverables are used under or pursuant to this Agreement (whether by the Joint Board, a member State or State agency, or an Other Joint Board eContractor, including but not limited to the ETC Contractor and the Operations Contractor, including at the toll bridges and roads and in the toll traffic lanes that utilize such System and at the Project Sites specified in this Agreement, and (ii) in the event of an Event of Default or the failure, unwillingness or inability of Contractor to perform its

obligations under the Contract Documents; and (iii) in the event of an Event of Default or the failure, unwillingness or inability of Contractor to perform its obligations under the Contract Documents, at any other toll roads, toll traffic lanes or similar or related facilities owned, operated or managed by the Joint Board, or a member StateStates' Party or Stateits agencyies. The foregoing rights and licenses shall survive the expiration or termination of this AgreementContract.

The Joint Board and its member the States States' Parties and their respective agencies may also use, reproduce and make derivative works from the Design Documents (including licensed Software) in connection with the Joint Board's and/or its member States' and their respective agencies design of future projects. The Contractor grants the Joint Board and its member States and their respective agencies any permissions or licenses pursuant to copyright or other intellectual property laws to effectuate this grant of permission. The Joint Board similarly may distribute or make the Design Documents available to third parties who will or may provide services to the Joint Board or its member States and their respective agencies on future projects. The Joint Board's and its' member States and their respective agencies' use of the Design Documents either by itself or by distribution to third parties without Contractor's involvement or on other projects is at the Joint Board's, its member States' and their respective agencies' sole risk.

Subject to any requirements in the Contract Documents for Contractor's delivery of specific documents, not less frequently than monthly, Contractor shall provide the Joint Board with a detailed written list of all Design Documents and Construction Documents prepared or received by Contractor, and Contractor shall furnish the Joint Board with copies of any such Design Documents and Construction Documents immediately upon the Joint Board's written request. Contractor shall maintain all other documents described in this <u>Section 14.4</u> and shall deliver copies to the Joint Board as required by the Contract Documents or upon request if not otherwise required to be delivered, with an indexed set delivered to the Joint Board as a condition to Formal Acceptance.

13.4.2.14.4.2. Commercial Off-the-Shelf Software

The Joint Board's interest in and obligations with respect to any COTS incorporated into the Project or System shall be determined in accordance with the standard license terms applicable to such Software; provided, however, that Contractor shall be solely responsible for all costs associated with such items and in no event shall Contractor change the terms of such agreements without the Joint Board's written approval.

13.4.3.14.4.3. Contractor's License

The Joint Board grants the Contractor a limited, exclusive license to use, reproduce and make derivative works of the Design Documents provided by the Contractor, however, the Contractor shall indemnify and hold harmless the Joint Board from and against any and all claims, damages, losses, costs and expenses arising out of the or resulting from

Contractor's use of the Design Documents. A similar exclusive license may be granted to Contractor's consultants and subconsultants.

If this Contract is terminated for convenience pursuant to <u>Section 10</u>, the Joint Board shall receive ownership of the property rights_, <u>except for copyrights in the Software and Source Code</u>, of the Design Documents upon current payment for all work associated with design performed in accordance with this Contract, at which time the Joint Board shall have the right to use, reproduce and make derivative works from the Design Documents to complete the work or as described above.

13.4.4.14.4.4. Source Code Escrow

[WHILE THE RFP INDICATES A STRONG PREFERENCE FOR COTS, IT SEEMS SOME NEW SOFTWARE CODE WILL NEED TO BE WRITTEN FOR THE PROJECT IN WHICH CASE AN ESCROW IS APPROPRIATE UNLESS IT WILL ALL BE OWNED BY AND DELIVERED IMMEDIATELY TO THE JOINT BOARD, WHICH IS UNLIKELY IN THESE TYPES OF TRANSACTIONS.]

The Joint Board and Contractor acknowledge that Contractor and/or Contractor's Software Suppliers may not wish to disclose release directly to the Joint Board the Software Source Code which is pre-existing works (not for Software owned by the Joint Board pursuant to Section 15.5.1), as public disclosure could deprive Contractor and/or Contractor's Software Suppliers of commercial value, but that the Joint Board must be ensured access to such Software Source Code in certain circumstances to permit it to realize the benefits contemplated by this Contract. Therefore, as a condition to payment of certain payment requests, and Substantial Completion Acceptance, Contractor shall place all the Software Source Code owned by Contractor, licensed to or by Contractor or with respect to which Contractor has a right to use in connection with the System or this Project in a Source Code Escrow with a mutually acceptable Code Escrow Agent engaged in the business of receiving and maintaining escrows of Software Source Code, related documentation, and other technology. With regard to Software and updates developed as part of the Maintenance Work, such disclosure and delivery to the Source Code Escrow shall occur at the earlier of (x) when required in connection with a Maintenance Payment request or (y) such time as the Software and updates are implemented. The terms of the Source Code Escrow shall be as set forth in Exhibit -K, provided that the Joint Board shall be assured access to release of the contents of the Software Source Code Escrow in the event any of the following events have occurred at the following times: (i) occurrence of a business failure (including cessation of business, bankruptcy, or insolvency) of the Contractor and/or Software Supplier; (ii) failure of the Contractor and/or Software Supplier has failed to provide or is unable or unwilling to provide the services as necessary to permit continued use of the Software by the Joint Board; the Joint Board Contractor that has acquired the System pursuant to the terms of this Contract) as contemplated by the Contract Documents; or (iii) in connection with a partial or complete termination of this Contract as a result of an Contractor Event of Default, or expiration of the Term. Contractor shall cause the Software Supplier to keep the Software Source Code up-todate by depositing all enhancements and modifications into the Source Code Escrow

during the term of its existence as such enhancements and modifications are developed, completed or implemented. The Contractor shall make all deposits into the Source Code Escrow in such manner as to cause all escrowed materials to reflect the current version of well as one (1) prior version of all materials in operation. All versions shall be clearly marked as to "current" versus "prior" version.

For purposes hereof, the "Software Source Code" means the software (b) written in programming languages, such as C and Fortran, including all comments and procedural code, such as job control language statements, in a form intelligible to trained programmers and capable of being translated into object or machine readable code for operation on computer equipment through assembly or compiling, and accompanied by documentation, including flow charts, schematics, statements of principles of operations, and architectural standards, describing the data flows, data structures, and control logic of the Software, including but not limited to the Contractor's version of compilers used in connection with the - Software -in sufficient detail to enable a trained programmer through study of such documentation to maintain and/or modify the Software without undue experimentation, and such other documentation and executables required to independently maintain the Software. and necessary information to build and replicate any specialized hardware, but excludes source code for Off-the-Shelf Software unless such source code is available to the Contractor. The term "Software Source Code" as it pertains to the documentation required to deposited into the Source Code Escrow means and includes everything necessary to allow a technical verification on the Software and other related assets to verify the Source Code files, executable files, database schema, tables, configuration files, batch files, documentation and directions are identical to that licensed to the Joint Board.

13.4.5. 14.4.5. Rights on Termination

If this Contract is terminated for any reason other than convenience pursuant to <u>Article 910</u>, the Joint Board shall receive ownership of the property rights, except for <u>Software</u> copyrights[?], of the Design Documents, at which time the Joint Board shall have the right to use, reproduce and make derivative works from the Design Documents to complete the work or as described above.

13.4.6.14.4.6. Open Records Laws

The Contractor acknowledges that the Joint Board is subject to Kentucky and Indiana Open Records laws and may be required to produce copies of all Design Documents upon request by the general public. The Contractor waives any and all claims against the Joint Board or its agents in connection with such duty by the Joint Board.

13.4.7. 14.4.7. Subcontracts

The Contractor shall obtain from its engineers and architects, as well as their Subcontractors and consultants of any tier, the rights of use that correspond to the rights given by the Contractor to the Joint Board in this Contract.

13.5.14.5. Information Supplied

The Joint Board has made available to the Contractor information which is described in the Contract Documents and certain reference documents (which are not included in the Contract Documents) regarding the Project and surrounding areas, and has allowed the Contractor access to the <u>Project Seite</u> for purposes of inspection and testing.

13.6.14.6. Responsibility for Design

The Contractor agrees that it has full responsibility for preparation and submission of the Design Documents:

- 1. The Contractor acknowledges its responsibilities to design the Project in conformance with the Project Scope of Work and the other Contract Documents.
- 2. The Contractor is not entitled to rely on and has not relied on any other documents or information provided by the Joint Board, except to the extent specifically permitted in the Contract Documents, or that the Contractor does so at its own risk.
- 3. The Contractor is responsible for correcting any errors through the design and/or construction process without any increase in the Contract Price or extension of the Contract Time.

13.7. 14.7. No Liability Regarding Reference Documents

The Joint Board does not represent or warrant that the information contained in the reference documents is either complete or accurate or that such information conforms with the requirements of the Contract Documents. The Contractor understands and agrees that the Joint Board shall not be responsible or liable in any respect for any loss, damage, injury, liability, cost or cause of action whatsoever suffered by any Contractorrelated entity by reason of any use of any information contained in the reference documents or any action or forbearance in reliance thereon, except to the extent that the Joint Board has specifically agreed herein that the Contractor shall be entitled to an increase in the Contract Price and/or extension of a Completion Deadline with respect to such matter. The Contractor further acknowledges and agrees that (a) if and to the extent the Contractor or anyone on Contractor's behalf uses any of said information in any way, such use is made on the basis that the Contractor, not the Joint Board, has approved and is responsible for said information, and (b) the Contractor is capable of conducting and is obligated hereunder to conduct any and all studies, analyses and investigations as it deems advisable to verify or supplement said information, and that any use of said information is entirely at Contractor's own risk and at its own discretion.

13.8. 14.8. Contractor Representations

To induce the Joint Board to enter into this agreement, the Contractor makes the following representations:

1. The Contractor has examined and carefully studied the Contract Documents and the other related data identified in the RFP;

- 2. The Contractor has visited the <u>Project sSite</u> and become familiar with and is satisfied as to the general, local, and <u>Project Site</u> conditions that may affect cost, progress, and performance of the <u>wW</u>ork;
- 3. The Contractor is familiar with and is satisfied as to all federal, state, and local Laws and regulations that may affect cost, progress, and performance of the wWork.
- 4. The Contractor has carefully studied all: (a) reports of explorations and tests of subsurface conditions at or contiguous to the site and all drawings of physical conditions relating to existing surface or subsurface structures at the Project_sSite, if any, that have been identified or made available by the Joint Board and (b) reports and drawings of environmental conditions, if any, at the Project_sSite that have been identified or made available by the Joint Board.
- 5. The Contractor is aware of the general nature of work to be performed by the Joint Board and others at the <u>Project sSite</u> that relates to the work as indicated in the Contract Documents.
- 6. The Contractor has considered the information known to the Contractor; information commonly known to contractors doing business in the locality of the Project; information and observations obtained from visits to the Project_sSite; the Contract Documents; and the Project_sSite; related reports and drawings identified in the Contract Documents, with respect to the effect of such information, observations, and documents on (1) the cost, progress, and performance of the work; (2) the means, methods, techniques, sequences, and procedures of construction to be employed by the Contractor, including any specific means, methods, techniques, sequences, and procedures of construction expressly required by the Contract Documents; and (3) Contractor's safety precautions and programs.
- 7. Based on the information and observations referred to above, the Contractor does not consider that further examinations, investigations, explorations, tests, studies, or data are necessary for it to enter into this Contract at the Contract Price, within the Contract Time, and in accordance with the other terms and conditions of the Contract Documents.
- 8. The Contractor has given the Joint Board written notice of all conflicts, errors, ambiguities, or discrepancies that the Contractor has discovered in the Contract Documents, and the written resolution thereof by the Joint Board is acceptable to the Contractor.
- 9. The Contract Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance and furnishing of the <u>wW</u>ork.

13.9. 14.9. Notification of Third-Party Claims

The Joint Board and the Contractor shall each provide timely written notification to the other Party of the receipt of any third-party claim relating to, arising out of, or connected with *Work performed by or on behalf of the Contractor under this Contract.

44.015.0 Obligations of the Contractor

14.1.15.1. Performance Requirements

14.1.1.15.1.1. Performance of Work

All materials, services and efforts necessary to achieve <u>sSubstantial eCompletion Acceptance</u>, <u>Punch List Acceptance and Formal Acceptance and Final Completion of the Operations Center, the Walk-in Centers, the Crossings and the System on or before the applicable <u>Completion dD</u>eadlines shall be <u>solely Contractor's sole</u> responsibility, except as otherwise specifically provided in the Contract Documents. The Contractor agrees that the costs of all such materials, services and efforts are included in the Contract Price.</u>

14.1.2.15.1.2. Performance as Directed

At all times during the term hereof, including during the course of, and notwithstanding the existence of, any \$\overline{\Delta}\text{dispute}\$, the Contractor shall perform as directed in writing by the Joint Board in a diligent manner and without delay, shall abide by the Joint Board's decision or order, and shall comply with all applicable provisions of the Contract Documents. If a \$\overline{\Delta}\text{dispute}\$ arises regarding such performance or direction, the \$\overline{\Delta}\text{dispute}\$ shall be resolved in accordance with Article 123.

14.2.15.2. General Obligations of the Contractor

The Contractor, in addition to performing all other requirements of the Contract Documents, shall:

- 1. (1)-furnish all design and other services, provide all materials and labor and undertake all efforts necessary or appropriate (excluding only those services, materials and efforts which the Contract Documents specify will be undertaken by other persons): (i) to construct the Project and maintain it during construction and installation and during the Maintenance Term in accordance with the Contract Documents, all laws, all Governmental Approvals, good industry practice, and all other applicable safety, environmental, licensing and other requirements falso list quality/safety plans required by the Contractor? of the Contract Documents, including but not limited to the Quality Assurance Plan and any other plans and procedures that Contractor is required to implement to ensure the quality and safety of the Work, so as to achieve Substantial Completion Acceptance, Punch List Acceptance and Formal Acceptance by the applicable Completion Deadlines; (ii) to construct the Project free from material defects except to the extent that such defects are inherent in prescriptive specifications included in the Contract Documents, unless (a) the Contractor has actual or constructive knowledge of such defects and (b) the Contractor fails to notify the Joint Board of the defect; and (iii) otherwise to do everything required by and in accordance with the Contract Documents:
- 2. at all times provide Contractor's Project Manager, who (i) will have full responsibility for the prosecution of the work, (ii) will act as agent and be a single point of contact in all matters on behalf of the Contractor, (iii) will be present (or its Approved designee will be present) at the Project sSite at all times that work is performed, and (iv) will have authority to bind the Contractor on all matters relating to the Project;

- 3. obtain all Governmental Approvals, including any revision, modification, amendment, supplement, renewal or extension of Governmental Approvals, required in connection with the Project or the Work;
- 4. comply with all conditions imposed by and undertake all actions required by and all actions necessary to maintain in full force and effect all Governmental Approvals, including implementation of all environmental mitigation measures required by the Governmental Approvals and Contract Documents, except to the extent that such responsibility is expressly assigned in the Contract Documents to another party;
- 5. provide such assistance as is reasonably requested by the Joint Board in dealing with any party and/or in prosecuting and defending lawsuits in any and all matters relating to the Project, which may include providing information and reports regarding the Project, executing declarations and attending meetings and hearings, but which shall in no event be deemed to require the Contractor to provide legal services;
 - 6. comply with any plans or submittals required by the Contract Documents:
- 7. cooperate with the Joint Board and other governmental entities with jurisdiction over the Project in the review and oversight of the Project and other matters relating to the work;
- 8. supervise and be responsible to the Joint Board for acts and omissions of all Contractor-Related Entities, as though all such entities were directly employed by the Contractor;
- 9. mitigate delay to the Project and mitigate damages due to delay to the extent possible, including by resequencing, reallocating or redeploying Contractor's forces to other work, as appropriate;
- 10. prior to delinquency pay all applicable federal, State and local sales, consumer, use and similar taxes, property taxes and any other taxes, fees, charges or levies imposed by governmental entities, whether direct or indirect, relating to, or incurred in connection with, the performance of the work; and
- 11. provide appropriate security for the Project <u>sSite</u>, and take all reasonable precautions and provide protection to prevent damage, injury, or loss to the work and materials and equipment to be incorporated therein, as well as all other property at the Site, whether owned by Contractor, the Joint Board or any other Person.

14.3. 15.3. Representations, Warranties and Covenants

The Contractor represents, warrants and covenants to the Joint Board as described in this Section. The representations, warranties and covenants in this Section shall survive expiration or earlier termination of this Contract.

14.3.1.15.3.1. Maintenance of Professional Qualifications; Qualifications of Employees

The Contractor and its design Subcontractor(s) have maintained, and throughout the term of this Contract and its design Subcontract(s) shall maintain, all required Joint Board, license status, professional ability, skills and capacity to perform the work, and

shall perform them in accordance with the requirements of the Contract Documents. If the Joint Board determines, in its sole discretion, that any Person employed by Contractor or any Subcontractor is not performing the work in a proper, desirable and skillful manner or is detrimental to the progress of the work and/or the Project, then, at the written request of the Joint Board, Contractor shall remove such Person from the Project and such Person shall not be reemployed on the Project without the prior written approval of the Joint Board.

14.3.2.15.3.2. Evaluation of Constraints

The Contractor has evaluated the constraints affecting delivery of the Project and has reasonable grounds for believing and does believe that the Project can be delivered within such constraints.

14.3.3.15.3.3. Feasibility of Performance

The Contractor has considered the information known to the Contractor; information commonly known to DBTs[?] dContractors doing business in the locality of the Project; information and observations obtained from visits to the Project sSite; the Contract Documents; and the Project sSite-related reports (which are not made a part of the Contract Documents) and drawings identified in the Contract Documents, with respect to the effect of such information, observations, and documents on (1) the cost, progress, and performance of the work; (2) the means, methods, techniques, sequences, and procedures of construction to be employed by the Contractor, including any specific means, methods, techniques, sequences, and procedures of construction expressly required by the Contract Documents; and (3) Contractor's safety precautions and programs. The Contractor has evaluated the feasibility of performing the work by the Completion Deadlines and for the Contract Price, and has reasonable grounds for believing and does believe that such performance is feasible and practicable.

14.3.4.15.3.4. Review of Site Information

To the extent the Contractor deems it necessary or advisable for submittal of a Proposal, the Contractor has, prior to submitting its Proposal, in accordance with good industry practice, undertaken appropriate activities sufficient to familiarize itself with surface conditions and subsurface conditions that may be discernible from the surface. Said activities have included reasonable inspection and examination of the Project Site and surrounding locations. Based on its review, inspection, examination and other activities, the Contractor is familiar with and accepts the physical requirements of the work, subject to the right to receive a Change Order for only those situations as specified in Article 4. The Contractor does not consider that further examinations, investigations, explorations, tests, studies, or data are necessary for it to enter into this Contract for the performance of the work, including all design, at the Contract Price, within the Contract Time, and in accordance with the other terms and conditions of the Contract Documents.

14.3.5.15.3.5. Review of Contract Documents

The Contract Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance and furnishing of the work. The Contractor has given the Joint Board written notice of all conflicts, errors, ambiguities, or discrepancies that the Contractor has discovered in the Contract Documents, and the written resolution thereof by the Joint Board is acceptable to the Contractor.

14.3.6.15.3.6. Governmental Regulations and Approvals

The Contractor is familiar with and is satisfied as to all federal, state, and local laws and regulations that may affect cost, progress, and performance of the work. After due consideration of all relevant factors, the Contractor believes that all Governmental Approvals will be granted in due course and will thereafter remain in effect so as to enable the work to proceed in accordance with the Contract Documents.

14.3.7.15.3.7. Organization

The Contractor is a ______ (type of legal entity) duly organized and validly existing under the laws of the State of _____ with all requisite power to own its properties and assets and carry on its business as now conducted or proposed to be conducted. The Contractor is duly qualified to do business, and is in good standing, in the Commonwealth of Kentucky and the State of Indiana, and will remain in good standing throughout the term of this Contract and for as long thereafter as any obligations remain outstanding under the Contract Documents.

14.3.8.15.3.8. Authorization

The execution, delivery and performance of this Contract have been duly authorized by all necessary actions of the Contractor, and, if applicable, Contractor's members, and will not result in a breach or a default under the organizational documents of any such entity or any indenture, loan, credit agreement, or other material agreement or instrument to which any such entity is a party or by which their properties and assets may be bound or affected.

14.3.9.15.3.9. Legal, Valid and Binding Obligation

The Contract constitutes the legal, valid and binding obligation of the Contractor and, if applicable, of each member of the Contractor, enforceable in accordance with its terms.

14.3.10.15.3.10. False or Fraudulent Statements and Claims

The Contractor recognizes that the requirements of the Program Fraud Civil Remedies Act of 1986, as amended, 31 U.S.C. section 3801 et seq., the USDOT regulations, "Program Fraud Civil Remedies," and 49 C.F.R. Part 31 apply to its actions hereunder. Accordingly, by signing the Contract, the Contractor certifies or affirms the truthfulness and accuracy of any statement it has made, it makes, or it may make pertaining to its performance of the work. In addition to other penalties that may be applicable under Kentucky or Indiana state law, the Contractor also acknowledges that if it makes a false, fictitious, or fraudulent claim, statement, submission, or certification, the federal government reserves the right to impose the penalties of the Program Fraud Civil

Remedies Act of 1986, as amended, on the Contractor to the extent the federal government deems appropriate.

14.3.11. **Design Work**

Any design Work furnished by Contractor shall be performed by personnel who are careful, skilled, experienced and competent in their respective trades or professions, who are professionally qualified to perform the Work in accordance with the Contract Documents and who shall assume professional responsibility for the accuracy and completeness of the Design Documents and Construction Documents prepared or checked by them.

14.3.12.15.3.12. Scheduling Work

Contractor shall at all times schedule and direct its work to provide an orderly progression of the work to achieve, Substantial Completion Acceptance, Punch List Acceptance and Formal Acceptance by the applicable Completion Deadlines and in accordance with the Schedule, including furnishing such employees, materials, facilities and equipment and working such hours, extra shifts, overtime operations, Sundays and holidays as are permitted by the Contract Documents and which may be necessary to achieve such goal, all at Contractor's sole cost, except as otherwise specifically provided in Section 4.

14.3.13.15.3.13. Rights in Software

With respect to Software, (a) Contractor has and will have sufficient right to assign or grant the rights and/or licenses granted in the Software pursuant to this Contract (and Contractor acknowledges that all Software that is developed in whole or in part for toll collection shall be covered by the licensing requirements of Section 14.4.1——, (b) all Software, except any preexisting works addressed in Section hereof, have not been and will not be used or published under circumstances which have caused or will cause a loss of copyright, patent rights, trademark or other intellectual property right therein IRFP INDICATES ALL SOFTWARE WILL BE COTS; VERIFY THAT NO PROVISIONS ARE REQUIRED FOR ORIGINAL WORKS OF AUTHORSHIP, AND THAT THE JOINT BOARD DOESN'T INTEND TO OWN EXCLUSIVE RIGHTS TO ANY SOFTWARE], and (c) all Software, including all preexisting works addressed in Section hereof, does not and will not infringe any patents, copyrights, trademarks or other intellectual property rights (including trade secrets), privacy or similar rights of any third party, nor is any claim (whether or not embodied in an action, past or present) of such infringement pending, been asserted or, to the best of Contractor's knowledge, been threatened against Contractor (or, insofar as Contractor is aware, any entity from which Contractor has obtained such rights). The representations and warranties set forth in this Section 15.3.13—— shall survive the Maintenance Term and shall survive the expiration or termination of this Contract

14.3.14.15.3.14. Protection of Customer Information

All customer information and data to which Contractor may have access or which Contractor may obtain in connection with performance of the work hereunder, and other

Project information generated in connection with this Contract, is and shall be the sole property of the Joint Board. Contractor shall at all times maintain the strict confidentiality of such information. Contractor shall have no right to sell, transfer, disclose or otherwise use such information for any purpose other than in performance of its duties hereunder. Contractor acknowledges and agrees that privacy of the customers is of paramount importance to the Joint Board and the traveling public and that Contractor will comply with all applicable Laws concerning privacy and confidentiality of personal information obtained in the course of the work performed under this Contract.

14.3.15.15.3.15. No Debarment

Contractor is not currently debarred, suspended, or otherwise excluded from or ineligible for participation in Federal Programs under Executive Order 12549 and further certifies that it will not do business with any Person that is currently debarred, suspended, or otherwise excluded from or ineligible for participation in Federal Assistance Programs under Executive Order 12549. Contractor shall require each Subcontractor to certify its eligibility to receive federal funds and, when requested by the Joint Board, shall promptly furnish a copy of such certification to the Joint Board.

14.3.16.15.3.16. Title

Subject to the provisions of Section with respect to intellectual property rights, Contractor warrants that it owns, or will own, and has, or will have, good and marketable title to all materials, equipment, tools and supplies furnished, or to be furnished, by it and its Subcontractors that become part of the Project or are purchased for the Joint Board for the operation, maintenance or repair thereof, free and clear of all Liens. Title to all of such materials, equipment, tools and supplies which shall have been delivered to the Project Site shall pass to the Joint Board, free and clear of all Liens, upon the sooner of (a) incorporation into the Project, or (b) payment by the Joint Board to Contractor of invoiced amounts pertaining thereto. Notwithstanding any such passage of title, Contractor shall retain sole care, custody and control of such materials, equipment, tools and supplies and shall exercise due care with respect thereto until the applicable Formal Acceptance Date, or until Contractor is removed from the Project.

14.4.15.4. Design Requirements

14.4.1. 15.4.1. Required Approval

Approval by the Joint Board is required prior to commencing any work that would necessitate a modification to the final design documentation or the project description contained in the Contract Documents, regardless of whether the modification is required by a Governmental Approval or is desired by the Contractor for its benefit or for any other reason.

14.4.2.15.4.2. Third Party Design Review Process

The Design Review Process shall be conducted as described in Section ____ of the RFP.

15.4.2.1.Design Reviews Required by Third Parties

The Contractor, working with the Joint Board, shall be responsible for giving and obtaining all design reviews required by utility owners, governmental entities, railroad owners and any other entities other than the Joint Board, as applicable.

14.5.15.5. Governmental Approvals and Permits

The Contractor shall procure all permits, licenses, inspections, and memberships; pay all charges, fees, and taxes; and give all notices necessary and incidental to the due and lawful prosecution of the work. The Joint Board or its members may have acquired environmental permits, including, but not limited to, U.S. Army Corps of Engineers Permit, IDNR Certificate of Approval of Construction in a Floodway, IDEM Section 401 Water Quality Certification, or a permit for construction of temporary pavement across a state line. If the Joint Board or its members have acquired one or more of such permits, the restrictions or conditions that were issued with such permits shall be made available to proposers prior to award. The Contractor shall prosecute the work in accordance with all such restrictions or conditions.

14.6.15.6. Design and Engineering Personnel.

All design and engineering work furnished by the Contractor shall be performed by or under the supervision of persons licensed to practice architecture, engineering or surveying (as applicable) in Kentucky for work to be performed in Kentucky and in Indiana for work to be performed in Indiana, and by personnel who are careful, skilled, experienced and competent in their respective trades or professions, who are professionally qualified to perform the work in accordance with the Contract Documents, and who shall assume professional responsibility for the accuracy and completeness of the design-related Contract Documents.

14.7.15.7. Progression of Work

Contractor shall at all times schedule and direct its work to provide an orderly progression of the work to achieve <u>sSubstantial eCompletion Acceptance</u>, <u>Punch List Acceptance and Formal Acceptance and Final Completion</u> by the applicable Completion Deadlines, including furnishing such employees, materials, facilities and equipment and working such hours (including extra shifts and overtime operations) as may be necessary to achieve such goals, all at Contractor's own expense, except as otherwise specifically provided in the Contract Documents.

14.8. 15.8. Contractor Responsible for Means and Methods

Contractor shall supervise, inspect, and direct the work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to provide the work in accordance with the Contract Documents. The Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of the work. Contractor shall be responsible to see that the completed work complies fully with the Contract Documents.

14.9.15.9. Services, Materials and Equipment

Unless otherwise specified in the Contract Documents, Contractor shall furnish or cause to be furnished and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start-up, and completion of the work.

14.10.15.10. Limitation on Use of Site and Other Areas

Contractor shall confine construction equipment, the storage of materials and equipment, and the operations of construction workers to the Project_sSite and other areas permitted by laws or regulations, and shall not unreasonably encumber the Project_sSite and other areas with construction equipment or other materials or equipment. Contractor shall assume full responsibility for any damage to any such land or area, or to owner or occupant thereof or of any adjacent land or areas, resulting from the performance of the work. Should any claim be made by any such owner or occupant because of the performance of work, Contractor shall first promptly notify the Joint Board of such claim in writing and then shall promptly settle with such other party by negotiation or otherwise resolve the claim by arbitration or other Delispute resolution proceeding or at law. Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall the Contractor subject any part of the work or adjacent property to stresses or pressures that will endanger it.

14.11.15.11. Safety and Protection

Contractor shall be solely responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety laws and regulations. The Contractor shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury, or loss to:

- 1. All persons on the <u>Project Site</u> or who may be affected by the work;
- 2. All the work and materials and equipment to be incorporated therein, whether in storage on or off the <u>Project Site</u>; and
- 3. Other property at the <u>Project Site</u> or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, utilities and underground facilities not designated for removal, relocation, or replacement in the course of construction.

Contractor shall comply with applicable Laws and regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and shall erect and maintain all necessary safeguards for such safety and protection. The Contractor shall notify owners of adjacent property and of underground

facilities and utility owners when prosecution of the work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property. Contractor shall comply with the applicable requirements of Kentucky's and Indiana's safety programs. The Contractor shall inform the Joint Board of the specific requirements of Contractor's safety program applicable to the Joint Board and its employees and representatives must comply while at the Project_sSite. All damage, injury, or loss to any property referred to in this Section 145.11 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor. Contractor's duties and responsibilities for safety and for protection of the construction shall continue until such time as all the work is completed and the Joint Board has made Formal Acceptance of the work.

14.12.15.12. Safety Representatives

Contractor shall designate qualified and experienced safety representatives located on Site whose duties and responsibilities shall be the prevention of accidents and the maintaining and supervising of safety precautions and programs.

14.13.15.13. Hazard Communications Programs

Contractor shall be responsible for coordinating any exchange of material safety data sheets or other hazard communication information required to be made available to or exchanged between or among employers at the Project_Site in accordance with laws or regulations.

14.14.15.14. Emergencies

In emergencies affecting the safety or protection of persons or the work or property at the <u>Project</u> Site or adjacent thereto, Contractor is obligated to act to prevent threatened damage, injury or loss.

14.15.15. Related Work at Project Site

The Joint Board and the Joint Board members have contracted with other entities to design and construct the Operations Center, Walk-in Centers and BridgesCrossing structures. Additionally the Joint Board and its members may perform other work at the Project Site. The Design-Builder and the Developer will afford the Contractor proper and safe access to the Project sSite and a reasonable opportunity for the introduction and storage of materials and equipment. The Contractor shall properly connect and coordinate the Work with other contractors working in the area. Unless otherwise provided in the Contract Documents, Contractor shall do all cutting, fitting, and patching of the Work that may be required to make its several parts come together properly and integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that the Contractor may cut or alter others' work with the written consent of the Joint Board and the others whose work will be affected. The duties and responsibilities of the Contractor

under this paragraph are for the benefit of such other contractors to the extent that there are comparable provisions for the benefit of Contractor in said direct contracts between the Joint Board and such other contractors. [DISCUSS/REVISE IN LIGHT OF PPA AND DB CONTRACT]

14.16.15.16. No Release of the Contractor Obligations

Contractor's obligation to perform and complete the work in accordance with the Contract Documents shall be absolute. None of the following will constitute an acceptance of work that is not in accordance with the Contract Documents or a release of Contractor's obligation to perform the work in accordance with the Contract Documents, or release the Contractor from its duty to correct defects:

- 1. Observations by the Joint Board;
- 2. The making of any progress or final payment;
- 3. The issuance of a certificate of \underline{sS} ubstantial \underline{eC} ompletion $\underline{Acceptance}$ or Formal Acceptance;
 - 4. Use or occupancy of the work or any part thereof by the Joint Board;
 - Any review and approval of a Ssubmittal;
 - 6. Any inspection, test, or approval by others; or
 - 7. Any correction of defective Construction by the Joint Board.

14.17.15.17. Maintenance During Construction

Contractor shall be responsible for maintenance of the wWork and the Project Site in accordance with Scope of Work Sections at all times prior to Substantial Completion Acceptance of the Operations Center, each Walk-in Center and each Crossing, and prior to System Substantial Completion Acceptance for any Work not included within a Crossing Substantial Completion Acceptance. Contractor shall maintain, rebuild, repair, restore or replace all wWork, including Design Documents, Construction Documents, materials, equipment, supplies and maintenance equipment which are purchased for permanent installation in, or for use during construction of the Project that is injured or damaged prior to the date Contractor's maintenance responsibility ends, regardless of who has title thereto under the Contract Documents and regardless of the cause of the damage or injury, at no additional cost to the Joint Board, except to the extent that the Joint Board is responsible for such costs in accordance with the express terms of this Contract. Contractor, at its cost, shall also have sole responsibility during such periods for rebuilding, repairing and restoring all other property within the ROW whether owned by Contractor, the Joint Board or any other Person.

If insurance proceeds with respect to any loss or damage for which Contractor is responsible for the rebuilding, repair or restoration thereof are paid to the Joint Board, then the Joint Board shall arrange for such proceeds to reimburse Contractor as repair or replacement work is performed by Contractor to the extent that the Joint Board has

not previously paid for such repair or replacement work; provided, however, that release of such proceeds to Contractor shall not be a condition precedent to Contractor's obligation to perform such replacement or repair work or indicate that such replacement or repair work has been approved and accepted by the Joint Board.

14.18. 15.18. Maintenance Obligations After System Acceptance

14.18.1.15.18.1. Contractor's Obligation to Maintain and Repair

During the Maintenance Term, Contractor shall be responsible for all maintenance of the Work and the System, and shall keep the same in good working order and condition, performing all preventative maintenance and routine maintenance and making all adjustments, repairs, rehabilitations, reconstructions, and replacements necessary to keep the System in operation in accordance with the System Performance Requirements set forth in Scope of Work Sections _____ RFP Appendix B Section B.23 at all times, including carrying out the approved Maintenance Plan developed pursuant to Section 145.18.2 below, scheduling and completing routine cleaning, inspection and servicing of all System components, provision and timely replenishment of all spares, supplies and consumables, and performance of scheduled preventative maintenance and corrective maintenance as required in the Contract Documents. Contractor shall maintain, rebuild, repair, restore or replace all work, including Design Documents, Construction Documents, Software, Hardware, materials, equipment, supplies and maintenance equipment which are purchased for permanent installation in, or for use during construction or operation of the Project that is injured or damaged prior to the date Contractor's maintenance responsibility ends, regardless of who has title thereto under the Contract Documents and regardless of the cause of the damage or injury, at no additional cost to the Joint Board, except to the extent that the Joint Board is responsible for such costs in accordance with the express terms of this Contract. Contractor acknowledges and agrees that the maintenance of the civil and structural portions of the System, including gantries, signs on tolling zone gantries, dynamic (or similar) message signs (including their structures and mounts) barriers (not including flexible delineators) that are installed or modified by Contractor in connection with the System, and pavement installed or modified by Contractor for the System to the extent that System sensors are located within such pavement are expressly included within its maintenance and repair obligations under this Section 14.5.18.

The Maintenance Term shall commence for each Crossing upon the expiration of the Warranty Period for such Crossing, and shall continue until the earlier to occur of the date that is four (4) years from the first anniversary of the System Formal Acceptance Date, or the date that is seven (7) years from the first Crossing Substantial Completion Acceptance Date to occur, plus any extension of such period due to the Joint Board's exercise of a Maintenance Option. The Joint Board shall have two Maintenance Options to extend the Maintenance Term for two years with each extension, up to a total of four years, for the Maintenance Option Prices specified in the Contractor's Price Proposal. The Joint Board shall exercise its Maintenance Options, if at all, by delivery of written notice to the Contractor of the Joint Board's election to exercise its Maintenance Option not fewer than

the then-current Maintenance Term. The Joint Board's failure to exercise the first such Maintenance Option shall constitute waiver of the Joint Board's right to exercise the second such option.

14.18.2.15.18.2. Maintenance Plan and Procedures

Contractor shall prepare and provide the Joint Board with a proposed written standard Maintenance Plan and procedures, including standards of performance with which Contractor and its employees and agents shall strictly comply. The Maintenance Plan shall include and cover all procedures of every kind necessary for the maintenance and repair of the System as specified in Scope of Work Section. Contractor shall develop procedures satisfactory to the Joint Board, in its sole discretion. Each Maintenance Plan shall also include a plan for inventory management and control, including all activities required to maintain an adequate supply of materials, supplies, spares, parts and equipment to maintain the System at all times. Once accepted, such Maintenance Plan shall be subject to change in whole or in part, at any time, at the sole discretion of the Joint Board pursuant to this Section 145.18.2.3 hereof.

Contractor shall furnish to the Joint Board the proposed Maintenance Plan for the Joint Board's review no later than 120 days prior to the Substantial Completion Deadline for the relevant Crossing or the System established pursuant to the then-current Project Schedule. The Joint Board shall conduct its review within 15 Days of receipt thereof. To the extent that the Joint Board objects to parts of the Maintenance Plan or directs changes therein, Contractor shall satisfy such objections and accommodate such change requests. Contractor shall submit the revised proposed Maintenance Plan for the Joint Board's review within 15 Days of receipt of the Joint Board's comments, whereupon the Joint Board shall review the revised Maintenance Plan within 15 Days of receipt thereof. Contractor shall continue to satisfy the Joint Board's objections and accommodate the Joint Board's change requests in similar fashion until the Joint Board accepts the Maintenance Plan. Neither the acceptance of any Maintenance Plan nor any direction of changes to a Maintenance Plan shall shift any liability or risk to the Joint Board for maintenance of the System or Contractor's obligations to comply with the Performance Requirements and the other requirements of the Contract Documents. Failure of the Joint Board to respond during any 15 Day review period shall not constitute acceptance thereof. In such event, Contractor shall deliver a written notice to the Joint Board expressly indicating the Joint Board's failure to respond. Should the Joint Board fail to respond within 15 Days of receipt of such written notice, such failure shall be deemed the Joint Board's acceptance of Contractor's submission.

The accepted Maintenance Plan shall be incorporated in a manual entitled "Maintenance Plan and Procedures, RBOC Toll System." Any changes to the Maintenance Plan approved by the Joint Board, in its sole discretion, shall be documented by changes to the manual. The manual will be kept up-to-date in a single bound volume (to the extent practicable) reflecting all provisions in effect at all times. The Maintenance Plan shall set forth a quality control and quality assurance program that will cause Contractor's maintenance and repair of the System to meet all

requirements hereof and will establish the procedures and methods of operation most likely to achieve quality and performance at minimum cost.

14.18.3. 15.18.3. Additional Maintenance Responsibilities

Contractor shall perform System Assurance Monitoring by collecting and analyzing data including System assurance data and System Functional Availability data, and perform diagnostic analysis of failures and trend analysis.

Contractor shall report monthly on System activities (and on Crossing activities prior to System Substantial Completion Acceptance), including operating statistics and System accuracy and Functional Availability, inventory, and other information as requested by the Joint Board. Contractor shall keep detailed maintenance records and inventory data to permit the Joint Board to ascertain Contractor's compliance with the Contract Document requirements and shall furnish copies of such documents upon request. The maintenance records shall include a summary of interruptions to normal Functional Availability and accuracy, explaining the duration and cause of such interruption. The procedures and forms for such recordkeeping shall be provided by Contractor and approved by the Joint Board in advance and as a condition to acceptance thereof. At a minimum, the monthly management reports shall include a summary of the month's activities, highlighting any special events and incidents, a summary of performance achieved, an accuracy and Functional Availability report, and such other information as may be required by the Contract Documents or requested by the Joint Board.

Contractor shall develop a capital asset replacement program, including capital asset condition monitoring and reporting thereon.

Contractor shall monitor the System from a security perspective, respond to emergencies, and report security issues and incidents to appropriate authorities.

Contractor shall train new maintenance and supervisory personnel prior to transition to maintenance of the System by the Joint Board as specified in <u>Section 145.18.5.1</u>.

Contractor shall be responsible for configuration management activities throughout the Maintenance Term for all System elements, including Hardware, Software and documentation, and support and maintain all System Software.

Contractor shall make no change in the design or configuration of the System as accepted by the Joint Board without the express written consent of the Joint Board, in its sole discretion.

14.18.4.15.18.4. Corrective Action

If the minimum System Functional Availability performance levels defined in ——the Performance Requirements in RFP Appendix B are not met for any calendar quartermonth (3 month period), or performance records (including any performance audit report) indicate either has not been or will not be met, Contractor shall promptly notify the Joint Board in writing and undertake a study to determine the causes and propose a plan to correct the problem. Contractor shall take immediate corrective

action to mitigate the cause at Contractor's sole cost and expense. The foregoing shall not modify, alter, limit or negate any of the Joint Board's rights or Contractor's obligations under <u>Section 98 and RFP Appendix B Section B.23</u>.

In addition to its rights under Section 14.18.4.1 above in the preceding paragraph, (i) within 7 days of receipt by Contractor (or such shorter period as specified in the Contract Documents or as reasonably required under the circumstances and specified by the Joint Board in its notice hereunder) of notice from the Joint Board specifying a failure of any aspect of the System or Contractor's failure to perform the Maintenance Work as required under the Contract Documents, or (ii) at any time, in the event of an emergency requiring immediate curative action or a situation which poses a significant safety risk to the public or significant potential revenue loss (as determined by the Joint Board, in its sole discretion), the Joint Board shall have the right, but not the obligation. to perform or have performed by third parties the necessary work, remedy, and the costs thereof shall be borne by Contractor. Alternatively, the Joint Board may deduct the amount of such costs and expenses from any sums owed by the Joint Board to Contractor pursuant to this Contract. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction. The foregoing shall not limit or modify the Joint Board's rights to any Stipulated Liquidated Damages that may arise out of such failure.

14.18.5.15.18.5. Transfer of Maintenance to the Joint Board at End of Maintenance Term

14.18.5.1.15.18.5.1. Training and Personnel

Contractor shall provide training of the Joint Board personnel or the personnel of a Joint Board-designated organization to maintain the Project to be initiated at least 6 months prior to the end of the Maintenance Term. Staff numbers and positions and training procedures will be in accordance with Contractor's staffing plan in the Maintenance Plan then in effect. Contractor shall design its classroom and on-the-job training with the objective of providing sufficient maintenance personnel competent to maintain the System. Contractor shall provide qualified instructors who will prepare the course material including the latest modifications, equipment changes, maintenance updates, and any revised maintenance procedures, and may enlist major Subcontractors or Suppliers to augment Contractor's own trainers. Literature, training aids, and equipment used in training shall be turned over to the Joint Board at the termination of the Maintenance Term. The training shall be designed with the objective that within 3 months of the start of the training, but no later than 90 Days prior to the end of the Maintenance Term, there shall be sufficient quantity of personnel trained so that the Project can be completely run without Contractor's personnel. It shall include classroom and hands-on experience.

14.18.5.2.15.18.5.2. Continued Stocking of Parts

At the end of the Maintenance Term, the spare parts, equipment, expendables and consumables inventory shall be fully stocked and complete (including substitution with suitable alternatives for any such spare parts, equipment and expendables that are obsolete) to the extent required by the Contract Documents, or as may be modified. In the event that such inventory is not fully stocked and complete, an amount equal to the cost of restocking and completing such inventory shall be deducted from any payments, if any, due Contractor. The Joint Board shall notify the Contractor promptly of any such deduction. To the extent that the cost of restocking and completing such inventory incurred by the Joint Board exceeds the payments due Contractor, such differential shall be paid by the Contractor to the Joint Board within 15 Days after the date of termination of the Maintenance Term.

45.016.0 Miscellaneous Provisions

15.1.16.1. Personal Liability of Public Officials

In carrying out any of the provisions of the Contract, or in exercising any power or Joint Board granted to them by or within the scope of the Contract, the Joint Board, the Joint Board members, and the Joint Board Representatives shall have no liability, either personally or as officials of the Commonwealth of Kentucky or of the State of Indiana; in all such matters they act solely as agents and representatives of the Commonwealth of Kentucky or of the State of Indiana.

45.2.16.2. No Waiver of Legal Rights

The Joint Board is not precluded or estopped by any measurement, estimate, or certificate made either before or after the completion, acceptance, Formal Acceptance of the work or payment for the work, from showing the true quantity and character of Contractor's work and materials furnished by the Contractor, or from showing that any such measurement, estimate, or certificate is untrue or incorrectly made, or that Contractor's work or materials that the Contractor furnishes do not conform to the Contract.

The Joint Board is not precluded or estopped, notwithstanding any such measurement, estimate, or certificate made either before or after the completion, acceptance, or Formal Acceptance of the work or payment for the work from recovering from the Contractor and its surety such damages as it may sustain by reason of Contractor's failure to comply with the terms of the Contract. Neither the Joint Board's acceptance, nor the acceptance of any representatives of the Joint Board, nor the Joint Board's Formal Acceptance, nor any payment for or acceptance of the whole or any part of the work, nor any extension of time, nor any the Joint Board possession of the work shall operate as a waiver of any portion of this Contract or of any power herein reserved, or any right to damages herein provided. A waiver of any breach of this Contract does not operate as a waiver of any other or subsequent breach. The Contractor, without prejudice to the terms of the Contract Documents, is liable to the Joint Board for latent defects, fraud or such gross mistakes as may amount to fraud, and the Joint Board's rights under any warranty or guaranty.

45.3.16.3. Policies, Procedures and Standards

15.3.1.16.3.1. Documents

The policies, procedures, standards, and forms of INDOT, KYTC, the Joint Board, and FHWA are included herein by reference. Additional policies, procedures, and standards have been noted elsewhere in the Contract Documents.

15.3.2. 16.3.2. Jurisdiction

Except as noted, the Joint Board is lead agency for the prosecution of this Project. The individual policies, procedures, and standards shall be determined by the location of the proposed work. Except as specified otherwise in the Scope of Work, Aany work in the State of Indiana shall be governed by the noted documents as issued by INDOT, and [Does this always work – are there areas of overlapping requirements where this might present an issue?] Aany work in the Commonwealth of Kentucky shall be governed by the noted documents as issued by the Joint Board. [DISCUSS THIS AND HOW IT RELATES TO SPECS, ETC.]

15.4.16.4. Conflicts of Interest

The Contractor shall not engage, on a full-time, part-time, or other basis during the Contract Time, professional or technical personnel who are or have been, during the Contract Time, in the employment of the Joint Board, except regularly retired employees, without the written consent of the Joint Board Representatives.

15.5. 16.5. Laws to be Observed

In all operations connected with the work, the Contractor shall strictly comply with all state, federal, and local ordinances, regulations, Laws, and bylaws controlling or limiting in any way the actions of those engaged in the work. The Contractor shall keep fully informed of federal and state laws; local laws; ordinances; and rules, regulations, orders, and decrees of bodies or tribunals having any jurisdiction or Joint Board which in any manner affect those engaged or employed on the work or which in any way affect the conduct of the work. The Contractor shall at all times observe and comply with all such laws, ordinances, rules, regulations, orders, and decrees. Contractor shall protect, indemnify, and exculpate the Commonwealth of Kentucky and the State of Indiana and its representatives, in accordance with Article 101, against civil claims or civil liabilities arising from or based on the violation of such Law, ordinance, rule, regulation, order, or decree, whether by itself or its employees.

15.6. 16.6. Reference Standards

Reference to standards, specifications, manuals or codes of any technical society, organization or association, or to Laws or Regulations, whether such reference be specific or by implication, shall mean the standard, specification, manual, code, or Laws or Regulations in effect on the Effective Date except as may be otherwise specifically stated in the Contract Documents. No provision of any such standard, specification, manual, or code, or instruction of a Supplier, shall be effective to change the duties and

responsibilities of the Joint Board, the Contractor, or any of their Subcontractors, consultants, agents, or employees from those set forth in the Contract Documents, nor shall it be effective to assign to the Joint Board or its officers, directors, members, partners, employees, agents, consultants, or Subcontractors any duty or authority to supervise or direct the furnishing or performance of the work or any duty or authority to undertake responsibility inconsistent with the provisions of the Contract Documents.

15.7.16.7. No Third Party Beneficiary

Except as provided below as to the <u>Joint Board membersStates' Parties</u>, the Contract Documents shall not be construed to create a contractual <u>(or third party beneficiary)</u> relationship of any kind between any persons or entities other than the Joint Board and the Contractor, including but not limited to any consultant retained by the Joint Board to prepare or review the Project Scope.

Nothing in the Contract Documents: (1) shall create for the benefit of any such Subcontractor, Supplier, or other individual or entity any contractual relationship between the Joint Board and any such Subcontractor, Supplier, or other individual or entity; or (2) shall create any obligation on the part of the Joint Board to pay or to see to the payment of any moneys due any such Subcontractor, Supplier, or other individual or entity except as may otherwise be required by Laws or Regulations.

As the Joint Board members States' Parties depend on the revenues from the facility and will be responsible for the operations and maintenance of the Project upon completion by the Contractor of construction and Formal Substantial Completion Acceptance [and expiration of the Warranty and Maintenance Term?], the Joint Board members States' Parties may therefore suffer direct damages by any material failure of the Contractor to comply with the terms of this Contract design and construct the System in accordance with the Contract Documents. Notwithstanding any other provision of Section 156.7, Contractor acknowledges and agrees that the States' Parties are intended (and direct) third party beneficiaries of this Contract, and as such, the States' Parties each have an independent right to enforce the terms and conditions hereof by asserting a from and after substantial completion, if any one of the Joint Board members suffers direct damages directly arising out of the material failure of the Contractor to design and construct the System in accordance with the Contract Documents, any one of the Joint Board members shall be deemed an expressly intended third-party beneficiary of this Contract. Contractor's duties to design and construct the Project in accordance with the Contract Documents also include the: (1) duty to perform all design and engineering work with such sufficient skill and expertise as are found generally in the industry by duly licensed professionals; and (2) duty to perform its work in a capable and workmanlike manner, free of significant defects. If any of the Joint Board members suffers such damages arising out of Contractor's material failure to design and construct the System in accordance with the Contract Documents, such failure by the Contractor shall give rise to a direct action on this Contract by that Joint Board member under Kentucky law against the Contractor for all reasonable, and direct losses sustained by the Joint Board member States' Party or States' Parties and caused by the failure of the Contractor to perform its contractual duties. Such losses

and damages specifically would include <u>but not be limited to</u>, to the extent allowed by Law, the actual, reasonable costs to repair, replace, redesign and/or reconstruct those portions of the System necessary to restore it to the functional status required initially for Substantial Completion. These duties and obligations are owed directly to each of the <u>Joint Board membersStates' Parties</u> and can only be waived or released by <u>aneach</u> individual <u>Joint Board member States' Party</u> itself. Contractor shall be afforded all notice and cure periods relating to any such failure as is set forth in this Contract. A <u>Joint Board member's States' Party's</u> action and rights shall not limit, diminish or derogate the rights and remedies of the Joint Board or the other <u>Joint Board members States' Parties</u> under the Contract Documents. The statute of limitations for any cause of action under this <u>Section 156.7</u> shall not begin to run until completion of construction and Formal Acceptance by the Joint Board, or such other date as may be provided by law, whichever is later.

45.8.16.8. Contractor's Responsibility For Work

Until the Joint Board makes Formal Acceptance of the work, the Contractor shall protect against injury or damage to any part of the work by the action of the elements, or from any other cause, whether arising from the execution, or from the nonexecution, of the work. Contractor shall Rrebuild, repair, and restore any portion of the work damaged by any of the above causes. The Contractor shall bear the expense of such repairs except for damages to the work which the Joint Board determines are due to Force Majeure Events.

45.9.16.9. Documents and Records

15.9.1.16.9.1. Escrowed Proposal Documents

Concurrently with the execution of the Contract, the Contractor delivered to the Joint Board one copy of all documentary information used in preparation of its Price Proposal for the Project (the "Escrowed Proposal Documents" or "EPD"), but excluding documents relating to other projects which may have been referenced by the Contractor in determining the Contract Price and the Maintenance Price. The EPD have been transferred to a locked fireproof cabinet in a locked room in the Operations Center, supplied by the Operations Contractor. The cabinet has two locks, with one key held by the Contractor and the other key held by the Joint Board. Concurrently with approval of each Change Order, if appropriate, one copy of all documentary information used in preparation of the Change Order shall be added to the EPD. The EPD will be held in such cabinet or otherwise maintained subject to Section 156.9.2 until all of the following have occurred: (a) 180 days have elapsed after expiration or earlier termination of the warranties or Maintenance Term, whichever is later, (b) all Disputes regarding this Contract have been settled, and (c) final payment of the Maintenance Price has been made by the Joint Board and accepted by the Contractor.

15.9.1.1.16.9.1.1. Review of EPD

The EPD shall be available during business hours for joint review by the Contractor and the Joint Board and any Dispute resolver in the resolution of Disputes, in connection with an audit (if the EPD is the subject of an audit) and as described in Section 156.9.7. The Joint Board shall be entitled to review all or any part of the EPD in order to satisfy itself regarding the applicability of the individual documents to the matter at issue. The Joint Board shall be entitled to make and retain copies of such documents as it deems appropriate in connection with any such matters, provided that the Joint Board has executed and delivered to the Contractor a confidentiality agreement specifying that, except as required by law or court order, all proprietary information contained in such documents will be kept confidential, that copies of such documents will not be distributed to any third parties other than -the Joint Board's agents, attorneys and experts, other dispute resolvers, and that all copies of such documents (other than those delivered to dispute resolvers) will be either destroyed or returned to the depository (or to the Contractor if the EPD have been returned to it) upon final resolution of the negotiations or disputes. The foregoing shall in no way be deemed a limitation on the Joint Board's discovery rights with respect to such documents. Nothing in this Section 156.9 shall prohibit the Joint Board from producing such documents it believes, in its discretion, it is required to produce in response to a request pursuant to the Kentucky Open Records Act or as otherwise required by law or court order. Nothing in this Section 156.9 shall restrict the Joint Board's rights to use the Design Documents and Technical Proposal Documents as provided in Section 134.5 and elsewhere in the Contract Documents.

15.9.1.2. 16.9.1.2. Property of the Contractor

The EPD are, and shall always remain, the property of the Contractor, subject to the Joint Board's right to review the EPD as provided herein. The Joint Board acknowledges that the Contractor considers that the EPD constitute trade secrets or proprietary information. This acknowledgment is based upon the Joint Board's understanding that the information contained in the EPD is not known outside Contractor's business, is known only to a limited extent and by a limited number of employees of the Contractor, is safeguarded while in Contractor's possession, and may be valuable to Contractor's business strategies, assumptions and intended means, methods and techniques. The Joint Board further acknowledges that the Contractor expended money in developing the information included in the EPD and further acknowledges that it would be difficult for a competitor to replicate the information contained therein. The Joint Board acknowledges that the EPD and the information contained therein are being provided to the Joint Board only because it is an express prerequisite to award of the Contract.

15.9.1.3. Representation and Warranty

The Contractor represents and warrants that the EPD provided concurrently with the Price Proposal documents constitute all of the information used in the preparation of Its Price Proposal and agrees that no other proposal preparation information will be considered in resolving Disputes. The Contractor also agrees that the EPD are not part of this Contract and that nothing in the EPD shall change or modify the Contract.

15.9.1.4.16.9.1.4. Contents of EPD

The EPD shall clearly itemize the estimated costs of performing the work required by the Contract Documents. All work shall be separated into sub-items as required to present a complete and detailed estimate of all costs. Crews, equipment, quantities and rates of production shall be detailed. Estimates of costs shall be further divided into Contractor's usual cost categories such as direct labor, repair labor, equipment ownership and operation, Expendable Materials, permanent materials and subcontract costs as appropriate. Plant and equipment and indirect costs shall also be detailed in Contractor's usual format. Contractor's allocation of plant and equipment, indirect costs, contingencies, markup and other items to each direct cost item shall be clearly identified. The EPD shall include all assumptions, quantity takeoffs, rates of production, the Contractor internal equipment rental rates and progress calculations, quotes from Subcontractors, consultants and Suppliers, memoranda, narratives and all other information used by the Contractor to arrive at the Proposal Price or Change Order price, as applicable.

15.9.1.5. 16.9.1.5. Format of EPD

The Contractor shall submit the EPD in the format actually used by the Contractor in preparing its Price Proposal. It is not intended that the Contractor perform any significant extra work in the preparation of these documents prior to the Contract Price Proposal due date. However, the Contractor represents and warrants that the EPD related to the proposal documents have been personally examined prior to delivery to escrow by an authorized officer of the Contractor and that they meet the requirements of Section 156.9.5 and are adequate to enable a complete understanding and interpretation of how the Contractor arrived at its Proposal Price. The Contractor further represents, warrants and covenants that the EPD related to each Change Order will be personally examined prior to delivery to escrow by an authorized officer of the Contractor and that they meet the requirements of Section 156.9.5 and will be adequate to enable a complete understanding and interpretation of how the Contractor arrived at its Change Order price.

15.9.1.6. 16.9.1.6. Compilation of Index

Within ten working days after the Effective Date (if not performed prior to Contract execution) and within ten working days after approval of any Change Order, representatives of the Joint Board and the Contractor shall review the EPD or supplemental EPD, as appropriate, to determine whether it is complete, and shall organize the EPD or supplemental EPD, as appropriate, and label each page so that it is obvious that the page is a part of the EPD and so as to enable a person reviewing the page to determine where it can be found within the EPD. The representatives shall also compile an index listing each document included in the EPD or supplemental EPD, as appropriate, and briefly describing the document and its location in the EPD. The Joint Board shall have the right to retain a copy of the Index. In the event that, following the initial organization, the Joint Board determines that the EPD is incomplete, the Joint Board may request the Contractor to supply data to make the EPD complete. The Contractor shall provide all such data within three working days of the request, and at that time it will be date stamped, labeled to identify it as supplementary EPD

Information, and added to the EPD. The Contractor shall have no right to add documents to the EPD except upon the Joint Board's request.

15.9.1.7.16.9.1.7. Confidentiality

The EPD shall at all times be treated as proprietary and confidential information and shall be used only for purposes described in <u>Section 156.9.1</u>. The Contractor shall label any EPD that the Contractor considers to be exempt from disclosure pursuant to the Indiana and Kentucky <u>Oopen Records Aacts</u>. At Contractor's request, confidentiality agreements shall be executed and delivered to the Contractor by the Joint Board's employees or agents who review or have access to the EPD.

15.9.1.8.16.9.1.8. Subcontractors

Contractor shall require the principal design Subcontractor, and all Major Subcontractors, to submit to Contractor a copy of all documentary information used in determining its Subcontract price (including the Subcontract Price included in any amendment and/or change order), immediately prior to executing the Subcontract and each change order and Subcontract amendment, to be held in the same manner as the EPDs and which shall be accessible by Contractor and its successors and assigns (including the Joint Board) and any dispute resolver, on terms substantially similar to those contained herein. Each such Subcontract shall contain a representation and warranty of the Subcontractor, for the benefit of Contractor and the Joint Board, that the EPDs include all the documentary information used in establishing its Subcontract price, and agreeing to provide a sworn certification in connection with each supplemental set of EPDs, that the information contained therein is accurate, complete and current. Each Subcontract that is not subject to the foregoing requirement shall include a provision requiring the Subcontractor to preserve all documentary information used in determining its Subcontract price and to provide such information to Contractor and the Joint Board in connection with any claim made by such Subcontractor.

15.9.2.16.9.2. Maintenance of, Access to and Audit of Records

The Joint Board and its members or their auditors may conduct an audit or investigation of any entity receiving funds directly under this Contract or indirectly through a Subcontract. Acceptance of funds directly under this Contract or indirectly through a Subcontract acts as acceptance of the authority of such auditors to conduct an audit or investigation in connection with those funds. An entity that is the subject of an audit or investigation must provide the auditor with access to any information the auditor considers relevant to the investigation or audit, including, without limitation, any information which the Joint Board is entitled to audit under this <u>Section 156.9.2</u>.

Contractor shall maintain at a Project administration office in [specify location] a complete set of all books and records prepared or employed by Contractor in its management, scheduling, cost accounting and other activities related to the work and the Project. Without limiting the provisions of <u>Section 156.9</u>, Contractor shall grant to the Joint Board and the relevant auditor such audit rights and shall allow the Joint Board and the relevant auditor such access to and the right to copy such books and records as

the Joint Board or the auditor may request in connection with the issuance of any NTP or with respect to any Change Orders, the resolution of Disputes, and such other matters as the Joint Board or the auditor reasonably deems necessary for purposes of verifying compliance with this Contract and applicable Law.

Where the payment method for any work is on a Force Account or force account basis, such examination and audit rights shall include all books, records, documents and other evidence and accounting principles and practices sufficient to reflect properly all direct and indirect costs of whatever nature claimed to have been incurred and anticipated to be incurred for the performance of such work. If an audit indicates Contractor has been overpaid under a previous progress report or progress payment, the excess payment will be credited against current progress reports or payments.

For cost and pricing data submitted in connection with the Project other than that which is covered by any Change Orders, auditors and their representatives shall have the right to examine all books, records, documents and other data of Contractor related to the negotiation of or performance of such work for the purpose of evaluating the accuracy, completeness and currency of the cost or pricing data submitted; provided, however, that the foregoing shall not apply to pricing based on adequate price competition, established catalog or market prices of commercial items sold in substantial quantities to the public or prices set by law or regulation, in each case, as determined by the Joint Board. Such right of examination shall extend to all documents deemed necessary by the Joint Board and its representative to permit adequate evaluation of the cost or pricing data submitted, along with the computations and projections used therein.

All claims filed against the Joint Board shall be subject to audit at any time following the filing of the claim. The audit may be performed by employees of the Joint Board or by an auditor under contract with the Joint Board. No notice is required before commencing any audit before 60 Days after the later of (i) the expiration of the Maintenance Term, or (ii) the termination of this Contract. Thereafter, The Joint Board shall provide 10 days notice to Contractor, any Subcontractors or their respective agents before commencing an audit. Contractor, Subcontractors or their agents shall provide adequate facilities, acceptable to the Joint Board, for the audit during normal business hours. Contractor, Subcontractors or their agents shall cooperate with the auditors. Failure of Contractor, Subcontractors or their agents to maintain and retain sufficient records to allow the auditors to verify all or a portion of the claim or to permit the auditor access to the books and records of Contractor, Subcontractors or their agents shall constitute a waiver of the claim and shall bar any recovery thereunder.

At a minimum, the auditors shall have available to them the following documents:

- 1. Daily time sheets and supervisor's daily reports;
- 2. Union agreements;
- 3. Insurance, welfare, and benefits records;
- 4. Payroll registers;

- 5. Earnings records;
- 6. Payroll tax forms;
- Material invoices and requisitions;
- 8. Material cost distribution work sheet;
- 9. Equipment records (list of company equipment, rates, etc.);
- 10. Subcontractors' (including Suppliers) invoices;
- 11. Subcontractors' and agents' payment certificates;
- 12. Canceled checks (payroll and Suppliers);
- 13. Job cost report;
- 14. Job payroll ledger;
- 15. General ledger;
- 16. Cash disbursements journal;
- 17. All documents that relate to each and every claim together with all documents that support the amount of damages as to each claim; and
- 18. Work sheets used to prepare the claim establishing the cost components for items of the claim including labor, benefits and insurance, materials, equipment, Subcontractors, all documents that establish the time periods, individuals involved, the hours for the individuals, and the rates for the individuals.

Contractor shall ensure that each Subcontract shall provide for the same audit rights in connection with such Subcontract as are provided under this Section 156.9.

Full compliance by Contractor with the provisions of this <u>Section 156.9</u> is a contractual condition precedent to Contractor's right to pursue a formal Dispute under <u>Section 123</u>.

Contractor represents and warrants the completeness and accuracy of all information it or its agents provides in connection with this <u>Section 156.9</u>.

45.10.16.10. Retention of Records

Contractor shall maintain all records and documents relating to the work, including copies of all original documents delivered to the Joint Board, and the Project in [specify location] at a convenient location within the greater Louisville Metropolitan area until five years after the later of (i) the expiration of the Maintenance Term, or (ii) the termination of this Contract. Contractor shall notify the Joint Board where such records and documents are kept. Notwithstanding the foregoing, all records which relate to Claims being processed or actions brought under the Dispute resolution provisions hereof shall be retained and made available until such actions and Claims have been finally resolved. Records to be retained include all books, electronic information and files and other evidence bearing on Contractor's costs under the Contract Documents. Contractor shall make these records and documents available for audit and inspection to the Joint Board, at Contractor's offices in Austin, Texas, at all reasonable times,

without charge, and shall allow such Persons to make copies of such documents, at no expense to Contractor. If approved by the Joint Board, photographs, microphotographs or other authentic reproductions may be maintained instead of original records and documents

15.11.16.11. Interest

Any interest due or determined to be due to the Contractor, including any pre-judgment interest, shall be at the prime rate as published in the Wall Street Journal at the time the underlying amounts were due.

15.12. 16.12. Severability

The invalidity in whole or in part of a provision of this Contract shall not void or affect the validity of all other provisions.

15.13.16.13. Entire Agreement

The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral.

16.017.0 Representatives 16.1.17.1. The Joint Board's Designated Representative the Joint Board Representatives are ______. The Joint Board Representatives shall be authorized to act on the Joint Board's behalf with respect to the Project.

16.2. 17.2. Contractor's Designated Representative

Contractor's Designated Representative is ______. Contractor's Designated Representative identified above shall be authorized to act on Contractor's behalf with respect to the Project.

46.3.17.3. Changing Designated Representative

Neither the Joint Board Representatives nor Contractor's Designated Representative may be changed without ten days written notice to the other party.

47.018.0 Access to Project Site; Utilities; Environmental Compliance 47.1.18.1. Access to Project Site

The Joint Board shall be responsible facquire necessary Project right of way. or providing Contractor shall coordinate with the Design Build Contractor and the Developer regarding with such access to the BridgesProject right of way as is necessary to perform the work. To the extent that Contractor has not been provided

with access to portions of the Bridges-right of way ("ROW")[Need to define ROW area available to the Contractor for its work and permanent improvements] prior to the date set forth on the Project Schedule, Contractor shall work around such area with the goal of minimizing delay to the completion of the Project.

All costs and expenses for the acquisition of any temporary right or interest in real property that Contractor determines necessary or desirable for its convenience in performance of the work, such as for work space, contractor laydown areas, materials or storage areas, or for any permanent interest in real property that Contractor may wish to acquire for its convenience which lies outside the ROW, shall be Contractor's sole responsibility, to be undertaken at Contractor's sole cost and expense. The Joint Board shall have no obligations or responsibilities with respect to the acquisition, maintenance or disposition of such rights or interests, and shall not use its powers of eminent domain in connection therewith. Contractor will comply with all applicable Laws in acquiring and maintaining or disposing of any such property rights or interests. Contractor shall cause the documentation of any such property interest to contain the grantor's express acknowledgment that the Joint Board shall have no liability with respect thereto.

17.2.18.2. Utilities

The Joint Board will ensure that all necessary Utilities enter the ROW of each BridgeCrossing, provided that the Joint Board may, in its sole discretion, require Contractor to perform such activities by including them in a Construction Change Directive. Contractor shall be responsible for verifying the location of all Utilities shown on any utility strip map provided by the Joint Board or otherwise related to any BridgeCrossing (including undertaking field inspections and Project Site studies, researching Utility Owner records and confirming Contractor's findings with the Joint Board and the Utility Owners), and for obtaining all Utility installations, hook-ups and service extensions that are required to operate the System, and for coordination with Utility Owners, at Contractor's cost. Contractor shall use its best efforts to minimize costs for which Contractor is entitled to compensation, including avoidance of an unidentified and misidentified Utility where feasible rather than its removal and/or reinstallation in a new location, and to minimize any delay for which Contractor is entitled to an extension in a Completion Deadline pursuant to this Section 178.2.

47.3.18.3. Hazardous Materials Management

17.3.1.18.3.1. Procedures and Compensation for Hazardous Materials Management

If during the course of the work, Contractor encounters Hazardous Materials—or potential Recognized Environmental Conditions, Contractor shall (a) promptly notify the Joint Board in writing and advise the Joint Board of any obligation to notify State or federal agencies under applicable Law; and (b) take reasonable steps, including design modifications and/or construction techniques, to avoid excavation or dewatering in areas with Hazardous Materials—or Recognized Environmental Conditions. Where excavation or dewatering of Hazardous Materials or Recognized Environmental Conditions—is unavoidable, Contractor shall utilize appropriately trained personnel and

shall select the most cost-effective approach to Hazardous Materials Management, unless otherwise directed by the Joint Board. Contractor's plan for Hazardous Materials Management shall be subject to the prior written approval of the Joint Board. Wherever feasible and consistent with applicable Law, contaminated soil and groundwater shall not be disposed off-site. All Hazardous Materials and potential Recognized Environmental Conditions—shall be managed, handled, remediated, transported (where applicable) and disposed of in accordance with applicable Law, Governmental Approvals, the Hazardous Materials Management Plan, and the approved Project Site Investigation Report.

Except where required to take an immediate action required by applicable Law, Contractor shall afford the Joint Board the opportunity to inspect sites containing Hazardous Materials or potential Recognized Environmental Conditions before any action is taken which would inhibit the Joint Board's ability to ascertain the nature and extent of the contamination.

Subject to the limitations and exceptions set forth in this and <u>Section 4</u>, Contractor shall be entitled to a Change Order as set forth in <u>Section 4-.8</u>— providing for additional compensation and/or a time extension with respect to costs and delays directly attributable to the discovery of Hazardous Materials within the ROW.

17.3.2.18.3.2. Hazardous Material Generator

As between Contractor and the Joint Board, the Joint Board shall be considered the generator of Hazardous Materials on any ROW as of the Effective Date; provided, however, that the foregoing shall not preclude or limit any rights or remedies that the Joint Board may have against third parties and/or prior owners, lessees, licensees and occupants of the ROW. As between Contractor and the Joint Board, Contractor shall be considered the generator of any Hazardous Materials on any ROW which result from (a) Release(s) of Hazardous Material attributable to the negligence, willful misconduct, or breach of applicable Law or contract by any Contractor-Related Entity; and (b) Release(s) of Hazardous Materials arranged to be brought onto ROW or elsewhere by any Contractor-Related Entity regardless of the cause of the Release of Hazardous Materials.

17.3.3.18.3.3. Hazardous Material Releases Caused by Contractor

Hazardous Materials Management costs, including assessment, containment, and remediation expenses, on, arising from or related to and/or which result from (a) Release(s) of Hazardous Material attributable to the actions, omissions, negligence, willful misconduct, or breach of applicable Law or contract by any Contractor-Related Entity; or (b) Release(s) of Hazardous Materials arranged to be brought onto any ROW or elsewhere by any Contractor-Related Entity shall not be compensable to Contractor, regardless of the cause of the Release of Hazardous Materials.

17.3.4.18.3.4. Hazardous Materials Brought to Final ROW by Contractor

Contractor shall be solely responsible for: (a) compliance with all Laws applicable to Hazardous Materials brought onto the Project Site by any Contractor-Related Entity; (b)

use, containment, storage, management, transport and disposal of all Hazardous Materials in accordance with the Contract Documents and all applicable Laws and Environmental Approvals; and (c) payment of all Losses associated with, arising out of or related to such Hazardous Materials.

17.3.5. 18.3.5. Environmental Approvals Relating to Hazardous Materials Management

Contractor shall be solely responsible for obtaining all Governmental Approvals relating to Hazardous Materials Management including federal and State surface water and groundwater discharge permits and permits for recycling or reuse of Hazardous Materials <u>required</u> for performance of the work. Contractor shall be solely responsible for compliance with such Governmental Approvals and applicable Laws, including those governing the preparation of waste profiles, waste manifests and bills of lading.

17.4.18.4. Environmental Compliance

Contractor shall be responsible for compliance with all conditions and requirements of the Environmental Approvals and similar Governmental Approvals specifically applicable to the Project, other than the mitigation requirements that are the Joint Board's responsibility in the Contract Documents. The Contract Price includes compensation for Contractor's performance of all environmental requirements and conditions relating to the Project, including mitigation measures.

18.019.0 Additional Federal and State Provisions

18.1.19.1. Certification Regarding Use of Contract Funds for Lobbying

Contractor certifies, to the best of its knowledge and belief (after due inquiry and investigation), that:

- 1. No Federal appropriated funds have been paid or will be paid, by or on behalf of Contractor, to any person for influencing or attempting to influence an officer or employee of any Federal agency, a member of Congress, an officer or employee of Congress, or an employee of a member of Congress in connection with the awarding of any Federal contract, the making of any Federal grant, the making of any Federal loan, the entering into of any cooperative agreement, and the extension, continuation, renewal, amendment, or modification of any Federal contract, grant, loan, or cooperative agreement.
- 2. If any funds other than Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any Federal agency, a member of Congress, an officer or employee of Congress, or an employee of a member of Congress in connection with this Federal contract, grant, loan, or cooperative agreement, the undersigned shall complete and submit Standard Form-LLL, "Disclosure Lobbying Activities," in accordance with its instructions.
- 3. This certification is a material representation of fact upon which reliance was placed when this transaction was made or entered into. Submission of this certification is a prerequisite for making or entering into this transaction imposed by 31

- U.S.C. § 1352. Any person who makes an expenditure prohibited under 31 U.S.C § 1352 shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such expenditure. Any person who fails to file or amend a declaration required to be filed or amended under 31 U.S.C. § 1352 shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure.
- 4. Contractor shall require that the language of this certification be included in all Subcontracts and that all Subcontractors shall certify and disclose accordingly.

18.2.19.2. Civil Rights; Equal Employment Opportunity and Certification

- 19.2.1. Contractor shall not, and shall cause the Subcontractors to not, discriminate on the basis of race, color, national origin or sex in the performance of the Work under the Contract Documents. Contractor shall carry out, and shall cause the Subcontractors to carry out, applicable requirements of 49 CFR Part 21 and 23 CFR Part 200 in the award and administration of FHWA-assisted agreements. Failure by Contractor to carry out these requirements is a material breach of this Contract, which may result in the termination of the Contract Documents or such other remedy as the Joint Board deems appropriate.
- 19.2.2. Contractor shall include <u>Section 189.2.1</u> in every Subcontract (including purchase orders and in every subcontract of any Contractor-Related Entity for Work), and shall require that they be included in all Subcontracts at lower tiers, so that such provisions will be binding upon each Subcontractor.
- <u>19.2.3.</u> Contractor confirms for itself and all Subcontractors that Contractor and each Subcontractor has an equal employment opportunity policy ensuring equal employment opportunity without regard to race, color, national origin, sex, age, religion or handicap; and that Contractor and each Subcontractor maintains no employee facilities segregated on the basis of race, color, religion or national origin. Contractor shall comply with all applicable Equal Employment Opportunity and nondiscrimination provisions set forth in <u>Exhibit</u> <u>L</u>, and shall require its Subcontractors to comply with such provisions.
- Contractor hereby certifies that it has participated in a previous contract or subcontract subject to the equal opportunity clause, as required by Executive Orders 10925, 11114, or 11246, and that, where required, it has filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance, a Federal Government contracting or administering agency, or the former President's Committee on Equal Employment Opportunity, all reports due under the applicable filing requirements.

CONTRACTOR ACKNOWLEDGES THAT THE FOREGOING CERTIFICATION WITH REGARD TO THE PERFORMANCE OF PREVIOUS CONTRACTS OR SUBCONTRACTS IS SUBJECT TO THE EQUAL OPPORTUNITY CLAUSE AND THE FILING OF REQUIRED REPORTS SHALL BE EXECUTED BY THE CONTRACTOR AND EACH NON-EXEMPT SUBCONTRACTOR.

19.3. Buy America

The Contractor shall comply with all Buy America requirements, including but not limited to those contained in 23 U.S.C. 313 and 23 CFR 635.410.

19.4. Prevailing Wage Rates

The Contractor shall comply with all applicable prevailing wage requirements, particularly those included in the schedule found in Exhibit XXXM, as updated.

19.5. Civil Rights Act of 1964

The KYTC, in accordance with the provisions of Title VI of the Civil Rights Act of 1964 (78 Stat. 252) and the Regulations of the Federal Department of Transportation (49 C.F.R., Part 21), issued pursuant to such Act, hereby notifies all bidders that it will affirmatively insure that the contract entered into pursuant to this advertisement will be awarded to the bidder whose proposal provides the best value to the Joint Board without discrimination on the ground of race, color, or national origin.

19.6. Notice To All Bidders

To report bid rigging activities, proposers shall call: 1-800-424-9071.

The U.S. Department of Transportation (DOT) operates the above toll-free "hotline" Monday through Friday, 8:00 a.m. to 5:00 p.m. eastern time. Anyone with knowledge of possible bid rigging, bidder collusion, or other fraudulent activities shall use the "hotline" to report such activities.

The "hotline" is part of the DOT's continuing effort to identify and investigate highway construction contract fraud and abuse and is operated under the direction of the DOT Inspector General. All information will be treated confidentially and caller anonymity will be respected.

19.7. Second Tier Subcontracts

Second Tier subcontracts on federally assisted projects shall be permitted. However, in the case of DBE's, second tier subcontracts will only be permitted where the other subcontractor is also a DBE. All second tier subcontracts shall have the consent of both the Contractor and the Engineer.

19.020.0 DBE Participation

Disadvantaged Business Enterprises shall have equal opportunity to participate in this Project. The Contractor shall use its best efforts to solicit bids from, and to contract with, DBE subcontractors to perform all types of work on the Project. This project shall have a DBE participation goal of _____3%. The Contractor shall comply with Title 49 CFR Part 26 in its efforts to meet the DBE participation goal for this Project. By submitting its Proposal, the Contractor represents that it will achieve the Project DBE participation goal. The Contractor shall demonstrate on an annual basis throughout the life of the Contract that the Project DBE participation goal is being achieved or that "good faith efforts" have been made to do so as described in 49 CFR 26.53. KYTC shall determine whether the Contractor has put forth adequate good faith efforts pursuant to paragraph____.

19.1.20.1. DBE Participation Plan

The Contractor's proposal shall include an initial DBE participation plan. The DBE participation plan shall demonstrate how the Contractor will use DBEs to meet or exceed the Project DBE participation goal for the life of the Project. The DBE participation plan shall identify and describe the planned utilization of DBEs in all subcontractor procurements. The Contractor shall include the subcontract DBE participation goals in the procurement documents for those procurements specified in the DBE participation plan. The Contractor shall maintain complete documentation of all procurements assigned DBE subcontractor goals.

19.1.1.20.1.1. Updates to the DBE Participation Plan

Every year, on the Contract award date, until Final Acceptance, the Contractor shall submit a revised DBE participation plan to the Joint Board for review, comment and approval. The Contractor shall also submit a yearly affidavit stating the level of DBE expenditure to date and describing planned efforts to meet the Project DBE participation goal through the duration of the Contract. On an annual basis, the Joint Board will verify that the Contractor has met its cumulative DBE participation goal and that it is complying with its DBE participation plan.

19.1.2.20.1.2. Affidavit

The Contractor's proposal shall include an affidavit stating that the Contractor will meet or exceed the Project DBE participation goal or shall submit proof of its good faith efforts to do so in compliance with 49 C.F.R. Part 26. It is the responsibility of the Contractor to obtain all required documentation from the DBE subcontractors and to provide the information to the Joint Board.

19.2.20.2. Commercially Useful Function

Proposed DBE participation shall not be counted toward the Project goal unless the DBE performs a commercially useful function as described in 49 CFR 26.55. To meet this requirement, DBE's must perform at least 30 percent of the total cost of its contract with its own work force as provided in 49 CFR 26.55(c)(3). The Contractor shall provide certified payrolls for all of its subcontractors. The certified payrolls for DBE

subcontractors will be reviewed by KYTC to ensure that the work in the payroll describes a commercially useful function.

19.3.20.3. Monthly Reports of DBE Participation

The Contractor shall also submit monthly reports throughout the duration of the Contract detailing DBE payments to date and payments anticipated for the remainder of the Project. The report shall also include, for the applicable reporting period, any and all payments the Contractor or any of its non-DBE consultants, subcontractors and/or suppliers have received (regardless of the form or manner) from any DBE for shared equipment, personnel, or facilities. For example, if the Contractor uses a DBE service, such as trucking, to satisfy a part or the Project DBE participation goal, the Contractor shall provide a copy of each canceled check issued to the DBE service provider upon request by the non-DBE contracting entity. DBE participation shall not be credited until documentation is received and approved by the Joint Board.

19.4.20.4. Replacement of Non-Performing DBEs

In the event that a DBE fails to perform the subcontracted work, the Contractor shall attempt to replace the non-performing DBE with another DBE. Replacement DBEs shall be given the same or an equivalent scope of work as the original DBE was to perform. A non-performing DBE is one that has been contracted to provide specific goods or perform services, but becomes unable or unwilling to provide the goods or services for which it was contracted. With the prior approval of the Joint Board, the Contractor may remove and replace a nonperforming DBE. The Contractor must submit a request for approval of the planned corrective action from KYTC, The Office for Civil Rights and Small Business Development, 200 Mero Street, 6th Floor West, Frankfort, KY 40622. The request must document the process, progress, and efforts being made or not made by the non-performing DBE. The request will then be submitted to the Joint Board for review.

19.5.20.5. 16.5 Waiver Process for DBE Project Goal

In the event the Contractor is unable to meet a portion of the Project DBE participation goal, the Joint Board will accept requests for partial waivers of the Project goal during the term of the Contract. Requests for partial waivers must include a full accounting of all good faith efforts expended to meet the Project goal and an explanation of the necessity of the waiver. Any such request shall be submitted at the time the Contractor's annual update of the DBE participation plan is due. Waiver requests shall be in writing and shall be submitted to KYTC, The Office for Civil Rights and Small Business Development, 200 Mero Street, 6th Floor West, Frankfort, KY 40622. The requests will then be submitted to the Joint Board, which shall individually evaluate all requests to determine if a waiver of any portion of the Project goal is warranted.

If an item of work subcontracted to a DBE firm is eliminated from the Project by the Joint Board, the Contractor may request a waiver of an equivalent portion of the Project goal. If the Contractor or the Joint Board proposes an alternate scope of work in place of the

eliminated work, then the Contractor shall expend good faith efforts to solicit DBE participation for the revised scope of work. If the Contractor is unable to obtain DBE participation for the revised scope of work it may then request a waiver of the equivalent portion of the Project goal.

If requesting a partial waiver of the Project DBE participation goal indicated in the DBE participation plan, the Contractor shall provide the following for Joint Board review and approval:

- 1. The dollar value and percentage of the Project goal the subcontract in question was intended to meet.
- 2. The dollar value and percentage of the reduction in the subcontractor DBE participation goal being requested.
- 3. The executed copy of the subcontract or purchase order between the Contractor/lower tier subcontractor and the non-performing DBE, if any.
- 4. Copies of all dated communications/solicitations, fax confirmations, personal contact sheets or meeting notes, negotiation summaries, and applicable follow-up communications.
- 5. Copies of all written communications, faxes, or emails that show that the Contractor or lower tier bidders solicited and provided DBEs with adequate information about the plans, specifications, and requirements of the Contract or subcontract in a timely manner to assist DBEs in responding to the solicitation efforts.
- 6. Copies of dated written communications, faxes, or emails for each non-competitive DBE quote that includes the dollar value of each item of work for which the DBE provided a quote. Copies of any written communications evidencing the unavailability of DBEs or their general unwillingness to provide a quote for the Project/subcontract opportunity.
- 7. Documentation of all negotiations with DBEs and reasons for rejecting bids received.
- 8. Documentation of efforts to utilize KYTC's Office for Civil Rights and Small Business Development or any minority or female business organization or association to meet the subcontractor DBE participation goal.
- 9. Such other factors the Contractor or its lower tier bidder/subcontractor determine may warrant consideration.

The Joint Board shall review the Contractor's waiver request and attached documentation and issue a written determination within ten business days. The Contractor may request administrative reconsideration of the Joint Board's written determination within ten business days of being notified that the Joint Board denied the waiver request.

The Contractor's efforts to obtain a waiver of DBE expenditure goals shall not be grounds for a change order and shall not affect the Project schedule.

19.6.20.6. Subcontractor Agreements

The Contractor shall provide a copy of any DBE subcontract agreement that the Contractor proposes to use to fulfill the Project DBE participation goal or for which the Contractor desires to obtain credit.

DBEs shall not perform services or supply goods, regardless of their lower tier status, unless an executed subcontract agreement or purchase order has been submitted to the Joint Board. The Contractor and lower tier non-DBE subcontractors with DBE participation responsibilities shall also submit copies of all change orders and contract modifications for their DBE subcontracts to the Joint Board. Each subcontract agreement shall at a minimum include all required federal provisions and the requirement that all lower tier subcontractors must cooperate with all compliance reviews.

19.7.20.7. Good Faith Requirements and Noncompliance

As describe above, Contractors shall submit a comprehensive, DBE participation plan describing how they will meet or exceed the DBE expenditure goal with the technical proposal. DBEs shall be utilized in all phases of the Project including consultants, subcontractors, and suppliers. If the written DBE participation plan does not adequately address how the DBE participation will meet or exceed the expenditure goal, or does not adequately document that it made good faith efforts to meet the goal, the Contractor submittal shall be deemed non-responsive.

The Contractor, on an annual basis, shall prepare an updated DBE participation plan consistent with the requirements for the original DBE participation plan, with the addition of demonstrating progress toward meeting or exceeding the Project DBE participation goal or that sufficient good faith efforts were made to meet the Project goal for the prior year. If the Contractor is not on track to meet the Project goal it shall submit an action plan to address shortfalls. Demonstration of good faith effort shall include:

- 1. Whether the Contractor attended any pre-bid meetings that were scheduled by KYTC or the Joint Board to inform DBEs of subcontracting opportunities;
- 2. Whether the Contractor provided solicitations through reasonable and available means:
- 3. Whether the Contractor provided written notice to all DBEs listed in the INDOT and KYTC DBE directories who are prequalified in the areas of work available on the Project;
- 4. Whether the Contractor followed up initial solicitations of interest by contacting DBEs to determine with certainty whether they were interested. If a reasonable amount of DBEs within the targeted area do not provide an intent to quote or no DBEs are prequalified in the subcontracted areas, the Contractor must notify the KYTC Disadvantaged Business Enterprise Liaison Officer of the lack of response;
- 5. Whether the Contractor selected portions of the work to be performed by DBEs to increase the likelihood of meeting the contract goal. This includes, where

appropriate, breaking out contract work items into economically feasible units to facilitate DBE participation, even when the Contractor might otherwise perform these work items with its own forces:

- 6. Whether the Contractor provided interested DBEs with adequate and timely information about the plans, specifications, and requirements of the Contract;
- 7. Whether the Contractor negotiated in good faith with interested DBEs not rejecting them as unqualified without sound reasons based on a thorough investigation of their capabilities. Any rejection should be so noted in writing with a description as to why an agreement could not be reached;
- 8. Whether quotes were received from interested DBEs, but were rejected as unacceptable and why the quotes were unacceptable. The fact that the DBEs' quote for the work is not the lowest quoted price received will not in itself be considered as a sound reason for rejecting the quote as unacceptable. The fact that the Contractor has the ability and/or desire to perform the contract work with its own forces will not be considered a sound reason for rejecting a DBE quote. Nothing in this provision shall be construed to require the Contractor to accept unreasonable quotes to satisfy the DBE goal;
- 9. Whether the Contractor specifically negotiated with subcontractors to assume part of the responsibility to meet the contract DBE goal when the work to be subcontracted includes potential DBE participation;
- 10. Whether the Contractor made any efforts and/or offered assistance to interested DBEs in obtaining the necessary equipment, supplies, materials, insurance, and/or bonding to satisfy the work requirements of the proposal; and
- 11. Any other evidence that the Contractor submits that may show that the Contractor has made reasonable good faith efforts to include DBE participation.
- 12. Where the information submitted includes repetitious solicitation letters, it is acceptable to submit a sample representative letter along with a distribution list of the firms solicited. Documentation of DBE quotations shall be a part of the good faith effort submittal as necessary to demonstrate compliance with the factors listed above which KYTC considers in judging good faith efforts. This documentation may include written subcontractors' quotes, telephone log notations of verbal quotes, or other types of quote documentation.

The Joint Board, in consultation with KYTC's Office for Civil Rights and Small Business Development, shall review all DBE participation plan submittals. In the event that the plan submitted does not sufficiently document qualifying good faith efforts, the Joint Board may consider withholding payment for the areas that are documented in the plan for DBE participation until such time the Contractor can demonstrate they are on track to meet or exceed the Project DBE participation goal.

19.7.1.20.7.1. Failure to Meet Good Faith Effort Requirement

Upon Project completion, if the Contractor fails to satisfy the DBE participation or to meet the Project DBE participation goal and the Joint Board determines that the Contractor failed to make sufficient efforts to meet the Project DBE participation goal,

the Contractor shall be assessed financial disincentives for failure to obtain sufficient DBE participation. The Joint Board will levy financial disincentives in the contract amount it would have required the Contractor to achieve the goal less what participation was achieved. The Joint Board will deduct this from the Contract or obtain payment through any other lawful method, including the withholding of payments due. The Contractor shall be offered the opportunity to meet in person for administrative reconsideration by submitting a request through KYTC.

19.7.2.20.7.2. Administrative Reconsideration

The Contractor shall be notified of the decision to reject a request for an allowance of good faith effort credit within 30 days. The notification shall be confirmed in writing. The Contractor shall have 30 days to request reconsideration. The reconsideration meeting shall be held within seven business days of the receipt of a request from the Contractor for reconsideration.

19.8.20.8. Prompt Payment

The prime contractor will be required to pay the DBE within seven (7) working days after it has received payment from KYTC for work performed or materials furnished.

19.9.20.9. Affidavit of subcontractor Payment

The Joint Board shall monitor and verify that work committed to the subcontractor participation goal within the DBE plan at the time the Contract is awarded is performed by DBEs, pursuant to 49 CFR 26.37(b). The Contractor shall submit an affidavit to the Joint Board stating the payments made to DBE firms. The Contractor shall submit the affidavit to the Joint Board within 10 calendar days of Final Acceptance or within 10 calendar days of when work is completed by a DBE, whichever is sooner. Each DBE firm shall sign the affidavit as well, verifying the amount of payment received. The blank spaces in the affidavit must be filled in correctly, where indicated. Completed and signed affidavit shall be mailed to the KYTC Office for Civil Rights and Small Business Development, 200 Mero Street, 6th Floor West, Frankfort KY 40622.

20.021.0 Workforce/EEO

The requirements of 41 CFR Part 60 shall apply to this Contract and the Contractor shall submit EEO certification with its proposal. In addition, within 30 days of award of the Contract, the Contractor shall file Standard Form 100 (EEO-1) with KYTC's Division of Construction Procurement, unless the Contractor has submitted the form within 12 months preceding the date of the award.

20.1.21.1. Records and Reports

The Contractor shall keep such records as are necessary to determine compliance with the Contractor's EEO obligations within the greater Louisville-Southern Indiana area. The records kept by the Contractor shall be designed to indicate:

- 1. The number of minority and non-minority group members and women employed in each work classification on the Project;
- 2. The progress and efforts being made in cooperation with unions to increase employment opportunities for minorities and women (applicable only to Contractors who rely in whole or in part on unions as a source of their work force);
- 3. The progress and efforts being made in locating, hiring, training, qualifying, and upgrading minority and female employees; and
- 4. The progress and efforts being made in securing the services of minority group subcontractors or subcontractors with meaningful minority and female representation among their employees.

All records outlined in this subsection shall be retained for a period of three years following Final Acceptance or longer if notice of an additional retention period is provided by the Joint Board. Records shall be available at reasonable times and places for inspection by authorized representatives of the Joint Board and FHWA.

The Contractor shall submit to the Joint Board a monthly report after notice to proceed and for the duration of the Project, indicating the number of minority, women, and non-minority group employees currently engaged in each work classification required by the contract work. This information is to be reported on KYTC Form PR 1391.

20.2.21.2. Prequalification

KYTC has waived its prequalification requirements for bidders for this project both because of the bi-state nature of the procurement and because it feels that the States and the Joint Board are adequately protected by the short listing process. Reporting participation when payments are made to subconsultants, including material suppliers, firms at all levels (firm, subconsultant, or subcontractor) shall provide KYTC's contract administrator (the addressee for invoices under these contracts) with an accounting of said payments. This information shall be submitted as part of the requests for payments made to the Joint Board.

21.022.0 Warranties **21.1.22.1.** Warranties

21.1.1.22.1.1. Warranty

(a) Except for Softw

(a) Except for Software, Contractor warrants that (a) all design, installation, integration and maintenance work furnished pursuant to the Contract Documents shall conform to all professional principles generally accepted as standards of the industry in the State, (b) the Project shall be free of defects, including design Errors, except to the extent that such defects are inherent in prescriptive specifications included in the Contract Documents, (c) the Project shall be fit for use for the intended function, (d) materials, equipment, and Hardware furnished under the Contract Documents shall be of good quality and new, and (e) the work shall meet all of the requirements of, and perform in accordance with, the Contract Documents.

- (b) Contractor warrants that (i) all Software shall be prepared in a workmanlike manner and with professional diligence and skill, (ii) all Software will function on the machines and with operating systems for which they are designed, (iii) all Software will conform to the specifications and functions set forth in the Contract Documents; and (iv) any third party software utilized by Contractor in connection with the System, when properly installed, will function together with the Software in an integrated manner to provide the Joint Board with the benefits contemplated by this Contract.
- (c) WITH RESPECT TO THE SOFTWARE ONLY, WARRANTIES EXPRESSED IN THE CONTRACT DOCUMENTS ARE IN LIEU OF ALL OTHER WARRANTIES, EXPRESS OR IMPLIED, INCLUDING WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE.

21.1.2.22.1.2. Warranty TermPeriod and Maintenance Term

The Warranty termPeriod that is included in the Warranty Price shall commence upon Substantial Completion Acceptance. Subject to extension under Section 19.2, Tthe Warrantiesy Price shall be full compensation to the Contractor for the Warranty during the Warranty Period, which shall remain in effect until one year after the Formal Acceptance Date for a Crossing, and the Warranty Period for parts of the System not covered by the Crossing Warranties shall remain in effect until one year after the System Formal Acceptance Date. The Maintenance Term shall commence for each Crossing and the System upon expiration of the applicable Warranty Period. Upon commencement of the Maintenance Term for a Crossing, Contractor's compensation for the Warranty for a Crossing shall be the Maintenance Price applicable to such Crossing, to be paid in equal monthly installments, in arrears, over the Maintenance Term. If the Joint Board determines that any of the work has not met the standards set forth in this Section 19.122.1.1 at any time within the Warranty period prior to the end of the Maintenance Term, then Contractor shall correct such work as specified below, even if the performance of such corrective work extends beyond the stated warranty periodend of the Maintenance Term. The Contract Price is deemed to include full compensation to Contractor for all warranty and maintenance obligations hereunder.

21.1.3. Remedy

Within 7 days of receipt by Contractor of notice from the Joint Board specifying a failure of any of the work to satisfy the Warranties or of the failure of any Subcontractor representation, warranty, guarantee or obligation which Contractor is responsible to enforce, Contractor and the Joint Board shall mutually agree when and how Contractor shall remedy such violation; provided, however, that (i) in case of an emergency requiring immediate curative action or a situation which poses a significant safety risk, Contractor shall implement such action as it deems necessary and shall notify the Joint Board in writing of the urgency of a decision; and (ii) the foregoing 7 day period shall not limit or modify the Joint Board's rights to pursue and obtain Stipulated_Liquidated Damages as set forth in this Contract during such 7 day period. Contractor and the Joint Board shall promptly meet in order to agree on a remedy. If Contractor does not use its best efforts to proceed to effectuate such remedy within the agreed time, or

should Contractor and the Joint Board fail to reach such an agreement within such 7 day period (or immediately in the case of emergency conditions, a situation which poses a significant safety risk or significant potential revenue loss, as determined by the Joint Board, in its sole discretion), the Joint Board shall have the right, but not the obligation, to perform or have performed by third parties the necessary remedy, and the costs thereof shall be borne by Contractor. Alternatively, the Joint Board may deduct the amount of such costs and expenses from any sums owed by the Joint Board to Contractor pursuant to this Contract and/or from any bond then held by the Joint Board. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction or draw. The Joint Board may agree to accept Nonconforming Work in accordance with Section 7.23.

21.1.4. 22.1.4. Permits and Costs

Contractor shall be responsible for obtaining any required encroachment permits and required consents from any other Persons in connection with Warranty Work. Contractor shall bear all costs of Warranty Work and Maintenance Work, including additional testing and inspections. In the event the Joint Board determines in its sole discretion that it is experiencing revenue loss necessitating Warranty Work or Maintenance Work, -Contractor shall pay to the Joint Board the expenses incurred by the Joint Board for independent quality assurance and/or quality control with respect to the Warranty Work and Maintenance Work and any lost revenue arising from or relating to such Warranty Work and Maintenance Work (provided, however, that Contractor's liability for revenue loss resulting from Warranty Work and Maintenance Work shall be as set forth in Section 8.7), in each case, within 10 days after Contractor's receipt of invoices therefore. Alternatively, the Joint Board may deduct such amounts from any sums owed by the Joint Board to Contractor pursuant to this Contract and/or from any bond then held by the Joint Board. The Joint Board shall notify the Contractor promptly of any determination by the Joint Board to make such deduction or draw.

21.2. Applicability of Warranties to Re-Done Work

21.2.1.22.2.1. Original Warranty Work

The Warranties shall apply to all Work re-done, repaired, corrected or replaced pursuant to the terms of this Contract. Following acceptance by the Joint Board of re-done, repaired, corrected or replaced work, the Warranties as to each re-done, repaired, corrected or replaced element of the work shall extend beyond the original warranty period in order that each element of the Project (including redone, repaired, corrected or replaced work) shall have at least a one year warranty period.

21.2.2.22.2. Maintenance Warranty Work

The Warranties shall also apply to all work re-done, repaired, corrected or replaced by Contractor in the performance of its obligation to maintain the Project during the Maintenance Term. The Warranties as to each such re-done, repaired, corrected or replaced element of the Work during the Maintenance Term shall remain in effect until one year after acceptance by the Joint Board of such element of the work.

21.3.22.3. Subcontractor Warranties

21.3.1.22.3.1. Warranty Requirements

Without in any way derogating the Warranties and Contractor's own representations and warranties and other obligations with respect to all of the Work, Contractor shall obtain from all Subcontractors and cause to be extended to the Joint Board, for periods at least coterminous with the Warranties, appropriate representations, warranties, guarantees and obligations with respect to design, installation, integration, materials, workmanship, equipment, tools and supplies furnished by such Subcontractors. All representations, warranties, guarantees and obligations of Subcontractors (a) shall be written so as to survive all Joint Board inspections, tests and approvals, and (b) shall run directly to and be enforceable by Contractor and/or the Joint Board and their respective successors and assigns. Contractor hereby assigns to the Joint Board all of Contractor's rights and interest in all extended warranties for periods exceeding the applicable Warranty period which are received by Contractor from any of its Subcontractors. To the extent that any Subcontractor warranty or guaranty would be voided by reason of Contractor's negligence in incorporating material or equipment into the work, Contractor shall be responsible for correcting such defect.

21.3.2.22.3.2. Enforcement

Upon receipt from the Joint Board of notice of a failure of any of the work to satisfy any Subcontractor warranty, representation, guarantee or obligation, Contractor shall enforce or perform any such representation, warranty, guaranty or obligation, in addition to Contractor's other obligations hereunder. The Joint Board's rights under this Section 1922.3.2 shall commence at the time such representation, warranty, guaranty or obligation is furnished and shall continue until the expiration of Contractor's relevant Warranty (including extensions thereof under Section 1922.2). Until such expiration, the cost of any equipment, Hardware, Software, material, labor or shipping shall be for the account of Contractor if such cost is covered by such a representation, warranty, guaranty or obligation and Contractor shall be required to replace or repair defective equipment, material or workmanship furnished by Subcontractors. provisions concerning Subcontractor warranties are intended to provide the Joint Board with an additional Person and source in which to seek recourse if work fails to meet the requirements of the Contract Documents. In no event shall the foregoing provisions be interpreted to modify, limit, discharge, release, negate or waive the Warranties or Contractor's obligations with respect to the work, and Contractor shall not be entitled to use the existence of Subcontractor warranties as a defense to Contractor's obligations under this Contract and the other Contract Documents.

21.4.22.4. No Limitation of Liability

Subject to <u>Section 8.7</u>, the foregoing Warranties and Subcontractor warranties are in addition to all rights and remedies available under the Contract Documents or applicable Law or in equity, and shall not limit Contractor's liability or responsibility imposed by the Contract Documents or applicable Law or in equity with respect to the Work, including liability for design defects, latent construction defects, strict liability,

breach, negligence, willful misconduct or fraud; provided, however, that upon expiration of the Warranties, Contractor shall have no further liability hereunder for patent construction defects.

21.5.22.5. Damages for Breach of Warranty

Subject to Section 8.7 and in addition to the Joint Board's other rights and remedies hereunder, at law or in equity, Contractor shall be liable for actual damages resulting from any breach of an express or implied warranty or any defect in the work, including the cost of performance of such obligations by others; provided, however, that Contractor's liability for revenue loss resulting from the failure of the System to perform in accordance with the Performance Requirements shall be as set forth in Section—8.

This Contract entered into as of the day and year first written above.

KENTUCKY TRANSPORTATION CABINET
BY:
Michael W. Hancock, P.E. Date
Secretary of the Transportation Cabinet
APPROVED AS TO FORM AND LEGALITY
BY:
Rebecca Goodman Date
Executive Director of the Office of Legal Services Transportation Cabinet The Joint Board Kentucky Transportation Cabinet
(<u>[Insert KYTC Signature Block]</u>
Signature) the The Contractor

[Insert Contractor Signature Block]

-(Signature)

<u>«»«»«»«»</u>

(Printed name and title) (Printed name and title)

IN WITNESS WHEREOF, we have hereunto set our hands this ____ day of ______, 2012.



Exhibit A: INSURANCE REQUIREMENTS

[Recommend review by risk management consultant. THIS DOES NOT CONTAIN SOME COVERAGES WE WOULD EXPECT TO SEE IN A TOLL SYSTEMS/TECH CONTRACT]

1. REQUIRED INSURANCE

Without limiting Contractor's obligations to indemnify the Joint Board and the Indemnified Parties, and prior to commencement of work, the Contractor shall obtain, at its own expense, and continuously maintain in full force and effect, the insurance coverages specified in this Exhibit A. Coverage shall be maintained through Formal Acceptance, or such longer or shorter time as may be specifically provided in this Exhibit A. The Contractor may use any combination of corporate insurance policies, project-specific policies, or a contractor controlled insurance program ("CCIP") provided that the selected policies comply with all of the requirements in this Exhibit A. All insurance required hereunder shall be procured from insurance companies with an A.M. Best and Company rating level of A-: VI or better and authorized or approved to do business in the State of Kentucky, or as otherwise approved by the Joint Board. All limits of insurance set forth below are in U.S. dollars. Each policy of insurance of the type and amounts described below shall in a form satisfactory to the Joint Board.

2. PROPERTY INSURANCE (COURSE OF CONSTRUCTION)

Upon commencement of construction and with approval of the Joint Board, the Contractor shall obtain and maintain a policy of builder's risk insurance for the Project as specified below. The policy may be a Contractor corporate master program policy or a stand-alone policy as long as all of the requirements of this <u>Section 2</u> are met. The insureds shall be the Contractor, all Subcontractors[consider deleting?] (excluding those solely responsible for design work) of any tier, the Joint Board, and the Indemnified Parties. Coverage extended to the Joint Board and the Indemnified Parties shall NOT be limited by use of the phrase "as their interests may appear."

2.1. Minimum Scope

A builder's risk insurance policy shall be on an "all risk" basis. Such coverage may be provided under a the Contractor master builder's risk program or may be provided under a project-specific policy, whichever provides the broadest coverage. Such insurance shall be on a replacement cost basis using a completed value form reasonably acceptable to the Joint Board to ensure adequacy of terms and sublimits. The policy shall cover all property, roads, buildings, bridge structures, other structures, fixtures, materials, supplies, foundations, pilings, machinery and equipment that are part of or related to the portions or elements of the Project, and the works of improvement, including permanent and temporary works and materials, and including goods intended for incorporation into the works located at the Project Right of Way, in storage or in the course of transit to the Project Right of Way and all improvements that are within the Project Right of Way. The policy shall not cover tools or equipment used by any

contractor to perform their work. Such tools and equipment are the sole responsibility of the contractor who owns or uses such tools.

The builder's risk policy must include coverage for:

- (a) any ensuing loss from faulty workmanship, Nonconforming Work, error, omission or deficiency in design or specifications;
 - (a) machinery accidents and operational testing, if applicable;
- (b) removal of debris, with a sub-limit of 25% of the loss or a sublimit of no less than \$25,000,000[?], and insuring the buildings, structures, machinery, equipment, materials, facilities, fixtures and all other properties constituting a part of the Project;
- (c) transit, including ocean marine coverage (unless insured by the Supplier or through a separate marine cargo policy), with sub-limits sufficient to insure the full replacement value of any key equipment item;
- (d) sub-limits sufficient to insure the full replacement value of any property or equipment stored either on or off the Site;
- (e) loss of revenue resulting from delayed opening and expense to reduce loss:
 - (f) collapse;
 - (a) terrorism:
 - (h) plans, blueprints and specifications;
- (i) demolition and increased cost of construction as required by law or ordinance with a sub-limit of no less than \$20,000,000[?]; and
- (j) soft costs expense (including costs of Governmental Approvals, mitigation costs, attorneys' fees, and other fees and costs associated with such damage or loss or replacement thereof). The soft cost expense sublimit may be included with the sublimit for delay.

There shall be no coinsurance penalty provision in any such policy. Deductibles or self-insured retentions shall be no greater than 5% of the total value insured at the time of loss subject to a minimum deductible no greater than \$250,000[?]. All deductibles or self-insured retentions shall be the responsibility of the Contractor.

The policy shall provide a "severability of interests provision," or "multiple insured's clause" or similar wording that the policy shall apply to each insured as if a separate policy had been issued to each insured except as to limits.

2.2. Damage to Property Coverage

Coverage shall be the broadest coverage commercially available for "all risks" of direct physical loss or damage. The policy shall provide coverage per occurrence up to the full replacement cost or a loss limit based on a Probable Maximum Loss (PML) study of the covered property loss, provided, however, that the policy may include appropriate sublimits for earthquake, earth movement, tsunami and flood but in no event less than \$50,000,000 aggregate each for earthquake and flood. If a PML option is used, then the study supporting the PML must be provided to the Joint Board, and the PML may only be used as an alternative if it is approved, in writing, by the Joint Board. At its option, the Joint Board may provide a PML obtained at the Joint Board's expense. If the Joint Board accepts the PML so obtained, the Joint Board may require the Contractor to obtain coverage with a loss limit less than full replacement cost. Any additional insurance premium shall be paid for by the Joint Board through a change order. Any reduction in insurance premium shall be credited to the Joint Board through a deductive change order.

2.3. Delay Coverage

The builder's risk insurance shall provide, up to the policy limit, delay coverage for loss of revenue due to physical damage to the Project, which shall apply only for the benefit of the Indemnified Parties. Valuation shall be based on estimated lost toll revenues with a period of indemnity limited to a period of no less than one year commencing on the date substantial completion would have occurred absent the loss occurrence. The minimum limit for this coverage shall be \$65,000,000. At its option, the Joint Board may provide a PML obtained at the Joint Board expense. If the Joint Board accepts the PML so obtained, the Joint Board may require the Contractor to obtain coverage with a loss limit greater than or less than \$65,000,000[?]. Unless provided otherwise in the Agreement, any additional insurance premium shall be paid for by the Joint Board through a Change Order. Any reduction in insurance premium shall be credited to the Joint Board through a deductive Change Order.

The policy shall not contain any coinsurance provision. The policy may include a "waiting period" deductible of no more than 60 days. No liquidated damages shall be deducted from the insurance recovery unless such damages are actually collected. Coverage shall include payment for expenses incurred for the purpose of reducing any loss amount to the extent such loss amount is actually reduced. Coverage shall include loss caused by action of civil Joint Board following loss due to an insured peril.

3. WORKERS' COMPENSATION AND EMPLOYER'S LIABILITY INSURANCE

During all phases of the Project, the Contractor shall obtain and maintain and shall require all Subcontractors of all tiers to obtain and maintain, a policy or policies of insurance providing workers' compensation statutory benefits and employer's liability in conformance with the laws of the State Indiana and the Commonwealth of Kentucky. Employer's liability limits shall be no less than \$1 million each accident, each employee, and policy limit, if scheduled under the excess or umbrella liability policies to reach a total of \$100,000,000. The workers' compensation policies shall provide the following: (a) a waiver of subrogation in favor of the Joint Board and the Indemnified parties; (b) a

provision extending coverage to all states operations; (c) c voluntary compensation endorsement; (d) an alternative employer endorsement; (e) coverage for liability under the United States Longshore and Harbor Workers' Compensation Act, as appropriate, by adding a Longshore and Harbor Workers' Compensation Act coverage endorsement (WC 00 01 06 A) to a standard workers compensation policy or by purchasing standalone coverage from a federally authorized insurer; (f) coverage for liability under Title 46 of the United States Code § 688 ("Jones Act") on an "if any" basis or as otherwise appropriate; and (g) an endorsement extending the policy to cover the liability of the insureds under the Federal Employer's Liability Act on an "if any" basis or as otherwise appropriate.

4. COMMERCIAL GENERAL LIABILITY INSURANCE

The Contractor shall obtain and maintain a policy or policies of commercial general liability insurance for bodily injury, property damage, personal injury and advertising injury specifically and exclusively for the Project, Project ROW, and Site, or shall obtain a separate project-specific general aggregate limit under Contractor's corporate program of insurance. Coverage shall be written on an occurrence form that shall provide coverage at least as broad as the coverage provided by Insurance Services Office (ISO) 2000 form CG 00.

The policy or policies shall be endorsed to remove exclusions pertaining to railroads. There shall be no "contractors limitation" endorsements as that term is defined as of the date of this agreement in the Glossary of Insurance and Risk Management Terms published by the International Risk Management Institute (http://www.irmi.com/online/insurance-glossary/default.aspx), that have not been reviewed and approved by the Joint Board or its representatives. There shall be no endorsement or modification of the CGL limiting the scope of coverage for liability assumed under an insured contract.

The commercial general liability insurance coverage shall have limits of not less than \$1,000,000 per occurrence, \$2,000,000 general aggregate and \$2,000,000 products/completed operations aggregate. The policy shall have a deductible of no greater than \$ _____. Completed operations coverage shall extend for as long as there is any exposure to liability under a statute of repose or any other applicable statute either through continuous maintenance of completed operations coverage in Contractor's corporate insurance program, including an endorsement providing completed operations coverage for additional insureds, or by purchase of extended completed operations coverage for a project-specific policy. If project-specific coverage through statutory exposure is not commercially available, completed operations coverage shall extend for at least ten years from project completion. If a project-specific policy is provided, the Contractor shall be the named insured and each of the Indemnified Parties shall also be an insured as to any loss or liability arising out of or in any way related to the Project, Project ROW, or Site. If Contractor's corporate program is used with a project-specific general aggregate limit, each of the Indemnified Parties shall be an insured under that policy.

5. AUTOMOBILE LIABILITY INSURANCE

The Contractor shall obtain and maintain and shall require all Subcontractors of all tiers to obtain and maintain a business auto liability policy or policies. Each policy shall cover accidental death, bodily injury and property damage liability arising from the ownership, maintenance, or use of all vehicles connected with performance of the work, including loading and unloading. The Contractor auto liability policies shall cover "any auto" (symbol 1) or shall be specifically endorsed to include liability coverage on an excess basis for vehicles owned or operated by Contractor's Subcontractors. Policies shall include as an insured anyone liable for the conduct of an insured as defined in the policy, or shall add as insureds, the Indemnified Parties. For any contractor of any tier, including the Contractor, who will be involved in any way with the transportation of Hazardous Materials using its own vehicles, pollution liability coverage at least as broad as that provided under the ISO pollution liability-broadened coverage for covered autos endorsement (CA 99 48) shall be provided and the automobile liability insurance policies shall be endorsed to include Motor Carrier Act Endorsement-Hazardous materials clean up (MCS-90) with a sublimit of no less than \$1,000,000.

Contractor's automobile liability coverage shall have a combined single limit per policy period of not less than \$100,000,000 and may be scheduled under the excess or umbrella liability policies to achieve the desired limit. Excess or umbrella policies shall cover "any auto" (symbol 1) or shall be specifically endorsed to include liability coverage on an excess basis for vehicles owned or operated by Contractor's Subcontractors.

The Contractor shall require and confirm that no automobile liability policy issued to a Subcontractor shall contain a self-insured retention exceeding \$50,000 per accident unless approved by the Joint Board. Deductibles, shall be no greater than \$ _____ in the aggregate.

6. UMBRELLA OR EXCESS LIABILITY INSURANCE

The Contractor shall obtain and maintain an umbrella or excess liability insurance policy with limits of not less than \$100,000,000 exclusive to the Project, that will provide bodily injury, personal injury and property damage liability coverage at least as broad as the primary coverages set forth above in Exhibit A, Section 4, including commercial general liability and employer's liability in excess of the amounts set forth above. The Contractor may use its corporate insurance program or a combination of corporate insurance and stand-alone policies to meet this requirement provided that the general aggregate limits available for this Project are not subject to erosion by losses on other projects not related to this Project.

Such policy or policies shall include the following terms and conditions: (a) policies shall contain a drop down feature requiring the policy to respond in the event that any primary insurance limits are exhausted or for occurrences covered by an umbrella policy but not covered in the underlying insurance; (b) policies shall provide coverage at least as broad as found in the underlying primary policies; and (c) there shall be no "contractors limitation" endorsements as described in Exhibit A, Section 4 above, that have not been reviewed and approved by the Indemnified Parties [Replace with the Joint Board?] or their representatives. The Joint Board and the Indemnified Parties shall

be included as insureds on the excess policy including coverage extension to all insureds for completed operations.

7. POLLUTION LIABILITY INSURANCE

The Contractor shall obtain and maintain contractor's pollution liability (CPL) insurance with a total limit of liability of no less than \$50,000,000 per loss and \$50,000,000 in the aggregate per policy period dedicated to this Project. The Contractor may use a corporate CPL program, provided that coverage extends for a minimum 10-year period as described in the next paragraph. The CPL shall be obtained on an occurrence basis for a policy term inclusive of the entire period of construction. The CPL shall provide for a minimum 10-year coverage period, including the construction period and an extended reporting period. If an occurrence based form is not available, coverage may be provided under a claims-made form provided that the 10 year-coverage and claim reporting period is provided either through an extended reporting period or through continuous maintenance of coverage with no advancement of a retroactive date that is no later than commencement of the work.

The CPL policy shall include coverage for investigation, removal, and remediation costs including monitoring or disposal of contaminated soil, surface water, groundwater or other contamination to the extent required by environmental laws caused by pollution conditions resulting from covered operations; third-party bodily injury and property damage, provided that the third-party property damage liability coverage includes loss of use of damaged property or of property that has not been physically injured or destroyed, resulting from pollution conditions caused by construction operations. The policy shall have no exclusions or limitations for loss occurring over water including but not limited to a navigable waterway.

Coverage as required in this paragraph shall apply to sudden and non-sudden pollution conditions resulting from the escape or release of smoke, vapors, fumes, acids, alkalis, toxic chemicals, liquids, or gases, waste materials, or other irritants, contaminants, or pollutants. The CPL shall also provide coverage for transportation and off-Site disposal of materials.

The Joint Board and each of the Indemnified Parties shall also be insureds. The policy shall not contain any provision or exclusion (including any so-called "insured versus insured" exclusion or "cross-liability" exclusion) the effect of which would be to prevent, bar, or otherwise preclude any insured or additional insured under the policy from making a claim which would otherwise be covered by such policy on the grounds that the claim is brought by an insured or additional insured against an insured or additional insured under the policy.

8. PROFESSIONAL LIABILITY INSURANCE

During all phases of the Project, the Contractor shall obtain and maintain or cause others, as appropriate, to obtain and maintain professional liability insurance, including design build contractor's professional liability and liability for design professional

services covering professional services performed in connection with this Contract, with limits not less than \$20,000,000 per claim and in the aggregate.

No self-insured retention for the Contractor or lead design entity shall exceed \$250,000 without prior written approval from the Joint Board, in its good faith discretion. Coverage shall apply specifically to professional activities performed under the Contract Documents. The policy(ies) shall have a retroactive date consistent with the inception of the first date of design or project or construction management activities, and no later than the date on which the RFP was issued.

The Contractor agrees to maintain this required coverage for a period of no less than five years after Substantial Completion or to purchase an extended reporting period for no less than five years after Substantial Completion. The coverage shall include all Contractor-Related Entities that are performing design services with respect to the Project. If the Contractor is working with a separate lead design entity, the Contractor shall require the lead design entity to agree to maintain this coverage for a period of no less than five years after Substantial Completion or to purchase an extended reporting period for no less than five years after Substantial Completion.

9. RAILROAD PROTECTIVE LIABILITY INSURANCE

The Contractor shall provide any coverage as may be required by any railroad as a condition of the railroad's consent for entry into railroad facilities or property. Such policy shall be effective during the period any work is being performed within 50 feet of any railroad right of way.

10. AIRCRAFT HABILITY INSURANCE

The Contractor shall provide, or cause to be provided, aircraft liability insurance, with a limit of not less than \$25,000,000 per accident or higher limits as may be required by the Joint Board, in all cases where any aircraft is used on the Project that is owned, leased, hired, or chartered by any Contractor Related Entity, protecting against claims for damages resulting from such use. For any aircraft intended for use in performance of the work, the aircraft crew, flight path and altitude, including landing of any aircraft on the Site or on any property owned by the Joint Board, shall be subject to review and written acceptance by the Joint Board prior to occurrence of any such usage. If any aircraft are leased or chartered with crew and/or pilot, evidence of non-owned aircraft liability insurance will be acceptable but must be provided prior to use of the aircraft. The Joint Board and the Indemnified Parties shall be added as insureds under any such policy.

11. WATERCRAFT LIABILITY INSURANCE

The Contractor shall provide, or cause to be provided, liability insurance covering the ownership, use, maintenance, loading or unloading of watercraft related to the performance of Contractor's work or any other operations contemplated under this Contract. Policies shall provide a limit of not less than \$25,000,000 per occurrence or higher limits as may be required by the Joint Board, in all cases where any watercraft is used on the Project that is owned, leased, hired, or chartered by any Contractor-Related

Entity. Coverage may be provided through any combination of commercial general liability, marine general liability, or protection and indemnity insurance provided that all watercraft operation is covered, regardless of watercraft size. Such coverage may be arranged in any combination of primary and excess policies, all of which shall include the Joint Board and the Indemnified Parties as insureds and shall explicitly waive subrogation against the Joint Board and the Indemnified Parties.

12. GENERAL REQUIREMENTS, CONDITIONS, AND AGREEMENTS PERTAINING TO INSURANCE

12.1. Premiums, Deductibles and Self-Insured Retentions

The Contractor shall be responsible for payment of premiums for all insurance required under this <u>Exhibit A</u>. The Joint Board and the Indemnified Parties have no obligation to pay any premium. The Contractor further agrees that for each claim, suit or action made against insurance provided hereunder, with respect to all matters for which the Contractor is responsible hereunder, the Contractor shall be solely responsible for all deductibles or self-insured retentions. Any self-insured retentions maintained by the Contractor over \$250,000 must be declared and approved by the Joint Board.

At the option of the Joint Board, the insurer shall either reduce or eliminate such self insured retentions with respect to the Joint Board, and the other Indemnified Parties; or the Joint Board in its good faith discretion, may require posting of collateral by the Contractor guaranteeing payment of losses and related investigations, claims administration and defense expenses.

12.2. Evidence of Insurance

Concurrently with Contractor's execution hereof or on such later date on which coverage is required to be provided hereunder, the Contractor shall deliver to the Joint Board a copy of each policy required to be provided by the Contractor under this Exhibit A, including any corporate policies used to satisfy the terms of this Exhibit A. If any required policy is not available at the time of Contract execution, the Contractor may submit a detailed binder for each required coverage, and/or a copy of the insurer's quote for each required coverage. The evidence provided must be adequate to allow the Joint Board to determine if all insurance requirements have been met. Newly issued policies shall be delivered to the Joint Board within 10 calendar days of receipt by the Contractor. This requirement does not apply to professional liability policies, worker's compensation policies and auto liability policies, for each of which a certificate of insurance is acceptable. The Joint Board shall have no duty to pay or perform under this Contract until such evidence of insurance, in compliance with all requirements of this Exhibit A has been provided.

12.3. Enforcement of Contract Provisions (non estoppel)

The Contractor acknowledges and agrees that any actual or alleged failure on the part of the Joint Board to inform the Contractor of non-compliance with any requirement

imposes no additional obligations on the Joint Board nor does it waive any rights hereunder.

12.4. Renewal Policies

The Contractor shall promptly deliver to the Joint Board evidence of insurance with respect to each renewal policy, as necessary to demonstrate the maintenance of the required insurance coverages for the terms specified herein. Such evidence shall be delivered to the Joint Board not less than 15 days prior to the expiration date of any policy, or such shorter period as approved in advance by the Joint Board.

12.5. Policy Endorsements and Waivers

All insurance policies required hereunder shall contain or be endorsed to comply with the following provisions:

- (a) For claims covered by the insurance specified herein, said insurance coverage shall be primary insurance with respect to the insureds, additional insureds, and their respective members, directors, officers, employees, agents, and consultants, except for coverage that by its nature cannot be written as primary, and shall specify that coverage continues notwithstanding the fact that Contractor has left the Site. Any insurance or self-insurance beyond that specified in this Contract that is maintained by an Indemnified Party, their directors, officers, employees, agents, and consultants shall be excess of such insurance and shall not contribute with it.
- (a) Liability insurance shall apply separately to each insured and additional insured against whom a claim is made or suit is brought, except with respect to the limits of the insurer's liability. If Contractor's liability policies do not contain the standard Insurance Services Office separation of insureds provision, or a substantially similar clause, such policies shall be endorsed to provide cross-liability coverage.
- (b) Policies shall not be suspended, voided, canceled, or reduced in coverage or in limits except after 30 days prior written notice has been given to the Joint Board by the Contractor or its insurance broker or insurer. Such endorsement shall not include any limitation of liability of the Insurer for failure to give such notice.
- (c) All endorsements adding insureds to required commercial general liability policies shall provide additional insureds with coverage for "completed operations," or a separate endorsement providing such coverage must be added to the policy in form acceptable to the Joint Board in its good faith discretion.
- (d) Each policy shall provide coverage on an "occurrence" basis and not a "claims made" basis (with the exception of professional liability) and no policy issued on an occurrence basis shall have any sunset clause requiring reporting within a specified period of time except as specified for pollution liability policies.

12.6. Waivers by the Parties

The Joint Board and the Contractor waive all rights against each other, against each of their agents and employees and their respective members, directors, officers, employees, agents and consultants for any claims to the extent covered by insurance obtained pursuant to this Exhibit A, except such rights as they may have to the proceeds of such insurance. The Contractor shall require all Subcontractors to provide similar waivers in writing, workers' compensation/employers liability policies shall include by endorsement or otherwise, a waiver of any right of subrogation against the Joint Board and the Indemnified Parties.

12.7. Changes in Insurance Requirements

The Joint Board shall notify the Contractor in writing of any changes in the requirements applicable to insurance to be provided by the Contractor. Except as set forth otherwise in this Contract, any additional cost from such change shall be paid by the Joint Board and any reduction in cost shall reduce the Contract Price pursuant to a Change Order.

12.8. No Recourse

There shall be no recourse against the Joint Board or the Indemnified Parties for payment of premiums or other amounts with respect to the insurance to be provided by the Contractor hereunder.

12.9. Non-Limitation of Insurance Requirements

The insurance coverage provided and limits required hereunder are minimum requirements and are not intended to limit Contractor's indemnification obligations under the Contract, nor do the indemnity obligations limit the rights of the insured parties to the coverage afforded by their insured status. Requirements of specific coverage features or limits contained in this <u>Exhibit A</u> are not intended as a limitation on coverage, limits or other requirements, or a waiver of any coverage normally provided by insurance.

Specific reference to a given coverage feature is for purposes of clarification only as it pertains to a given issue and is not intended by any party or insured to be all inclusive, or to the exclusion of other coverage, or a waiver of any type. With the exception of any insurance required by a railroad, liability coverage will not be limited to the specific location designated as the Site except that if the Contractor arranges project-specific general liability, excess liability, or workers' compensation coverage, limitations of coverage to the project site will be permitted subject to the Joint Board approval and use of the broadest available site-specific endorsements. No liability policy shall contain any provision or definition that would serve to eliminate so-called "third-party-over action" claims, including any exclusion for bodily injury to an employee of the insured or of any contractor or Subcontractor.

12.10. Commercial Unavailability of Required Coverages

If, through no fault of the Contractor, any of the coverages required in this <u>Exhibit A</u> (or any of the required terms of such coverages, including policy limits) become unavailable or are available only with commercially unreasonable premiums, the Joint Board will

consider in good faith alternative insurance packages and programs proposed by the Contractor, with the goal of reaching agreement on a package providing coverage equivalent to that specified herein. The Contractor must demonstrate to the Joint Board's satisfaction that it has used diligent efforts in the global insurance markets to place the required insurance coverages, and shall advise the Joint Board of the specific results of those efforts. The Contractor shall not be entitled to any increase in the Contract Price for increased costs resulting from the unavailability of coverage and the requirement to provide acceptable alternatives. The Joint Board shall be entitled to a reduction in the Contract Price if the Joint Board, in its sole discretion, agrees to accept alternative policies providing less than equivalent coverage.

12.11. Notice and Prosecution of Claims

The Joint Board shall have the right, but not the obligation, to submit the Joint Board's claims and tenders of defense and indemnity under applicable insurance policies. Unless otherwise directed by the Joint Board in writing with respect to the Joint Board's insurance claims, the Contractor shall be responsible for reporting and processing all potential claims by the Joint Board or the Contractor or tenders for defense and indemnity under the appropriate insurance policies. The Contractor agrees to report timely to the insurer(s) under such policies any and all matters which may give rise to an insurance claim by or the Joint Board and to promptly and diligently pursue such insurance claims in accordance with the claims procedures specified in such policies, whether for defense or indemnity or both. The Contractor shall enforce all legal rights against the insurer under the applicable insurance policies and applicable Governmental Rules in order to collect thereon, including pursuing necessary litigation and enforcement of judgments.

The Contractor shall immediately notify the Joint Board, and thereafter keep the Joint Board fully informed, of any incident, potential claim, claim or other matter of which the Contractor becomes aware that involves or could conceivably involve an Indemnified Party as a defendant. The Contractor will cooperate with the Joint Board, and shall require its liability insurers to agree in writing to work with the Joint Board to assure compliance with all regarding timely response to claims.

The Joint Board agrees to promptly notify the Contractor of the Joint Board's incidents, potential claims against the Joint Board, and matters of which the Joint Board is actually aware which may give rise to an the Joint Board insurance claim or to a right of defense and indemnification under the Contract. Delivery of any such notice will constitute a tender of the Joint Board's defense of the claim to the Contractor and the insurer under any applicable Insurance Policies, subject to the Joint Board's rights to control its own defense to the extent provided in this Contract or by applicable Governmental Rules. The Joint Board shall cooperate with the Contractor as necessary for the Contractor to fulfill its duties hereunder, including providing the Contractor a copy of all written materials the Joint Board receives asserting a claim against the Joint Board that is subject to defense by an insurer under an Insurance Policy or by the Contractor under the Contract. If, in any instance, the Contractor has breached its obligations respecting insurance coverage set forth in the Contract Documents or is unable to enforce and

collect any such insurance for failure to assert claims in accordance with the terms of the Insurance Policies or to prosecute claims diligently, then for purposes of determining damages resulting from the breach or inability to enforce or collect, on or determining reductions in compensation due from the Joint Board to the Contractor, the Contractor shall be treated as if it has elected to self-insure up to the full amount of insurance coverage that would have been available had the Contractor performed such obligations or not committed such failure. Nothing in this Exhibit A or elsewhere in this Exhibit A shall be construed to treat the Contractor as electing to self-insure where the Contractor is unable to collect due to the bankruptcy or insolvency of any insurer which at the time the insurance policy is written meets the rating qualifications set forth in this Exhibit A. In the event that an insurer providing any of the insurance policies becomes the subject of bankruptcy proceedings, becomes insolvent, or is the subject of an order or directive limiting its business activities given by any Governmental Entity, including the State Department of Insurance, the Contractor shall exercise best efforts to promptly, and at its sole cost and expense, secure alternative coverage in compliance with the insurance required under this Exhibit A, has furnished original evidence of insurance for the required coverage as required under this Exhibit A, and such insurance has been approved by the Joint Board, nor shall the Contractor allow any Subcontractor to commence work under its Subcontract until the then the Joint Board shall have the right but not the obligation, to purchase such insurance or to suspend Contractor's right to proceed until proper evidence of insurance is provided. Any amounts requirements contained in this Exhibit A so as to avoid any lapse in insurance coverage.

12.12. Commencement of Work

The Contractor shall not commence work under this Contract until it has obtained the insurance required under this Exhibit A, has furnished original evidence of insurance for the required coverage as required under this Exhibit A, and such insurance has been approved by the Joint Board, nor shall the Contractor allow any Subcontractor to commence work under its Subcontract until the insurance required of the Subcontractor has been obtained and approved by the Contractor. No delay in securing such insurance, certificates of insurance or approvals shall extend any time period or Completion Deadline or provide any other relief or entitlement for Change Order.

12.13. Contractor's Failure to Comply

If the Contractor or any Subcontractor fails to provide and maintain insurance as required herein, paid by the Joint Board (plus an administrative charge equal to 10% of the cost) shall, at the Joint Board's sole option, be deducted from amounts payable to the Contractor or reimbursed by the Contractor upon demand, plus interest thereon from the date of payment by the Joint Board to the reimbursement date, at the lesser of (a) 10% per annum or (b) the maximum rate allowable under applicable Governmental Rules. Nothing herein shall preclude the Joint Board from exercising its rights and remedies under this <u>Exhibit A</u> as a result of the failure of the Contractor or any Subcontractor to satisfy the obligations of this <u>Exhibit A</u>.

If on account of Contractor's failure to comply with the provisions of this <u>Exhibit A</u>, the Joint Board is adjudged to be responsible for all or any portion of a judgment, loss or settlement (through admission or stipulation by the Contractor or court decision) that would have been covered by insurance but for non-compliance with this <u>Exhibit A</u>, then any loss or damage it shall sustain by reason thereof shall be borne by the Contractor, and the Contractor shall immediately pay the same to the Joint Board, upon receipt of written demand therefor and evidence of such loss or damage.

12.14. Subcontractor Insurance Requirements

The Contractor shall cause each Subcontractor to provide insurance that complies with requirements for the Contractor-provided insurance set forth in this Exhibit_A in circumstances where Subcontractor acts or omissions are not covered by the Contractor-provided insurance, including automobile liability. Except a otherwise specified in this Exhibit_A, the Contractor shall have sole responsibility for determining the limits of coverage required to be obtained by Subcontractors, which determination shall be made in accordance with reasonable and prudent business practices. The Contractor shall cause each such Subcontractor to include the Joint Board and the Indemnified Parties as additional insureds under such Subcontractor's general liability and excess liability insurance policies. If requested by the Joint Board, the Contractor shall promptly provide certificates of insurance evidencing coverage for each Subcontractor. The Joint Board shall have the right to contact the Subcontractors directly in order to verify the required coverage.

12.15. Disclaimer

The Contractor and each Subcontractor shall have the responsibility to make sure that their insurance programs fit their particular needs, and it is their responsibility to arrange for and secure any insurance coverage that they deem advisable, whether or not specified herein.

The Joint Board makes no representation or warranty that the coverage, limits of liability or other terms specified for the insurance policies to be carried pursuant to this Exhibit A are adequate to protect the Contractor against its undertakings under the Contract Documents or its liability to any third party or preclude the Joint Board from taking any actions as are available to it under this Contract or otherwise at law. The Joint Board shall not be limited to the amount of the insurance premium not paid in the proof of any damages it may claim against the Contractor arising out of or by reason of failure of the Contractor to provide and keep in force the insurance policies required by and on the terms of this Exhibit A, but the Joint Board shall instead be entitled to recover the full amount of damages available.

EXHIBIT B: PAYMENT BOND

Contractor:
Address:
Surety:
Address:

KNOW ALL MEN BY THESE PRESENTS, that we	
with a business address of	, as
Principal, and	, a corporation
organized and existing under the laws of the State of	with its
principal place of business at	, as
Surety, are held and firmly bound unto the Joint Board for the Louis	ville Southern-
Indiana Ohio River Bridges ("the Joint Board") in its official capacity as	s agent for the its
members, as Obligee in the sum of	
(\$), being
100% of the contract sum under the terms of the design-build contra	act between the
Principal and the Joint Board dated ("the Contract") for the work on t	
system for the Louisville-Southern Indiana Ohio River Bridges Project	("the Project"), for
the payment whereof Principal and Surety bind themselves as well as	
heirs, executives, administrators, successors and assigns, jointly a	and severally.
	•

NOW THEREFORE, the condition of this obligation is such that if the Principal shall promptly make payment to all Claimants, as that term is defined below, for all labor, materials, equipment, and supplies used, or reasonably required for use, in the performance of work on the Project, then this obligation shall be null and void; otherwise, it shall remain in full force and effect.

Further, the Principal and Surety agree to defend, indemnify and hold harmless the Joint Board and its members from any and all claims, demands, liens or suits by any person or entity which asserts a claim or demand or files a lien for the payment for labor, materials and supplies used, or reasonably required for use, in the performance of work on the Project, whether or not the claim is found to be valid.

A "Claimant" is defined as follows:

(a) A person or entity which has a direct contract with or is employed by the Principal to supply labor, materials, equipment or supplies used, or reasonably required for use, in the performance of work on the Project. The term "labor, materials, equipment or supplies" shall also include, without limitation, that part of any water, gas, fuel, power, light, heat, oil, gasoline, communication equipment or rental equipment used in the performance of work on the Project. The terms shall also include architectural and engineering services required for performance of work on the Project by the Principal or the Principal's Subcontractors, and such other items for which a Mechanic's Lien may

be asserted against the funds payable for the work on the Project under the laws of the Commonwealth of Kentucky or the State of Indiana. A person or entity which has a direct contract with the Principal shall give written notice to the Surety, with a copy sent to the Joint Board, stating its intent to file a claim under the terms of this Payment Bond, along with supporting documentation, stating the amount of the claim. Such Notice of Intent shall be provided within 90 days of when the Claimant last performed work on the Project.

- (b) A person or entity which does not have a direct contract with the Principal shall have a claim under the terms of this Payment Bond under the following conditions:
- 1. The person or entity shall have furnished written notice of its intent to assert a claim ("the Notice") to the Principal with a copy being sent to the Joint Board and the Surety within 90 days of having last performed labor or furnished materials or equipment to the Project.
- 2. In the Notice, the claimant shall have provided reasonable documentation as to the amount of its claim and documentation to establish that the labor, materials, equipment, or supplies were used, or reasonably required for use, in the performance of work on the Project.
- B. When the Surety receives written notification from a Claimant, either one having a direct contract with the Principal or one not having a direct contract with the Principal ("the Notice"), the Surety shall (a) acknowledge in writing receipt of such claim within 14 calendar days of receipt of such Notice from the Claimant; (b) send a copy of the written acknowledgement sent to the Claimant to the Joint Board; (c) promptly commence an independent investigation of the claim; (d) provide a written report and confirmation to the Claimant, with a copy sent to the Joint Board, within a reasonable time after having received Notice of the claim as to any undisputed portion of the claim; (e) arrange for payment of any undisputed portion of the claim within an additional 45 days of completing its independent investigation; and (f) send to the Joint Board evidence of any payments made to the Claimant.
- C. No suit or action shall be commenced by a Claimant under the terms of this Payment Bond after expiration of a period of one (1) year from either (a) the date on which the Claimant provided the required Notice to the Surety as set forth above or (b) the date on which the last labor, materials, equipment or service was performed or provided by the Claimant for which a claim was made, whichever occurs first.
- D. Upon a written request from any person or entity which may be a potential beneficiary under the terms of this Payment Bond, the Principal shall promptly furnish a copy of this Payment Bond or shall permit a copy of this Payment Bond to be made at the Principal's expense.

Signed and spaled this	day of	20	
Olgrica ana scalca inis	day or	 	

Witnesses:
PRINCIPAL
Title
Address
Witnesses:
SURETY
Title
Address

EXHIBIT C: PERFORMANCE BOND

Contractor:
Address:
Surety:
Address:

with a business address of , as Principal, and , a corporation organized and existing under the laws of the State of ______ with its principal place of business at , as Surety, are held and firmly bound unto the Joint Board for the Louisville-Southern Indiana Ohio River Bridges Project("the Joint Board") in its official capacity as agent for the its members, as Obligee in the sum of (\$), being 100% of the contract sum under the terms of the design-build contract between the Principal and the Joint Board dated ("the Contract") for the work on the toll collection system for the Louisville-Southern Indiana Ohio River Bridges Project ("the Project"), for the payment whereof Principal and Surety bind themselves as well as their respective heirs, executives, administrators, successors and assigns, jointly and severally.

WHEREAS the Principal has entered into the Contract with the Joint Board/Obligee for the design and construction of the Project which contract is by reference incorporated herein and made a part hereof as if fully rewritten.

NOW THEREFORE, THE CONDITION OF THIS OBLIGATION IS SUCH THAT, if the Principal shall well and truly perform and fulfill all of the undertakings, covenants, terms, conditions, warranties and obligations of the contract for the work on the Project together with any extensions thereof that may be granted by the Joint Board, the Principal and Surety shall have no obligation under this Performance Bond and it shall be null and void; otherwise, this Performance Bond shall remain in full force and effect. If there is no default by the Joint Board, the Surety's obligation under this Performance Bond shall arise under the following conditions:

1. The Joint Board has notified the Principal and the Surety in writing at their addresses shown below that the Joint Board is considering declaring the Principal in default and has requested a meeting to be attended by the Principal and the Surety and to be held not later than twenty (20) days after receipt of such written notice to discuss the circumstances that have caused the Joint Board to consider declaring the Principal in default. If after such meeting the Joint Board, the Principal and the Surety mutually agree, the Principal shall be allowed such reasonable time as specified in writing to correct the conditions identified by the Joint Board as the basis for its decision to consider declaring the Principal in default, but such an agreement shall not waive or modify the Joint Board's right to subsequently declare the Principal in default. The Joint Board may provide one or more such written notices of its intent to declare the Principal in default and conduct one or more meeting as described above without altering or amending any rights of the Joint Board set forth in this Performance Bond.

- 2. The Joint Board has declared the Principal in default in writing and formerly terminated the Principal's right to complete work on the Project. Such written declaration of default by the Joint Board shall not be issued earlier than thirty (30) days after the Principal and the Surety have received notice of the Joint Board's intent to consider the Principal in default as specified in the paragraph 1 above.
- 3. The Joint Board shall send written notice as specified above in paragraphs 1 and 2 via certified mail, return receipt requested to the Principal and the Surety at the following addresses:

Surety:

Principal:

- 4. Upon declaration of the Principal's default, the Joint Board has agreed in writing to pay the Balance of the contract price as defined below to the Surety in accordance with the terms of the Contract and the Construction Documents with the Principal.
- 5. When the Joint Board has satisfied the conditions set forth above, the Surety shall promptly commence and conduct an independent investigation at its sole expense and take the following actions at its sole expense within a reasonable time after receiving written Notice that the Joint Board has declared the Principal in default:
 - Correct any deficiencies or defects in the work by the Principal; and
- b. Pay the costs associated with design professionals, delay costs and attorneys' fees resulting from the Principal's default; and
- c. Pay any liquidated damages as set forth in the Contract due to the Principal's default; and
 - Select one of the following options.
- 1. If agreed to by the Joint Board in writing, arrange for the Principal to complete the work on the Project as the Subcontractor to the Surety which shall be the Contractor; or
- 2. Arrange for completion of work on the Project through independent contractors which are approved by the Joint Board with the Surety as the Contractor, or
- 3. Obtain bids or negotiated contracts with qualified contractors acceptable to the Joint Board to complete work on the Project in accordance with its terms and conditions and secured by a Performance Bond and a Payment Bond in the full amount of the contract to perform the remaining work, naming the Joint Board as obligee, executed by sureties acceptable to the Joint Board, and pay to the Joint Board the difference between the remaining contract funds and the contract price with the replacement contractor.

- 6. If the Surety does not proceed as provided for in Paragraph 5 above within one hundred twenty (120) days of the declaration by the Joint Board that the Principal is in default, the Surety shall be deemed to be in default of its obligations under this Performance Bond. Under such circumstances, the Joint Board shall be entitled to enforce any remedy available to the Joint Board, at law or in equity. Under such circumstances, the Surety shall remain liable to the Joint Board in an amount up to and including the full penal sum of this Performance Bond for all damages incurred by the Joint Board, including but not limited to costs for design professionals, delay cost and attorneys' fees.
- 7. No right of action shall accrue on this Performance Bond to any person or entity other than the Joint Board or its members, executors, administrators or successors.
 - 8. The Surety hereby waives notice of any change to the Contract or the Construction Documents, including changes in the time of performance,
 - 9. The Surety hereby agrees that the penal sum set forth above shall be increased to equal the amount of any change orders in the work on the Project once all change orders exceed, in the aggregate, the sum of One Million Dollars (\$1,000,000) and further agrees to issue one or more Riders as requested by the Joint Board during the course of the work on the Project to acknowledge increase(s) in the penal sum.
- 10. Any and all proceedings, legal or equitable, to enforce the terms of this Performance Bond shall be instituted in the Franklin Circuit Court in Frankfort, Kentucky. The Surety hereby consents to jurisdiction before the Franklin Circuit Court in Frankfort, Kentucky to adjudicate any and all Disputes regarding the terms of this Performance Bond. The Surety hereby waives any defense of personal jurisdiction or forum non-conveniens for actions commenced by the Joint Board or its members to enforce the terms of this Performance Bond in the Circuit Court of Frankfort Kentucky by the Joint Board or its members.
- 11. Any proceeding, legal or equitable, to enforce the terms of this Performance Bond shall be instituted within two (2) years (a) after the Joint Board declares the Principal in default, (b) after the Principal ceases work on the Project, or (c) after the Surety refuses or fails to perform its obligations under the terms of this Performance Bond, whichever occurs first.
- 12. The term "Balance of the contract price" means the following: The total amount payable by the Joint Board to the Principal under the terms of the Contract for the work on the Project after all adjustments to the original contract price have been made, including any modifications to the Contract to increase or decrease the original contract price, less all payments made to the Principal by the Joint Board under the terms of the Contract, and less (1) the costs to correct any defective or deficient work, (2) the costs associated with work by the design professionals, delay costs and attorneys fees

resulting from or related to the Principal's default, and (3) any liquidated damages payable under the Contract due to the Principal's default.

Signed and Sealed this	day of	, 2 0
	Witnesses:	
		=
	PRINCIPAL	
		=
	Title	
		=
	Address	-
	Witnesses:	
		_
	SURETY	_
	Title	_
		=
	Address	

EXHIBIT XXXX

Obtain Official wage rate schedule from the Department of Labor for Jefferson County and for Clark County, Indiana.

